What is The Family of the Law? The Influence of the Nuclear Family Model

Alan Brown

Law School, University of Strathclyde For the degree of: Doctor of Philosophy August 2016 This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyrights Acts as qualified by University of Strathclyde regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed:			
Date:			

Acknowledgments

There are several people who have my gratitude for helping me get to the end of this (seemingly impossible) task. To both my supervisors Dr Mary Neal and Professor Kenneth Norrie, who throughout have been consistent sources of valuable knowledge, helpful advice and invaluable support. To my family and friends for enduring the tedium of updates on my (non) progress. To my mum, for believing in me, for reading various drafts on far more occasions than would have seemed possible when this process began and for spotting all those errant commas. Finally, to Lynsey, for her patience, for listening to another 'philosophical question' and for being there through the ups and downs (mostly downs) of pretty much this entire process, step by step with me, I could not have done it without you.

Abstract

This thesis argues that the legal understanding of 'family' is underpinned by a particular idealised image of the family; the 'nuclear family', comprising the nexus of the conjugal relationship and the 'parent/child' relationship. I contend that this model of family is premised upon the traditional, distinct, gendered roles of 'father as breadwinner' and 'mother as homemaker', which in turn are associated with the historical, liberal understanding of the 'public/private' divide and the orthodox construction of the legal subject as rational, autonomous and self-interested. The influence of the nuclear family is noted in several different contexts: various specific legal definitions of 'family', the legal regulation of adult, conjugal relationships, the attribution of legal parenthood and the construction of the role of the 'parent' within the law.

This examination of the law's model of the 'family' has been prompted by the substantial reforms undertaken in family law in recent decades and the significant evolution in both social attitudes and familial practices that has occurred in parallel over that time. Ultimately, this thesis concludes that while these reforms have resulted in additional categories of relationship coming to be situated within the nuclear family model (notably unmarried cohabitants and same-sex couples), there has not, as yet, been any fundamental alteration of the underpinning concept of the nuclear family itself.

Table of Contents

Acknowledgmentsii		
Abstract	iv	
Chapter 1: Introduction	5	
1.1. Purpose and Scope of the Thesis	5	
1.2. Original Contribution to Knowledge		
1.3. Thesis Structure and Outline	14	
1.3.A. Part 1	14	
1.3.B. Part 2	15	
1.3.C. Part 3		
1.3.D. Conclusion.	17	
Part 1: The Family	18	
Chapter 2: What is the Law's 'Family'?	19	
Introduction	19	
2.1. 'Family' as a Social Concept	21	
2.2. The Law's 'Definition(s)' of 'Family'	24	
2.2.A. The Definition of 'Family' Under the Rent Acts		
2.2.A.1. The Role of Judicial Interpretation	28	
2.2.A.2. 'Family' as Defined by the 'Ordinary Man' Test		
2.2.A.3. The Role of the 'De Facto Familial Nexus'		
2.2.A.3.i. The Scope of the 'De Facto Familial Nexus'		
2.2.A.3.ii. The Boundaries of the 'De Facto Familial Nexus'		
2.2.A.4. Unmarried Cohabitants		
2.2.A.5. Same-Sex Cohabitants	38	
Convention on Human Rights	41	
2.2.C. The 'Formalistic' Definition of 'Family Member' in EU Law		
2.2.D. The Influence of the Nuclear Family Model within these Definition		
'Family'		
Conclusion	48	
Chapter 3: The Historical and Philosophical Underpinnings of the 'Nucle Family' Model		

Introduction	49
3.1. The 'Family' and the 'Legal Subject'	50
3.1.A. 'Family' and the Subject in Historical Societies	
3.1.A.1. The Classical Societies	
3.1.A.2. The Medieval and Renaissance Periods	
3.1.B. The Development of the 'Public/Private' Divide	
3.1.B.1. The 'Public/Private Divide' within the Law	
3.1.C. The Orthodox Understanding of the 'Legal Subject'	
3.1.C.1. The Normative Nature of the Orthodox 'Legal Subject'	
2.2. Cuitiguing the Outhodox Construction of the Legal Subject	61
3.2. Critiquing the Orthodox Construction of the Legal Subject	
3.2.A. The Apparent Objectivity Cloaks Values and Assumptions	
3.2.A.1. The Legal Subject Favours Specific Individuals	
3.2.B. The Legal Subject Masks Inequality	09 70
3.2.B.1. The Masculinity of the Orthodox 'Legal Subject'	
3.2.C. 'The Persons Cases': The Rejection of Women as Legal Subjects	12
3.3. The Nuclear Family as the Natural Model of 'Family'	
3.3.A. The Legal Subject and the Nuclear Family	
3.3.B. The Historical Significance of the Nuclear Family	
3.3.B.1. The Nuclear Family in Historical Judicial Reasoning	80
3.3.B.2. The Nuclear Family as the Dominant Social Understanding	82
Conclusion	83
Part 2: The Conjugal Relationship	85
t art 2. The Conjugar Kelationsinp	05
Chapter 4: The Legal Regulation of Conjugal Relationships	86
Introduction	86
4.1. What is Marriage in UK Law?	
4.1.A. The Legal Definition of Marriage	
4.1.A.1. The Judicial Understanding of Marriage	
4.1.A.2. The Legal Effects of Marriage	92
4.1.B. Is the Understanding of Marriage Changing?	
4.1.B.1. Same-Sex Marriage and the Understanding of Marriage	97
4.2. The Centrality of Conjugality	101
4.2.A. The Legal Regulation of Cohabitation	102
4.2.A.1. Cohabitation as a 'Marriage-Like' Relationship	106
4.2.B. The Justification for the Centrality of Conjugality	
4.2.B.1. The Central Nexus: 'Stability' and the Conjugal Couple Form	
Conclusion	114
Part 3: Parenthood	116

Chapter 5: The Attribution of Legal Parenthood within UK Law	. 117
Introduction	. 117
5.1. Natural Reproduction	. 119
5.2. Assisted Reproduction	. 123
5.2.A. Motherhood	. 126
5.2.B. Non-Gestational Female 'Parents'	. 127
5.2.C. Fatherhood	. 130
5.2.D. The Influence of the Nuclear Family Model	. 134
5.3. Surrogacy	
5.3.A. Motherhood	. 138
5.3.B. Fatherhood	. 139
Conclusion	. 141
Chapter 6: The Legal Understanding of the Parental Role	. 143
Introduction	. 143
6.1. What is the 'Parent' in the Law?	. 144
6.1.A. The Content of Parental Responsibilities and Parental Rights	
6.1.A.1. Scotland	
6.1.A.2. England and Wales	. 148
6.1.B. The Role of the 'Parent'	
6.2. The Gendered Parenting Roles of the Nuclear Family	
6.2.A. What is the 'Mother' in the Law?	
6.2.A.1. The Judicial Preference for Mothers as the Carers of Children	
6.2.A.2. The 'Natural' Construction of Motherhood	
6.2.B. What is the 'Father' in the Law?	
6.2.B.1. The Shift in the Construction of Fatherhood	
6.2.B.2. Fatherhood Outside of the Traditional Nuclear Family	
6.2.C. The Gendered Parenting Roles within the Welfare Principle	
6.2.C.1. The Significance of Contact with the Non-Resident Parent	
6.2.C.2. Contact and Gender Neutrality	. 169
6.3. The Role of Lesbian 'Mothers' or 'Parents'	. 173
6.3.A. The Judicial Approach to Lesbian Mothers	. 173
6.3.A.1. Disputes Between Lesbian Couples	
6.3.A.2. 'Known Donor' Disputes	
6.3.B. The Construction of the Role of the Non-Gestational Female 'Parent'	180
Conclusion	183

Chapter 7: Conclusion	185
7.1. Thesis Summary	185
7.2. Key Conclusions	187
7.3. The Family of the Law in the Future	188
References	190
Books	190
Chapters in Edited Collections	195
Journal Articles	
Reports	209
Newspaper Articles	
Internet Resources	212
List of Cases	213
List of Legislation	222
Domestic Legislation	222
International Legislation	

Chapter 1: Introduction

This thesis argues that the legal understanding of 'family' in the United Kingdom continues to default to the traditional 'nuclear family', which comprises the nexus of the conjugal relationship and the 'parent/child' relationship, and that this traditional archetype sits uneasily against the diversity and complexity of family forms, structures and practices in contemporary UK society. The traditional nuclear family model retains significant normative and rhetorical power and authority, in spite of the seemingly progressive legal reforms of recent decades, the vast changes in societal demographics, and the resultant prevalence and acceptance of a greater variety of family forms. Moreover, the values upon which that idealised image of family is premised are perpetuated; in particular, the separation of gender roles ('man as breadwinner' and 'woman as homemaker') which is intrinsic to the traditional nuclear family. Indeed, while the nuclear family model now encompasses relationships (same-sex couples and unmarried opposite-sex cohabitants) which were previously located outside its boundaries, the values and assumptions of the archetypical, nuclear family have not been displaced or significantly altered in the process.

Ultimately, this thesis will conclude that regardless of substantive legislative reforms, changes in social demographics and the growing diversity of familial practices, the legal understanding of the 'family' continues to be centred around the idealised image of the traditional 'nuclear family', consisting in the nexus of the conjugal relationship and the parent/child relationship, and that despite the acceptance of a greater diversity of family forms the legal system continues to conceptualise all families through the prism of the nuclear family.

1.1. Purpose and Scope of the Thesis

There have been significant changes in the familial demographics of UK society

throughout the 20th and early 21st centuries: fewer marriages, ¹ a rise in divorce, ² an increase in cohabitation outside marriage ³ and a higher number of single parent families. ⁴ Alongside these social changes, there have been substantial and diverse reforms to various aspects of Scots and English family law, including: the liberalisation of the divorce regime, ⁵ legislation governing assisted reproduction, ⁶ increasing legal recognition of unmarried cohabiting couples, ⁷ reform of the adoption regime, ⁸ the introduction of civil partnership ⁹ and the subsequent extension of marriage to same sex couples. ¹⁰ Writing extra-judicially, the President of the Family Division Sir James Munby has observed that '[p]erhaps in no other area of law have there been such changes in recent decades as in family law. ¹¹

¹ Office for National Statistics (ONS), 'Marriages in England and Wales (Provisional), 2012', (June 2014), at 2-3, shows a significant decline in marriages since the late 1960s (from over 400,000 per year to under 250,000 per year), available at - http://www.ons.gov.uk/ons/dcp171778_366530.pdf. Similar statistics are available for Scotland, which show a decline from over 40,000 marriages per year in the 1960-70s to under 30,000 per year in recent years, National Records of Scotland, 'Marriage and Civil Partnership Time Series Data', (August 2015), available at - http://www.nrscotland.gov.uk/statistics-and-data/statistics/statistics-by-theme/vital-events/marriages-and-civil-partnerships/marriages-time-series-data.

² ONS, 'Divorces in England and Wales, 2012', (February 2014), at 2, shows a substantial increase in divorce both generally from the 1930s onwards (in 1931 there were only 3764 instances of divorce) but also specifically in the period from the late 1960s to the 1980s, after the reform of the divorce regime (from under 24,000 divorces in 1960 compared with a peak of around 160,000 in 1985), available at -http://www.ons.gov.uk/ons/dcp171778_351693.pdf. Similar statistics are available for Scotland, showing a general increase from only 468 divorces in 1934, with a similar peak of 13,365 in 1985, National Records of Scotland, 'Divorces Time Series Data', (March 2013), available at -http://www.nrscotland.gov.uk/statistics-and-data/statistics/statistics-by-theme/vital-events/vital-events-divorces-and-dissolutions/divorces-time-series-data.

ONS, 'Short Report: Cohabitation in the UK, 2012', (November 2012), at 1, shows that there were 5.9 million people cohabiting in the UK in 2012, which is around double the figure from 1996, available at - http://www.ons.gov.uk/ons/dcp171776_284888.pdf.

⁴ ONS, 'Social Trends: Households and Families' (No. 41, February 2011), at 7, shows that between 2001-2010 the percentage of people in 'Married Couple Families' has fallen from 72.4% to 68% and that there has been a corresponding increase in 'Cohabiting Couple Families' (12.5% to 15.3%) and 'Lone Parent Families' (14.8% to 16.2%) during that period, available at - http://www.ons.gov.uk/ons/rel/social-trends-rd/social-trends/social-trends-41/index.html.

⁵ Divorce Reform Act 1969 and Divorce (Scotland) Act 1976.

⁶ Human Fertilisation and Embryology Act 2008, which amended the Human Fertilisation and Embryology Act 1990.

⁷ Through both the specific statutory regime in s.25-30 Family Law (Scotland) Act 2006 and the ad hoc inclusion of cohabitants within other statutory provisions, e.g. the Child Support Act 1991.

⁸ Adoption and Children Act 2002 and Adoption and Children (Scotland) Act 2007.

⁹ Civil Partnership Act 2004, also notable is the Gender Recognition Act 2004, which provided the means for legal recognition of a change of gender, subsequent to the decision in *Bellinger v Bellinger* [2003] 2 AC 467.

¹⁰ Marriage (Same Sex Couples) Act 2013 and the Marriage and Civil Partnership (Scotland) Act 2014.

¹¹ Sir James Munby, 'Years of Change: Family Law in 1987, 2012 and 2037' [2013] 43 (3) Fam. Law 278, at 278. See further e.g. John Eekelaar, 'Then and Now - Family Law's Direction of Travel' (2013) 35 (4) Journal of Social Welfare and Family Law 415.

However, as Kaganas and Day Sclater have commented, 'it has long been recognised that law is a clumsy tool for managing complex family problems, let alone intimate relationships or emotional trauma. The law is clumsy because it deals in generalities and is ill-equipped to take full account of the subtleties of emotion and motivation.' Thus, the complexity and variability of 'family' structures and forms in 21st century society clashes with the law's desire for clarity and certainty and this ongoing tension underpins the arguments advanced within this thesis. As a result of this tension, as Chan observes, 'it is clear that what counts as a family, for legal purposes, is itself a legal construct.' It is that construct, the legal understanding of the 'family', which this thesis examines. I argue that the law in the UK continues to be underpinned by the idealised image of the traditional nuclear family and that this is problematic for the regulation of the diversity of family forms and practices within 21st century UK society.

In order to evidence this central argument, the thesis will consider how the law defines and understands the 'family'. This thesis will demonstrate that there is no single, overarching *definition* of the 'family' within the law, but that instead there is a determinate and archetypical *idealised image* of the 'family' which underpins and influences legal regulation of familial relationships and is employed by the law across different contexts. This idealised image is that of the traditional 'nuclear family', consisting in the nexus of the conjugal relationship and the parent/child relationship. The elements of the nuclear family image will be examined throughout the thesis, in order to reveal the underlying assumptions and values which underpin the law's construction and understanding of 'family.' A number of contexts in which there is a specific legal definition of 'family' will be examined, in order to illustrate how the idealised image of the nuclear family has influenced the judicial interpretation of these definitions, drawing the boundary of 'family' around adult conjugal relationships and (at least de facto) parent/child relationships, and excluding

¹² Felicity Kaganas and Shelley Day Sclater, 'Contact Disputes: Narrative Constructions of 'Good' Parents' (2004) 12 (1) Feminist Legal Studies 1, at 6.

¹³ Winnie Chan, 'Cohabitation, Civil Partnership, Marriage and the Equal Sharing Principle' (2013) 33 (1) Legal Studies 46, at 46.

all other forms of long-term, interdependent and 'caring' relationships from the law's understanding of 'family'.

Subsequently, I will consider the legal understanding and regulation of the core familial relationships that comprise the central nexus of the nuclear family. This will involve detailed consideration of both (i) the law's approach to recognising and regulating adult personal relationships and (ii) the legal understanding of parenthood and parenting. I will observe that legal regulation of adult personal relationships remains centred on 'marriage-like' conjugality, with traditional, heterosexual marriage still the law's model or 'gold standard'. Non-marital relationships (those between same-sex couples and unmarried opposite-sex cohabitants) are recognised and regulated by the law insofar as they share sufficient features and characteristics in common with marriage to be considered capable of performing similar societal functions. Since relationships between same-sex partners and unmarried heterosexual cohabitants can also be located within the central nexus of the nuclear family model (the conjugal relationship and the parent/child relationship), their recognition and regulation does not alter the fundamental reliance upon that idealised image, nor affect its normative power in shaping the legal understanding of the 'family'.

I will also illustrate the continuing centrality of the nuclear family model by examining the legal regulation of the parent/child relationship and showing how, despite the increasing use of the gender-neutral language of the 'parent', the law still deploys a gendered, binary model of parenting and parenthood wherein the gendered parenting roles of the mother and the father are understood and constructed differently. This will be demonstrated via a consideration of two distinct aspects of the legal regulation of the parent/child relationship: the attribution of legal parenthood, and the legal understanding of the role of the 'parent'. Regarding the former, I will observe that the law's approach to the attribution of legal parenthood is premised upon a binary, two-parent model, ideally consisting of one mother and one father. It will be shown that in determining parenthood, the law continues to attach significance to marriage and thereby to be influenced by the traditional, nuclear family model. This is particularly apparent in the contexts of assisted reproduction

and surrogacy, where more complex, 'alternative' forms of reproduction are disruptive to traditional understandings and thus are potentially transformative, but are nevertheless forced within the confines of the binary, two-parent model of the nuclear family. I will build on this in considering the construction of the parental role, arguing that in spite of the appearance of the usage of the gender-neutral language of 'parent', the law continues to rely upon the traditional, gendered constructions of 'mother as natural carer' and 'father as breadwinner' associated with the nuclear family model, in its understanding of the parental role. I will illustrate this by exploring judicial understandings of the parental role and showing the continued normative influence of the gendered parenting roles upon these understandings.

Ultimately, I will conclude that, in spite of the substantial volume of family law reform over the past several decades, and the significant developments in social and familial demographics within UK society, the legal understanding of the 'family' continues to be underpinned by the idealised image of the traditional nuclear family, comprising the nexus of the conjugal relationship and the parent/child relationship.

1.2. Original Contribution to Knowledge

Recent reforms in family law have created the space within which this thesis makes its distinctive and original contribution to knowledge. It does so in two main ways: first, by sustained analysis and critique of the continuing hold of the traditional nuclear family model on the legal understanding of 'family'; and second, by describing and understanding the 'nuclear family' as comprising the nexus of these two core familial relationships. The overarching conclusion of the thesis is that, despite radical shifts in family profiles and a considerable volume of legal reforms responding to these shifts, there has *not* been the radical shift that might appear at first sight in the law's understanding of 'family', since the latter remains firmly premised upon the nuclear family model. This conclusion advances knowledge in the field by considering the legal meaning of 'family' in light of the considerable number

of recent reforms in family law.

To that end, this research builds upon the critical examinations of the legal understanding of 'the family' in the work of earlier writers, including Boyd, 14 Dewar, 15 Glennon 16 and Millbank. 17 The consideration of the 'family' in this thesis has been particularly influenced by two works which pre-date much of the aforementioned legislative reform; Diduck's, Law's Families¹⁸ and O'Donovan's, Family Law Matters. 19 In Law's Families, Diduck observed that, '[t]he family on which family law and policy is based bears only slight resemblance to the way we do family inside and outside the home on a day-to-day basis'. 20 This tension between the legal construct of the 'family' and the everyday reality of family life and familial practices continues to be apparent. In Family Law Matters, O'Donovan comments that, '[f]amily law is dynamic, it is in flux. But it is embedded in the legal tradition from which it springs.'²¹ This statement remains as pertinent now as it was 25 years ago, and this thesis explores the contradiction between the need for family law to respond dynamically to changes in family forms and practices and the continuing dominance of traditional ideals and values about the nature and structure of the 'family'.

My central argument is that different types of relationship have been subsumed into the existing, dominant model of family: the nuclear family, comprising the nexus of the conjugal relationship and the parent/child relationship. As such, this thesis is firmly grounded in the existing literature, but it also develops and builds upon the

-

¹⁴ Susan Boyd, 'What is a 'Normal' Family? *C v C (A Minor) (Custody: Appeal)*' (1992) 55 (2) Modern Law Review 269.

¹⁵ John Dewar, 'The Normal Chaos of Family Law' (1998) 61 (4) Modern Law Review 467.

¹⁶ Lisa Glennon, 'Fitzpatrick v Sterling Housing Association Ltd. - An Endorsement of the Functional Family?' (2000) 14 (3) International Journal of Law, Policy and the Family 226.

¹⁷ Jenni Millbank, 'The Limits of Functional Family: Lesbian Mother Litigation in the Era of the Eternal Biological Family' (2008) 22 (2) International Journal of Law, Policy and the Family 149.

¹⁸ Alison Diduck, *Law's Families*, (Markham, 2003), see also e.g. Alison Diduck, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood' [2007] 19 (4) Child and Family Law Quarterly 458, Alison Diduck, 'Shifting Familiarity' (2005) 58 (1) Current Legal Problems 235 and Alison Diduck, 'What is Family Law For?' (2011) 64 (1) Current Legal Problems 287.

¹⁹ Katherine O'Donovan, *Family Law Matters*, (Pluto Press, 1993), see also Alison Diduck and Katherine O'Donovan (eds.), *Feminist Perspectives on Family Law*, (Abingdon, 2006).

²⁰ Diduck, *Law's Families*, at 212.

²¹ O'Donovan, Family Law Matters, at xiv.

insights of earlier authors given the changing context of legislative reform, societal demographics and family forms and practices.

1.2.A. Themes of Analysis

As well as the aforementioned literature which considers the 'family' as an overarching legal concept, my work relies upon and is situated within the literature which identifies and critiques the gendered nature of the dominant legal and cultural understanding of parenthood.²² My exploration of the legal construction of parenthood in Chapters 5 and 6 is underpinned by the work of various writers who examine the legal understanding of the gendered parenting roles of 'mother' and 'father' and explore the continuing influence of these traditional roles on the way the role of the 'parent' is currently understood within the law.

Moreover, when considering the legal understanding of 'motherhood', I join the critical literature which challenges the dominant construction of motherhood as a 'natural', indivisible status, fundamentally linked to the process of gestation.²⁵ I also engage with the body of work which explores the legal understanding of lesbian parenthood²⁶ in order to consider the continuing influence of the dominant, 'natural',

_

²² See e.g. Emily Jackson, 'What is a Parent?', in Diduck and O'Donovan (eds.), *Feminist Perspectives on Family Law*, and Kim Everett and Lucy Yeatman, 'Are Some Parents More Natural Than Others?' [2010] 22 (3) Child and Family Law Quarterly 290, Susan Boyd, 'Gendering Legal Parenthood: Bio-Genetic Ties, Intentionality and Responsibility' (2007) 25 Windsor Yearbook of Access to Justice 63 and Kaganas and Day Sclater, 'Contact Disputes: Narrative Constructions of 'Good' Parents'.

²³ See e.g. Katherine O'Donovan, 'Constructions of Maternity and Motherhood in Stories of Lost Children' in Jo Bridgeman and Daniel Monk (eds.), *Feminist Perspectives on Child Law*, (Cavendish, 2000)

²⁴ See e.g. Sally Sheldon and Richard Collier, *Fragmenting Fatherhood: A Socio-Legal Study*, (Hart, 2008).

²⁵ See e.g. Katherine O'Donovan and Jill Marshall, 'After Birth: Decisions about Becoming a Mother', in Diduck and O'Donovan (eds.), *Feminist Perspectives on Family Law*, Gillian Douglas, 'The Intention to Be a Parent and the Making of Mothers' (1994) 57 (4) Modern Law Review 636 and O'Donovan, 'Constructions of Maternity and Motherhood in Stories of Lost Children' in Bridgeman and Monk (eds.), *Feminist Perspectives on Child Law*.

²⁶ See e.g. Sarah Beresford, 'Get Over Your (Legal) 'Self': A Brief History of Lesbians, Motherhood and the Law' (2008) 30 (2) Journal of Social Welfare and Family Law 95, Fiona Kelly, '(Re)Forming Parenthood: The Assignment of Legal Parentage Within Planned Lesbian Families' (2009) 40 (2) Ottawa Law Review 185, Robert Leckey, 'Law Reform, Lesbian Parenting, and the Reflective Claim' (2011) 20 (3) Social and Legal Studies 331 and Leanne Smith, 'Tangling the Web of Legal Parenthood: Legal Responses to the Use of Known Donors in Lesbian Parenting Arrangements'

indivisible construction of the role of the 'mother' despite recent legislative reforms which enable the recognition of a non-gestational female 'parent' in lesbian couples.²⁸ Similarly, my discussion of the legal understanding of 'fatherhood' draws upon the developing body of literature²⁹ which has identified the 'fragmentation of fatherhood';30 this literature observes the tension within the law between the traditional construction of the 'father', based upon the public, 'breadwinner' role, and the developing construction of a private, caring 'fatherhood', based around greater involvement in day-to-day parenting.³¹

The theoretical framework of this thesis draws on the work of feminist writers who have critiqued the gendered and patriarchal nature of law itself, including the works of Pateman, 32 Moller Okin, 33 Boyd 34 and O'Donovan. 35 In discussing the underpinning values of the nuclear family in Chapter 3 this thesis focuses upon those feminist theorists who have critiqued the dominant, orthodox, liberal construction of the 'legal subject' as intrinsically 'masculine'. Naffine, 37 for example, has

^{(2013) 33 (3)} Legal Studies 355.

27 Within this, the thesis engages with the empirical literature on the parenting of lesbian couples, see e.g. Gillian Dunne, 'Opting into Motherhood: Lesbians Blurring the Boundaries and Transforming the Meaning of Parenthood and Kinship' (2000) 14 (1) Gender and Society 11, Jacqui Gabb, 'Lesbian M/Otherhood: Strategies of Familial-linguistic Management in Lesbian Parent Families' (2005) 39 (4) Sociology 585 and Leanne Smith, 'Is Three a Crowd? Lesbian Mothers Perspectives on Parental

Status in Law' [2006] 18 (2) Child and Family Law Quarterly 231.

28 Under the 'status provisions' for parenthood in cases of assisted reproduction in s.42-48 Human Fertilisation and Embryology Act 2008.

²⁹ See e.g. Richard Collier, *Masculinity, Law and the Family*, (Routledge, 1995), Deborah Lupton and Lesley Barclay, Constructing Fatherhood: Discourses and Experiences, (SAGE, 1997), Sally Sheldon, 'Fragmenting Fatherhood: The Regulation of Reproductive Technologies' (2005) 68 (4) Modern Law Review 523 and Leanne Smith, 'Clashing Symbols? Reconciling Support for Fathers and Fatherless Families After the Human Fertilisation and Embryology Act 2008' [2010] 22 (1) Child and Family Law Quarterly 46.

³⁰ See Sheldon and Collier, Fragmenting Fatherhood: A Socio-Legal Study.

³¹ See e.g. Felicity Kaganas, 'A Presumption that 'Involvement' of Both Parents is Best: Deciphering Law's Messages' [2013] 25 (3) Child and Family Law Quarterly 270, Scott Coltrane, Family Man: Fatherhood, Housework and Gender Equality, (Oxford University Press, 1996), Anna Dienhart, Reshaping Fatherhood: The Social Construction of Shared Parenting, (SAGE, 1998) and Barbara Hodson (ed.), Making Men into Fathers: Men, Masculinities and the Social Politics of Fatherhood, (Cambridge University Press, 2002).

² Carole Pateman, *The Sexual Contract*, (Polity, 1988).

³³ Susan Moller Okin, Women in Western Political Thought, (Virago, 1980).

³⁴ Susan Boyd (ed.), Challenging the Public/Private Divide: Feminism, Law and Public Policy, (Toronto University Press, 1997).

³⁵ Katherine O'Donovan, Sexual Divisions in Law, (Weidenfeld and Nicholson, 1985).

³⁶ See e.g. Susan James and Stephanie Palmer (eds.), Visible Women: Essays on Feminist Legal Theory and Political Philosophy, (Hart, 2002) and Nicola Lacey, Unspeakable Subjects: Feminist Essays in Legal and Social Theory, (Hart, 1998). This thesis also engages with the more general critique of the

extensively critiqued the orthodox construction of the legal subject and who has characterised that subject as the 'man of law'. 38

The consideration of the legal and social understanding of the 'family' within this thesis is also influenced by the growing body of literature on the moral and ethical significance of 'care', ³⁹ particularly the work of Held, ⁴⁰ Tronto ⁴¹ and Noddings. ⁴² This literature is premised upon recognition of the 'relational subject', as Held describes, '[t]he ethics of care...characteristically sees persons as relational and interdependent, morally and epistemologically ⁴³ and consequently this literature focuses on the centrality of relationships and interdependence. ⁴⁴ Most relevantly for the analysis here, the literature on 'care' has been developed by Herring, in *Caring and the Law*, ⁴⁵ toward an alternative normative approach ⁴⁶ which seeks to diminish

orthodox, liberal construction of the legal subject, see e.g. Costas Douzinas and Adam Geary, *Critical Jurisprudence: The Political Philosophy of Justice*, (Hart, 2005), Anthony Carty (ed.), *Post-Modern Law: Enlightenment, Revolution and the Death of Man*, (Edinburgh University Press, 1990), Costas Douzinas and Ronnie Warrington, *Justice Miscarried: Ethics, Aesthetics and the Law*, (Harvester Wheatsheaf, 1994) and Zygmunt Bauman, *Postmodern Ethics*, (Blackwell, 1993).

³⁷ See e.g. Ngaire Naffine, Law and the Sexes: Explorations of Feminist Jurisprudence, (Allen and Unwin, 1990), Ngaire Naffine and Rosemary J. Owens (eds.), Sexing the Subject of Law, (LBC Information Services, 1997), Ngaire Naffine, Law's Meaning of Life: Philosophy, Religion, Darwin and the Legal Person, (Hart, 2009), Ngaire Naffine, 'Who are Law's Persons: From Cheshire Cats to Responsible Subjects' (2003) 66 (3) Modern Law Review 346 and Ngaire Naffine, 'Can Women Be Legal Persons?' in James and Palmer (eds.), Visible Women: Essays on Feminist Legal Theory and Political Philosophy.

³⁸ Naffine, *Law and the Sexes: Explorations of Feminist Jurisprudence*, states, at 100, '[t]he legal model of the person, it will be argued, is a man, not a woman. He is a successful middle-class man, not a working class male. And he is a middle-class man who demonstrates what one writer has termed a form of 'emphasised' middle-class masculinity. In short, he is a man; he is a middle class man; and he evinces the style of masculinity of the middle class.'

³⁹ See e.g. Carol Gilligan, *In a Different Voice: Psychological Theory and Women's Development*, (Harvard University Press, 1982), Marilyn Friedman, *Liberating Care*, (Cornell University Press, 1993) and Ruth Groenhout, *Connected Lives: Human Nature and an Ethics of Care*, (Rowman and Littlefield, 2004).

⁴⁰ Virginia Held, *The Ethics of Care*, (Oxford University Press, 2006).

⁴¹ Joan Tronto, Moral Boundaries: a Political Argument for an Ethic of Care, (Routledge, 1993).

⁴² See e.g. Nel Noddings, *Caring: A Feminine Approach to Ethics and Moral Education*, (University of California Press, 1984) and Nel Noddings, *Starting at Home: Caring and Social Policy*, (University of California Press, 2002).

⁴³ Held, *The Ethics of Care*, at 13.

⁴⁴ Jonathan Herring, 'The Disability Critique of Care' (2014) 8 Elder Law Review 1, observes, at 3, that '[e]thics of care is based on the belief that people are relational. People understand themselves in terms of their relationships.'

⁴⁵ Jonathan Herring, *Caring and the Law*, (Hart, 2013).

⁴⁶ Jonathan Herring, 'The Legal Duties of Carers' (2010) 18 (2) Medical Law Review 248, at 254, states, '[r]elying on an ethic of care, I would promote a vision of the law which sees people with interdependent relationships as the norm around which legal and ethical responses should be built.'

the law's reliance on the abstract, individualised subject. Herring suggests that this understanding of the law 'starts with a norm of interlocking mutually dependent relationships, rather than an individualised version of rights. I will engage with this literature throughout my consideration of the legal understanding of the 'family' in Chapter 2, my exploration of the legal regulation of the conjugal relationship in Chapter 4 and my examination of the legal understanding of the role of the parental role in Chapter 6. However, it is not my aim in this thesis to develop such an alternative normative approach to understanding the 'family', but rather to elucidate, and demonstrate the continuity of, the law's preference for the 'nuclear family' form.

1.3. Thesis Structure and Outline

The thesis is structured in three parts, each of which explores a different familial relationship.

1.3.A. Part 1

Part 1, which consists of Chapters 2 and 3, considers the 'family' as a legal concept. To begin with, Chapter 2 briefly considers 'family' as a wider social concept, noting the lack of consensus as to its meaning. Subsequently the chapter observes that the law in the United Kingdom similarly lacks a single overarching definition of the 'family', either in statute or under common law. However, the chapter will identify three specific contexts in which legal definitions of family have apparently been developed: succession by 'family members' to private sector tenancies under the Rent Acts; the right to 'family life' in Article 8 of the European Convention on Human Rights; and the free movement of 'family members' under the EU 'Citizenship Directive'. The chapter will show that, in each of these contexts, the idealised image of the traditional, nuclear family (comprising the nexus of the

⁴⁷ The relationship between the literature on 'care' and the orthodox construction of the 'legal subject' as rational, autonomous and self-interested will be considered below at Chapter 3, section 3.2, 'Critiquing the Orthodox Construction of the Legal Subject'.

⁴⁸ Herring, *Caring and the Law*, at 46.

conjugal relationship and the parent/child relationship) is central to how these definitions are interpreted and understood by the judiciary.

Chapter 3 builds upon Chapter 2 by exploring why this idealised image of the nuclear family has come to exert such significant influence upon the legal understanding of the family. The chapter will argue that the combination of the following factors; the historical recurrence and dominance of the nuclear family form, the orthodox, liberal construction of the 'legal subject' and the continuing influence of the separate and distinct gender roles of male 'breadwinner' and female 'homemaker' of the 'public/private' divide, has resulted in the traditional, nuclear family being positioned as the 'natural' and 'common sense' construction of the 'family' within social and cultural discourse. Consequently, I will argue that this positioning has resulted in the nuclear family coming to underpin the law's definitions and understanding of the 'family'.

1.3.B. Part 2

Part 2, which consists solely of Chapter 4, examines one of the core relationships within the central nexus of the nuclear family: the conjugal relationship. This chapter shows how the legal regulation of adult personal relationships is premised around marriage and 'marriage-like' conjugal relationships. The chapter considers the continuing centrality of marriage within the law, suggesting that the historical judicial definition retains significant influence upon contemporary judicial understandings of marriage, in spite of the recent legislation extending marriage to same-sex couples. Subsequently, the chapter explores the centrality of conjugality to the legal regulation of adult relationships. I will argue that the law privileges the conjugal couple form because it is constructed as performing important social functions, both for its members themselves and in relation to the upbringing of children. Within this, I will focus upon the legal regulation of cohabitation and note the extension of the marriage model to other relationships. I will argue that this extension has occurred on the basis that such relationships are understood as being sufficiently 'marriage-like'; they share characteristics and features (with particular

emphasis given to their conjugality) with traditional, heterosexual marriage such that they are positioned within the idealised image of the nuclear family.

1.3.C. Part 3

Part 3, comprising Chapters 5 and 6, examines the other core relationship within the central nexus of the nuclear family: the 'parent/child' relationship. These chapters explore the law's approach to parenthood and parenting, focusing upon the attribution of legal parenthood and the legal understanding of the parental role. Chapter 5 considers how legal parenthood is determined and assigned in a variety of contexts: natural reproduction, medically assisted reproduction and surrogacy. Although the chapter will set out the substantive differences between the regimes which determine legal parenthood in cases of natural reproduction and assisted reproduction, I will argue that across these diverse contexts legal parenthood is consistently premised upon a binary, two-parent model, which ideally envisages one mother and one father as the legal parents.

Chapter 6 builds upon this and explores the legal construction of the role of the 'parent'. The chapter will argue that the law has a relatively opaque understanding of the role of the 'parent'. As a result of this the legal understanding of parenthood continues to rely upon the archetypical gendered parenting roles of 'mother' and 'father', which are understood as representing the 'natural' and 'common-sense' parental roles. The chapter also examines the legal interpretation of the welfare of the child, arguing that this interpretation provides evidence of the influence of these gendered parenting roles. Additionally, the chapter will consider the legal understanding of the role of the non-gestational female parent in lesbian couples. I will observe that there has been limited legislative or judicial engagement with the nature of this parental role because it lacks the readily understood 'natural' construction possessed by the gendered parenting roles. This will be taken to support the conclusion that the legal understanding of the role of the 'parent' is premised upon the distinct, gendered parenting roles of 'mother' and 'father' of the nuclear family.

1.3.D. Conclusion

It will be concluded, in Chapter 7, that the legal understanding of 'family' in the United Kingdom continues, notwithstanding radical law reform over the past four decades, to be premised upon the idealised image of the traditional 'nuclear family', comprising the nexus of the conjugal relationship and the 'parent/child' relationship. The normative power of the idealised image of the nuclear family is evident in the fact that legal regulation of adult relationships is centred upon the conjugal relationship. Moreover, the gendered parenting roles of 'mother' and 'father', which are embedded in the nuclear family, continue to possess normative influence in the judicial and legislative understanding of the role of the 'parent'. Accordingly, the overall conclusion of this thesis will be that despite legislative reforms and changes in familial demographics, the traditional nuclear family remains the idealised image upon which the legal understanding of the 'family' is based.

Part 1: The Family

Chapter 2: What is the Law's 'Family'?

Introduction

Diduck has observed that, "[t]he Family" has almost iconic status in popular and "official" discourses, even though there is no official or universal definition of it. It means different things to different people, and meets different needs for different people.' Indeed, social and cultural understandings of 'family' are shifting and 21st century UK society comprises of a diversity of family structures and forms. These shifts are reflected in the language used by both the government and the judiciary when they consider and discuss families. The foundational document of the previous coalition government stated that, '[t]he Government believes that strong and stable families of all kinds are the bedrock of a strong and stable society. Similar rhetoric appears within judicial language; in Fitzpatrick v Sterling Housing Association Ltd. Lord Clyde observed that, '[t]he concept of the family has undergone significant development during recent years, both in the United Kingdom and overseas... Social groupings have come to take a number of different forms. Thus, both the government and the judiciary now acknowledge a greater variety of relationships as 'family'.

[.]

¹ Diduck, *Law's Families*, at 1.

² Zygmunt Bauman, *Liquid Love: On the Frailty of Human Bonds*, (Polity Press, 2003), observes, at 91, '[a]n unprecedented fluidity, fragility and in-built transience (the famed "flexibility") mark all sorts of social bonds which but a few dozen years ago combined into a durable, reliable framework inside which a web of human interactions could be securely woven.'

³ See e.g. ONS, 'Social Trends: Households and Families', No. 41, ONS, 'Short Report: Cohabitation in the UK, 2012' and ONS, 'Marriages in England and Wales (Provisional), 2012' for evidence of the shifts in demographics within UK society.

⁴ This is especially notable when compared to the 'family values' rhetoric of the Conservative Governments of the late 1980s and early 1990s.

⁵ 'The Coalition: Our Programme for Government', (HM Government, May 2010), at 19, emphasis mine, available at -

 $[\]underline{https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/78977/coalition_programme_for_government.pdf.}$

⁶ [2001] 1 AC 27, see further e.g. Ghaidan v Godin-Mendoza [2004] 2 AC 557, Re G (Children) (Residence: Same-Sex Partner) [2006] 2 FLR 629 and Re P (Adoption: Unmarried Couples) [2008] 2 FLR 1084.

⁷ ibid, per Lord Clyde, at 51.

⁸ Deborah Chambers, A Sociology of Family Life: Change and Diversity in Intimate Relationships, (Polity Press, 2012), at 53, observes that, '[w]hat we define as "family" is now more flexible and dynamic, and embraces new kinds of intimacies that were once ignored or condemned.'

However, in spite of this apparent shift in the understanding of 'family' and the range of recent legislation concerning various aspects of the 'family', there are relatively few statutes which utilise the term 'family' itself - at least as a concept activating legal consequences¹⁰ - and the law in the United Kingdom lacks a single, overarching definition of 'family'. In Fitzpatrick¹¹ Lord Nicholls of Birkenhead observed that:

Family is a word with several different meanings. In some contexts family means children ("when shall we start a family?") In other contexts it means parents and child ("accommodation suitable for families"). It may mean all persons connected however remotely by birth, marriage or adoption ("family tree"). 12

Because of the variability and indeterminacy inherent in the concept of 'family', I will argue that the judiciary - and the law itself - draws upon an archetypical, but determinate image of 'family'; the traditional, nuclear family. To understand how and why the law relies upon this idealised image of 'family', it is necessary to first consider the meaning of 'family' as a wider social concept.

Accordingly, this chapter will begin in section 2.1, by considering 'family' as a social concept, observing the lack of an overarching definition within the literature. In section 2.2, I will consider various legal definitions of 'family' which exist within specific contexts. I will argue that, in spite of the legal recognition of diverse family forms, legal definitions of 'family' remain predominantly based around the idealised image of the traditional, heterosexual, nuclear family: the nexus of the conjugal relationship and the parent/child relationship.

⁹ e.g. Adoption and Children Act 2002, Civil Partnership Act 2004, Human Fertilisation and Embryology Act 2008, Marriage (Same Sex Couples) Act 2013 and Children and Families Act 2014.

¹⁰ See below at section 2.2, 'Law's Definition(s) of 'Family'', for a consideration of three such areas. ¹¹ [2001] 1 AC 27.

¹² ibid, per Lord Nicholls of Birkenhead, at 41.

2.1. 'Family' as a Social Concept

'Family' is central to our understanding and experience of human existence. As Bernardes states, '[m]ost people in Western industrialised societies, and probably most people worldwide, consider family living as the most important aspect of their lives.' However, in spite of this importance of our families in shaping our identity, it is apparent from the sociological literature that the term 'family' is not easily definable; as Coltrane observes, 'we can never be quite sure what family means unless we can understand the context in which it is used.' This reflects the evolving nature of the cultural understanding of family; Leeder has commented:

The family has been around since the beginning of humankind and clearly will exist in some form forever...The family in the world is in process: resilient the family copes with the forces acting on it and adapts in an ongoing manner that makes it a highly elastic and changeable form.¹⁶

The suggestion from the literature is that 'family' is not a fixed concept with a readily identifiable, simple definition. In this section, I will consider sociologists' attempts to understand the meaning of 'family' and conclude that the lack of a fixed meaning has resulted in the law lacking a readily available and identifiable social construction of 'family' to underpin its understanding.

Given the lack of a readily apparent and straightforward definition of 'family',¹⁷ some theorists have sought to understand and illuminate the meaning of family by reference to a variety of characteristics, features and connections which are said to be

¹³ Jon Bernardes, Family Studies: An Introduction, (Routledge, 1997), at 1.

¹⁴ Scott L. Coltrane, *Gender and Families*, (Rowman and Littlefield, 1998), at 5.

¹⁵ Diana Gittins, *The Family in Question: Changing Households and Familiar Ideologies*, (2nd edition, MacMillan, 1993), at 3, has observed that, 'an amalgam of discourses combine to create a dominant representation of what a family should be like. This representation changes over time, but nonetheless is presented as something universal.'

¹⁶ Elaine Leeder, *The Family in Global Perspective: A Gendered Journey*, (SAGE, 2003), at 2.

¹⁷ See e.g. Charles B. Hennon and Stephan M. Wilson, *Families in a Global Context*, (Routledge, 2008), for an exploration of the differences and similarities between families across different countries and cultures.

consistent across different family forms.¹⁸ Such theoretical understandings of the 'family' are often premised upon the emotional bonds between individuals, with these connections being articulated using a variety of terms; for example, intimacy,¹⁹ love²⁰ or trust.²¹ In this regard, Cheal suggests that 'a family is considered to be any group which consists of people in intimate relationships which are believed to endure over time and across generations.'²² Other writers have attempted to provide a slightly more detailed description of the shared characteristics of families. McKie and Callan, for example, identify three 'principles and ideas'²³ which families are constructed as sharing: 'values', 'memories' and 'spaces and places'.²⁴ These understandings of 'family' propose that it is the shared experiences and the quality of the relationships involved that are the significant, defining features of the 'family'.

Related to this emphasis on characteristics, experiences and relationships, some theorists have also attempted to understand the 'family' by focusing upon the social role performed by families and identifying the core functions of families.²⁵ Silva and Smart suggest that the 'family' is not underpinned by a specific structure or form, but instead, '[i]n this context of fluid and changing definitions of families, a basic core remains which refers to the sharing of resources, caring, responsibilities and

_

¹⁸ See e.g. David Morgan, *Family Connections: An Introduction to Family Studies*, (Polity Press, 1996), at 11, who suggests that the term 'family' should be employed, 'to refer to sets of practices which deal in some way with ideas of parenthood, kinship and marriage and the expectations and obligations which are associated with these practices.'

¹⁹ Diduck, 'What is Family Law For?', at 289, suggests that, '[w]hat makes a relationship familial to me then is not necessarily a biological, legal or conjugal connection, rather it is what people do in it, it is a relationship characterized by some degree of intimacy, interdependence, and care.'

²⁰ Herring, 'The Disability Critique of Care', equates 'love' and 'care', stating, at 2, '[c]are is the manifestation of that most basic moral value: love. It involves meeting the needs of others, which is a primary good.'

²¹ Trudy Covier Dilament of Taylor Covier Dilament

²¹ Trudy Govier, *Dilemmas of Trust* (McGill's-Queen's University Press, 1998), at 68, emphasises that, '[g]ood enough families are founded, not on heterosexuality and stereotypical gender roles, not on male providers, not on biological reproduction, but on trust between people who live together in a home, trust each other, and are committed to building a life together.'

²² David Cheal, *Sociology of Family Life*, (Palgrave, 2002), at 4.

²³ Linda McKie and Samantha Callan, *Understanding Families: A Global Introduction*, (SAGE, 2012), at 23.

²⁴ ibid, at 23-24. In addition, at 21-23, they identify five 'common characteristics' of families, these are 'a common identity', 'economic co-operation and ownership', 'reproduction of the next generation' 'care work and domestic labour' and 'co-residence'. However, it is suggested that these characteristics are illustrative of the problems involved in attempting to provide greater detail, beyond reference to 'values' or 'principles and ideals', because, it is submitted that these characteristics would exclude some family forms from this suggested understanding of 'the family'.

²⁵ The use of a 'function-based definition' of 'family' within the law will be considered below in section 2.2, 'The Law's 'Definition(s)' of 'Family''.

obligations. What a family is appears intrinsically related to what it does.'26 Moreover, there is a growing and developing body of work which seeks to ground a definition of 'family' within the literature on the moral and ethical significance of 'care'.²⁷ Herring provides a definition of 'family' that is premised upon the centrality of 'care', as '[p]eople providing each other with a substantial amount of care in a relationship marked by commitment.'²⁸ Young similarly suggests that 'family' should be understood, 'as people who live together and/or share resources necessary to the means for life and comfort; who are committed to caring for one another's physical and emotional needs to the best of their ability.'²⁹ These definitions combine references to the emotional connections within relationships with the identification of core characteristics that are understood as being central to all familial relationships. I observe that all of these different theoretical approaches are united by their attempts to define 'family' without emphasising specific family forms or valorising particular categories of relationship.

However, I argue that the various theoretical attempts to define the 'family', whether based on 'care', on identifying the social role of families, or on articulating shared familial characteristics and experiences, face similar problems, since all of these approaches result in an ambiguous understanding of the 'family' and do not provide a simple, easily understandable definition. I suggest that this is due to a the lack of consensus regarding which functions or characteristics are fundamental to all families,³⁰ and due to the vagueness which is seemingly inherent in definitions based

_

²⁶ Elizabeth B. Silva and Carol Smart, 'The 'New' Practices and Politics of Family Life' in Elizabeth B. Silva and Carol Smart (eds.), *The New Family?*, (Thousand Oaks, 1999), at 7.

²⁷ As discussed above in Chapter 1, subsection 1.2.A, 'Themes of Analysis'. See e.g. Herring, *Caring and the Law*, who suggests, at 11, '[c]aring is a most basic human need' and Held, *The Ethics of Care*, who states, at 15, '[a]n ethic of care focuses on attentiveness, trust, responsiveness to need, narrative nuance, and cultivating caring relations.'

²⁸ Herring, Caring and the Law, at 194.

²⁹ Iris Marion Young, *Intersecting Voices*, (Princeton University Press, 1997), at 106.

³⁰ While some of the central functions of a family may be apparent, I suggest that the boundaries of family functions are less clear. In this regard, compare the description given by Cheal, *Sociology of Family Life*, at 7, who comments, '[t]he things that family members do are easy to identify. They give and lend money, they get children ready to go to school, they prepare and share food, they have sex and express love in other ways, and so on', and that of McKie and Callan, *Understanding Families: A Global Introduction*, at 23, who refer to, '[f]urther characteristics and issues that come to mind when we think of families include childcare and working parents, solo childrearing, care for sick or older relatives, the legal and social implications of cohabitation, life after divorce for parents and children, and home-based care for the terminally ill.' These two descriptions illustrate the wide diversity of

upon 'care', experiences, or emotional connections. Noddings acknowledges this ambiguity in the context of care, observing that, '[m]ost people agree that the world would be a better place if we all cared more for one another, but despite that initial agreement we find it hard to say exactly what we mean by *caring*.'³¹ Nevertheless, this thesis should not be understood as suggesting that the ambiguity associated with understanding 'the family' in terms of 'care', common characteristics or core functions is necessarily any more normatively problematic than the 'nuclear family' model which I will discuss subsequently. Instead, my suggestion is that embracing such ambiguity does not provide a simple, unified definition of 'family' as a social concept and thus does little to further our understanding of the law's underlying model of family, which this thesis aims to elucidate.

From this overview of the literature, it is apparent that 'family' lacks a precise definition as a social concept.³² As Cheal observes, '[t]here is no single concept of the family which is true for all historical periods and in all places.'³³ Consequently the law does not have a simple, unified definition of 'the family' as a social concept to underpin its approach to defining and understanding family.³⁴

2.2. The Law's 'Definition(s)' of 'Family'

The lack of a single, unified definition of 'family' as a social concept is reflected within legal understanding; Diduck and Kaganas have observed that, '[t]here is no statutory definition of family, and there is really no common law definition either. How can law, which depends on certainty, cope with this lacuna?'³⁵

characteristics and functions that are potentially identifiable as being central to 'family' and therefore the difficulties of using these approaches to provide a precise, overarching, definition of 'family' as a social concept.

³¹ Noddings, *Starting at Home: Caring and Social Policy*, at 11. See further e.g. Herring, *Caring and the Law*, at 13, who similarly notes that '[p]roducing a definition of care is far from straightforward.'

³² See e.g. Carol Smart, *Personal Life: New Directions in Sociological Thinking*, (Polity Press, 2007), for an attempt to escape from the problems caused by the conceptual category of 'the family'.

³³ Cheal, *Sociology of Family Life*, at 4.

³⁴ O'Donovan, *Family Law Matters*, at 11, argues that, 'the lack of definition of the notion of family allows a subtext of values, such as those derived from patriarchy, to control.'

³⁵ Alison Diduck and Felicity Kaganas, *Family Law, Gender and the State*, (3rd edition, Hart, 2012), at 21.

Given this apparent lack of an overarching definition, how then does the law understand and define 'family'? In his textbook Family Law, ³⁶ Herring suggests five possible approaches the law could adopt: 1. 'The person in the street's definition', which he describes as being based on 'common usage' of the term 'family'. 37 2. 'An idealised definition', of which he states '[i]n our society many would see this as a married couple with children'. 38 3. 'A function-based definition', which 'examines the functions of families in our society' and determines family based upon whether relationships perform those functions.³⁹ 4. 'A formalistic definition', which is based upon 'whether the group of individuals in question has certain observable traits that can be objectively proved. 40 5. 'A self-definition approach', which 'would state "you are a family if you say you are". 41

The fact that these substantively different approaches are all intellectually sustainable illustrates the complexities of attempting to provide a legal definition of 'family'. Herring observes that '[t]he law...in defining families uses a combination of a formalist and function-based approach.'42 However, in the consideration of the law's attempts to define 'family' which follows, I will observe that in addition to these two approaches there has been significant reference to 'the person in the street's definition' of 'family'. I will argue that within this approach there is evidence of an 'idealised definition' of 'family', based upon the traditional, nuclear family, which has exerted significant influence on the law's definitions of 'family'. 43 Herring has observed that '[m]any people have a stereotypical image of what the "ideal family" is like - a mother, a father and two children.'44 Indeed, Bernardes describes this archetype of the 'nuclear family' in the following terms:

³⁶ Jonathan Herring, *Family Law*, (7th edition, Pearson Longman, 2015).

³⁷ ibid, at 3.

³⁸ ibid, at 4.

³⁹ ibid, at 3.

⁴⁰ ibid, at 3.

⁴¹ ibid, at 4.

⁴² ibid, at 6.

⁴³ Therefore, I suggest that of Herring's five approaches, only the 'self-definition approach' has not exerted influence on the legal attempts to define 'family'. Arguably, this reflects the difficulties of giving legislative effect to such a self-definition approach.

44 Herring, *Family Law*, at 2.

[A] common and popular image of "the nuclear family" portrays a young, similarly aged, white, married heterosexual couple with a small number of healthy children living in an adequate home. There is a clear division of responsibilities in which the male is primarily the full-time breadwinner and the female primarily the caregiver and perhaps a part-time or occasional income earner.⁴⁵

The nuclear family is an idealised image of family, which provides a normative vision of family life;⁴⁶ Muncie and Sapsford comment that, 'the idea of the nuclear family clearly retains a potency such that all other forms tend to be defined with reference to it.'⁴⁷ Boyd has previously claimed that the 'normal family' of law 'is heterosexual, nuclear, generally white and middle class, and usually involves a dependent role for the women who has more responsibility for home and childcare than the man, and who preferably remains outside the workforce.'⁴⁸ Thus, I will argue that the 'idealised definition' of 'family', within legal understanding, is premised upon the traditional, nuclear family, which comprises the nexus of the conjugal relationship and the parent/child relationship.⁴⁹

This section considers three different contexts in which there exists a legal definition of 'family': succession by 'family members' to private sector tenancies under the Rent Acts (subsection 2.2.A); the right to respect for 'family life' under Article 8 of the European Convention on Human Rights (subsection 2.2.B) and free movement for the 'family members' of EU citizens under Directive 2004/38/EC ('the

⁴⁵ Bernardes, Family Studies: An Introduction, at 2-3.

⁴⁶ Fiona Williams, *Rethinking Families*, (Calouste Gulbenkian Foundation, 2004), at 18, has observed, '[t]he nuclear family of the post-war world, with its male breadwinner, was a construction of what family life *should* look like.'

⁴⁷ John Muncie and Roger Sapsford, 'Issues in the Study of 'The Family" in John Muncie, et all (eds.), *Understanding the Family*, (2nd edition, SAGE, 1997), at 10, see further Bernardes, *Family Studies: An Introduction*, at 3, who similarly observes, 'the idea of the "nuclear family" is remarkably powerful'.

⁴⁸ Boyd, 'What is a 'Normal' Family? *C v C (A Minor) (Custody: Appeal)*', at 276. It is worth noting that this description was expressed over 20 years ago and it is suggested that there have been some shifts in this 'normal family' of law in the time since.

⁴⁹ See e.g. Carol Smart, Bren Neale and Amanda Wade, *The Changing Experience of Childhood: Families and Divorce*, (Polity, 2001), who observe, at 10, 'children have been fused with their parents into an idealised, inseparable family unit. In the process, the diverse identities and interests of individual family members have been concealed.'

Citizenship Directive') (subsection 2.2.C). Of these contexts, two (the Rent Acts and ECHR Art.8) leave family undefined, which has resulted in the definition being developed through judicial decisions, while the other (the EU Citizenship Directive) provide a formalistic definition of 'family', by specifying a fixed list of relationships. An examination, in subsequent sections, of how the case law under the Rent Acts, has considered and expanded the definition of 'family', will reveal consistent reference to Herring's 'person in the street's definition' of 'family'. 50 Within this case law and those decisions considering 'family life' under Art.8 ECHR, as well as the Citizenship Directive's definition of 'family member', I will contend that there is substantial influence of, in Herring's terms, an 'idealised definition' of 'family'. I will argue that this 'idealised definition' of 'family' is based around the traditional, heterosexual, nuclear family model; comprising the nexus of the conjugal relationship and the parent/child relationship. Moreover, in the following sections it will be shown how additional categories of relationship (unmarried cohabitants and same-sex couples) have been brought within these two judicial definitions of 'family' (under the Rent Act and Art.8 ECHR), on the basis that those relationships share sufficient similarities and functions with this idealised, nuclear family. This shows that even where, in Herring's terms, a 'function-based definition' of 'family' is utilised, the functions which are considered significant reflect the idealised image of the traditional nuclear family.

2.2.A. The Definition of 'Family' Under the Rent Acts

To begin, the foremost example within domestic law of the courts explicitly considering and interpreting the definition of 'family' at various points throughout the 20th century was under the Rent Acts.⁵¹ These Acts contained the statutory regime that governed the area of 'private sector' tenancies, prior to 1989.⁵² The most recent

-

⁵⁰ See e.g. *Brock v Wollams* [1949] 2 KB 388.

⁵¹ Rent Act 1977 (as amended by Housing Act 1988) being the most recent. Since the Increase of Rent and Mortgage Interest (Restrictions) Act 1920 s.12 (1) (g) some form of statutory protection was provided for a 'member of the tenant's family', under these Acts. See also Rent (Scotland) Act 1984 Sch. 1 Para 3.

⁵² Housing Act 1988 and Housing (Scotland) Act 1988 replaced the regime of 'protected tenancies' and 'statutory tenancies' under the Rent Acts with those of 'assured tenancies', which provide a much

legislation in England and Wales, the Rent Act 1977, provides that, 'a person who was a member of the original tenant's family'⁵³ will be 'entitled to an assured tenancy'⁵⁴ upon the original tenant's death;⁵⁵ this exists in combination with provision being made available for a 'surviving spouse'.⁵⁶ Crucially, there is no guidance within the Rent Acts themselves as to the meaning of 'family', which is not limited to specific categories of relationship. Wikeley has observed that, 'since the early days of the Rent Acts the concept of "family", as a means of delineating the framework for defining succession to statutory tenancies, has proved to be remarkably problematic.'⁵⁷ This lack of legislative clarity as to the precise definition of 'family' has necessarily led to significant judicial consideration of the definition since the original statutory provision was enacted.⁵⁸ As a consequence the definition of 'family', under the Rent Acts, has been developed through judicial decisions.

2.2.A.1. The Role of Judicial Interpretation

The judgments in the cases which consider the definition of 'family' under the Rent Acts invariably make clear that the discussion of the definition of 'family' is specific to this particular statutory context.⁵⁹ This approach is encapsulated by the statement of Lord Slynn in *Fitzpatrick v Sterling Housing Association Ltd.*⁶⁰ that, '[i]n other

_

more limited regime of statutory protection for tenants. The provisions of the Rent Acts will still apply to tenancies created prior to the commencement of these Acts.

⁵³ Rent Act 1977 Sch. 1 Para 3 (1), see also Rent (Scotland) Act 1984 Sch. 1 Para 3.

⁵⁴ ibid, this is an 'assured tenancy' subject to the new tenancy regime of Housing Act 1988. Prior to that Act, the surviving family member would have been entitled to succeed to the 'statutory tenancy', s.2.(1) (b) Rent Act 1977.

s.2 (1) (b) Rent Act 1977.

55 Subject to the requirement that they were, 'residing with him [in the dwelling-house] at the time of and for the [period of 2 years] immediately before his death'.

⁵⁶ Sch. 1 Para 2. The two major distinctions are that the 'spouse' does not require to have been resident for the 2-year period and that they are entitled to a 'statutory tenancy' as opposed to an 'assured tenancy', which is advantageous to the spouse because it continues under the more favourable Rent Act regime, see also Rent (Scotland) Act 1984 Sch. 1 Para 2.

⁵⁷ Nick Wikeley, 'Fitzpatrick v Sterling Housing Association Ltd.: Same-Sex Partners and Succession to Rent Act Tenancies' [1998] 10 (2) Child and Family Law Quarterly 191, at 191.

⁵⁸ An early example of this need for clarity is provided in *Price v Gould* [1930] All ER 389, in which it was held that brothers and sisters of the tenant were included within 'family'.

⁵⁹ It is notable that the limited cases considering 'family' under s.87 Housing Act 1985 (a statute concerning succession to public sector tenancies, which employs a formalistic definition, specifying a fixed list of relationships) also stress that the definition only applies to that specific context, see e.g. Ward LJ in *Sheffield City Council v Personal Representatives of Wall* [2011] 1 WLR 1342, at 1350. ⁶⁰ [2001] 1 AC 27.

statutes, in other contexts, the words may have a wider or narrower meaning.'⁶¹ Notably, this statutory entitlement for 'family members' to succeed to a private sector tenancy no longer exists for tenancies created after January 1989.⁶² However, in spite of these issues, I argue that the development throughout the cases of this definition of 'family' is potentially illustrative of the law's understanding of 'family' more generally, particularly given the paucity of consideration of this concept otherwise within the law.

Cretney and Reynolds have argued that, '[t]he interpretation of this expression...has caused the courts difficulty over the years, and the decisions were not easy to reconcile.' However, through the development of the definition of 'family' in these decisions, I will argue that it is possible to see repeated reference to, in Herring's terms, 'the person in the street's definition' of 'family'. I will also argue that this definition contains within it, in Herring's terms, an 'idealised definition' of 'family', based around the traditional, heterosexual, nuclear family; comprising the 'nexus' of the conjugal relationship and the parent/child relationship. I will further suggest that there is evidence within the judicial decisions of the usage of, in Herring's terms, a 'function-based definition' of 'family', which is based upon replicating the functions of this idealised image of the nuclear family, rather than upon reflecting an alternative understanding of 'family', premised upon the centrality of 'care', mutuality and interdependence.

2.2.A.2. 'Family' as Defined by the 'Ordinary Man' Test

From the initial reported cases it is apparent that 'family', in this context, is to be given its 'popular meaning' and determined on the basis of the understanding of the

-

⁶¹ ibid, at 32.

⁶² Although both Housing Act 1988 (s.17) and Housing (Scotland) Act 1988 (s.31) retain an entitlement to succession of the 'assured tenancy' for the 'surviving spouse', which is now defined in both Acts to include Civil Partners and both opposite-sex and same-sex cohabitants.

⁶³ Stephen M. Cretney and F.M.B Reynolds, 'Limits of the Judicial Function' (2000) 116 (Apr) Law Quarterly Review 181, at 182.

Gammans v Ekins [1950] 2 KB 328, at 331, per Asquith LJ, or as Evershed MR expressed it in Langdon v Horton [1951] 1 KB 666, at 669, quoting Shakespeare, 'the word "family" is used here, to borrow the words used of the soldier in King Henry V, in a sense "base, common, and popular".

'ordinary man', or as Herring has expressed it, 'the person in the street's definition' of family. In *Brock v Wollams*, ⁶⁵ Cohen LJ formulated the test as, '[w]ould an ordinary man, addressing his mind to the question whether Mrs. Wollams was a member of the family or not, have answered "yes" or "no"?' This test was thereafter consistently referred to and affirmed in subsequent cases. ⁶⁷ However, O'Donovan has pointed out that the 'ordinary man' test is problematic because it:

[D]oes not require a careful sampling of public opinion. Nor even is a jury polled for its opinion. Not only does it assume that the judiciary is in touch with public opinion, but it leaves open to the discretion of the judge the definition of popular morality, which may be confused with personal morality.⁶⁸

Reliance on 'the person in the street's' understanding of family has allowed the judiciary to obscure the influence of an 'idealised definition', based upon the nuclear family, because this definition is presented as 'natural' and 'common sense' through the usage of the 'ordinary man' test. Nevertheless, over the course of the second half of the 20th century, the 'common sense' definition was extended by the courts to cover a wider variety of relationships than under the original interpretation, as the common sense understanding of 'family' has been held to have changed over time. ⁶⁹ The types of relationships that have been held to be 'family', together with those relationships that have not, establish the boundaries of the courts' understanding of 'family', as well as identifying the characteristics which are viewed as fundamental to a familial relationship.

⁶⁵ [1949] 2 KB 388.

⁶⁶ ibid, at 395.

⁶⁷ See e.g. Lord Diplock in *Joram Developments Ltd. v Sharratt* [1979] 1 WLR 928, at 931, '[t]his test, which does no more than say that "family" where it is used in the Rent Acts is not a term of art but is used in its ordinary popular meaning, has been repeatedly referred to and applied in subsequent cases.'

⁶⁸ O'Donovan, *Family Law Matters*, at 35. Similarly, Diduck and Kaganas, *Family Law, Gender and the State*, at 27, comment, '[l]aw's normative vision of "the family" with marriage as the benchmark, is reproduced each time a court is called upon to decide whether a particular living arrangement is "familial" or not, but this vision is said to be that of the "ordinary public" rather than that of the court.' ⁶⁹ This will be considered below at subsection 2.2.A.4, 'Unmarried Cohabitants' and 2.2.A.5, 'Same-Sex Cohabitants'.

Initially, based on the understanding of the 'ordinary man' test, the definition of 'family' under the Rent Acts was interpreted narrowly by the courts. However, in spite of the repeated references to this 'ordinary man' test, there was some further judicial elaboration as to the scope of 'family' under this provision. The understanding of 'family' of these cases is summarised by Asquith LJ in Gammans v Ekins⁷⁰ where he stated, 'the material decisions limit the membership of the same "family" to three relationships: first, that of children; secondly, those constituted by way of legitimate marriage, like that between husband and wife; thirdly, relationships whereby one person becomes in loco parentis to another.'71 It should be noted that at this point in history only surviving 'widows' were specifically referred to as a separate category within the applicable statutory provision⁷² and thus surviving male spouses were considered 'family' at that time. 73 Therefore, this judicial statement seems to limit 'family' to that situation of 'widowers' and to parent/child relationships, 74 which is viewed as including de facto parent/child relationships. 75 I observe that, under the judicial interpretation of 'the person in the street's' understanding, the definition of 'family' was limited to the two relationships (the conjugal and the parent/child) which comprise the central nexus of the idealised nuclear family, identified above.

2.2.A.3. The Role of the 'De Facto Familial Nexus'

The limits of the 'ordinary man' definition of 'family' were explored by Russell LJ nearly 15 years after *Gammans*, in *Ross v Collins*, ⁷⁶ when he stated that:

It still requires, it seems to me, at least a broadly recognisable de facto

⁷

⁷⁰ [1950] 2 KB 328.

⁷¹ ibid, at 331.

⁷² s.12 (g) Increase of Rent and Mortgage Interest (Restrictions) Act 1920.

⁷³ As decided in *Salter v Lask* [1925] 1 KB 584. Surviving male spouses were not given equivalent protection to surviving female spouses until the definition was amended by s.76 (1) Housing Act 1980 and s.13 Tenant's Rights (Scotland) Act 1980.

⁷⁴ Jones v Trueman [1949] EGD 277, having held that 'illegitimate children' were to be included within the definition of 'family'.

⁷⁵ Brock v Wollams [1949] 2 KB 388, having held that if a child was adopted in fact, but not in law, this would be considered 'family'.

⁷⁶ [1964] 1 WLR 425.

familial nexus. This may be capable of being formed and recognised as such by the ordinary man where the link would be strictly familial had there been a marriage, or where the link is through adoption of a minor, de jure or de facto, or where the link is "step", or where the link is "in-law" or by marriage.⁷⁷

In Ross it was held that a non-sexual relationship between an elderly man and a much younger woman, who were unrelated by blood and who had lived together for around 22 years, ⁷⁸ could not be considered 'family' under this definition. This passage was quoted in full and affirmed by the House of Lords in Joram Developments Ltd. v Sharratt, 79 which concerned relatively similar facts to Ross, except that Joran Developments involved an elderly woman and a significantly younger man: they too were not considered 'family' under this definition. Lord Diplock was explicit in his affirmation of the Ross reasoning, stating, '[a]s for my reason for dismissing the instant appeal, I would not seek to improve upon what was said there by my noble and learned friend (then Russell LJ).'80 Taken together, these two cases seem to suggest that this 'broadly recognisable de facto familial nexus' is the boundary beyond which the judicial definition of 'family', based upon 'the person in the street's' understanding, will not go; given the refusal of the courts to apply the term 'family' to these two relatively similar factual situations. In Ross it is expressly stated that, 'two strangers cannot, it seems to me, ever establish artificially for the purposes of this section a familial nexus by acting as brothers or as sisters, even if they call each other such and consider their relationship to be tantamount to that.'81 In this way, the boundaries of the 'ordinary man' test, which represents the 'common sense' understanding of 'family', are set by this 'de facto familial nexus'. Indeed, I argue that the positioning of this boundary, by the decisions in Ross and Joran Developments, illustrates the intertwining, within judicial reasoning, of 'the person in the street's' understanding and the idealised image of the nuclear family.

7

⁷⁷ ibid, at 432.

⁷⁸ Miss Collins having originally sublet a room, before later becoming an unpaid 'housekeeper' in lieu of rent and subsequently acting as the tenant's primary carer after the death of his wife.

⁷⁹ [1979] 1 WLR 928, the House of Lords limited this judgment to its own specific facts, per Lord Diplock, at 930.

⁸⁰ ibid, at 931.

⁸¹ Ross v Collins [1964] 1 WLR 425, per Russell LJ, at 432.

Moreover, Russell LJ's statement in *Ross*, quoted above, suggests that this 'de facto familial nexus' seems to be premised upon the existence of a marriage and/or a parent/child or de facto parent/child relationship. In other words, the 'de facto familial nexus' reflects the traditional, nuclear family, comprising the conjugal relationship (originally limited to married couples) and the parent/child relationship (widely defined to include 'de facto' relationships). Thus, I argue that it is the presence of these particular relationships that is central to establishing the existence of the 'de facto familial nexus', rather than either a 'function-based definition' or a definition premised around examining the interdependence and 'care' involved in any individual relationship. Herring has recognised the primacy of conjugal relationships to the law's understanding of 'family', observing, '[c]are is at the centre of family life. Yet it is sexual relationships which have, for a long time, dominated family law and been regarded as the focus of the definition of a family and the marker for legal intervention.' 82

Thus, I submit that within the judicial consideration of 'family' under the Rent Acts, which is consistently asserted as being based upon the understanding of the 'person in the street', the 'de facto familial nexus' represented an 'idealised definition' of 'family' and this had as its basis the nuclear family. ⁸³ An examination of the cases which have extended the definition of 'family' under the Rent Acts, as well as some cases which have fallen outside the boundaries of the 'de facto familial nexus', illustrates more clearly the centrality of this idealised image of the nuclear family.

2.2.A.3.i. The Scope of the 'De Facto Familial Nexus'

The case of *Sefton Holdings v Cairns*⁸⁴ illustrates how strictly the boundary of the 'de facto familial nexus' and thus 'the person in the street's' understanding of family is interpreted. In this case a woman (then aged 23) had moved in with the family of a friend after the death of her boyfriend in World War Two (her own parents both

⁸² Herring, Caring and the Law, at 187.

84 [1988] 2 FLR 109.

⁸³ See e.g. the description, quoted above, given by Bernardes, *Family Studies: An Introduction*, at 2-3.

having predeceased her). Forty-five years later the daughter of the family died. The court determined that the defendant was not a member of the tenant's family, because the requisite 'de facto familial nexus' was not present in this case. Comparison was drawn with the factual circumstances in *Ross* and *Joram Developments*, with Lloyd LJ stating:

But the fact remains that when the defendant was taken in nearly 50 years ago she was taken in, to use the language of Russell L.J., as a stranger; and however long she may have lived with the family and however kindly they may have treated her, and however close their friendship may have become, the defendant did not, and in my judgment could not, have become a member of Ada's family.⁸⁵

This case makes clear that without the presence of either a conjugal relationship or a de facto parental relationship (as the defendant was an adult when she began residing in the household) the 'de facto familial nexus' could not be established and therefore there cannot be 'family' between adults who are unrelated by blood. The presence of these *relationships* appears to be the decisive criterion and the absence of such relationships removes the claim to the term 'family' under the Rent Acts' definition. This applies regardless of the mutuality and interdependence of the relationship itself, starkly illustrating the boundary of the 'de facto familial nexus' and the significance within this definition of the idealised image of the nuclear family.

2.2.A.3.ii. The Boundaries of the 'De Facto Familial Nexus'

The two contrasting decisions of *Jones v Whitehilt*⁸⁶ and *Langdon v Horton*⁸⁷ further demonstrate the extent to which the courts have interpreted 'family' as being based around the existence of this 'de facto familial nexus'. In *Jones* it was held that a niece who had lived with her aunt and uncle for around the last 2 years of their lives was part of their 'family', whereas in *Langdon* it was held that first cousins who had

⁸⁶ [1950] 2 KB 204.

⁸⁵ ibid, at 111.

⁸⁷ [1951] 1 KB 666.

lived together for around 30 years were not 'family', with Singleton LJ stating that, 'I do not see that it is possible to say that a cousin is a member of another cousin's family merely because she is a cousin.'88

While on first glance these cases appear to have somewhat contradictory results, I argue that they both support the idea of the 'de facto familial nexus' as consisting of the nexus of the conjugal relationship and the parent/child relationship. In *Jones* the court was able to view the relationship between a niece and her uncle and aunt as suitably analogous to a 'parent/child' relationship and thus capable of falling within the boundaries of the 'de facto familial nexus'. In his judgment, Evershed MR placed considerable significance on the fact that the defendant assumed, 'out of natural love and affection, the duties and offices peculiarly attributable to members of a family of going to live with her uncle and aunt to look after them in their declining years.'89 It is submitted that the court treated these 'duties and offices' as equivalent to those performed by children for their ageing parents and effectively constructed a de facto parent/child relationship on this basis. Thus, I observe that only once the existence of the relationship had been established, the 'caring' provided within that relationship then appeared to take on significance within judicial reasoning. In Langdon, on the other hand, the absence of this de facto parental relationship⁹⁰ meant that while the parties were related, they were not considered 'family' and therefore any interdependence or 'care' that existed within the relationship was not considered to be relevant. I suggest that this case was effectively treated as analogous with the factual circumstances in Ross, Joram Developments and Sefton Holdings, in spite of the existence of the extended familial relationship (that of first cousins) and therefore the relationship was considered to fall outside the boundary of the 'de facto familial nexus'.

I suggest that these two cases illustrate how the court's interpretation of this 'de facto familial nexus', within 'the person in the street's' understanding of family, reflects an

-

⁸⁸ ibid, at 672.

⁸⁹ Jones v Whitehill [1950] 2 KB 204, per Evershed MR, at 207.

⁹⁰ As well as the absence of a conjugal relationship, which is not applicable in these factual circumstances due to the blood relationship of first cousins.

'idealised definition' of family, based upon the traditional nuclear family. This is because those relationships which are viewed as capable of being located within the boundaries of this 'de facto familial nexus' are those which the courts consider to share sufficient similarities and characteristics with the archetypical nuclear family, either through the presence of a conjugal relationship⁹¹ or through the existence of a de facto parent/child relationship.

2.2.A.4. Unmarried Cohabitants

The evolution in the judicial treatment of unmarried cohabiting couples shows how the definition of 'family' has been extended, by reference to 'the person in the street's' understanding, while still relying on the 'de facto familial nexus', which itself is unaltered by the extension of 'family'. There has been a clear development since the decision in Gammans v Ekins, 92 where it was held that a childless unmarried couple who had lived together for a significant period of time were not 'family'. The court believed that it would 'be an abuse of the English language' to describe such a couple as a 'family', due to the fact that they were 'living in sin'.94 However, relatively soon after the Gammans decision the definition of 'family' was extended in *Hawes v Evenden*⁹⁵ to include the survivor of an unmarried couple of 12-13 years, who had 2 children. 96 The existence of a connected parent/child relationship, completing the central nexus of the nuclear family, seems to have been important in influencing the judicial view of the 'familial' character of the unmarried cohabitants, with Somervell LJ stating, 'in a case where the evidence justifies a finding that they all lived together as a family, then...I think that the mother is a member of the family'. 97 This statement suggests that the presence of children was

⁹¹ The role of conjugality in regulating adult, personal relationships will be explored further below in Chapter 4, section 4.2., 'The Centrality of Conjugality'.

⁹² [1950] 2 KB 328.

⁹³ ibid, per Asquith LJ, at 331.

⁹⁴ ibid, the moralistic determination of the judgment in this case is apparent, with Evershed MR going on to state, at 334, '[i]t may not be a bad thing that by this decision it is shown that, in the Christian society in which we live, one, at any rate, of the privileges which may be derived from marriage is not equally enjoyed by those who are living together as man and wife but who are not married.'

⁹⁵[1953] 1 WLR 1169.

⁹⁶ Or as she was described by Somervell LJ, ibid, at 1170, 'his mistress'.

⁹⁷ ibid, at 1171.

the decisive factor which led to the expansion of the definition of 'family' in this case. ⁹⁸ Indeed, the importance placed upon the parent/child relationship as foundational to the definition of 'family' in the initial decisions is illustrated by the statement of Bucknill LJ in *Brock v Wollams* ⁹⁹ that, '[t]he primary meaning of the word "family" therefore is "children." ¹⁰⁰

The position of unmarried cohabitants without children was reconsidered by the Court of Appeal around 25 years after these cases in *Dyson Holdings Ltd. v Fox.*¹⁰¹ As in *Gammans*, this case concerned an unmarried cohabiting couple who had no children and here they had lived together for around 40 years. In *Dyson Holdings*, however, the court determined that this cohabiting couple were 'family'. This was done on the basis that the operation of the 'ordinary man' test allowed for changes in social attitudes to be reflected in this definition of 'family'. James LJ stated that, '[t]he popular meaning given to the word "family" is not fixed once and for all time. I have no doubt that with the passage of years it has changed James LJ added: 'it is, I think, not putting it too high to say that between 1950 and 1975 there has been a complete revolution in society's attitude to unmarried partnerships of the kind under consideration. Although there was some initial judicial hesitance about the scope of this extension of the definition, to subsequently held to be 'family', until finally cohabiting couples were brought within the statutory

⁹⁸ Notably only 2 years earlier the court had decided that a male survivor of a cohabiting couple with a child was not 'family' in *Perry v Dembrowski* [1951] 2 KB 420. Although the judgment of Somervell LJ, at 422-423, left open the possibility that in different factual circumstances a surviving 'illegitimate father' may be capable of being considered 'family'.

⁹⁹ [1949] 2 KB 388.

ibid, at 349.

¹⁰¹ [1976] 2 QB 503.

Lord Denning MR went further in his judgment, ibid, stating at 509, 'it appears to me that Gammans v Ekins [1950] 2 KB 328 was wrongly decided.'

¹⁰³ ibid, at 511.

¹⁰⁴ ibid, at 512.

¹⁰⁵ Helby v Rafferty [1979] 1 WLR 13, where a cohabitation of 5 years was held to lack the permanence and stability', per Stamp LJ, at 18, required to be considered 'family'.

Watson v Lucas [1980] 1 WLR 1493, where one of the parties to a 19-year long cohabitation was, in fact, married to another woman throughout this relationship, the couple were still considered 'family'.

¹⁰⁷ Chios Property Investment Co Ltd. v Lopez (1988) 20 HLR 120 held that a relationship of only around 2 years duration was enough to be considered 'family'.

definition of 'spouse' for the purposes of the legislation at issue in these cases. ¹⁰⁸

None of these decisions subsequent to Dyson Holdings altered the basic 'de facto familial nexus' described above in Ross; nor did they shatter the nuclear family paradigm. This line of authority simply lessens the significance that had previously been attached to marriage. The extension of the definition of 'family' in these cases reflects, in Herring's terms, a 'function-based' approach to defining 'family', in that a clear analogy is drawn between marriage and those 'marriage-like' cohabiting relationships which had previously been refused recognition by the courts. 109 The extension is based upon the functions that those cohabiting relationships are seen to share with marriage, which remains understood as the paradigm conjugal relationship within the idealised image of the nuclear family. However, it appears to be the existence of a conjugal relationship that remains the decisive factor; thus the engagement with Herring's 'function-based definition' does not extend beyond the boundaries of the 'de facto familial nexus'. Consequently, those individuals in the factual circumstances exemplified by Ross v Collins 110 and Joram Development Ltd. v Sharratt¹¹¹ remain situated outside the boundaries of that 'de facto familial nexus' and therefore are not considered to be 'family'.

2.2.A.5. Same-Sex Cohabitants

In Fitzpatrick v Sterling Housing Association Ltd., 112 the House of Lords held that a same-sex couple who had cohabited for 18 years came within the definition of 'family'. 113 The decision emphasised the changing social attitudes that had occurred

¹⁰⁸ Housing Act 1988 s.39 (2) and Sch. 4 Para 1 (2), inserted Sch. 1 Para 2 (2) into the Rent Act 1977. 109 The legal approach towards cohabitation more generally will be considered below in Chapter 4,

subsection 4.2.A, 'The Legal Regulation of Cohabitation'. ¹¹⁰ [1964] 1 WLR 425.

¹¹¹ [1979] 1 WLR 928.

^{112 [2001] 1} AC 27. Prior to this decision, in a different statutory context (involving public sector tenancies under the Housing Act 1980) in Harrogate Borough Council v Simpson (1985) 17 HLR 205, it was held that the survivor of a same-sex female couple could not be described as having been 'living together as husband and wife' (one of the statutory definitions of 'family' under s.50 (3) (b) of the Housing Act 1980) with their deceased partner and therefore they could not be considered 'family' under that definition.

¹¹³ Prior to Sch. 8 Para 13 Civil Partnership Act 2004, which amended Rent Act 1977 Sch. 1 Para 2 to expressly include both surviving civil partners and surviving same-sex cohabitees alongside surviving

throughout the 20th century. 114 In his judgment, Lord Clyde stated that:

[T]he meaning of the word family in its sense of a group united by some tie or bond such as blood, marriage or personal affection may not have, as a matter of language, altered. What has changed are the precise personal associations to which the concept may now be applied.¹¹⁵

In other words, the 'de facto familial nexus' was still determining which relationships were considered 'family', but, in the same way as opposite sex cohabitants, same-sex relationships were now regarded as capable of forming part of that nexus, due to developments in 'the person in the street's' understanding of 'family'. This decision can be seen as a further extension of what is considered an acceptable form of conjugal relationship within the 'de facto familial nexus'. Moreover, Lord Slynn sought to describe some of the features or characteristics of a relationship that would give rise to it being considered as familial when he observed that, '[t]he hall marks of the relationship were essentially that there should be a degree of mutual interdependence, of the sharing of lives, of caring and love, of commitment and support.' In bringing same-sex conjugal relationships within the definition of 'family' on the basis that they share many of the core characteristics of marriage, which remains the 'gold standard' of conjugal relationships, the court seemed to have adopted, in Herring's terms, a 'function-based' approach to defining 'family'. Thus, the development of the definition of 'family' in *Fitzpatrick* shows that neither

spouses. This followed on from the House of Lords decision in *Ghaidan v Godin-Mendoza* [2004] 2 AC 557, decided prior to the 2004 Act coming into force; which held that due to s.3 Human Rights Act 1998, Rent Act Sch. 1 Para 2 (2) should be read in such a way as to include same-sex cohabiting couples under the definition of 'spouse'. As the distinction between same-sex and opposite-sex couples in this context was considered to be a violation of ECHR Art.14 in conjunction with Art.8.

However, see Ralph Sandland, 'Not 'Social Justice': The Housing Association, The Judges, The Tenant and His Lover' (2000) 8 (2) Feminist Legal Studies 227, who observes, at 227, 'it is a case with, to say the least, a mixed message.'

¹¹⁵ Fitzpatrick v Sterling Housing Association Ltd. [2001] 1 AC 27, at 50.

ilio ibid, at 39, Lord Slynn stated, 'I prefer to say that it is not the meaning which has changed but that those who are capable of falling within the words have changed.'

¹¹⁷ ibid, per Lord Slynn, at 38.

¹¹⁸ This will be considered further below in Chapter 4, subsection 4.1.B.1, 'Same-Sex Marriage and the Understanding of Marriage'.

¹¹⁹ See e.g. Glennon, 'Fitzpatrick v Sterling Housing Association Ltd. - An Endorsement of the Functional Family?'

marriage nor heterosexuality is a prerequisite for the existence of the 'de facto familial nexus'.

However, regardless of the judicial reference to 'mutual interdependence' and 'caring and love', I contend that the existence of a conjugal relationship remains the crucial factor in determining whether a relationship is considered 'family' under the Rent Acts. First, the examination of the development of the definition of 'family' throughout the case law, set out above, has established the centrality of the presence of the 'de facto familial nexus' to that definition and has illustrated how strictly the boundary of that 'nexus' is drawn. Secondly, in his judgment, Lord Clyde refers to 'the common bond in a partnership of two adult persons which may entitle the one to be in the common judgment of society a member of the other's family', ¹²¹ which reflects the emphasis on 'common sense' understandings and Herring's 'the person in the street's definition' of family. His lordship continues, by stating that:

It would be difficult to establish such a bond unless the couple were living together in the same house. It would also be difficult to establish it without an active sexual relationship between them or at least the potentiality of such a relationship. If they have or are caring for children whom they regard as their own they would make the family designation more immediately obvious, but the existence of children is not a necessary element. 122

I argue that it follows from this that any significance being granted to the mutuality and interdependence within relationships is secondary to those relationships being capable of being located within the central nexus of the idealised nuclear family. 123

Moreover, the importance of the traditional nuclear family was reaffirmed by Lord

¹²⁰ Or in different factual circumstances the existence of a 'de facto' parent/child relationship.

¹²¹ Fitzpatrick v Sterling Housing Association Ltd. [2001] 1 AC 27, at 51.

¹²² ibid.

ibid, at 44, Lord Nicholls states, '[w]here sexual partners are involved, whether heterosexual or homosexual, there is scope for the intimate mutual love and affection and long-term commitment that typically characterise the relationship of husband and wife.'

Nicholls in *Fitzpatrick*,¹²⁴ when he stated that, 'the paradigm family unit was, and still is, a husband and wife and their children.' As Glennon observes, 'the term "family" was not updated as such or given a different meaning to accommodate changing social parameters. Instead, acknowledgment of changed social conditions enlarged the category of person entitled to be included within the definition of the family'. I endorse this observation and submit that the nuclear family, based around a heterosexual marriage, remains the idealised image of 'family', within the judicial definition under the Rent Acts, but same-sex conjugal couples are being brought within the boundaries of this idealised image, in much the same way as unmarried heterosexual couples were in the *Dyson Holdings* decision.

2.2.B. The Definition of 'Family Life' under Article 8 of the European Convention on Human Rights

Another context where there has been significant judicial consideration of the definition of 'family' has been under Article 8 of the European Convention on Human Rights (ECHR), which provides that: 'Everyone has the right to respect for his private and family life, his home and his correspondence.' As with the Rent Acts, the term 'family life' is not defined in the convention and instead has been judicially interpreted over time by the European Court of Human Rights.

This understanding of 'family life' under Article 8 has been held to encompass various close familial relationships, including those between siblings, ¹²⁸ grandparents

¹²⁴ [2001] 1 AC 27.

¹²⁵ ibid, per Lord Nicholls, at 42. Furthermore, given how much it is relied on in the early cases, it is notable that his lordship criticises the usage of the 'ordinary man' test, stating, at 45, '[t]his oft-quoted test has tended to bedevil this area of law...Contrary to what seems implicit in this form of question, the expression family does not have a single, readily recognisable meaning.'

Glennon, 'Fitzpatrick v Sterling Housing Association Ltd. - An Endorsement of the Functional Family?', at 235.

¹²⁷ Article 8 (1) European Convention on Human Rights. For a comparison of the approaches of the European Court of Justice and the European Court of Human Rights, see Helen Stafford, 'Concepts of Family Under EU Law - Lessons from the ECHR' (2002) 16 (3) International Journal of Law Policy and the Family 410.

¹²⁸ Moustaquim v Belgium (1991) 13 EHRR 802, at 813, Para 36.

and grandchildren¹²⁹ and even uncle and nephew.¹³⁰ The meaning and scope of 'family life' under Article 8 has undergone a development across the case law which broadly parallels that visible in the Rent Acts cases described above; beginning with an understanding of 'family life' primarily based around marriage and the traditional nuclear family in the early cases, the Court has gradually moved toward a broader and more inclusive understanding in the later cases. The Court has noted that 'the notion of the "family" in this provision is not confined solely to marriage-based relationships and may encompass other *de facto* "family" ties where the parties are living together outside of marriage.' The 'illegitimate family' of mother and child, unmarried cohabitants with children, fathers who did not reside with their children, childless unmarried cohabitants, the relationship between a female to male transsexual and a child born to his female partner through artificial insemination of 'family life' under Article 8.

Despite these developments, 'de facto' relationships (those that are not registered formally with the state) do not *automatically* amount to 'family life', and the court has made clear that, 'a number of factors may be relevant, including whether the couple live together, the length of their relationship and whether they have demonstrated their commitment to each other by having children together or by any

¹²⁹ Marckx v Belgium (1979) 2 EHRR 330, at 348, Para 45, see also Vermeire v Belgium (1993) 15 EHRR 488.

¹³⁰ Boyle v UK (1995) 19 EHRR 179.

¹³¹ Keegan v Ireland (1994) 18 EHRR 342, at 360, Para 44, this principle has consistently been reaffirmed. Strikingly similar words were used in the more recent case of *Schalk and Kopf v Austria* (2011) 53 EHRR 20, at 701, Para 90.

¹³² Marckx v Belgium (1979) 2 EHRR 330, at 346, Para 40.

¹³³ Johnston v Ireland (1987) 9 EHRR 203.

First, when there had previously been a marriage between mother and father in *Berrehab v Netherlands* (1989) 11 EHRR 322 and subsequently in all circumstances e.g. *Boughameni v France* (1996) 22 EHRR 228.

¹³⁵ Keegan v Ireland (1994) 18 EHRR 342.

¹³⁶ X, Y and Z v UK (1997) 24 EHRR 143.

¹³⁷ Schalk and Kopf v Austria (2011) 53 EHRR 20 at 702, Para 94, '[i]n view of this evolution the Court considers it artificial to maintain the view that, in contrast to a different-sex couple, a same-sex couple cannot enjoy "family life" for the purposes of art.8.' It is notable that the court and the commission had previously rejected several opportunities to bring same-sex couples within the definition of 'family life', see e.g. Karner v Austria (2004) 38 EHRR 24, Mata Estevez v Spain [2001] ECHR 896 and Simpson v United Kingdom (1986) 47 D&R 274.

other means.' However, in this context the use of, in Herring's terms, a 'function-based definition' of 'family' can only act negatively to exclude some conjugal relationships, it cannot be employed to transform non-conjugal interdependent 'caring' relationships into 'family': this is similar to the approach taken under the Rent Acts described above. Nevertheless, one textbook on the ECHR has observed that '[f]amily life is now understood as extending beyond formal relationships and the family based on marriage... The Court has taken into account increasingly the substance and reality of relationships, acknowledging developments in social practices and the law in European states.'

Despite the trend of gradual extension, the Court has continually acknowledged, throughout this development, 'that protection of the family in the traditional sense is, in principle, a weighty and legitimate reason which might justify a difference in treatment.' The repeated restatement of this principle suggests that the idealised image of the traditional nuclear family continues to exert significant influence on the manner in which the definition of 'family life' has evolved under Art.8. As White and Ovey have observed, '[t]he core family within the case-law of the Strasburg Court is very much a man and a woman with children of whom they are the parents'; 143 in other words, the central nexus of the conjugal relationship and the parent/child relationship. Thus, in spite of the appearance within the ECHR jurisprudence of some elements of Herring's 'function-based definition' of 'family', I contend that in defining 'family life' within this jurisprudence there continues to be reliance upon an 'idealised definition' of 'family', based around the traditional, nuclear family.

¹³⁸ *X, Y and Z v UK*, at 166, Para 36.

¹³⁹ Throughout subsection 2.2.A, 'The Definition of 'Family' Under the Rent Acts', but particularly at subsection 2.2.A.4, 'Unmarried Cohabitants' and 2.2.A.5, 'Same-Sex Cohabitants'.

 ¹⁴⁰ D.J. Harris, M. O'Boyle, E.P. Bates and M. Buckley, *Law of the European Convention on Human Rights*, (Oxford University Press, 2009).
 ¹⁴¹ ibid, at 371.

¹⁴² Karner v Austria (2004) 38 EHRR 24, at 537, Para 40, this reflects the judgment 25 years previously in *Marckx*, where it was stated, at 346, Para 40, '[t]he Court recognises that support and encouragement of the traditional family is in itself legitimate or even praiseworthy.'

Robin C.A. White and Clare Ovey, *Jacobs, White and Ovey: The European Convention on Human Rights*, (5th edition, Oxford University Press, 2010), at 337.

2.2.C. The 'Formalistic' Definition of 'Family Member' in EU Law

European Union law provides for certain rights in relation to freedom of movement and residence.¹⁴⁴ Directive 2004/38/EC ('the Citizenship Directive') provides, in Article 1(a) for 'the right of free movement and residence within the territory of the Member States by Union citizens and their family members'.¹⁴⁵ Article 2 (2) of the directive contains a formalistic definition of 'family member' as follows: '(a) the spouse, (b) the partner with whom the Union citizen has contracted a registered partnership,¹⁴⁶ (c) the direct descendants who are under the age of 21 or are dependants,¹⁴⁷ (d) the dependent direct relatives in the ascending line'.¹⁴⁸ Therefore 'family member' under this definition is essentially restricted to registered conjugal relationships and parent/child relationships,¹⁴⁹ which reflects the central nexus of relationships that constitutes the nuclear family. It is notable that the relatively strict approach of this directive actually widened, if only very slightly, the original definition of 'family member' contained within the previous regulation governing freedom of movement for workers,¹⁵⁰ which did not include the category of 'registered partnerships'.¹⁵¹

The case of *Netherlands v Reed*, ¹⁵² which considered the previous definition, illustrates how rigidly the preceding provision was interpreted by the European Court of Justice (ECJ). In this case an unmarried cohabitant was not considered a 'family

_

¹⁴⁴ As a consequence of the result of the recent referendum on British membership of the European Union, these rights will no longer be part of UK law at some point in the future. However due to the uncertainty regarding the process and time scale for the British exit from the EU, these rights will remain a part of UK law for the foreseeable future.

¹⁴⁵ Directive 2004/38/EC, Article 1 (b) additionally provides, 'the right of permanent residence in the territory of the Member States for Union citizens and their family members'.

¹⁴⁶ Subject to the further requirement in Art.2 (2) (b) that the 'registered partnership' must be, 'on the basis of the legislation of a Member State, if the legislation of the host Member State treats registered partnerships as equivalent to marriage and in accordance with the conditions laid down in the relevant legislation of the host Member State'.

¹⁴⁷ Art.2 (2) (c) also specifies that this category includes, 'those of the spouse or partner as defined in point (b)'.

Art.2 (2) (d) specifies similar provision for 'those of the spouse or partner as defined in point (b)'. The meaning of 'dependency' was explored in *Jia v Migrationsverket* [2007] OB 545 (Case C-1/05).

Category (d) is dealing with a version of the parent/child relationship, where the elderly 'parent' has become a dependant of the 'child'.

¹⁵⁰ Regulation 1662/68/EEC, 'On Freedom of Movement for Workers within the Community'.

¹⁵¹ ibid, Article 10 (1).

¹⁵² [1987] 2 CMLR 448 (Case 59/85).

member' because such relationships are not specifically mentioned in the definition and the court stated, 'it must be held that the term "spouse" in Article 10 of the regulation refers to a marital relationship only. The formalistic approach of the ECJ to the term 'spouse', was also evident, in a different context, in *D* and Sweden *v* Council of the European Union, where it was held that a registered same-sex partnership could not be considered equivalent to a marriage, with the court stating clearly, 'that concept is distinct from marriage'. Gaffney-Rhys has noted that 'the European Court of Justice has adopted a very traditional interpretation of the terms spouse and marriage'. Given the ECJ's strict approach to defining 'spouse' and the fact that there is no specific provision for unmarried cohabitants within the definition, such relationships are not covered by this formalistic definition of 'family member'.

It should be noted, however, that the 'Citizenship Directive' does provide for the right of entry and residence for 'the partner with whom the Union citizen has a durable relationship', ¹⁵⁸ but crucially this is qualified on the basis that, '[t]he host Member State shall undertake an extensive examination of the personal circumstances'. ¹⁵⁹ The factual circumstances of a cohabiting couple will be examined to determine whether their relationship is 'durable' before the right is granted, whereas if they were deemed to be 'family members' the right would apply automatically, as a result of that status. Therefore, a 'function-based definition', which considers the 'durability' of relationships is only utilised as a means of excluding some non-marital conjugal relationships. Consequently, under the 'Citizenship Directive' any consideration of the characteristics or the mutuality and 'caring' within individual relationships is subordinate to the existence of relationships situated within the central nexus of the nuclear family. It is notable that

¹⁵³ ibid, Para 15, at 465.

¹⁵⁴ This is also shown by the case of *Diatta v Land Berlin* [1986] 2 CMLR 164 (Case 267/83), where a married couple were separated and expecting to divorce but were still considered 'spouses' because their marriage had not been formally dissolved.

¹⁵⁵ [2003] 3 CMLR 9 (Joined Cases C-122/99P and C-125/99P).

ibid, Para 31, at 260.

¹⁵⁷ Ruth Gaffney-Rhys, 'EU Directive on Freedom of Movement' (2006) International Family Law 65, at 66

¹⁵⁸ Directive 2004/38/EC Article 3 (2) (b).

¹⁵⁹ ibid.

this approach shares similarities with that taken in the interpretation of 'family life' under Art.8 ECHR described above. 160

On the basis of the definition of 'family member' prior to the 'Citizenship Directive', McGlynn suggested that, '[t]his "model European family" is a reproduction of the traditional "nuclear" family: that of the heterosexual married union, in which the husband is head of the family and principal breadwinner and the wife is the primary childcarer.' Thus, the extension of the definition of 'family member' in the 'Citizenship Directive' has been limited and the definition continues to be based around registered conjugal relationships and related parent/child relationships. We can see, therefore, that the nuclear family, comprising the nexus of the conjugal relationship and the parent/child relationship, remains the 'idealised definition' of 'family' that underpins the 'formalistic definition' of 'family member' in EU Law.

2.2.D. The Influence of the Nuclear Family Model within these Definitions of 'Family'

The development of the definitions of 'family' under the Rent Acts and 'family life' under Art.8 has followed a similar pattern. Both began with a fairly restrictive definition, based around traditional understandings of marriage and the nuclear family. This archetype of 'family' subsequently became an idealised image within these definitions. Thereafter, both approaches extended their understanding of 'family' gradually and ended up encompassing a much wider variety of relationships and family forms, seemingly through the use of elements of Herring's 'function-based' approach to defining the 'family'. However, in spite of this process of extension, within both approaches it is possible to identify a hierarchy of family forms, as Herring observes:

1

¹⁶⁰ See above at subsection 2.2.B, 'The Definition of 'Family Life' under Article 8 of the European Convention on Human Rights'.

Clare McGlynn, 'The Europeanisation of Family Law' [2001] 13 (1) Child and Family Law Quarterly 35, at 47, see also e.g. Lorna Woods, 'Family Rights in the EU - Disadvantaging the Disadvantaged?' [1999] 11 (1) Child and Family Law Quarterly 17.

¹⁶² See e.g. the decisions regarding the Rent Acts in *Brock v Wollams* [1949] 2 KB 388 and *Gammans v Ekins* [1950] 2 KB 328.

Despite these developments recognising a variety of family forms it can be argued that there is a hierarchy of families in family law: with the top position being taken by married couples, with civil partners, then unmarried heterosexual couples and then unpartnered same-sex couples below them. Certainly the closer the relationship is to the "ideal" of marriage the more likely it is to be recognised as a family. 163

In the absence of a precise or detailed definition of 'family' in either the Rent Acts or ECHR Art.8, the courts have turned to the 'ordinary' understanding of the 'person in the street', represented through the 'de facto familial nexus' 164 or 'de facto family ties', 165 to fill the gap. These definitions reflect an idealised image of the traditional, heterosexual, nuclear family. The approach taken to 'family member' under the Citizenship Directive provides further evidence that it is the existence of the central nexus of the conjugal relationship and the parent/child relationship that is the decisive factor in granting the label of 'family' within the law. The influence of this idealised image is illustrated by the way that definitions of 'family' and 'family life' have been expanded, by the judiciary, to include relationships which share enough similarities to, and therefore can be situated within the boundaries of, the nuclear family model. Thus, I suggest that Herring's 'function-based definition' of 'family', as well as any alternative approach to defining 'family' premised upon caring, mutuality and interdependence, are subordinate to the idealised image of the nuclear family. Thus, the consideration of functions only occurs for conjugal relationships and consequently, the 'care' and interdependence of relationships cannot transform those relationships outside the central nexus into 'family' under these legal definitions.

Herring, Family Law, at 6.
 See e.g. Ross v Collins [1964] 1 WLR 425.

¹⁶⁵ See e.g. Schalk and Kopf v Austria (2011) 53 EHRR 20, at 701, Para 90.

Conclusion

This chapter considered how 'family' is understood and defined within the law. The chapter began by exploring 'family' as a social and cultural concept, observing from the relevant literature that there is no consensus as to its meaning and no single, universally applicable definition of 'family'. I observed that this lack of overarching definition is reflected in the law. However, in this chapter I identified three specific legal contexts in which 'family' is defined, which spanned both domestic and European law. Within these definitions, I noted two contrasting approaches: a formalistic approach, which limits 'family' to a list of particular relationships and excludes all other relationships from the definition; and an alternative approach, which provides no specific guidance as to the extent of 'family' and instead allows the definition to be interpreted and developed judicially.

I then argued that an idealised image of the 'family' underpins these judicial definitions, irrespective of the approach taken to interpretation. This is the image of the traditional 'nuclear family', comprising the nexus of the conjugal relationship and the parent/child relationship. I argued that this ideal type of family has exerted significant influence on the expansion of the definition of 'family' in these different legal contexts. This has resulted in those relationships which are viewed as sufficiently resembling the archetypical nuclear family being brought within the parameters of the legal definitions of 'family'. Consequently, relationships possessing 'marriage-like conjugality' (same-sex and opposite-sex cohabitants) are now considered 'family'. Overall, in this chapter, I have aimed to illustrate the continuing influence of the traditional nuclear family, characterised by the central nexus of the conjugal relationship and the parent/child relationship, upon the legal understanding of the 'family'.

Chapter 3: The Historical and Philosophical Underpinnings of the 'Nuclear Family' Model

Introduction

In Chapter 2, I argued that the legal understanding of the 'family' was underpinned by an idealised image of 'family': 1 the traditional nuclear family, comprising the nexus of the conjugal relationship and the parent/child relationship. This chapter will explore the reasons for the nuclear family being positioned as the 'natural' or 'normal' family of law. It will consider both the nature of historical family forms and the values which underpin western society and the legal system: the divide between the public and private spheres of society and the orthodox construction of the rational and autonomous legal subject. I will argue that the nuclear family model's significance inevitably arises from the intersection of these values and the historical prevalence of the nuclear family form.

In section 3.1, I will describe how the nuclear family has been a recurrent and dominant form of family throughout history, and how the image of the nuclear family aligns with the law's image of the individual - the 'legal subject'. Relatedly, I will argue that the delineation of the male and female gender roles within the 'public/private divide' has contributed to the gendered roles (the man as 'breadwinner' and the woman as 'homemaker') embodied within the traditional nuclear family. In section 3.2, I will consider some of the theoretical critique of the orthodox construction of the legal subject that questions its supposed objectivity and self-evidence, as well as relying upon the feminist literature which examines the nature of its underpinning values. I will use this critique to illustrate some of the issues caused by the application of this orthodox construction of the legal subject. In section 3.3, I will argue that the combination of the historical recurrence and dominance of the nuclear family form, the influence of the gendered roles of the public/private divide and the values of the orthodox legal subject, have resulted in

¹ See Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family''.

the idealised image of the nuclear family being positioned and understood as the 'natural' and 'common-sense' understandings of 'family', which has in turn influenced the *legal* understanding of 'family' as described in Chapter 2.

3.1. The 'Family' and the 'Legal Subject'

In this section, I will explore how the idealised image of the nuclear family has come to exert significant gravitational force upon the law by considering the recurrence of the nuclear family form throughout history and examining the values and ideals which underpin this archetype of 'family'.

This section will consider the orthodox construction of the 'legal subject'.² The characteristics that are encapsulated within the 'person' of law determine how individuals are understood by the law, and I will argue that this construction exerts significant influence upon the legal understanding of the 'family' composed of such individuals. As Naffine and Owens observe, '[t]he legal person, or legal subject, plays an absolutely critical role in law. The attributes accorded by law to its subject serve to justify and rationalise law's very forms and practices.'³ The orthodox understanding is that law employs a neutral and objective approach to its subject.⁴ However this claim of impartiality obscures the fact that law has in mind a particular type of individual with particular characteristics when it constructs the legal subject. I will argue that the values and ideals privileged by law as it conceptualises and defines its subject support the dominant understanding of the nuclear family and its

² Richard Tur, 'The 'Person' in Law' in Arthur Peacocke and Grant Gillett (eds.), *Persons and Personality in Contemporary Inquiry*, (Basil Blackwell, 1987) suggests at 123, that, '[t]here is no general law of persons, but rather, a series of rules concerning relationships and liabilities.' Therefore, it is argued that the 'legal subject' is not necessarily a simple, unitary ideal, but rather that there exist various constructions of the subject throughout the law, employed in different areas and contexts. However, these 'subjects' share many similarities and are representative of the same underlying values and ideals, which are reinforced throughout law. Consequently for the purposes of this chapter the generic term 'legal subject' will be used.

³ Ngaire Naffine and Rosemary J. Owens, 'Introduction: Sexing Law' in Naffine and Owens (eds.), *Sexing the Subject of Law*, at 7.

⁴ For examples of the orthodox construction of the subject, see Bryant Smith, 'Legal Personality' (1928) 37 (3) Yale Law Journal 283, D.P. Derham, 'Theories of Legal Personality' in Leicester Webb (ed.), *Legal Personality and Political Pluralism*, (Melbourne University Press, 1958) and Peter Cane, *Responsibility in Law and Morality*, (Hart, 2002).

positioning as the common-sense and 'natural' image of family within the law's definitions of 'family'.

To begin, I will trace the historical conceptions of 'family', from classical societies through the Middle Ages and into the Enlightenment, observing that the nuclear family form is evident across these different historical contexts (subsection 3.1.A). I will explore how the recurrence of this idealised image throughout history reflects the persisting notion of a division between the public and private spheres of society (subsection 3.1.B). I will then consider how these historical family forms are influenced by the dominant, orthodox construction of the legal subject (subsection 3.1.C).

3.1.A. 'Family' and the Subject in Historical Societies

3.1.A.1. The Classical Societies

The roots of the philosophical, political and legal systems of western society are found in the ideas and systems of Ancient Greece and Rome. Consequently, the family forms and structures, as well as the cultural values of those societies, possess a continuing relevance upon our understandings of the traditional and the 'natural'. On that basis, regarding the 'family' in Ancient Greece, the historian Lacey states that:

The smallest unit of the state is the family, the oikos, which is comprised of the three elements, the male, the female and the servant...Male and female have a natural instinct to procreate themselves successors says Aristotle...and this introduces into the family a fourth element, the children.⁵

This basic family structure is not radically different from contemporary constructions, with its principal nuclear core. However, it is noteworthy that the

⁵ W.K. Lacey, *The Family in Classical Greece*, (The Author, 1980), at 15.

central familial unit of Greek society was understood to include servants. The Greek conception of family ('the oikos') may more accurately be viewed as having encompassed the entirety of those living within a household. Furthermore, regarding the conception of the relationship between husbands and wives, Harrison observed, '[t]here can be no doubt that a woman remained under some sort of tutelage during the whole of her life. She could not enter into any but the most trifling contract, she could not engage her own hand in marriage, and, she could not plead her own case in court.'6 Historians have observed that the legal status of men and women was substantively different in Ancient Greece, 7 it was the case that 'only adult male citizens could exercise the privileges of membership in the community to their full extent.'8 It is apparent that this was a society which was characterised by male dominance of public office and affairs. Consequently, this was a society which employed a rigid division between the 'public' sphere (for men) and the 'private' sphere (for women). Indeed, it was engagement with the public sphere, which was limited to men, that was viewed as giving rise to the rights and duties associated with being a full legal subject. On this basis, Dickenson has argued that the nature of the division between the genders in Ancient Greece went further than splitting society into separate public and private realms for men and women respectively, commenting, '[i]n Aristotle women's absence from the public sphere is mirrored by their subordination in the private realm...I think Aristotle's male is the "symbol" of norm in both private and public arenas. In both realms women are less than full subjects.'10 Thus, in Ancient Greece the female role was characterised by a dual subordination in both the public and private spheres.

⁶ A.W. Harrison, *The Law of Athens: The Family and Property*, (Oxford University Press, 1968), at 108

⁷ Much of the material written about the family of Ancient Greece primarily concerns the Athenian perspective (due to the extensive writings of Aristotle). While there would have been unifying themes and ideals across the city states, there would also evidently have been specific differences. While Lacey, *The Family in Classical Greece*, focuses primarily on Athens, she considers the significant differences of the Spartan model, at 194-208.

⁸ Raphael Sealey, *The Justice of the Greeks*, (University of Michigan Press, 1994), at 133.

⁹ The presence of slavery was a fundamental feature of Ancient Greek society and therefore the Greeks had little difficulty with the notion that people could be considered as non-subjects or as mere property. Harrison, *The Law of Athens: The Family and Property*, states at 163, '[f]rom many points of view slaves in fourth century Athens were chattels. Aristotle describes a slave as "a live possession" or "a live tool".

¹⁰ Donna Dickenson, *Property, Women and Politics: Subjects or Objects?*, (Polity Press, 1997), at 46.

The Roman conception of the 'family' followed a similar basic structure to that of the Greeks, which the historian Rawson details as follows:

The nuclear family was small, but what the Romans meant as *familia* could be much larger. The Roman *familia* consisted of the conjugal family plus dependants (i.e. a man, his wife, and their unmarried children), together with the slaves and sometimes freedmen and foster-children who lived in the same household.¹¹

It is apparent that this Roman *familia* had a central core, based around the nuclear family, which resembles the 21st century understandings of 'family'. However, like the Greek 'oikos', the Roman *familia* represented a family structure which was wider in scope than our 21st century conception including, as it did, all dependants, freedmen and slaves.¹²

This Roman *familia* was underpinned by the concept of the *pater familias*, who exercised effective control over the behaviour and property of all other members of that family; Du Plessis describes the *pater familias* as, 'the eldest male ancestor of a specific family. He had in his power (potestas) all descendants traced through the male line. The paterfamilias was *sui inuri*, i.e. legally independent - he could not be in anyone else's power.' This legal framework, based around the almost absolute power of the (male) *pater familias*, determined the relationship between husbands and wives in Roman society. The 19th century scholar De Colquhoun stated: '[t]he Old Roman law conferred on the husband an almost absolute power over the wife by the *conventio in manum*; all she acquired vested in him, and he acquired the same rights over her person and property as if she had been his natural daughter.' Women were therefore legally subordinate to both their husbands and (as with all other

1

¹¹ Beryl Rawson, 'The Roman Family' in Beryl Rawson (ed.), *The Family in Ancient Rome: New Perspectives*, (Croom Helm, 1986), at 7.

¹² Similar to Ancient Greece, Roman society was premised on the existence of widespread slavery, Gaius in his 'Institutes of Gaius', Francis de Zulueta (ed.), (Clarendon Press, 1946-53), wrote, '[c]ertainly, the great divide in the law of persons is this: all men are either free men or slaves', (D.1.5.2) (cf. Inst.Gai. 1. 9).

¹³ Paul du Plessis, *Borkowski's Textbook on Roman Law*, (Oxford University Press, 2010), at 110.

¹⁴ Patrick Mac Chombaich De Colquhoun, *A Summary of the Roman Civil Law*, Volume 1, (V. and R. Stevens and Sons, 1849), at 501.

dependants) to their *pater familias*. The structuring of the Roman *familia* around the authority of the *pater familias* acted as a clear limit on women's participation in the public sphere, so that, as Frier and McGinn describe, '[i]n the early Roman Empire...women were citizens but were nonetheless legally barred from voting, holding magistracies, serving as jurors, and generally performing what were thought of as public duties.' This limited public role combined with the ultimate authority of the *pater familias* in the private realm reflects the dual subordination of women described above by Dickenson in relation to Ancient Greek society and this illustrates the gendered nature of legal subjecthood in the classical societies. While the political and legal systems in the UK emerged later, independently of the Greek and Roman empires and thus reflected different cultural norms, it is contended that the notion of delineated gendered roles remained a constant feature of Western philosophical and social understanding. Indeed, this distinction continues to be reflected in the understanding of the 'natural' gender roles within the nuclear family. In the nuclear family.

3.1.A.2. The Medieval and Renaissance Periods

It has been observed by family historians¹⁷ that by the end of the medieval period the nuclear family form had come to possess an even more central role in English society¹⁸ than it had enjoyed in the classical societies.¹⁹ The historian Houlbrooke

¹⁵ Bruce W. Frier and Thomas A.J. McGinn, *A Casebook on Roman Family Law*, (Oxford University Press, 2004), at 453.

¹⁶ The significance of these gendered roles within the legal understanding of the parental role will be considered below in Chapter 6, section 6.2, 'The Gendered Parenting Roles of the Nuclear Family'.

¹⁷ For example, see the works of anthropologist and historian Alan MacFarlane, *The Origins of English Individualism: The Family, Property and Social Transition*, (Blackwell, 1978) and *Marriage and Love in England: Modes of Reproduction*, 1300-1840, (Blackwell, 1986).

¹⁸ Some historians subscribe to an alternative narrative of far greater change in family forms, which views the dominance of the nuclear family form as occurring more recently and describes the 'family' of the medieval period as being based on a wider kinship structure. A leading example of this approach is presented by Lawrence Stone in *The Family, Sex and Marriage in England 1500-1800*, (Penguin, 1979). Although it is worth noting that Stone's work has been the subject of significant academic criticism, e.g. Alan MacFarlane, 'Review of Stone' (1979) 18 (1) History and Theory 103, where he stated, at 106 '[h]is massive effort to fit the material into an inadequate scheme provides a compendium of the distortions produced when a tenacious but false paradigm blinds the historian.'

¹⁹ However, the writing of notable 17th century philosopher Thomas Hobbes reflected the Greek and Roman construction and understanding of family. In his seminal work, *Leviathan*, (First published 1651, reprinted by Cambridge University Press, 1991), at 172, he described a family as consisting of,

states that '[b]etween the fifteenth and eighteenth centuries there was little change in familial forms and functions. The nuclear family was dominant and wider ties of kinship were relatively weak.'20 The significance of the nuclear family form is observed to have continued throughout the Renaissance, and Houlbrooke observes that, 'the momentous developments of this period, though certainly affecting family life, brought no fundamental change in familial forms, functions and ideals.'21 Thus, the nuclear family continued to represent the dominant form of family, in England, in spite of significant cultural and societal changes.

The division between the public and private spheres of society and the associated view of women as lacking full legal subjecthood is also evident throughout medieval history, as Perkin has argued, '[t]he subjection of women was enshrined in English law and custom for nine hundred years. Common Law reflected rather than caused that subjection.'22 The division gained legal expression in England through the common law doctrine of coverture, which was summarised by the writer Blackstone as follows: '[b]y marriage, the husband and wife are one person in law: that is, the very being or legal existence of the woman is suspended during the marriage, or at least is incorporated and consolidated into that of the husband'. 23 This gendered division dominated throughout Christendom in the medieval period and it also subsequently survived the cultural upheaval of the Renaissance; as the early 20th century historian Goodsell observed, '[i]f the social position of the more favoured women was raised during the Renaissance, if they were educated and held in higher esteem, it yet remains true that little advancement was made in freeing them from the financial and legal disabilities of the Middle Ages.'24 Therefore, both the nuclear family form and a gendered division of social roles were evident throughout the medieval and renaissance periods.

²⁰ Ralph A. Houlbrooke, *The English Family: 1450-1700*, (Longman, 1984), at 253.

^{&#}x27;a man and his children, or of a man and his servants; or of a man, and his children, and servants together: wherein the Father or Master is the sovereign.'

²¹ ibid, at 16, see further Richard Grassby, Kinship and Capitalism: Marriage, Family and Business in the English-Speaking World, 1580-1740, (Cambridge University Press, 2001), who notes, at 2, '[f]amily historians have emphasized the continuity of communal forms, that the nuclear family had a long history and co-existed with kinship'.

²² Joan Perkin, Women and Marriage in Nineteenth-Century England, (Routledge, 1988), at 1.

²³ Sir William Blackstone, Commentaries on the Laws of England, Volume 1, (4th edition, J. Exshaw,

²⁴ Willystine Goodsell, *A History of Marriage and the Family*, (2nd edition, MacMillan, 1934), at 257.

3.1.B. The Development of the 'Public/Private' Divide

While the notion of a division between the public and private spheres of society had existed throughout history,²⁵ and persisted throughout the supposedly 'transformative' developments that took place during the Enlightenment period;²⁶ Bottomley and Bronitt have noted that, '[f]rom the earliest liberal philosophers there came the belief that social life was to be seen as divided into public and private spheres. The private, or domestic, sphere was an area that should be none of "law's business" where "the King's writ did not run".'²⁷ This divide was adopted as a key tenet of liberal philosophy, with John Stuart Mill writing in *On Liberty*, that, '[i]n England...there is considerable jealousy of direct interference, by the legislative or the executive power, with private conduct'.²⁸ Therefore, this particularly liberal understanding of the 'public/private' divide has been a crucial feature in the subsequent development of western society.²⁹ O'Donovan describes this liberal construction of the division between public and private spheres as referring:

[T]o two distinct social realms constituted within liberal social philosophy divided from one another by legal regulation. The public realm is presented as that of state, market and politics, and is the world of men; the private realm, associated primarily with women, is

²⁵ The role of the liberal understanding of the public/private divide in shaping the nuclear family will be discussed further throughout the remaining sections of Chapter 3.

²⁶ Peter Gay, *The Enlightenment: An Interpretation - The Rise of Modern Paganism*, (W.W. Norton, 1995), states at 3, '[t]he men of the Enlightenment united on a vastly ambitious program, a program of secularism, humanity, cosmopolitanism, and freedom, above all, freedom in its many forms - freedom from arbitrary power, freedom of speech, freedom of trade, freedom to realise one's talents, freedom of aesthetic response, freedom in a world of moral men to make his own way in the world.'

²⁷ Stephen Bottomley and Simon Bronitt, *Law in Context*, (4th edition, Federation Press, 2012), at 8. ²⁸ John Stuart Mill, *On Liberty and Other Writings*, Stefan Collins (ed.), (Cambridge University Press, 1989), at 12.

²⁹ Susan Boyd, 'Challenging the Public/Private Divide: An Overview' in Boyd (ed.), *Challenging the Public/Private Divide: Feminism, Law and Public Policy*, at 8, states, '[t]his divide, which has long informed dominant Western ways of knowing and being, denotes the ideological division of life into apparently opposing spheres of public and private activities, and public and private responsibilities. It has prevailed particularly over the past two centuries, as with the industrialisation of Western capitalist societies, people's lives were increasingly divided into public and private spheres at both material and ideological level.'

the world of family. The values prevalent in the former are those of individualism; in the latter self-sacrifice and altruism are idealised.³⁰

The onset of the enlightenment was heralded by some as the 'age of reason'³¹ and it concerned itself with the freedom and liberty of man³² but it did little, at least initially, for the legal or social position of women.³³ In England and Wales, the historical doctrine of coverture was not reformed until the various Married Women's Property Acts of the late 19th century.³⁴ These Acts granted women the capacity to exercise legal personhood separate from that of their husbands, allowing them to own and control their own property.³⁵ However, even this reform did not alter the legal reliance on the 'public/private' divide nor did it reduce its social power and it was not until further social and legal developments during the 20th century³⁶ that women began to be seen as having social roles outside of the domestic context.³⁷

I suggest that the allocation of gender roles within the idealised image of the nuclear family is premised upon the ideals derived from the particularly liberal understanding

_

³⁰ O'Donovan, Family Law Matters, at 23.

³¹ Ernst Cassirer, *The Philosophy of the Enlightenment*, (Beacon Press, 1955, translated from the original German edition, 1932), states at 6, '[t]he eighteenth century is imbued with a belief in the unity and immutability of reason. Reason is the same for all thinking subjects, all nations, all epochs and all cultures.'

³² See e.g. the much quoted words of Jean-Jacques Rousseau, '[m]an is born free, and everywhere he is in chains. Those who think themselves the masters of others are indeed greater slaves than they.' In *Of the Social Contract, Or Principles of Political Right*, (First published 1762, reprinted by Dent, 1973).

³³ The historical lack of legal subjecthood granted to women is considered below at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'.

³⁴ The Married Women's Property Acts 1870, 1882, 1884 and 1893. See also the Married Women's Property (Scotland) Acts 1881 and 1920.

³⁵ However, Albie Sachs and Joan Hoff Wilson, *Sexism and the Law: A Study of Male Beliefs and Legal Bias in Britain and the United States*, (Martin Robertson, 1978), at 137, suggest that, '[t]he Married Women's Property Act 1882, is often held out as a milestone in the march of women to equality, but in reality it did little more than save wealthy women from the irksome restraints of holding property through trustees.'

³⁶ See e.g. the extension of the franchise, through the Representation of the People Act 1918 (granted the right to vote to some women over 30) and Representation of the People (Equal Franchise) Act 1928 (granted women the same voting rights as men), the Guardianship of Infants Act 1925, which introduced the 'welfare principle' and therefore removed the power of legal guardianship from the father and the reform of Divorce (Divorce Reform Act 1969 and Divorce (Scotland) Act 1976).

³⁷ Moller Okin, *Women in Western Political Thought*, at 290, observes, '[a]s most present-day feminists agree, the political emancipation of women brought with it very little of substance with respect to their economic and social position, and their actual life experience.' The contemporary understanding of the roles of men and women within the family and society is considered further below at in Chapter 6, section 6.2, 'The Gendered Parenting Roles of the Nuclear Family'.

of the 'public/private' divide.³⁸ O'Donovan has noted that, '[t]he division of labour whereby one spouse works for earnings and the other for love encapsulates the public/private split.'³⁹ This particular division of labour is reflected in the delineation between 'breadwinner' and 'homemaker' within the traditional nuclear family model.⁴⁰ Moller Okin expands upon this, commenting that:

[T]he existence of a distinct sphere of private, family life, separated off from the realm of public life, leads to the exaggeration of women's biological differences from men, to the perception of women as primarily suited to fulfil special "female" functions within the home, and consequently to the justification of the monopoly by men of the whole outside world.⁴¹

The separation of gender roles within the archetype of the traditional nuclear family reflects the historical significance of the 'public/private' divide, because as Boyd observes '[t]he public/private divide is intrinsically connected to the familial ideology that dominates capitalist societies.'42

3.1.B.1. The 'Public/Private Divide' within the Law

Historically, law has embraced the idea of a public/private divide, distinguishing between areas that were thought to be appropriate for legal regulation ('the public') and those that were not ('the private'), as O'Donovan observes:

This division is not confined to distinguishing relations between individual and state from relations between individuals. It also draws a line dividing the law's business from what is called private.

See the description of the nuclear family above in, Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family''.

³⁸ This connection will be explored further below at subsection 3.3.A, 'The Legal Subject and the Nuclear Family'.

³⁹ O'Donovan, Sexual Divisions in Law, at 9.

⁴¹ Moller Okin, Women in Western Political Thought, at 275.

⁴² Boyd, 'Challenging the Public/Private Divide: An Overview' in Boyd (ed.), *Challenging the Public/Private Divide: Feminism, Law and Public Policy*, at 17.

Although the boundary between the private and public shifts over time, the existence of the distinction and the notion of boundary are rarely questioned.⁴³

The idea of a clear division between the public and private realms was encapsulated, in the early 17th century, by the seminal statement in *Semayne's Case*⁴⁴ that, 'the house of every one is to him as his castle and fortress', ⁴⁵ which came to be expressed in the maxim, 'an Englishman's home is his castle'. This clear separation of the public and private spheres continued to be reflected in judicial language over 250 years later in *Re Agar-Ellis*, ⁴⁶ where Bowen LJ stated, '[b]oth as regards the conduct of private affairs, and of domestic life, the rule is that Courts of Law should not intervene except upon occasion. It is far better that people should be left free'. ⁴⁷ As recently as 40 years ago, the House of Lords stated in *Charter v Race Relations Board*, ⁴⁸ 'the natural antithesis to "public" is "private." O'Donovan has observed that '[t]he view is that the ongoing family and marriage should be left alone, so long as conflict does not cause breakdown. ⁵⁰ Indeed, the law's reluctance to interfere in the private realm results in the logical consequence that the regulation of the home and the family is understood as being inherently out-with the law's concern.

The significance of the public/private divide in historical judicial reasoning is illustrated again in the early 20th century decision in *Balfour v Balfour*,⁵¹ which concerned whether an agreement between a husband and wife regarding the payment of ongoing maintenance should be considered contractual. Atkin LJ clearly stated of such arrangements that, 'they are not contracts, and they are not contracts because the parties did not intend that they should attend legal consequences.' Subsequently such arrangements between husbands and wives have been described judicially as

⁴³ O'Donovan, Sexual Divisions in Law, at 8.

⁴⁴ (1604) 77 ER 194.

⁴⁵ ibid, at 196.

⁴⁶ (1883) 24 Ch. 317.

⁴⁷ ibid, per Bowen LJ, at 335.

⁴⁸ [1973] AC 868.

⁴⁹ ibid, per Lord Cross of Chelsea, at 906, see also the judgment of Lord Hodson, at 897.

⁵⁰ O'Donovan, Sexual Divisions in Law, at 13.

⁵¹ [1919] 2 KB 571.

⁵² ibid, per Atkin LJ, at 579.

'merely a matter of convenience', and the House of Lords has stated that, 'the court will be slow to infer legal obligations from transactions between husband and wife in the ordinary course of their domestic life.' This presumption of a lack of intention to create legal relations has been held to include similar agreements between those in other close familial relationships, including mother and daughter in *Jones v Padavatton*. Salmon LJ went on to state that '[this general presumption] derives from experience of life and human nature which shows that in such circumstances men and women usually do not intend to create legal rights and obligations, but intend to rely solely on family ties of mutual trust and affection.'

In his judgment in *Balfour*,⁵⁹ Atkin LJ essentially framed the decision in terms of recognising the divide between the public and private spheres, stating, '[i]n respect of these promises each house is a domain into which the King's writ does not seek to run, and to which his officers do not seek to be admitted.'⁶⁰ Arrangements between husband and wife (or parents and children) are considered to be firmly within the 'private sphere' and therefore not subject to legal interference unless this presumption can be overturned.⁶¹ These cases illustrate the influence exerted historically upon English judicial reasoning by the idea of a division between the public and private spheres, and by the understanding that the private sphere should not be subject to legal regulation or scrutiny.

⁵³ Spellman v Spellman [1961] 1 WLR 921, per Danckwerts LJ, at 926.

⁵⁴ Pettitt v Pettitt [1969] 2 WLR 966.

⁵⁵ The basic principle was repeatedly upheld by the Court of Appeal, see e.g. *Gould v Gould* [1970] 1 QB 275 and *Merritt v Merritt* [1970] 1 WLR 1211, two cases which considered whether the principle could be applied to married couples who were separated. The principle was referenced by Baroness Hale more recently in *MacLeod v MacLeod* [2010] 1 AC 298, at 315. It may be noted that this presumption against the enforceability of agreements between family members is not made by Scots law.

Notably the presumption was held not to apply to a 'man and his mistress' in *Diwell v Farnes* [1959] 1 WLR 624, showing the power of traditional marriage at that point in time.

⁵⁷ [1969] 1 WLR 328, in his judgment Danckwerts LJ states, at 332, 'there is no doubt that the same principles apply to dealings between other relations, such as father and son'.
⁵⁸ ibid, at 332.

⁵⁹ [1919] 2 KB 571.

⁶⁰ ibid, at 579.

⁶¹ Parker v Clark [1960] 1 WLR 286 and Collier v Hollinshead (1984) 272 EG 941, provide examples of cases where the court departed from this general presumption, on the basis that in the particular factual circumstances the actions of the parties suggested an intention to create legal relations.

3.1.C. The Orthodox Understanding of the 'Legal Subject'

The orthodox understanding holds that the legal subject represents a neutral and abstract standard⁶² and consequently that the 'person' of law is not directly relatable to the physical human being.⁶³ The mid-20th century scholar Nekam commented that, '[t]here is nothing in the notion of the subject of rights which in itself, would necessarily, connect it with human personality.'⁶⁴ Similarly, the positivist legal philosopher Kelsen observed that, '[t]he person exists only insofar as he "has" duties and rights; apart from them the person has no existence whatsoever.'⁶⁵ The critical theorist Naffine, who has written extensively on the construction of the legal subject, has observed that, '[i]t is the conventional legal view, supported by many learned treatises, which asserts that law does not operate with a specific individual in mind, that it favours no one type of person. In its purported objectivity and neutrality, law is supposed to adopt a common approach to all.'⁶⁶

Under this dominant conception of the legal subject, the view is that law is not actually engaged in an ongoing process of constructing the subject, but rather that the 'legal subject' amounts to an objective and fixed standard, ⁶⁷ which is based upon logical and rational principles. ⁶⁸ Building on this conception, in a subsequent work, Naffine went on to describe how this orthodox view of the nature of the legal subject is premised on the belief that:

_

⁶² Naffine, Law and the Sexes: Explorations of Feminist Jurisprudence, at 64-84.

⁶³ Arguably this lack of connection between the legal subject and an embodied human being is most clearly illustrated by the doctrine of 'separate legal personality' of corporations, associated with the case of *Salomon v Salomon & Co Ltd.* [1897] AC 22, which endows a corporation with legal personality distinct from the individuals who control it, see e.g. Nicholas James, 'Separate Legal Personality: Legal Reality and Metaphor' (1993) 5 (2) Bond Law Review 217.

⁶⁴ Alexander Nekam, *The Personality Conception of the Legal Entity*, (Harvard University Press, 1938), at 26.

⁶⁵ Hans Kelsen, General Theory of Law and State, (Russell and Russell, 1945), at 94.

⁶⁶ Naffine, Law and the Sexes: Explorations of Feminist Jurisprudence, at ix.

⁶⁷ The orthodox construction of the legal subject is described as the 'individuating, abstract form' of law, by David Sugarman and G.R Rubin, 'Introduction: Towards a New History of Law and Material Society in England 1750-1914' in G.R. Rubin and David Sugarman (eds.), *Law, Economy and Society, 1750-1914: Essays in the History of English Law,* (Professional Books, 1984), at 49.

⁶⁸ Naffine, *Law and the Sexes: Explorations of Feminist Jurisprudence* at 24-29, see also e.g. Roger Cotterrell, *The Sociology of Law: An Introduction*, (Butterworths, 1992) and Alison Jaggar, *Feminist Politics and Human Nature*, (Rowman and Littlefield, 1988).

The construct of the legal person and its legal personality is purely a matter of legal expediency. The legal person has no moral or empirical content, thus defined. It relies neither on biological or psychological predicates; nor does it refer back to any particular social or moral idea of a person and it is completely distinguished from these philosophical conceptions of the person which emphasise the importance of reason or intrinsic human value.⁶⁹

The orthodox construction views the legal subject as 'an abstraction'⁷⁰ and a necessary (legal) fiction,⁷¹ whose existence allows the law to operate effectively and efficiently.⁷² Kelsen famously describes the legal subject as being, 'not a natural reality but a social construction created by the science of law'.⁷³ The subject is conceptualised as a purely legal standard which does not, and should not, visibly promote or preference any moral or ethical ideals or values in its application,⁷⁴ because as Naffine comments, '[t]here is no moral essence to the legal person, in this view, and those who have sought to find one are misguided.'⁷⁵ Therefore, it is valuable to consider the values of the orthodox legal subject because the construction of the individual within the law can influence the legal understanding of the 'family'.

3.1.C.1. The Normative Nature of the Orthodox 'Legal Subject'

As has already been observed, the development of the law has been significantly influenced by the values of Enlightenment liberalism.⁷⁶ Similarly, I argue that the characteristics of the legal subject reflect the values that are privileged by

⁶⁹ Naffine, Law's Meaning of Life: Philosophy, Religion, Darwin and the Legal Person, at 35.

⁷⁰ Smith, 'Legal Personality', at 293.

⁷¹ Derham, 'Theories of Legal Personality' in Webb (ed.), *Legal Personality and Political Pluralism*, described the legal subject as, 'artificial, fictional', at 6.

⁷² FH Lawson, 'The Creative Use of Legal Concepts' (1957) 32 (5) New York University Law Review 909, stated at 913, that it was, 'created to serve certain purposes' and 'to conform to predetermined definitions.'

⁷³ Hans Kelsen, *Pure Theory of Law*, (University of California Press, 1967), at 174.

⁷⁴ The possibility of moral character of the legal subject has been considered in the context of corporate personality by Peter French, 'The Corporation as a Moral Person' (1979) 16 (3) American Philosophical Quarterly 207.

⁷⁵ Naffine, Law's Meaning of Life: Philosophy, Religion, Darwin and the Legal Person, at 37.

⁷⁶ In subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

construction of the individual within that same liberal philosophical tradition, which has underpinned the subsequent development of western society.⁷⁷ The historical strength of the ideology of liberalism, premised upon the centrality of reason, reflects a time in history when much of the world was beyond the explanation and comprehension of man.⁷⁸ Douzinas and Geary suggest that, '[t]he belief in reason's power to explain the world and the assumption that law can regulate our lives are psychological defences against the horrors of chaotic existence.⁷⁹ The certainty provided by the values of reason, logic and rationality, transferred from the sciences through the growth of analytical legal positivism,⁸⁰ helped endow the law with a renewed sense of fundamental authority, after the decline of the once dominant natural law tradition.⁸¹ Despite its purported and self-avowed neutrality and objectivity, the legal subject is constituted in a form that reflects the subject of the dominant philosophical theory of western capitalist society.⁸²

Different writers have suggested a range of features as being essential to the conception of the liberal subject. Lukes has observed that, '[t]he idea of the independent, rational citizen is a central presupposition of classical liberal

-

⁷⁷ Bottomley and Bronitt, *Law in Context*, describe liberalism as 'the dominant ideology of modern western society dating from about the 17th century', at 5, see also Andrew Vincent, *Modern Political Ideologies*, (3rd edition, Wiley-Blackwell, 2009), at 23-83 and C.B McPherson, *The Political Theory of Possessive Individualism: Hobbes to Locke*, (Clarendon Press, 1962).

⁷⁸ See above in subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

⁷⁹ Douzinas and Geary, Critical Jurisprudence: The Political Philosophy of Justice, at 43.

⁸⁰ Associated with the legal philosopher John Austin, see e.g. *Lectures on Jurisprudence or the Philosophy of Positive Law*, revised and edited by Robert Campbell, (3rd edition, John Murray, 1869).

⁸¹ In the face of the decline of the power of the Church to control knowledge during the Enlightenment and the work of philosophers such as Thomas Hobbes, *Leviathan*, on the nature of state power and David Hume, *A Treatise of Human Nature*, (First published 1739, reprinted by Clarendon Press, 1896).

on the 'is/ought distinction', who had questioned some of Natural Law's core assumptions.

This reflection of capitalist values is illustrated by the aforementioned doctrine of 'separate legal personality', in essence under this doctrine corporations are considered capable of possessing the characteristics of the 'legal subject'. On the issue of the subjecthood of corporations more generally, see e.g. Nicola Lacey, "Philosophical Foundations of the Common Law': Social not Metaphysical' in Jeremy Horder (ed.), Oxford Essays in Jurisprudence, (4th series, Oxford University Press, 2000). However, some critical theorists would argue that some of the values support by the orthodox 'legal subject' can actually be traced back to a much earlier point in history than the Enlightenment and the development of liberalism. See e.g. Mark C. Taylor, Deconstruction in Context: Literature and Philosophy, (University of Chicago Press, 1986), at 4, who argued that, '[f]rom its inception in Greece, Western philosophy has, for the most part, privileged oneness and unity at the expense of manyness and plurality. Accordingly the Western philosophical project can be understood as the repeated effort to overcome plurality and establish unity by reducing the many to the one.'

democratic theory.'83 Bottomley and Bronitt have suggested that '[1]iberalism at its heart posits the view of rational person or legal subject - perhaps famously embodied in the standards imposed by the "reasonable person" that inhabits the law of negligence and criminal law doctrine.'84 Moller Okin notes that 'the liberal tradition assumes that the behaviour of its political actors will be based on self-interest',85 while Dietz has suggested that 'the liberal individual might be understood as the competitive entrepreneur'.86 O'Donovan has provided the following description of the liberal legal subject:

The ideal legal subject in liberal theory is a rational, choosing person, capable of decision, an autonomous individual. This individual is without particularities of identity such as gender. Such a figure of neutrality is a deliberate legal creation to overcome differences - whether of cultural origin, race, gender, or other particularities.⁸⁷

Thus, the liberal individual is typically characterised as a rational, autonomous, self-determining and self-interested being⁸⁸ and the orthodox understanding of the legal subject reflects these values.

3.2. Critiquing the Orthodox Construction of the Legal Subject

The orthodox understanding of the legal subject has not been universally accepted by theorists; instead it has been subjected to significant critique.⁸⁹ This critique

⁸³ Steven Lukes, *Individualism*, (Blackwell, 1973), at 139, who additionally writes, at 56, 'autonomy is a value that has always been central to liberalism.'

⁸⁴ Bottomley and Bronitt, *Law in Context*, at 1.

⁸⁵ Moller Okin, Women in Western Political Thought, at 284.

Mary Dietz, 'Context is All: Feminism and Theories of Citizenship' (1987) 116 (4) Daedalus 1, at 5.
 Katherine O'Donovan, 'With Sense, Consent, or Just a Con?: Legal Subjects in the Discourse of

Autonomy', in Naffine and Owens (eds.), *Sexing the Subject of Law*, at 47.

88 When this liberal individual is translated into the legal subject, it finds expression as the 'reasonable man' of the common law; for example, the 'reasonable man' is used to determine the standard of care

expected within the law of negligence, see e.g. *Blyth v Birmingham Waterworks Co* (1856) 11 Ex. 781. ⁸⁹ The orthodox construction of the legal subject has undergone significant and varied critique, the consideration of which here will be necessarily brief, for a fuller exploration of this critique; see e.g. Naffine, *Law's Meaning of Life: Philosophy, Religion, Darwin and the Legal Person*.

questions the claims of the inherent objectivity and neutrality of the subject, as well as criticising the liberal values upon which the subject is premised; it also involves a rejection of the notion that the legal subject amounts to an abstract standard that applies equally and fairly to all individuals. This critique serves to unpick the assumptions of the legal subject; and thus the assumptions from which the 'natural' understanding of the nuclear family derives.

Thus, in this section I will consider some of the critical scholarship which disputes the orthodox construction's claims to objectivity and self-evidence and questions the applicability of the values that it is premised upon (subsection 3.2.A). I will then explore the feminist critique that the legal subject is based upon and privileges, characteristics and values historically associated with men and an idealised form of masculinity (subsection 3.2.B). To conclude, I will focus upon the historical 'persons' cases to illustrate some of the issues caused by the values and ideals of the orthodox construction of the legal subject (subsection 3.2.C).

3.2.A. The Apparent Objectivity Cloaks Values and Assumptions

It is argued that the orthodox construction's claim as to its own inherent objectivity and neutrality allows that construction of the legal subject to justify its own particular values, choices and assumptions on the basis that they amount to self-evident truths, ⁹⁰ Naffine has observed:

The legal person's very abstractness, his paradigm quality, serves to show that the legal approach is not even an approach in fact but an inherently neutral way of organising and arbitrating relations between human beings. By his any-personess, the legal man of law demonstrates the appropriateness and essential rightness of the law's particular approach to the world - the way that is not a way.⁹¹

⁹⁰ Naffine, Law and the Sexes: Explorations of Feminist Jurisprudence, at 44-47.

⁹¹ ibid, at 123.

I suggest that the continued influence of the orthodox construction derives, to a significant extent, from this abstractness. The apparent objectivity of the legal subject has been a significant contributor in helping to sustain the dominance of the orthodox construction of the legal subject. The portrayal of the subject as inherently neutral and objective obscures its basis in contested values which reflect the dominant, liberal philosophical traditions of western society. Sample of the orthodox construction of the legal subject.

The consistent privileging of these specific characteristics and values within the conceptualisation of the legal subject has led to the perception that the ideals embodied by the orthodox understanding are fundamental and inherent to the idea of the subject itself, rather than the result of a choice between competing values.⁹⁴ These liberal values can be seen to recur throughout law itself; Davies has observed that, 'the idea of a unified actor who is independent and rational forms the basis not only of many areas of substantive law, but also of the idea of law itself, as it has been traditionally presented.'⁹⁵ Consequently, the legal subject's strong association with the liberal values of rationality, autonomy and self-interest is presented by the orthodox construction as an inescapable feature of law itself, rather than as the outcome of any kind of value-judgement.⁹⁶ Herring has observed that, '[m]uch of the law is based on the assumption that we are competent, detached, independent people who are entitled to have our rights of self-determination and autonomy fiercely protected.'⁹⁷ Going further, Naffine and Owens have contended that '[l]aw has always assumed and constituted a subject who is deemed to act in certain ways, to

⁹² Kathy Laster and Padma Ranan, 'Law for one and one for all? An Intersectional Legal Subject', in Naffine and Owens, *Sexing the Subject of Law*, observe, at 197, '[t]he strength of the 'reasonable man' ideology lies in its universalism. The 'reasonable man' test is a convenient ruse. It allows the common law to pretend that its standard of judgement is based on the 'rationality' of Western Enlightenment philosophy, while still allowing realistic concessions to human frailty.'

For a critique of these dominant moral and ethical values and traditions, see e.g. Bauman, *Postmodern Ethics* and the work of Emmanuel Levinas, e.g. *Alterity and Transcendence*, (Linton, 1999), *Totality and Infinity*, (Duquesne University Press, 1999) and *The Levinas Reader*, Sean Hand (ed.) (Blackwell, 1989).

Margaret Davies, Asking the Law Question, (3rd edition, Lawbook Co., 2008), at 357, describes these notions as amounting to, 'an assumption, or even a dogmatic assertion, which cannot be demonstrated.'

⁹⁵ ibid, at 330.

⁹⁶ Michael Gardiner, 'Alterity and Ethics: A Dialogical Perspective' (1996) 13 (2) Theory, Culture & Society 121, argues, at 133, that due to the nature of the orthodox construction of the legal subject, 'the state cannot be genuinely pluralistic; it cannot register and protect difference.'

⁹⁷ Herring, 'The Disability Critique of Care', at 2.

wield certain rights and to assume certain responsibilities. And law has engaged in this act of creation quite self-consciously, fully aware that it is constituting a subject.'98

3.2.A.1. The Legal Subject Favours Specific Individuals

While under the orthodox construction the subject is presented as an abstract standard, lacking in any moral content, critics argue that the reality of its construction is that it favours a certain type of individual, who possesses specific characteristics.⁹⁹ Lukes observes that such privileging of a specific ideal type is unsurprising given that the "individuals" involved here – whether natural, or utilitarian or economic men – always turn out on inspection to be social and indeed historically specific. "Human nature" always in reality belongs to a particular kind of social man." As described above, the legal subject is based upon the ideal of the liberal individual; a rational and autonomous being. 101 Consequently, the construction of the subject is presented as neutral and self-evident, because it is based upon dominant cultural ideals that are presented as being 'natural' or 'common sense'. Indeed, on a similar basis the nuclear family is subsequently positioned as the 'natural' and 'common sense' understanding of family. 102 The literature on 'care' suggests that the focus on the abstract 'individual' itself is normatively problematic, because it diminishes the importance of relationships between individuals to the notion of subjecthood. 103 Herring and Foster suggest that 'it is impossible to speak atomistically about an

⁹⁸ Naffine and Owens, 'Introduction: Sexing Law' in Naffine and Owens (eds.), Sexing the Subject of Law, at 7.

⁹⁹ It has been suggested that the orthodox subject possesses characteristics typically associated with men and 'masculinity', see below at subsection 3.2.B.1, 'The Masculinity of the Orthodox 'Legal Subject''.

¹⁰⁰ Lukes, *Individualism*, at 75.

¹⁰¹ Herring, 'The Disability Critique of Care', at 3, presents a normative critique of this construction of the individual, arguing that, '[e]thics of care is based on the belief that people are relational. People understand themselves in terms of their relationships. They do not seek to promote only their own interests, not because they are "selfless", but because their interests are tied up with the interests of others.'

¹⁰² See further below at section 3.3, 'The Nuclear Family as the Natural Model of 'Family''.

¹⁰³ Jonathan Herring, 'Where are the Carers in Healthcare Law and Ethics?' (2007) 27 (1) Legal Studies 51, at 68, argues that, '[o]ne of the most attractive aspects of an ethic of care approach is that it seeks to move away from an atomistic picture of individuals, with rights that compete against each other, to a model that emphasises the responsibilities of people towards each other in mutually supporting relations.'

"individual". We are such quintessentially relational creatures that we should (and for all practical purposes do) abandon the legal fiction of a person who is an island unto herself."

The presentation of the subject as abstract and objective has the consequence of measuring all individuals against the same standard. It is submitted that this is problematic because, as Sugarman and Rubin suggest, 'this "individuality" is an abstraction, shorn of its particular context – and its relative position and power in society. As a result, substantial differences between relevant actors are obscured and ignored.'105 Thus the standard of the legal subject acts to remove or conceal the actual differences between people; the subject is disinterested in the particular characteristics of individuals. As Barron observes, '[w]ithin a formally rational legal order, the legal subject is an abstract entity, striped of contingent empirical determinations and thereby revealed as the essential sameness which characterises all human beings.'106 Under the orthodox construction, the legal subject amounts to a totalising standard within which the particular behaviours, circumstances and characteristics of specific individuals are flattened out and judged in terms of the 'objective' standard of the (fictitious) rational, autonomous and self-interested liberal individual. Consequently, '[c]hildren, married women, bankrupts, lunatics, Jews and foreigners have all been assigned a distinct legal status within the history of the common law, distinguishing their legal position from that of the adult male, solvent, sane, Christian citizen.' Throughout history, then, various different categories of person have been excluded from the status of 'legal subject'. 108

-

¹⁰⁴ Jonathan Herring and Charles Foster, 'Welfare Means Relationality, Virtue and Altruism' (2012) 32 (3) Legal Studies 480, at 497.

Sugarman and Rubin, 'Introduction: Towards a New History of Law and Material Society in England 1750-1914' in Rubin and Sugarman (eds.), *Law, Economy and Society, 1750-1914: Essays in the History of English Law*, at 49.

Anne Barron, 'Legal Discourse and the Colonisation of the Self in the Modern State', in Carty (ed.), *Post-Modern Law: Enlightenment, Revolution and the Death of Man*, at 113.

¹⁰⁷ John Dawson, 'The Changing Legal Status of Mentally Disabled People' (1994) 2 Journal of Law and Medicine 38, at 41.

¹⁰⁸ The example of the historical exclusion of women from full legal subjecthood is considered below at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'.

3.2.B. The Legal Subject Masks Inequality

This aspect of the orthodox construction of the legal subject - that it almost completely reduces differences between individuals to abstract sameness - allows law to ignore the substantial inequalities of circumstance of those individuals that come before it. 109 The paradox of the orthodox construction of the legal subject is that while it is traditionally presented as an objective standard, it is in reality laden with the subjective values of the philosophical, political and economic systems which underpin western society. 110 Within the standard of the legal subject, Douzinas and Warrington observe, '[t]he suffering face of the outsider is "translated" into the reasonable man of the common law and on this basis is found not to be suffering at all, not to be in need of the very protection that the law is supposed to provide for the weak and the disadvantaged.'111 The values of the legal subject are not impartial, but rather are premised upon a liberal vision of human nature which is contested. 112 Naffine has observed that the subject can be 'described as an undesirable caricature of a human being: impossibly self-possessed and self-reliant, will-driven, clinically rational and individualistic.' 113 By subjecting all people to this abstract standard, and holding that all people are equal before the law, the reality of difference between rich and poor, man and woman, black and white, young and old, is concealed within the apparent 'objectivity' of the legal subject. 114 The formal equality provided by the legal subject benefits those who conform to its characteristics and share its values; however those individuals who do not fit within the ideal type of the liberal individual are significantly disadvantaged by the application of this standard. 115

¹⁰⁹ Douzinas and Warrington, *Justice Miscarried: Ethics, Aesthetics and the Law*, at 165, suggest that, '[e]quality is not equality but absolute dissymmetry'.

¹¹⁰ As described above at subsection 3.1.C.1, 'The Normative Nature of the Orthodox 'Legal Subject''.

¹¹¹ Douzinas and Warrington, Justice Miscarried: Ethics, Aesthetics and the Law, at 23.

¹¹² For the Marxist critique of the subject and of law generally see e.g. Evgenii Pashukanis, *Law and Marxism: A General Theory*, Chris Arthur (ed.), (Pluto Press, 1989) and Hugh Collins, *Marxism and Law*, (Oxford University Press, 1982).

¹¹³ Naffine, 'Who are Law's Persons: From Cheshire Cats to Responsible Subjects', at 365.

¹¹⁴ This approach evokes the oft-quoted line from Anatole France's, *Le Lys Rouge*, (Calmann-Levy, 1894); that the poor must, 'labour in the face of the majestic equality of the law, which forbids the rich as well as the poor to sleep under bridges, to beg in the streets and to steal bread.'

¹¹⁵ The 'persons' cases described below at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects', illustrate how this applied in relation to women in the 19th and early 20th century. Additionally, Douzinas and Warrington, *Justice Miscarried: Ethics, Aesthetics and the Law*, at 218-232, provide the example of the 'objective' test for the construction of 'refugee', from the cases

3.2.B.1. The Masculinity of the Orthodox 'Legal Subject'

The type of individual envisaged by the dominant conception of the legal subject has been critiqued¹¹⁶ on the basis that this individual is endowed with characteristics which are traditionally associated with men (and more precisely, with a particular idealised form of masculinity). 117 The denial of subjecthood to women in the 'persons cases', considered below, reflects the fact that historically in the UK, the dominant, orthodox construction of the legal subject was understood as excluding women. 118 This exclusion was justified on the basis of 'natural' or 'common-sense' understandings about distinctions between the capabilities, characteristics and roles of the genders. 119 I submit that this distinction is a consequence of the liberal understanding of the 'public/private' divide and the duality of gender roles this promotes, with man as the 'breadwinner' and woman as the 'homemaker'. Indeed, I argue that this construction of separate, distinct gender roles continues to manifest itself in the dominant, contemporary legal understanding of the 'family'. 120 The values of the orthodox legal subject are also the values that are associated with the public realm, which was traditionally considered the sole preserve of men. 121 Pateman has argued that, '[o]nly masculine beings are endowed with the attributes and capacities necessary to enter into contracts, the most important of which is ownership of property in the person; only men, that is to say, are "individuals". 122 The intertwining of the values of the orthodox legal subject and those understood as characteristically 'masculine' is unsurprising because the orthodox subject (based

of R v Secretary of State for the Home Department, Ex p. Sivakumaran and Conjoined Appeals [1988] AC 958 and R v Secretary of State for the Home Department, Ex p. Bugdaycay [1987] AC 514.

¹¹⁶ For more on the feminist critique of the subject see e.g. James and Palmer (eds.), Visible Women: Essays on Feminist Legal Theory and Political Philosophy and Lacey, Unspeakable Subjects: Feminist Essays in Legal and Social Theory.

See R.W. Connell, *Masculinities*, (University of California Press, 2005), for an exploration of the complexity surrounding the usage of the term 'masculinity'.

¹¹⁸ See below at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'. This denial of subjecthood to women was also apparent in the classical societies, as discussed above at subsection 3.1.A.1, 'The Classical Societies'.

¹¹⁹ For example, see the judgment of Lord Neaves in *Jex-Blake v Senatus of University of Edinburgh* (1873) 11 M 784, below at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'.

As will be shown throughout the remainder of this thesis, particularly in the course of Part 3, 'Parenthood'.

¹²¹ See above at subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

¹²² Pateman, *The Sexual Contract*, at 5.

upon the liberal individual) and the public/private divide were both presented as inherent aspects of the 'neutral' and 'natural' social and legal order. 123

Naffine has argued that the legal person should be understood as 'the man of law': 124

The legal model of the person, it will be argued, is a man, not a woman. He is a successful middle-class man, not a working class male. And he is a middle-class man who demonstrates what one writer has termed a form of "emphasised" middle-class masculinity. In short, he is a man; he is a middle class man; and he evinces the style of masculinity of the middle class. 125

In a subsequent work, Naffine observes that in addition to embodying these 'masculine' characteristics, the subject is premised upon an explicit rejection of the features traditionally assigned to women, stating that, 'he seemed to take his nature from the positive exclusion of those characteristics which have traditionally been associated with women and from the consignment to women of those qualities and activities which would represent a diminution of his person.' As previously argued the 'objective' standard of the legal subject permits law to deem certain characteristics and behaviours as inherently appropriate and correct. Therefore, because of the orthodox subject's construction around these 'masculine' values, this will clearly favour the rational, middle-class, 'economic man' of liberalism, described by Naffine as 'the man of law', while consequently being detrimental to the interests of women. This feminist critique argues that despite the eventual

.

¹²³ Moller Okin, *Women in Western Political Thought*, at 5, observes more generally, that, '[i]t must be recognised at once that the great tradition of political philosophy consists, generally speaking of writings by men, for men and about men.'

¹²⁴ Naffine, Law and the Sexes: Explorations of Feminist Jurisprudence, at 100-123.

ibid, at 100. The description of Lord Denning in *Wachtel v Wachtel (No.2)* [1973] 2 WLR 366, at 376, that, '[t]he husband will have to go out to work all day and must get some woman to look after the house - either a wife, if he remarries, or a housekeeper, if he does not', illustrates the influence upon judicial reasoning of the construction of the legal subject as a successful middle-class man.

Naffine, 'Can Women Be Legal Persons?' in James and Palmer (eds.), Visible Women: Essays on Feminist Legal Theory and Political Philosophy, at 81.

¹²⁷ ibid, at 82-83, Naffine uses the example of the legal regulation of pregnant women, both through abortion laws and judicial decisions approving caesarean sections against the wishes of those women to illustrate the 'maleness' of the legal subject, she suggests that, at 83, 'legal persons do not reproduce and divide in this manner. Certainly they are never pregnant.'

inclusion of women within the legal definition of 'person', the legal subject retains elements of this embodiment as a male being of a particular socio-economic class and continues to be constructed on the basis of these idealised 'masculine' characteristics of rationality and self-interest.

3.2.C. 'The Persons Cases': The Rejection of Women as Legal Subjects

Cases containing detailed judicial discussion of the nature of the legal subject are limited in number because of the historical dominance of the orthodox construction of the subject, reflecting Naffine's observation that '[s]ometimes the nature of the legal person is presumed and implicit rather than expounded or defended.'128 However, judicial consideration of the subject of law is found in a succession of cases from the late 19th and early 20th centuries. These cases determined whether or not the use of the word 'person'129 in a variety of statutory provisions 130 relating to the ability and capacity to perform certain public functions included women as well as men. In these cases the judiciary was effectively considering, in a variety of contexts, the extent of the public role which was legally available to women at that point in history. While these cases do not involve disputes regarding the family, nevertheless they assist us by clarifying how the law (at this particular time in history) framed women's role as firmly within the private sphere as wives, mothers and 'homemakers' - a role constructed in binary opposition to the public role of men as legal subjects.

The decisions in these cases consistently denied women the right to exercise public functions that were held, at the time, to be the exclusive domain of men. This

-

¹²⁸ Naffine, Law's Meaning of Life: Philosophy, Religion, Darwin and the Legal Person, at 15.

¹²⁹ Or in some statutes (e.g. s.3 Representation of the People Act 1867) whether the word 'men' included women, due to s.4 Interpretation Act 1850 (Lord Brougham's Act), stating, 'words importing the masculine gender shall be deemed and taken to include females...unless the contrary as to gender...is expressly provided'.

¹³⁰ See e.g. s.27 Representation of the People (Scotland) Act 1868 and s.11 (1) Municipal

See e.g. s.27 Representation of the People (Scotland) Act 1868 and s.11 (1) Municipal Corporations Act 1882.

¹³¹ See Sachs and Wilson, Sexism and the Law: A Study of Male Beliefs and Legal Bias in Britain and the United States, at 4-66 for a detailed exploration of these cases, which are referred to as 'The Male Monopoly Cases'.

included women being prohibited from registering to vote, ¹³² women being prevented from attending university, ¹³³ women being barred from serving as local councillors, ¹³⁴ women being prevented from serving as either 'law agents' ¹³⁵ or solicitors ¹³⁶ and women being held to be unable to sit in the House of Lords. ¹³⁷ Collectively these decisions amounted to a denial that women were full legal subjects with full access to the public sphere of society. ¹³⁸ The explicit rejection of the subjecthood of women contained in the judgments demonstrates the rhetorical power that the strict 'public/private' divide and its restriction of women to the private sphere of the home and family held over social and legal understanding at that time in history. ¹³⁹

Examining the judgments in these cases reveals certain unifying strands and recurring themes. First, the cases contain repeated and clear rejections of the idea that women should be included within the word 'person' and this exclusion is expressed as self-evident and uncontroversial, reflecting the asserted self-evidence of the legal subject described above. This is exemplified by the judgment of Bovill CJ in *Wilson v Town Clerk of Salford*, ¹⁴⁰ where he simply stated that, '[i]t is clear that only males are "persons" within the meaning of [the relevant provision]...we have therefore no authority to hear this case. ¹⁴¹ In *Chorlton v Lings*, ¹⁴² Willes J described women as having a 'legal incapacity to vote at elections'. ¹⁴³ The use of this language seems to deny that the legal personhood of women is a legitimate issue, by suggesting that

¹³² Chorlton v Lings (1868) LR 4 CP 374 and Brown v Ingram (1868) 7 M 281.

¹³³ Jex-Blake v Senatus of University of Edinburgh (1873) 11 M 784, which is referred to as 'Septem contra Edine' – The Edinburgh Seven Case.

¹³⁴ Beresford-Hope v Lady Sandhurst (1889) 23 QBD 79, however ironically and somewhat absurdly in De Souza v Cobden [1891] 1 QB 687, the court was prepared to sanction criminal penalties for those women who 'purported' to act as local councillors.

¹³⁵ Hall v Incorporated Society of Law Agents (1901) 9 SLT 150.

¹³⁶ Bebb v Law Society [1914] 1 Ch. 286.

¹³⁷ Viscountess Rhondda's Claim [1922] 2 AC 339.

¹³⁸ The 'masculinity' of the legal subject was explored above at, subsection 3.2.B.1, 'The Masculinity of the Orthodox 'Legal Subject'.

This reflects the historical emphasis within the law on the strict public/private divide discussed above at subsection 3.1.A, "Family' and the Subject in Historical Societies' and subsection 3.1.B.1, 'The 'Public/Private' Divide within the Law'.

^{140 (1868)} LR 4CP 398.

ibid, per Bovill CJ, at 399.

¹⁴² (1868) LR 4 CP 374.

¹⁴³ ibid, per Willes J, at 388.

their exclusion is uncontroversial and thus not worthy of any detailed judicial consideration.

Second, there are references within the judgments to the previous customary exclusion of women from these activities or functions justifying the continued prohibition of women, unless there is express inclusion of women to the contrary by Parliament. This was exemplified by Lord Robertson in *Nairn v University of St. Andrews and Others*, who stated that the central fact in the present appeal is that from time immemorial men only have voted in parliamentary elections. Recourse to such language allowed the judges to disregard the wider moral implications of their decisions. It is submitted that this judicial statement reflects the orthodox construction of the legal subject as representing an objective standard, which is equally applicable to all, regardless of differences between individuals.

Third, the language and tone of the judgments regarding women is particularly striking; some of the decisions suggest that the exclusion of women from these public functions provided evidence of the *privileged* position women held in society, rather than an indication of their lack of status. Sachs and Wilson observe that throughout the judgments in these cases, '[t]he words that constantly recur in describing their attitudes towards women are decorum, respect and propriety. In their view, this respect for women did not hold women back, but shielded them from the harsh vicissitudes of public life.' This language is typified by Wiles J in *Chorlton v Lings*, who stated that, 'I must protest against its being supposed to arise in this country from any underrating of the sex either in point of intellect or worth. That would be quite inconsistent with one of the glories of our civilization, - the respect and honour in which women are held.' The prohibitions were described, in other

¹⁴⁴ See e.g. Swinfen Eady LJ in *Bebb v Law Society* [1914] 1 Ch. 286, who stated, at 297, 'it is sufficient to rest this case upon the inveterate practice of the centuries that, ever since attorneys as a profession have existed, women have been admitted to the office, and, in my opinion, that shews what the law is and has been.'

¹⁴⁵ [1909] AC 147.

¹⁴⁶ ibid, at 164.

¹⁴⁷ Sachs and Wilson, Sexism and the Law: A Study of Male Beliefs and Legal Bias in Britain and the United States, at 53.

¹⁴⁸ (1868) LR 4 CP 374.

¹⁴⁹ ibid, at 388.

cases, as existing to protect women from the stresses of the public sphere, which they were not believed to be suited to for reasons of the supposedly 'natural' distinctions between the capacities of the genders. This is encapsulated by the judgment of Lord Neaves in *Jex-Blake v Senatus of University of Edinburgh*, where he stated, '[t]he powers and susceptibilities of women are as noble as those of men; but they are thought to be different, and, in particular, it is considered that they have not the same power of intense labour as men are endowed with.' I argue that this notion of a 'natural' distinction between the genders continues to be evident within the nuclear family model.

These legal restrictions on women's participation in the public sphere were not removed until the Sex Disqualification Removal Act 1919. ¹⁵³ Inevitably, after Parliament had legislated to eliminate the majority of these explicit exclusions, the judiciary then reversed its previous position and held, in *Edwards v Attorney General Canada*, ¹⁵⁴ that women were after all capable of being 'persons'. ¹⁵⁵ The rationale of the earlier court decisions was seemingly questioned by Lord Sankey, who stated, '[t]he exclusion of women from all public offices is a relic from days more barbarous than ours, but it must be remembered that the necessity of the times often forced on man customs which in later years were not necessary. ¹⁵⁶ Regarding the motivation for this complete revision in judicial opinion, Sachs and Wilson have suggested that, '[i]n the absence of any other satisfactory explanation, the conclusion becomes inescapable that what had changed was not the meaning of the word "person", nor

¹⁵⁰ (1873) 11 M 784.

ibid, at 833, he further added to this passage, stating, 'I confess that, to some extent, I share in this view, and should regret to see our young females subjected to the severe and incessant work which my own observation and experience have taught me to consider as indispensable to any high attainment in learning'.

¹⁵² See the description of the 'nuclear family' given by Bernardes, *Family Studies: An Introduction*, at 2-3, set out above in Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family'.

¹⁵³ s.1 stated, '[a] person shall not be disqualified by sex or marriage from the exercise of any public function, or from being appointed to or holding any civil or judicial office or post, or from entering or assuming or carrying on any civil profession or vocation, or for admission to any incorporated society'.

¹⁵⁴ [1930] AC 124

ibid, at 138, Lord Sankey stated, '[t]he word "person"...may include members of both sexes, and to those who ask why the word should include females the obvious answer is why should it not? In these circumstances the burden is upon those who deny that the word includes women to make out their case.'

¹⁵⁶ ibid, at 128.

the modes of reasoning appropriate to lawyers, but the conception of women and women's position in public life held by the judges.' 157 It is submitted that the judgments in these cases illustrate how problematic the self-asserted abstractness, impartiality and dispassion of the legal subject appears, once the tangible history of legal decision-making is examined and scrutinised. 158

I contend that the historical exclusion of women from the legal definition of 'person' has influenced the construction of the legal subject and the values which that subject embodies, particularly regarding how these values continue to reflect the historical dominance within western society of the liberal understanding of the public/private divide and its associated gender roles. 159 I argue that the remnants of these values and ideals continue to influence the nuclear family model.

3.3. The Nuclear Family as the Natural Model of 'Family'

In this section, I will show that the recurrence of the nuclear family form throughout history has combined with the influence of the construction of the liberal legal subject and the ideals of the 'public/private' divide to influence the positioning of the nuclear family as the 'natural' and 'common sense' model of the family. I will argue that this construction of the legal subject as rational, autonomous and self-interested, in combination with the historical exclusion of women from the public sphere through the liberal understanding of the 'public/private' divide, have influenced the values of the traditional nuclear family (subsection 3.3.A). I will further argue that the nuclear family has become the law's idealised image of 'family' as a result of its aforementioned historical dominance, which has allowed it to appear and to be presented as the natural and 'common-sense' understanding of 'family' (subsection 3.3.B).

¹⁵⁷ Sachs and Wilson, Sexism and the Law: A Study of Male Beliefs and Legal Bias in Britain and the United States, at 41.

¹⁵⁸ Lord Sankey in Edwards v Attorney General Canada [1930] AC 124, at 134, observed that, '[c]ustoms are apt to develop into traditions which are stronger than law and remain unchallenged long after the reason for them has disappeared.'

¹⁵⁹ As considered above at subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

3.3.A. The Legal Subject and the Nuclear Family

Moller Okin has observed that '[t]he traditional, supposedly indispensable, nuclear family is used as the connecting link by which the basic biological differences between the sexes are expanded into the entire set of ascribed characteristics and prescribed functions which make up the conventional female sex role.' Thus, historically the private sphere of home and family was positioned as the appropriate realm for women. As O'Donovan observes, '[t]he insistence on the idea that women belong in the private sphere is part of the cultural superstructure which has been built on biological foundations.' Within the historical, liberal conception of the public/private divide, the private sphere of the home and the family was conceptualised as the place where the liberal individual would be able to escape from the issues, pressures and problems of the public sphere, as Naffine describes:

Liberal theory describes quite another world operating in the private sphere. To this place the liberal citizen can retreat and feel secure from the interventions of other persons and from the state, indeed the vital liberty of the person is thought to depend on the security of the private realm, where the individual can relax, express emotions, love and be loved.¹⁶³

Traditionally the law was largely content to ignore the private sphere,¹⁶⁴ which was seen as embodying an entirely different set of values from those of the liberal individual or legal subject,¹⁶⁵ Moller Okin has commented that, '[t]heorists who have assumed a high degree of egoism to determine relations between individuals in the sphere of the market, have assumed almost total altruism to govern interfamilial

^{1.0}

¹⁶⁰ Moller Okin, Women in Western Political Thought, at 293.

¹⁶¹ As considered above in subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

¹⁶² O'Donovan, Sexual Divisions in Law, at 16.

¹⁶³ Naffine, Law and the Sexes: Explorations of Feminist Jurisprudence, at 69.

¹⁶⁴ This approach was exemplified by the judgment of Bowen LJ in *Re Agar-Ellis* (1883) 24 Ch. 317, at 335, where he stated, 'I for one should deeply regret the day, if it ever came, when Courts of Law or Equity thought themselves justified in interfering more than is strictly necessary with the private affairs of the people of this country.' See also the cases discussed above in subsection 3.1.B.1, 'The 'Public/Private' Divide within the Law'.

¹⁶⁵ O'Donovan, *Family Law Matters*, at 23, observes that in the private sphere, 'self-sacrifice and altruism are idealised.'

relationships.' As the critical theorist Unger observes, '[i]n our public mode of being we speak to common language of reason, and live under laws of the state, the constraints of the market, and the customs of different social bodies to which we belong. In our private incarnation, however, we are at the mercy of our own sense impressions and desires.' There is a clear association of the private sphere with women, which reflects the 'masculinity' of the legal subject; O'Donovan suggests that, '[t]hose areas such as the personal, sexuality, biological reproduction, family home, which are particularly identified socially as the women's domain, are also seen as private.'

Following on from this, I argue that this construction of the rational and autonomous subject encourages the emphasis upon family form and structure within legal definitions of 'family' (represented through reliance upon the central nexus of the conjugal and parent/child relationships),¹⁶⁹ rather than an approach to understanding 'family' premised upon 'care' and relationality. Herring suggests that '[t]he values that are promoted within an ethic of care are not isolated autonomy or the pursuance of individualised rights, but rather those of promoting caring, mutuality and interdependence.' These values of mutuality and interdependence promoted by the 'ethic of care' are in opposition to those of the orthodox, autonomous, self-interested, liberal legal subject. Therefore, I submit that it is predictable that 'caring' is not privileged within the legal understanding of the 'family'. 172

Moreover, the separation of gender roles between breadwinner and homemaker within the traditional nuclear family could be understood as encompassing the distinction between the legal subject (the man) and the other(s)¹⁷³ (the woman and

-

¹⁶⁶ Moller Okin, Women in Western Political Thought, at 284.

¹⁶⁷ Roberto Unger, *Knowledge and Politics*, (Free Press, 1975), at 59.

¹⁶⁸ O'Donovan, Sexual Divisions in Law, at 3.

¹⁶⁹ As described above in Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family''.

¹⁷⁰ Herring, 'Where are the Carers in Healthcare Law and Ethics?', at 66.

¹⁷¹ See e.g. Gilligan, In a Different Voice: Psychological Theory and Women's Development, Groenhout, Connected Lives: Human Nature and an Ethics of Care, Held, The Ethics of Care and Noddings, Caring: A Feminine Approach to Ethics and Moral Education.

¹⁷² As observed above in Chapter 2, section 2.2., 'The Law's 'Definition(s)' of 'Family''.

For a detailed theoretical exploration of the role of 'the other' in ethics and morality, see e.g. Bauman, *Postmodern Ethics* and the work of Emmanuel Levinas, e.g. *Alterity and Transcendence*, *Totality and Infinity*, and *The Levinas Reader*, Sean Hand (ed.).

any children), who were not considered full legal subjects.¹⁷⁴ Thus, in addition to these non-subjects being unable to gain access to the public sphere, they were also deemed to be subordinate to the subject within the family context.¹⁷⁵ Pateman observes that:

The antinomy private/public is another expression of natural/civil and women/men. The private, womanly sphere (natural) and the public, masculine sphere (civil) are opposed but gain their meaning from each other, and the meaning of the civil freedom of public life is thrown into relief when counterposed to the natural subjection that characterises the private realm. ¹⁷⁶

Thus, I contend that the orthodox construction of the legal subject has impacted upon the form and values of the nuclear family, particularly influencing the separate and distinct gender roles envisaged within that idealised image of family.

3.3.B. The Historical Significance of the Nuclear Family

Western society, since the classical societies, has been based around a model of family centred on the 'nuclear' core, 177 comprising the nexus of the conjugal relationship and the parent/child relationship. Gittins has commented that, 'ideals of family relationships have become enshrined in our legal, social, religious and economic systems which, in turn, reinforce the ideology and penalise those who transgress it. Thus there are very real pressures on people to behave in certain ways,

1

¹⁷⁴ See e.g. the repeated judicial denials of the personhood of women, discussed above at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'.

Ancient Rome (reflecting the concept of *paterfamilias*), discussed above at subsection 3.1.A, "Family' and the Subject in Historical Societies'. This is further shown by the almost complete legal power historically possessed by a father over his children, see e.g. Brett MR in *Re Agar-Ellis* (1883) 24 Ch. 317, at 326, who described the position as follows, 'the father has the control over the person, education, and conduct of his children until they are twenty-one years of age. That is the law.' This power has been gradually diminished since the Guardianship of Infants Act 1925, which will be further considered below in Chapter 6, 'The Legal Understanding of the Parental Role'.

¹⁷⁶ Pateman, *The Sexual Contract*, at 11.

¹⁷⁷ See above at subsection 3.1.A, "Family' and the Subject in Historical Societies'.

¹⁷⁸ As described above in Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family''.

to lead their lives according to acceptable norms and patterns.' ¹⁷⁹ The recurrence of the nuclear family as the central family form since the classical period, as described above, has resulted in the nuclear family being positioned not only as the form of family that was traditionally dominant in society but also as the 'natural' or 'common-sense' model of the family. Torrant has observed that, 'the nuclear family form - that is, its dominance - was underpinned by an interlocking matrix of assumptions that together, constituted an ideology of this family form as "natural". ¹⁸⁰

3.3.B.1. The Nuclear Family in Historical Judicial Reasoning

As well as exerting influence upon the legal definitions of 'family' described above, ¹⁸¹ the positioning of the nuclear family as 'natural' and the assumption of a separation between the gender roles in that idealised image of 'family' is apparent in other historical judicial reasoning. ¹⁸² One such example, from the early 20th century, is found in *Short v Poole Corporation*, ¹⁸³ where Pollock MR referred to 'the incidence of domestic duties which may be presumed to fall with greater weight upon a married woman than upon a single woman. ¹⁸⁴ This understanding of a clear division of responsibilities between the genders within (indeed created by) the marital relationship is also evident in cases concerning divorce and judicial

¹⁷⁹ Gittins, The Family in Question: Changing Households and Familiar Ideologies, at 72.

¹⁸⁰ Julie P. Torrant, *The Material Family*, (Sense, 2011), at ix, Torrant goes on to further state that, '[t]hese assumptions linked a specific sex/gender/sexuality/procreation relation and understood these as natural and transhistorical.'

See the discussion of the role of the 'ordinary man' test with the judicial consideration of the Rent Act definition of 'family member', above in Chapter 2, subsection 2.2.A.2, "Family' as Defined by the 'Ordinary Man' Test'.

This is in addition to the cases showing the influence of the orthodox construction of the legal subject considered above at subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'.

¹⁸³ (1926) Ch. 66, this was a case concerning a local authority policy to terminate the employment of all married women teachers, the validity of the policy was upheld by the judgment, see e.g. *Price v Rhondda Urban District Council* (1923) 2 Ch. 372, for another case concerning a similar factual situation.

¹⁸⁴ ibid, per Pollock MR, at 86. Warrington LJ used similar words in his judgment at 92, referring to, 'the privilege and the burden of domestic ties'.

separation, from the late 19th century¹⁸⁵ into the mid-20th century.¹⁸⁶ This distinction of roles was given explicit judicial expression in *Wachtel v Wachtel (No 2)*,¹⁸⁷ where Lord Denning MR stated:

When a marriage breaks up, there will thenceforward be two households instead of one. The husband will have to go out to work all day and must get some woman to look after the house - either a wife, if he remarries, or a housekeeper, if he does not. He will also have to provide maintenance for the children. The wife will not usually have so much expense. She may go out to work herself, but she will not usually employ a housekeeper. She will do most of the housework herself, perhaps with some help. Or she may remarry, in which case her new husband will provide for her. 188

This statement clearly associates men with the public sphere and women with the private sphere; it assumes distinct roles for men and women, with husbands as breadwinners and wives as homemakers within the family. Forty years ago, then, judicial language reflected the traditional, gendered conception of the nuclear family and this clear division of gender roles within the 'family' was presented matter-of-factly as the 'common sense' or 'natural' understanding of 'family'.

¹⁸⁵ See e.g. *Wood v Wood* [1891] P 272, *Dean v Dean* [1923] P 172 and *Gilbey v Gilbey* [1927] P 197, in these cases emphasis is placed upon the husband's duty to provide maintenance for his wife and the marital conduct of the wife is a significant factor in determining the extent of her entitlement.

¹⁸⁶ See e.g. Pearce J in *Duchense v Duchense* [1951] P 101, who stated, at 114, '[t]he rights of a wife who has always been selfish and unloving ought not to be identical with those of a wife who has always been loving and unselfish.' See further e.g. *Schlesinger v Schlesinger* [1960] P 191, *Porter v Porter* [1969] 1 WLR 1155 and *Ackerman v Ackerman* [1972] 2 WLR 1253.

¹⁸⁷ [1973] 2 WLR 366.

ibid, at 376, per Lord Denning MR.

The influence of the division of gender roles will be considered in relation to the legal understanding of parenthood and parenting below in Chapter 6, section 6.2, 'The Gendered Parenting Roles of the Nuclear Family'.

¹⁹⁰ However, it should be noted that the approach set out in *Wachtel* was not followed by the Court of Appeal in *Dart v Dart* [1996] 2 FLR 286 and subsequently the judicial language in divorce cases has departed from explicit endorsement of the traditional division of gendered roles, see e.g. *White v White* [2001] 1 AC 596.

3.3.B.2. The Nuclear Family as the Dominant Social Understanding

The dominance of the ideology of the nuclear family is noticeable in the work of the early and mid-20th century sociologists and anthropologists who studied the family and its organization. The understanding of the nuclear family as 'natural' is exemplified by Linton, who suggested that '[t]he ancient trinity of father, mother, and child has survived more vicissitudes than any other human relationship. It is the bedrock underlying all other family structures. Within this dominant ideology, evidence of the historical incidence of the nuclear family form is combined with an appeal to the rhetorical power of tradition, in an effort to present the nuclear family as something that stands apart from particular social contexts. Murdock observed the existence of the nuclear family across cultures and concluded that '[t]he nuclear family is a universal human social grouping. Either as the sole prevailing form of family or as the basic unit from which more complex familial forms are compounded, it exists as a distinct and strongly functional group in every known society.

Thus, I argue that there is a certain circularity involved in the dominant and orthodox understanding: the nuclear family is constructed as the 'natural' model of family, which is largely premised on its consistent historical importance. On the basis of this 'natural' construction, I suggest that the nuclear family is positioned as the idealised image which underpins the legal understanding of the 'family', which is held to be based upon 'the person in the street's' understanding of 'family', as set out

_

¹⁹¹ For a more detailed consideration of the historical development of sociological and anthropological work on the family, see e.g. Faith Robertson Elliot, *The Family: Change of Continuity?*, (MacMillan, 1986).

¹⁹² Ralph Linton, 'The Natural History of the Family' in Ruth Nanda Anshen (eds.), *The Family: Its Function and Destiny*, (Revised edition, Harper, 1959), at 52.

¹⁹³ However, it is worth noting that this approach was not universally accepted within these disciplines at this point in the 20th century, see e.g. C.C. Harris, *The Family: An Introduction*, (Allen and Unwin, 1969), who argued that the nuclear family resulted from particular social and cultural processes rather than possessing a 'natural' basis.

¹⁹⁴ George P. Murdock, 'The Universality of the Nuclear Family' in Norman W. Bell and Ezra F. Vogel (eds.), *A Modern Introduction to The Family*, (Revised edition, Free Press, 1968), at 38. Notably, these empirical claims of the universality of the nuclear family were questioned within the same volume, by Kathleen E. Gough, 'Is the Family Universal? - The Nayar Case', further suggesting that the dominant ideology has always been subjected to criticism.

¹⁹⁵ See above at subsection 3.1.A, "Family' and the Subject in Historical Societies'.

above in Chapter 2.¹⁹⁶ In this way, I argue that the dominance of the nuclear family model effectively sustains itself, because it is presented as the 'natural' model of family and derives its continued significance from that positioning.

Conclusion

In Chapter 2, I argued that the traditional nuclear family, comprising the nexus of the conjugal and parent/child relationships, underpins the law's understanding and definition(s) of 'family'. In this chapter, I have considered the reasons for the normative centrality of the nuclear family within legal understanding.

In this chapter I have observed that the nuclear family form recurs as a central family unit throughout history and that this historical prominence of the nuclear family coincides with the historical prevalence of a division of society into public and private spheres. I have argued that the liberal notion of the 'public/private' divide influenced the orthodox construction of the legal subject, and that this orthodox construction embedded the values of the nuclear family, particularly the delineation of gendered roles (the man as 'breadwinner' and the woman as 'homemaker'). I have further argued that the nuclear family model possessed this dominance within the legal understanding of the 'family' because its historical position as a central family form resulted in it becoming viewed as the 'common sense' or 'natural' model of family. Through the judicial adoption of 'common sense' understandings, the nuclear model has also become the idealised image of 'family' within the law, as discussed above in Chapter 2.

The remaining three chapters of this thesis will explore the continuing normative significance of the traditional, nuclear family, given the context (set out above in the introduction) of legislative reforms and changes in familial demographics and practices; by focusing upon the legal regulation of the two relationships (the conjugal

 $^{^{196}}$ As considered above in Chapter 2, subsection 2.2.A.2, "Family' as Defined by the 'Ordinary Man' Test'.

relationship and the parent/child relationship) which form the central nexus of the nuclear family.

Part 2: The Conjugal Relationship

Chapter 4: The Legal Regulation of Conjugal Relationships

Introduction

This chapter will explore the regulation of adult relationships, and I will argue that the normative dominance of the nuclear family can be shown by the continuing centrality of marriage and 'marriage-like' conjugality to the legal recognition and regulation of adult personal relationships.

Historically marriage was central to the understanding of family both in society and in the law. However gradual shifts have taken place in the past century, with fewer people getting married, divorces increasing and much greater numbers of unmarried cohabitating couples. Today, cohabitation is understood in public attitudes as a legitimate alternative to marriage. Additionally, over the past 50 years there has been an ongoing process of recognition and regulation of same-sex relationships, from decriminalisation, through the judicial extension of existing

¹ As can be seen from the judicial attitude towards unmarried cohabitants in the cases concerning the Rent Act definition of 'family', discussed above in Chapter 2, subsection 2.2.A.4, 'Unmarried Cohabitants', see e.g. *Gammans v Ekins* [1950] 2 KB 328.

² ONS, 'Marriages in England and Wales (Provisional), 2012', at 2, shows that there has been a gradual decline from over 400,000 marriages per year to around 250,000 per year in 2012. Statistics showing a similar trend for marriages in Scotland, from 1971-2013, are found in 'Scotland's Population 2013 - The Registrar General's Annual Review of Demographic Trends 159th edition' (August 2014), at 49, available at - http://www.nrscotland.gov.uk/files//statistics/annual-review-2013/rgar-2013.pdf.

³ ONS 'Divorces in England and Wales 2012', at 2, shows the trends in divorce from the 1930s onwards, first showing a gradual increase, before a significant rise from the 1960s onwards (there were under 24,000 divorces in 1960, compared to nearly 120,000 in 1972 and a peak of around 165,000 in 1993). Statistics showing a similar trend for divorces in Scotland from 1971-2011 are available in, 'Scotland's Population 2011 - The Registrar General's Annual Review of Demographic Trends 157th edition' (August 2012), at 59, available at - http://www.gro-scotland.gov.uk/files//statistics/annual-review-2011/rgar-2011.pdf. See further ONS, 'Marriages in England and Wales (Provisional), 2012', at 9, which shows that in 34% of marriages at least one of the parties had been married previously with 15% being remarriages for both parties.

⁴ ONS, 'Short Report: Cohabitation in the UK, 2012', at 1, shows that there were 5.9 million people cohabiting in the UK in 2012, which is around double the figure from 1996.

⁵ Alison Park and Rebecca Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30, (2013), at 7-14, illustrates the shift in attitudes towards marriage and cohabitation amongst the UK population.

⁶ Through the partial decriminalisation of s.1 Sexual Offences Act 1967 and s.80 Criminal Justice (Scotland) Act 1980.

statutory provisions to same-sex relationships,⁷ the subsequent explicit inclusion of same-sex couples within legislative provisions,⁸ to the specific statutory regulation of same-sex relationships first as civil partnerships⁹ and more recently the extension of marriage to same-sex couples.¹⁰ Barlow and James comment that 'British society today comprises a plurality of family structures which our research confirms are considered acceptable personal lifestyle choices. Marriage, while still highly valued at least in the abstract, has lost its monopoly on sexual intimacy and childbearing in Britain.¹¹ In spite of these developments, I will argue that marriage retains a central position within UK law. Moreover, I will contend that the extension of legal regulation to non-marital adult personal relationships is premised upon the understanding that it is the 'marriage-like' characteristics and features of these relationships, particularly their conjugality that is used to warrant such regulation. Thus, I will argue that marriage and the traditional nuclear family retain significant normative power in the construction of adult (conjugal) relationships within the legal understanding of the 'family'.

In section 4.1, I will consider the question, 'what is marriage in UK law?' The legal definition of marriage will be examined and the continuing influence of historical judicial constructions of marriage will be explored. I will consider whether the contemporary understanding of marriage is changing; exploring the impact of the extension of marriage to same-sex couples upon the dominant understanding of marriage. In section 4.2, I will build upon this by examining the centrality of the conjugal relationship to the regulation of adult relationships. I will consider the legal regulation of cohabitation, arguing that relationships between cohabitees are recognised and regulated by the law on the basis that they are sufficiently 'marriage-

⁷ e.g. Same-sex couples being considered 'family' for the purposes of the Rent Act in *Fitzpatrick v Sterling Housing Association Ltd.* [2001] 1 AC 27, as discussed in Chapter 2, subsection 2.2.A.5, 'Same-Sex Cohabitants'.

⁸ e.g. Same-sex couples being allowed to adopt children as a couple, s.50 Adoption and Children Act 2002 and s.29 (3) Adoption and Children (Scotland) Act 2007, the inclusion of female same-sex couples within the legal parenthood provisions in cases of assisted reproduction, s.42-48 Human Fertilisation and Embryology Act 2008 and the extension of 'parental orders in cases of surrogacy to all same-sex couples, s.54 Human Fertilisation and Embryology Act 2008.

⁹ s.1 Civil Partnership Act 2004.

¹⁰ Marriage Same-Sex Couples Act 2013 and Marriage and Civil Partnership (Scotland) Act 2014.

¹¹ Anne Barlow and Grace James, 'Regulating Marriage and Cohabitation in 21st Century Britain' (2004) 67 (2) Modern Law Review 143, at 172.

like' and that this comparison with marriage occurs primarily as a result of their common conjugality.

4.1. What is Marriage in UK Law?

O'Donovan observes that, '[m]arriage has contractual and institutional elements, but it is also sui generis, a law unto itself. ¹² Marriage has been described, both judicially and otherwise, as an 'institution' or as a 'status', and as discussed above in Chapters 2 and 3, it occupies a central position within the traditional, nuclear family model. 15 As Diduck and Kaganas comment, '[m]arriage... is accorded a privileged social status which provides a place for the legitimate expression of heterosexual desires, imbuing other types of sexual activity and other relationships with a lesser status.¹⁶ In the 1970s in *Campbell v Campbell*. marriage was described as being, 'essential to the well-being of our society.' This historical emphasis on the societal importance of marriage was re-affirmed by the coalition government that took office in 2010; its response to the consultation on equal marriage 19 stated: 'Marriage is a hugely important institution in this country. The principles of long-term commitment and responsibility which underpin it bind generations together, and make our society strong.'20 I will suggest that marriage is still viewed within legal understanding as

¹² O'Donovan, Family Law Matters, at 44.

¹³ See e.g. Ormrod J in *Corbett v Corbett* [1970] 2 All ER 33, at 48 and Lord Nicholls of Birkenhead in Bellinger v Bellinger [2003] 2 AC 467, at 480, for judicial descriptions of marriage as an 'institution'. Former Prime Minister David Cameron was quoted as describing marriage as 'a great institution', Christopher Hope, 'We Will Legalise Gay Marriage by 2015, says David Cameron', 24^{th} (Daily Telegraph, London, July 2012), available at http://www.telegraph.co.uk/news/politics/9425174/We-will-legalise-gay-marriage-by-2015-says-David-Cameron.html.

 $[\]overline{^{14}}$ See e.g. Lord Hoffmann in Re P (Adoption: Unmarried Couples) [2008] 2 FLR 1084, at 1088, his lordship also described marriage as a 'very important institution' at 1090.

¹⁵ See particularly Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family''.

¹⁶ Diduck and Kaganas, Family Law, Gender and the State, at 73.

¹⁷ [1977] 1 All ER 1.

¹⁸ ibid, per Sir George Baker P, at 6.

^{19 &#}x27;Equal Marriage: The Government's Response', (HM Government, December 2012), available at https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/133262/consultationresponse 1 .pdf.

20 ibid, Ministerial Foreword of the Rt Hon Maria Miller MP, at 4.

representing the pinnacle or 'gold standard' of conjugal relationships and family forms; throughout this chapter I argue that marriage remains the standard against which the law measures all other relationships. 22

In this section, I will consider the legal definition of marriage (subsection 4.1.A), examining the development of the judicial understanding of marriage and exploring how this has evolved over time (subsection 4.1.A.1). I will then examine whether the social and legal understanding of marriage is changing (subsection 4.1.B), considering whether the recent extension of marriage to same-sex couples has affected the fundamental meaning of legal marriage (subsection 4.1.B.1).

4.1.A. The Legal Definition of Marriage

There is no statutory definition of marriage,²³ and Herring suggests that '[i]t is impossible to provide a single definition of marriage.'²⁴ Historically in England and Wales,²⁵ the Church of England had control over matrimonial law and regulation.²⁶ As a consequence, even after the creation of civil marriage,²⁷ the Christian

²¹ See e.g. Jo Miles, Fran Wasoff and Enid Mordaunt, 'Reforming Family Law - The Case of Cohabitation: 'Things May Not Work Out as You Expect' (2012) 34 (2) Journal of Social Welfare and Family Law 167, at 168 for academic usage of this description.

In some jurisdictions the centrality of marriage is illustrated through explicit constitutional protections, see e.g. the Constitution of Ireland, Article 41 (3), which states, '[t]he State pledges itself to guard with special care the institution of Marriage, on which the Family is founded, and to protect it against attack' and the Basic Law for the Federal Republic of Germany, Article 6 (1), which states, '[m]arriage and the family shall enjoy the special protection of the state.' Furthermore, the European Convention on Human Rights, Article 12, states, '[m]en and women of marriageable age have the right to marry and to found a family.'

²³ Neither the Marriage Act 1949 nor the Matrimonial Causes Act 1973 provides a definition of the term 'marriage'; this is also the case with the Marriage (Scotland) Act 1977.

²⁴ Herring, *Family Law*, at 70.

²⁵ In Scotland the religious control of marriage operated differently, due to the different role of the Church of Scotland, which was not an organ of the state, as the Church of England was.

²⁶ The Ecclesiastical Courts retained authority over matrimonial matters until the Matrimonial Causes Act 1857, which set up the Court for Divorce and Matrimonial Causes.

²⁷ In England and Wales civil marriage was introduced by the Marriage Act 1836 and in Scotland it was not introduced until over 100 years later in the Marriage (Scotland) Act 1939. In 2012, Civil marriages accounted for around 70% of marriages, ONS, 'Marriages in England and Wales (Provisional), 2012', at 6.

conception of marriage²⁸ represented the legal understanding of marriage.²⁹ I will argue that the remnants of this dominance are still apparent in the influence of traditional ideals within the legal understanding of marriage.³⁰

4.1.A.1. The Judicial Understanding of Marriage

The starting point for the judicial definition of marriage is the judgment of Lord Penzance³¹ in Hyde v Hyde,³² in which he stated: 'I conceive that marriage, as understood in Christendom, may for this purpose be defined as the voluntary union for life of one man and one woman, to the exclusion of all others.³³ It is also apparent that historically the judicial understanding of marriage was premised upon the construction of separate and distinct gender roles.³⁴ associated with the public/private divide and the traditional nuclear family.³⁵ The husband was constructed as the head of the household and breadwinner, while the wife was constructed as the dependent homemaker. This understanding is demonstrated by the judgment of Sir James Hannen P, in *Durham v Durham*, ³⁶ who stated that marriage involved 'protection on the part of the man, and submission on the part of the

²⁸ e.g. As found in the 'Book of Common Prayer: The Form of Solemnization of Matrimony', available at - https://www.churchofengland.org/prayer-worship/worship/book-of-common-prayer/theform-of-solemnization-of-matrimony.aspx.

Hyde v Hyde (1866) LR 1 P&D 130.

³⁰ See e.g. the judgment of Sir Mark Potter P in Wilkinson v Kitzinger (No. 2) [2007] 1 FLR 295, which refused to recognise a Canadian same-sex marriage as a marriage in the UK, instead converting it to a civil partnership. Indeed, it is notable that the law in the UK still provides for both religious marriage (Part II Marriage Act 1949 and s.9-16 Marriage (Scotland) Act 1977) and civil marriage (Part III Marriage Act 1949 and s.17-20 Marriage (Scotland) Act 1977) even though the legal effects of each are the same, with the distinctions relating to the formalities of the marriage and the wedding ceremony.

³¹ However, see Rebecca Probert, 'Hyde v Hyde: Defining or Defending Marriage?' [2007] 19 (3) Child and Family Law Quarterly 322, at 325, where she argues that, 'Lord Penzance's dictum should not be regarded as a definition of marriage but as a defence, and that it has been used as a defence rather than a definition by subsequent judges.'

³² (1866) LR 1 P&D 130.

³³ ibid, per Lord Penzance, at 130, see further *Re Bethell* (1887) 38 Ch. D 220.

³⁴ See e.g. Pretty v Pretty [1911] P 83, where Bargrave Deane J stated, at 88, '[s]ome people think that, in such matters, you must treat men and women on the same footing. But this Court has not taken, and, I hope, never will take, that view. I trust that, in dealing with these cases, it will ever be remembered that the woman is the weaker vessel'.

³⁵ As discussed above in Chapter 3, subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

³⁶ (1885) 10 PD 80.

woman.'37

The definition of marriage provided by Lord Penzance underpinned the judicial understanding of marriage throughout the 20th century³⁸ and the definition has continued to be referenced judicially in recent times.³⁹ This was illustrated by the judgment of Sir Mark Potter P in *Wilkinson v Kitzinger (No 2)*,⁴⁰ who commented that '[t]his definition has been applied and acted upon by the courts ever since.⁴¹ Later in his judgment, the President went on to observe that:

It is apparent that the majority of people, or at least of governments, not only in England but Europe-wide, regard marriage as an age-old institution, valued and valuable, respectable and respected, as a means not only of encouraging monogamy but also the procreation of children and their development and nurture in a family unit (or "nuclear family").⁴²

This statement shows the ongoing influence of the traditional conception of marriage and its interrelationship with the nuclear family upon contemporary judicial understanding.⁴³ This continuing significance was noted by Sir James Munby, who commented extra-judicially that '[u]ntil very recently, family law was concerned largely, if not exclusively, with the family wrought in the image of Lord Penzance's famous definition of marriage'.⁴⁴ An example is the description of marriage given by

3

³⁷ ibid, per Sir James Hannen P, at 82.

³⁸ See e.g. *Corbett v Corbett* [1970] 2 All ER 33. However, the continued use of this definition was subject to academic critique, see e.g. Sebastian Poulter, 'The Definition of Marriage in English Law' (1979) 42 (4) Modern Law Review 409.

³⁹ See e.g. Ward LJ's describing it as 'the hallowed definition of marriage' in *Bibi v Chief Adjudication Officer* [1998] 1 FLR 375, at 379 and Munby J's reference to 'Lord Penzance's famous definition' in *Sheffield City Council v E and S* [2005] 1 FLR 965, at 998.

⁴⁰ [2007] 1 FLR 295.

⁴¹ ibid, per Sir Mark Potter P, at 301.

⁴² ibid, at 329.

⁴³ However, it should be noted that the President's reliance on the traditional definition has been the subject of academic criticism, see e.g. Rosemary Auchmuty, 'What's So Special About Marriage? The Impact of Wilkinson v Kitzinger' [2008] 20 (4) Child and Family Law Quarterly 475, at 480, who observed, in regard to the above quoted passage, that, '[t]his sentence is contentious in almost every aspect.'

⁴⁴ Munby, 'Years of Change: Family Law in 1987, 2012 and 2037', at 279.

Lord Millett in *Ghaidan v Godin-Mendoza*:⁴⁵ '[m]arriage is the lawful union of a man and a woman. It is a legal relationship between persons of the opposite sex. A man's spouse must be a woman; a woman's spouse must be a man. This is the very essence of the relationship, which need not be loving, sexual, stable, faithful, long-lasting or contented.'⁴⁶ This judicial statement invokes the language of Lord Penzance's definition and illustrates the continuing significance of the historical and traditional definition of marriage within judicial understanding.⁴⁷ These various judicial comments show that the definition of marriage encapsulated in *Hyde v Hyde*, and the ideals it embodies, retain significant influence upon the contemporary judicial conception of marriage.

4.1.A.2. The Legal Effects of Marriage

In *Re P (Adoption: Unmarried Couples)*,⁴⁸ Baroness Hale of Richmond stated that '[m]arriage brings with it legal rights and obligations between the couple which unmarried couples do not have.'⁴⁹ Historically, due to the doctrine known in English law as coverture,⁵⁰ marriage resulted in a wife having no legal personality separate from that of her husband.⁵¹ This doctrine was gradually eliminated by the Married Women's Property Acts of the late 19th century,⁵² which allowed married women to own their own property, effectively eliminating this most substantial of all the legal

-

⁴⁵ [2004] 2 AC 557.

⁴⁶ ibid, per Lord Millett, in his dissenting judgment, at 588.

⁴⁷ In addition, the final sentence provides significant insight into the judicial understanding of the marital relationships, because it makes clear that, in Lord Millet's opinion, law has little interest in the nature or quality of the relationship between the parties to an on-going marriage. See further e.g. *Vervaeke v Smith* [1982] 2 All ER 144, where Lord Halisham stated, at 147, '[a]lthough valid in point of form that marriage with Smith was not in any sense an ordinary one. There was no intention to cohabit as man and wife.'

⁴⁸ [2008] 2 FLR 1084.

⁴⁹ ibid, per Baroness Hale, at 1121.

For details on the Ancient Scots law of marriage, see e.g. the transcription of the 15th century lectures of William Hay, John C. Barry, *William Hay's Lectures on Marriage*, (Stair Society, 1967) and Patrick Fraser, *Treatise on Husband and Wife, According to the Law of Scotland*, (2nd edition, T & T Clark 1876-78)

⁵¹ Sir William Blackstone, *Commentaries on the Laws of England*, Volume 1, at 442.

⁵² The Married Women's Property Acts 1870, 1882, 1884 and 1893 and See also the Married Women's Property (Scotland) Acts 1881 and 1920.

consequences of marriage.⁵³ Other previously important consequences of marriage have also now been eliminated, including the husband being the sole legal guardian of children born to the marriage,⁵⁴ the 'legitimate' status of children born during a marriage,⁵⁵ and the common law rule that a husband could not be found guilty of raping his wife.⁵⁶

With these historical effects abolished,⁵⁷ the main legal consequences of marriage are now the financial and economic entitlements and benefits it brings; including access to favourable taxation regimes,⁵⁸ consequences for succession⁵⁹ and access to the regime of financial provision on divorce.⁶⁰ Thus, regardless of how marriage is understood in cultural discourse, the legal consequences of marriage are now primarily financial. Despite the diminution in the legal effects of marriage, Hibbs, Barton and Beswick observe that many people who get married have 'misperceptions about the law relating to marriage and cohabitation, and [are] anyway little concerned with the legal consequences of their decision.'⁶¹ In other words, individuals do not generally make decisions about whether or not to get married on the basis of the legal effects of marriage.

⁵³ In *Midland Bank Trust Co Ltd. v Green* [1982] Ch. 529, at 538, Lord Denning MR observed that, 'medieval lawyers held that husband and wife were one person in law: and that the husband was that one. It was a fiction then. It is a fiction now.'

⁵⁴ Which was effectively abolished by 'welfare test', first established in s.1 Guardianship of Infants Act 1925.

⁵⁵ The distinction between children born within marriages and those 'illegitimate' children born outside marriage was effectively abolished by the Family Law Reform Act 1987 and the Law Reform (Parent and Child) (Scotland) Act 1986.

⁵⁶ Abolished in England and Wales in *R v R (Rape: Marital Exemption)* [1992] 1 FLR 217 and in Scotland in *S v HM Advocate* 1989 SLT 469.

⁵⁷ For the abolition of other historical consequences of marriage, see e.g. the Law Reform (Husband and Wife) (Scotland) Act 1984 and the more recent abolition of the wife's defence to a charge of reset in s.7 Marriage and Civil Partnership (Scotland) Act 2014.

⁵⁸ e.g. the spousal exemption for inheritance tax, s.18 Inheritance Tax Act 1984 and the exemption from capital gains tax on transfers between spouses, s.58 Taxation of Chargeable Gains Act 1992.

⁵⁹ e.g. rights for spouses on intestate succession, in England and Wales, s.46 Administration of Estates Act 1925 and prior and legal rights for spouses in Scotland, provided by Part 2 Succession (Scotland) Act 1964.

⁶⁰ The provisions for 'ancillary relief', in England and Wales, are provided by Part 2 Matrimonial Causes Act 1973; see also s.8-17 Family Law (Scotland) Act 1985 for the Scottish regime of financial provision on divorce.
⁶¹ Mary Hibbs, Chris Barton and Joanne Beswick, 'Why Marry? - Perceptions of the Affianced'

Mary Hibbs, Chris Barton and Joanne Beswick, 'Why Marry? - Perceptions of the Affianced' [2001] 31 (3) Fam. Law 197, at 207, this statement is based on the findings of empirical research, carried out by the authors, with couples preparing to get married.

Moreover, in view of the relatively limited effects of marriage in contemporary society, Herring has observed that, '[t]he law has had much to say about who can marry whom and how the relationship can be ended, but says very little explicitly about the content of the relationship itself.' Against this background, it is perhaps unsurprising that the traditional 'natural' construction of marriage (and the associated gendered roles) has filled the gap. The next section will explore the continued centrality of these traditional ideals and consider whether the social and legal understanding of marriage is changing.

4.1.B. Is the Understanding of Marriage Changing?

There is some recent evidence of judicial recognition that the understanding of marriage may be changing, ⁶³ reflecting a movement away from the language of Lord Penzance in *Hyde v Hyde*. ⁶⁴ In *Sheffield City Council v E and S* ⁶⁵ Munby J stated that, 'marriage, whether civil or religious is a contract, formally entered into. It confers on the parties the status of husband and wife, the essence of the contract being an agreement between a man and a woman to live together, and to love one another as husband and wife, to the exclusion of all others.' ⁶⁶ While this description still reflects elements of the traditional definition, ⁶⁷ the emphasis placed on the contractual nature of marriage ⁶⁸ and the corresponding omission of any reference to

⁶² Herring, *Family Law*, at 72, the complex provisions governing the formalities of marriage are found in the Marriage Act 1949 and the Marriage (Scotland) Act 1977, in addition s.11-16 Matrimonial Causes Act 1973 provide the rules governing nullity of marriage and the distinction between void and voidable marriages, see further Rebecca Probert, 'The Evolving Concept of Non-Marriage' [2013] 25 (3) Child and Family Law Quarterly 314.

⁶³ See e.g. the dissenting judgment of Thorpe LJ in the Court of Appeal in *Bellinger v Bellinger* [2001] 2 FLR 1048, where he acknowledged, at 1082, 'the world that engendered those classic definitions has long since gone...The intervening 130 years have seen huge social and scientific changes...Illegitimacy with its stigma has been legislated away: gone is any social condemnation of cohabitation in advance of or in place of marriage.'

⁶⁴ (1866) LR 1 P&D 130. See e.g. Probert, 'Hyde v Hyde: Defining or Defending Marriage?', who has been critical of the judicial reliance upon *Hyde*, observing, at 336, '*Hyde* should be seen for what it is: a case of considerable historical interest, that tells us a great deal about the attitudes of mid-Victorian England - but nothing about how marriage should be defined today.'

^{65 [2005] 1} FLR 965.

⁶⁶ ibid, per Munby J, at 1004.

⁶⁷ Particularly through limiting marriage to 'one man and one woman' and retaining 'to the exclusion of all others'.

⁶⁸ The contractual nature of marriage was also emphasised by Thorpe LJ in *Bellinger v Bellinger* [2001] 2 FLR 1048, who stated, at 1082, '[m]arriage has become a state into which and from which

the institutional character of marriage, illustrates an apparent shift in understanding away from the dominance of the historical definition. This ostensible shift in the judicial understanding of marriage is also evident from the judgments in *Re P* (*Adoption: Unmarried Couples*),⁶⁹ where Lord Hoffman stated that '[i]t is clear that being married is a status',⁷⁰ and Baroness Hale observed that, 'marriage is not just a contract; it is also a status',⁷¹

Given this judicial recognition, combined with the trends in societal demographics⁷² and changing social attitudes toward marriage and cohabitation,⁷³ some academic commentators⁷⁴ have suggested that cultural understandings of marriage are evolving.⁷⁵ Auchmuty has argued that, '[t]hough the institution retains residual value for some cultural groups...for most of us it is *simply a lifestyle choice*.'⁷⁶ On the basis of their empirical research, Eekelaar and MacLean suggest that '[i]t becomes increasingly difficult to identify being married in itself as *necessarily*, or even *characteristically*, constituting a significant source of personal obligations in the eyes of the participants in such relationships.'⁷⁷ Therefore, for many, marriage is now being constructed as a 'partnership of equals'⁷⁸ and this has been reflected in some

_

people choose to enter and exit. Thus I would now redefine marriage as a contract for which the parties elect but which is regulated by the state'.

⁶⁹ [2008] 2 FLR 1084.

⁷⁰ Ibid, at 1088.

⁷¹ ibid, at 1120. It is not the purpose or intention of this thesis to explore the academic literature which considers the contractual nature of marriage.

⁷² See e.g. ONS, 'Marriages in England and Wales (Provisional), 2012' and ONS, 'Short Report: Cohabitation in the UK, 2012'.

⁷³ See e.g. Park and Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30.

⁷⁴ See e.g. Carol Smart, 'Law and Family Life: Insights from 25 Years of Empirical Research' [2014] 26 (1) Child and Family Law Quarterly 14, who observed, at 29, '[m]arriage remains important but only in as much as it is part of a process of forming kin, alongside other methods of so doing.'

⁷⁵ However, other commentators have been more critical, see e.g. Heather Brook, 'Zombie Law: Conjugality, Annulment and the (Married) Living Dead' (2014) 22 (1) Feminist Legal Studies 49, who argued, at 50, that marriage, 'is something of a zombie category.'

⁷⁶ Rosemary Auchmuty, 'Law and the Power of Feminism: How Marriage Lost its Power to Oppress Women' (2012) 20 (2) Feminist Legal Studies 71, at 74.

⁷⁷ John Eekelaar and Mavis MacLean, 'Marriage and the Moral Bases of Personal Responsibility' (2004) 31 (4) Journal of Law and Society 510, at 536.

⁷⁸ See e.g. Anne Barlow, 'Solidarity, Autonomy and Equality: Mixed Messages for the Family?' [2015] 27 (3) Child and Family Law Quarterly 223, who observes, at 223, '[a]dult personal relationships are increasingly characterised as equal partnerships where the partners should be at liberty jointly to exercise their autonomy around decision-making on family issues.'

judicial language.⁷⁹ Within this construction of marriage, individuals are given the opportunity for personal fulfilment and as a mechanism to express love,⁸⁰ show commitment⁸¹ and provide care⁸² for each other. Barlow and James observe that:

[M]arriage has simply lost its power to hold people together and to entice them into lifelong partnerships in the first place. The social structures which gave it this power have been greatly weakened by the lack of religion, women's financial independence, state support for lone parents, separation of sex from marriage and of childbearing from marriage, ease of divorce, ease of cohabitation.⁸³

It would seem, therefore, that while the *idea* of marriage retains significant normative power, the understanding of the relationship, for some couples, has shifted away from the traditional construction.⁸⁴ However, it is also clear that in spite of this rhetorical shift, the traditional gendered roles of 'breadwinner' and 'homemaker' still exert influence on wider social and familial practices.⁸⁵ This continuing significance has been acknowledged judicially; in *Bellinger v Bellinger* Lord Nicholls noted

⁷⁹ See e.g. Lord Emslie's statement in *S v HM Advocate* 1989 SLT 469, at 473, that, 'A husband and wife are now for all practical purposes equal partners in marriage' and Munby J's comment in *Sheffield City Council v E and S* [2005] 1 FLR 965, at Para 131, that, '[t]oday both spouses are the joint, co-equal heads of the family.'

Hibbs, Barton and Beswick, 'Why Marry? – Perceptions of the Affianced', at 200, state regarding the individuals they interviewed, '[n]o one claimed, or admitted to, financial reasons for marriage, love was the single highest-ranking reason given for marriage.'

⁸¹ Diduck, *Law's Families*, at 33, suggests that, 'while older married couples spoke of obligations, often stemming from the marriage vows, younger ones spoke of commitments.'

⁸² Herring, *Caring and the Law*, at 199, suggests that, '[m]arriage should be about the promotion of caring and ensuring justice within the sharing of caring obligations. So understood limitations based on the gender of the parties or their sexual behaviour would be irrelevant.'

⁸³ Barlow and James 'Regulating Marriage and Cohabitation in 21st Century Britain', at 176.

⁸⁴ See e.g. Park and Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30, at 1-32. Within this, at 4, the authors show there has been a substantial decrease in religious identification in the UK in the past 30 years, finding that 48% of the UK population, in 2012, describe themselves as having no religious affiliation (this figure is down slightly from 51% in 2009), up from 31% in 1983.

⁸⁵ See e.g. ONS, 'Time Use Survey 2005', (July 2006), which suggested that men spend significantly less time per day than women on both childcare and housework, as well as Rosemary Crompton and Clare Lyonette, 'Who Does the Housework? The Division of Labour within the Home', British Social Attitudes Survey 24, (SAGE, 2008) and Jacqueline Scott and Elizabeth Clery, 'Gender Roles: An Incomplete Revolution?', British Social Attitudes Survey 30, (2013), available at http://www.bsa.natcen.ac.uk/media/38723/bsa30 full report final.pdf. These issues will consider further below in Chapter 6, subsection, 6.2.B.1, 'The Shift in the Construction of Fatherhood'.

⁸⁶ [2003] 2 AC 467.

that, '[m]arriage is an institution, or relationship, deeply embedded in the religious and social culture of this country.' Writing extra-judicially, Baroness Hale has commented that '[w]e do not need to regard marriage as a religious sacrament to believe it is more than a mere individual contract.' As O'Donovan observes:

The sacred character of marriage as an institution calls on a past, understood and shared tradition, and on an eternal future, a perpetuity. Marriage is an emblem of continuity, of reproduction of the race, of the recruitment of new members, of the creation of new units of generation from one generation to another.⁸⁹

Thus, I contend that these two contrasting conceptions of marriage (the traditional status-based understanding premised upon distinct gendered roles and the modern contract-based understanding of a 'partnership of equals') are both operating simultaneously within contemporary legal and cultural understandings of marriage. However, as noted above, ⁹⁰ the traditional image of marriage and the gendered roles it reflects still retains substantial normative influence upon the legal understanding of the family.

4.1.B.1. Same-Sex Marriage and the Understanding of Marriage

The clearest example of a change in the legal definition of marriage is the opening of the relationship to same-sex couples⁹¹ through recent legislation.⁹² The introduction

⁻

⁸⁷ ibid, per Lord Nicholls, at 480, for a consideration of the decision in *Bellinger* see e.g. Stephen Gilmore, '*Bellinger v Bellinger* - Not Quite Between the Ears and Between the Legs - Transsexualism and Marriage in the Lords' [2003] 15 (3) Child and Family Law Quarterly 295.

⁸⁸ Brenda Hale, 'Equality and Autonomy in Family Law' (2011) 33 (1) Journal of Social Welfare and Family Law 3, at 4.

⁸⁹ O'Donovan, Family Law Matters, at 47.

⁹⁰ In subsection 4.1.A.1, 'The Judicial Understanding of Marriage'.

⁹¹ Some differences remain between marriage for same-sex and opposite-sex couples; one of these relates to the availability of certain types of religious marriage. In England and Wales, marriage under the rites of the Church of England will not be available to same-sex couples (s.1 (2) Marriage (Same-Sex Couples) Act 2013), while other religious bodies are required to 'opt-in' if they wish to perform same-sex marriages (s.5 Marriage (Same-Sex Couples) Act 2013). Similarly, in Scotland the legislation requires religious bodies to 'opt-in' (s.12 Marriage and Civil Partnership (Scotland) Act 2014 amends s.8 Marriage (Scotland) Act 1977) to solemnising same-sex marriage. Therefore, in both jurisdictions, those religious bodies who do not wish to allow same-sex marriage are not obligated to

of this legislation epitomises the remarkable transformation of the legal approach to same-sex relationships in the UK.⁹³ This is especially notable given the governmental⁹⁴ and judicial⁹⁵ language used about homosexuality and same-sex relationships as recently as 25-35 years ago, as well as social attitudes towards same-sex relationships at that time.⁹⁶ The legislation opening marriage to same-sex couples follows⁹⁷ the ad hoc extension of various statutory provisions to same-sex couples⁹⁸ and the legal recognition of same-sex relationships through the Civil Partnership Act 2004.⁹⁹

Some commentators have described the transformative potential¹⁰⁰ of the inclusion of same-sex couples in marriage on the institution itself.¹⁰¹ Their contention is that

solemnise same-sex marriages (s.2 Marriage (Same-Sex Couples) Act 2013 and s.12 Marriage and Civil Partnership (Scotland) Act 2014). The Religious Society of Friends (Quakers), the Unitarian Church, the Movement for Reform Judaism and Liberal Judaism all supported the legislative approach and will perform same-sex marriages.

⁹² In Scotland, through the Marriage and Civil Partnership (Scotland) Act 2014 and in England and Wales, through the Marriage Same-Sex Couples Act 2013, which states in s.1 (1), 'Marriage of same sex couples is lawful'.

⁹³ Given that consensual sexual intercourse between men was only decriminalised within the past 50 years.

Most infamously in s.28 of the Local Government Act 1988 which prohibited Local Authorities from promoting homosexuality and described it as a 'pretended family relationship'.

⁹⁵ See e.g. *Re D (An Infant) (Adoption: Parent's Consent)* [1977] AC 602, at 623, where Lord Wilberforce spoke of, 'homosexual tendencies, or of any similar abnormal conduct'. See further the child custody cases involving lesbian mothers, which will be discussed below in Chapter 6, subsection 6.3.A, 'The Judicial Approach to Lesbian Mothers'.

⁹⁶ In 1987 64% of those surveyed believed that sexual relations between two adults of the same-sex were 'always wrong', in 2012 the figure was only 22% of those surveyed, Park and Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30, at 14-17.

⁹⁷ The approach in the UK follows the 'standard sequence' of legislative steps, described by Kees Waaldijk, 'Standard Sequences in the Legal Recognition of Homosexuality - Europe's Past, Present and Future' (1994) 4 Australasian Gay and Lesbian Law Journal 50, see further e.g. Lisa Glennon, 'Displacing the 'Conjugal Family' in Legal Policy - A Progressive Move' [2005] 17 (2) Child and Family Law Quarterly 141.

⁹⁸ e.g. The succession entitlements under the Rent Act, discussed above in Chapter 2, and the inclusion within those couples who can adopt jointly in s.50 Adoption and Children Act 2002 and s.29 (3) Adoption and Children (Scotland) Act 2007.

⁹⁹ See e.g. Nicholas Bamforth, "The Benefits of Marriage in all but Name"? Same-Sex Couples and the Civil Partnership Act 2004' [2007] 19 (2) Child and Family Law Quarterly 133, for the argument that civil partnership was explicitly designed to replicate the features of opposite-sex marriage.

¹⁰⁰ See e.g. Thomas B. Stoddard, 'Why Gay People Should Seek the Right to Marry' in Mark Blasius and Shane Phelan (eds.), *We Are Everywhere: A Historical Sourcebook of Gay and Lesbian Politics*, (Routledge, 1997), Cheshire Calhoun, *Feminism, the Family, and the Politics of the Closet: Lesbian and Gay Displacement*, (Oxford University Press, 2000) and for a review of some of the literature in this area see Victoria Clarke, 'Lesbian and Gay Marriage: Transformation or Normalisation?' (2003) 13 (4) Feminism and Psychology 519.

the presence of same-sex couples within the institution of marriage could accelerate the aforementioned evolution of the understanding of marriage; 102 because as Harding argues, '[o]nce marriage becomes a formal relationship between two persons (rather than "one man and one woman") the place of gender roles (wife/husband; mother/father) within the family are necessarily disrupted. However, other scholarship has been more cautious about the consequences and implications of same-sex marriage, suggesting that the inclusion of same-sex couples within marriage could instead act to normalise those same-sex relationships towards heteronormative standards. On the basis of recent research, Auchmuty has noted that, '[g]endered roles, albeit modified as a result of feminist successes, often seem to be the only roles younger gays and lesbians recognise. Thus, there exists a tension between this potential for transformation and the lived experiences of (particularly younger) same-sex couples which suggests the continuing power and influence of traditional (heterosexual) understandings.

As discussed above, the perception of a division of gender roles continues to exist within a retained and surviving traditional, heterosexual construction of marriage. ¹⁰⁷ These distinct gender roles, of male 'breadwinner' and female 'homemaker', were used to define appropriate behaviour for men and women in the private sphere of

¹⁽

¹⁰¹ ONS Release, 'Marriages in England and Wales (Provisional), for Same Sex Couples Q1 and Q2 2014', (August 2014), states that between March 29th and June 30th 2014 1,409 marriages of same-sex couples took place within England and Wales, therefore suggesting that so far same-sex marriages have made up a small amount of all marriages, available at http://www.ons.gov.uk/ons/rel/vsob1/marriages-in-england-and-wales--provisional-/for-same-sex-couples-q1-and-q2-2014/index.html.

To See above at subsection 4.1.B, 'Is the Understanding of Marriage Changing?'

Rosie Harding, 'Sir Mark Potter and the Protection of the Traditional Family: Why Same Sex Marriage is (Still) a Feminist Issue' (2007) 15 (2) Feminist Legal Studies 223, at 232.

Rosie Harding, 'Recognising (and Resisting) Regulation: Attitudes to the Introduction of Civil Partnership' (2008) 11 (6) Sexualities 740, at 748, notes that, '[t]he creation of a legal framework for same-sex relationships becomes a mapping out of how things are supposed to go – two people meet, fall in love, enter into a civil partnership, become more emotionally mature, become financially dependent on each other, and so on – mirroring heteronormative discourse around societal expectations of marriage and life.'

¹⁰⁵ Rosemary Auchmuty, 'Dissolution or Disillusion: The Unravelling of Civil Partnerships' in Nicola Barker and Daniel Monk (eds.), *From Civil Partnership to Same-Sex Marriage: Interdisciplinary Reflections*, (Routledge, 2015), at 216.

¹⁰⁶ Charlotte Bendall, 'A Break Away from the (Hetero)norm?: *Lawrence v Gallagher* [2012] 1 FCR 557; [2012] EWA Civ 394' (2013) 21 (3) Feminist Legal Studies 303, considers the influence of such heteronormative ideals on the first Court of Appeal decision regarding the dissolution of a civil partnership.

¹⁰⁷ See above at subsection 4.1.A.1, 'The Judicial Understanding of Marriage'.

marriage, the home and the family. 108 In contrast, there was an understanding that same-sex couples were not so constrained by the pressure to replicate these socially constructed roles and thus were able to develop more individualised conceptions of the roles within their own families. 109 Diduck has previously observed that 'Illesbian women and gay men...create their identities outside the norm, without the constraints it imposes on gender expectations and gender practices, arguably rendering their partnerships and familial identities freely chosen in a way that would be impossible for heterosexual individuals.'110 However, more recent empirical research suggests that the traditional gendered roles are exerting influence upon younger same-sex couples in construction of their identities. Heaphy, Smart and Einarsdottir observe that younger same-sex couples 'have a strong sense of the ordinariness of same-sex relationships, and Auchmuty cautions that, '[a]t worst, what we may see (and my findings suggest are already seeing) is same-sex couples behaving "just like" heterosexuals because these are the only spaces they can occupy that will be recognised and protected by law.'112

Thus, there seems to be some evidence that the formal equality granted by the legal recognition and regulation of same-sex relationships, 113 as well as the more positive societal attitudes towards same-sex relationships, 114 has shaped the identities of the

¹⁰⁸ See Chapter 3 above where this issue is explored in some detail, see further e.g. the Canadian Law Commission Report, 'Beyond Conjugality: Recognising and Supporting Close Adult Personal Relationships' (December 2001), at 16, which observes, '[c]lose personal relationships between women and men have been and still are marked by unequal distributions of income, wealth and

power.' Available at - http://www.samesexmarriage.ca/docs/beyond_conjugality.pdf.

See e.g. Jeffrey Weeks, Brian Heaphy and Catherine Donovan, Same Sex Intimacies: Families of Choice and Other Life Experiences, (Routledge 2001). Moreover, research on the parenting practice of lesbian families suggests that the traditional gender roles are not replicated by the parenting of lesbian couples, see e.g. Gabb, 'Lesbian M/Otherhood: Strategies of Familial-linguistic Management in Lesbian Parent Families' and Dunne, 'Opting into Motherhood: Lesbians Blurring the Boundaries and Transforming the Meaning of Parenthood and Kinship'. This research is considered further below in Chapter 6, section 6.3, 'The Role of Lesbian 'Mothers' or 'Parents''.

¹¹⁰ Diduck, *Law's Families*, at 30.

¹¹¹ Brian Heaphy, Carol Smart and Anna Einarsdottir, Same Sex Marriages: New Generations, New Relationships, (Palgrave MacMillan, 2013), at 4.

¹¹² Rosemary Auchmuty, 'The Experience of Civil Partnership Dissolution: Not 'Just Like Divorce" (2016) 38 (2) Journal of Social Welfare and Family Law 152, at 172.

¹¹³ For an examination of the role of formal and substantive equality in the debates on same-sex marriage, see e.g. Robert Leckey, 'Must Equal Mean Identical? Same-sex Couples and Marriage' (2014) 10 (1) International Journal of Law in Context 5.

114 See e.g. Park and Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and

Parenthood', British Social Attitudes Survey 30, at 14-17.

younger generation of same-sex couples towards the 'ordinary' and an understanding of a closer equivalence with their opposite-sex peers. The traditional construction of marriage with its separate gendered roles retains significant normative authority within both legal and cultural understandings and these traditional (heteronormative) ideals and values appear to be influencing younger same-sex couples construction of their own relationships. As Heaphy, Smart and Einarsdottir observe, '[c]ompared to previous generations of same-sex relationships as reported by a number of studies, the younger couples we studied appeared to be more actively invested in convention than in radical relational experimentation.' 117

It is submitted that as a consequence of the continuing significance of the traditional 'natural' construction of marriage, a fundamental transformation of the legal and cultural understanding of marriage remains an aspiration. Moreover, given the influence of these traditional roles upon the attitudes and identities of younger same-sex couples the extent to which the inclusion of same-sex couples within marriage can radically alter the institution of marriage itself remains unclear. Thus, it appears that tension will continue between the transformative potential of the inclusion of same-sex couples within marriage and the rhetorical power of the traditional understanding of marriage to normalise same-sex relationships towards heteronormative standards and roles.

4.2. The Centrality of Conjugality

As discussed above, in spite of the decrease in the number of marriages taking place in the UK, ¹¹⁸ marriage remains central to the legal understanding of the 'family'. ¹¹⁹ I

-

¹¹⁵ Auchmuty, 'Dissolution or Disillusion: The Unravelling of Civil Partnerships' in Barker and Monk (eds.), *From Civil Partnership to Same-Sex Marriage: Interdisciplinary Reflections*, at 203, observes that, '[y]ounger lesbians, like gay men, have no shared memory of the oppressiveness of marriage: marriage for them is simply a lifestyle choice. Similarly, younger gay men have no shared memory of outlaw status. Both sexes act upon the confident, if sometimes misplaced assumption that their relationships will be "equal" with those of heterosexuals.'

As discussed above at subsection 4.1.B, 'Is the Understanding of Marriage Changing?'

Heaphy, Smart and Einarsdottir, *Same Sex Marriages: New Generations, New Relationships*, at 4. ONS, 'Marriages in England and Wales (Provisional), 2012', at 2.

¹¹⁹ As discussed above at subsection 4.1.B, 'Is the Understanding of Marriage Changing?'

will argue that legal regulation is extended to other cohabiting, interdependent, monogamous, sexual relationships on the basis that those relationships are regarded as sufficiently 'marriage-like' because they share enough of the features and characteristics traditionally associated with marriage. In drawing comparisons with marriage, I will suggest that emphasis is placed upon one particular feature of the relationship; the existence of a conjugal or sexual element. ¹²⁰ I will contend that the centrality of conjugality (understood as the combination of a sexual relationship, living together and interdependence) within the legal regulation of adult relationships illustrates the continuing influence of the nuclear family upon the legal understanding of the family, because relationships are regulated on the basis that they can be located within the central nexus of that nuclear family.

In this section, I will begin by considering some of the justifications for the legal regulation of cohabitation, before arguing that cohabitation is being regulated on the basis that it is a 'marriage-like' relationship (subsection 4.2.A). I will then explore the justification for the centrality of conjugality to the legal regulation of adult personal relationships (subsection 4.2.B), arguing that the conjugal couple form is understood as providing greater social stability particularly in regard to the upbringing of children (subsection 4.2.B.1). I will conclude by contending that this emphasis on stability reflects the central nexus of the conjugal relationship and the parent/child relationship within the nuclear family model.

4.2.A. The Legal Regulation of Cohabitation

Over the past 50 years there has been a significant demographic shift, away from the

-

¹²⁰ The centrality of conjugality to the regulation of other adult relationships is noteworthy, given that conjugality is not a requirement within marriage itself. In English Law, marriage merely requires to be consummated to be considered valid; marriage is voidable if it has not been consummated due to both 'incapacity' s.12 (a) Matrimonial Causes Act 1973 and 'wilful refusal' s.12 (b) Matrimonial Causes Act 1973, see e.g. *Clarke v Clarke* [1943] 2 All ER 540 and *R v R* [1952] 1 All ER 1194. In Scots Law it is the *potential* for conjugality that is important, as a marriage is voidable on the ground of incurable impotency, see e.g. *L v L* 1931 SC 477 and *J v J* 1978 SLT 128. Therefore, in neither jurisdiction does marriage need to involve an ongoing sexual relationship. However, I argue that the dominant understanding of marriage is premised upon a conjugal relationship subsisting between the parties.

dominance of marriage and toward an increase in unmarried cohabitation. 121 Once a small minority, cohabiting couples (and their children) now make up around 15% of families in the UK. 122 Over the same period there has been a corresponding increase in the social acceptance of unmarried cohabitation. ¹²³ Barlow and James suggest that 'the social and cultural norms which made people feel that they had or ought to get married in order to be accepted as a decent member of society, seem to have disappeared. Cohabitation is now a perfectly acceptable family form'. 124 The change in the cultural understanding of cohabitation is apparent from the language of the government, which now endorses a range of family forms. This is exemplified by the previous coalition's 'Programme for Government', which states: 'The Government believes that strong and stable families of all kinds are the bedrock of a strong and stable society.'125 In spite of this rise in cohabitation, the law in the UK was initially reluctant to acknowledge and regulate these relationships. 126 Indeed, even after such acknowledgment, the subsequent legal responses to cohabitation have been varied. ranging from the inclusion of cohabitants within certain statutory rights 127 on an ad hoc basis, ¹²⁸ through to essentially ignoring the relationship and treating the parties to it as (legal) strangers, ¹²⁹ to the creation in Scotland of a statutory regime

1/

¹²¹ ONS, 'Short Report: Cohabitation in the UK, 2012', at 1.

ONS, 'Social Trends: Households and Families' (2011) No. 41, at 7.

Park and Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30, at 7-14.

¹²⁴ Barlow and James, 'Regulating Marriage and Cohabitation in 21st Century Britain', at 160, see also Auchmuty, 'Law and the Power of Feminism: How Marriage Lost its Power to Oppress Women', who goes further, suggesting, at 81, '[t]he tipping point has been reached; not only has cohabitation ceased to be a preparation for marriage, and become an alternative to it, but it will soon be the *preferred* alternative.'

¹²⁵ 'The Coalition: Our Programme for Government', (May 2010), at 19.

See e.g. the evolution of judicial attitudes concerning the inclusion of cohabitants within the definition of 'family' under the Rent Acts, between the 1950s and the 1970s, discussed above in Chapter 2, subsection 2.2.A.4, 'Unmarried Cohabitants'.

See e.g. Child Support Act 1991, Part IV Family Law Act 1996, s.1 (3) (b) Fatal Accidents Act 1976 and s. s.18 Matrimonial Homes (Family Protection) (Scotland) Act 1981.

¹²⁸ Rebecca Probert, 'Cohabitation in Twentieth Century England and Wales: Law and Policy' (2004) 26 (1) Law and Policy 13, at 23, observes that in the 1970s, '[p]arliament was beginning to include cohabitants within the scope of legislation, but in a very ad hoc fashion.'

As occurred in relation to the absence of financial provision upon relationship breakdown in England and Wales, see e.g. *Burns v Burns* [1984] Ch. 317, where May LJ stated, at 334, 'the resolution of these disputes must depend upon the ascertainment according to normal principles of the respective property rights between the man and the woman.' Although this position has been altered by the more recent judicial developments regarding constructive trusts, see *Stack v Dowden* [2007] 1 FLR 1858 and *Jones v Kernott* [2012] 1 FLR 45.

The inclusion of cohabitants within various statutory provisions suggests recognition that cohabiting couples should be subject to some legal regulation. Scots law differs from English law on whether such regulation should take the form of a statutory regime specifically for cohabitants. ¹³¹ A significant reason for the lack of consensus is that, as Baroness Hale observed in *Stack v Dowden*, ¹³² '[c]ohabitation comes in many different shapes and sizes.' ¹³³ Such relationships range from those where the parties have actively chosen to reject marriage in order not to subject their relationships to legal regulation, ¹³⁴ to those relationships where the parties erroneously believe themselves to have formed a 'common law marriage', ¹³⁵ to those couples that view their cohabitation as a temporary or transitional arrangement, which could be converted to a marriage at an (as yet undetermined) future point. ¹³⁶ Consequently, Chan observes that '[t]he understandings that underpin de facto partnerships find no natural, unitary expression to be captured by law.' ¹³⁷

Given this diversity of relationships, on what basis does the law seek to regulate

¹³⁰ s.25-30 Family Law (Scotland) Act 2006.

¹³¹ The Law Commission Report, 'Cohabitation: The Financial Consequences of Relationship Breakdown', (Law Com No 307, July 2007), available at - https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/228881/7182.pdf, proposed a statutory regime for England and Wales similar to the Scottish approach. However, there are currently no plans from the government to act upon the recommendations of the report. In a written Ministerial Statement on the 6th September 2011, the Parliamentary Under-Secretary of State, Ministry of Justice, Jonathan Djanogly, stated that, '[w]e do not therefore intend to take forward the Law Commission's recommendations for reform of cohabitation law in this parliamentary term.'

⁽Hansard HC Col 16WS). ¹³² [2007] 1 FLR 1858.

ibid, per Baroness Hale, at 1874, see further Lord Hoffman in *Re P (Adoption: Unmarried Couples)* [2008] 2 FLR 1084, at 1090, who stated of cohabitation, these, 'relationships may vary from quasi-marital to ephemeral.'

¹³⁴ Anne Barlow, Carole Burgoyne, Elizabeth Clerly and Janet Smithson, 'Cohabitation and the Law: Myths, Money and the Media', British Social Attitudes Survey 24, (SAGE, 2008), at 33, observe that 31% of those currently cohabiting had previously been married, compared to 17% of those now married and suggests that, 'in some instances, cohabitation might be a refuge for those disillusioned by previous experiences of marriage.'

¹³⁵ See Rebecca Probert, 'Common-law Marriage: Myths and Misunderstandings' [2008] 20 (1) Child and Family Law Quarterly 1, for a consideration of the development of a common law marriage myth within law.

Barlow, Burgoyne, Clerly and Smithson, 'Cohabitation and the Law: Myths, Money and the Media', British Social Attitudes Survey 24, at 33, observed that, in 2006, 56% of cohabiting relationships ended in marriage.

¹³⁷ Chan, 'Cohabitation, Civil Partnership, Marriage and the Equal Sharing Principle', at 55.

cohabitation? I will argue that while the law is influenced by both changes in social attitudes and a paternalistic desire to protect vulnerable (female) cohabitants, the central justification for legal regulation of cohabitation is acknowledging that cohabitation is a 'marriage-like' relationship, based primarily upon the shared conjugality which features in these different types of relationship.

The aforementioned changes in British social attitudes have resulted in a far closer equivalence being drawn between marriage and cohabitation than was apparent historically. 138 Eekelaar and MacLean have observed, '[o]ur evidence shows that married and unmarried people who are living together share many values. Indeed, the similarities in the normative determinants of their behaviour may be greater than the dissimilarities.'139 The dominant cultural understanding now envisages significant similarities between the two relationships and their functions. 140 These shifts have been, and continue to be, a significant factor supporting further legal regulation of cohabitation.¹⁴¹

The normative resonance of the unfairly treated, 'deserving' cohabitant (the 'Mrs Burns figure')¹⁴² has resulted in a paternalistic motive for legal regulation; to provide some protection for economically vulnerable cohabitants. This is visible in the Scottish Law Commission Report, 143 upon which the statutory regime for cohabitants was based, which stated that the proposed regime 'should be confined to the easing of certain legal difficulties and the remedying of certain situations which are widely

¹³⁸ Anne Barlow, 'Cohabitation Law Reform - Messages from Research' (2006) 14 (2) Feminist Legal Studies 167, at 173, observes that, 'the social norms whereby most people use marriage and cohabitation as a personal lifestyle choice have diverged from the legal norms which continue to privilege marriage over cohabitation in many significant ways, although there is no real understanding of this by the British public, it seems.'

¹³⁹ Eekelaar and MacLean, 'Marriage and the Moral Bases of Personal Responsibility', at 538.

Park and Rhead, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30, at 9, observe that only 42% of respondents agree with the statement, 'people who want children ought to get married', this figure is down from 71% in 1989. ¹⁴¹ See e.g. The Law Commission Report, 'Cohabitation: The Financial Consequences of Relationship Breakdown' (2007), Para 1.6, at 2, which states, '[c]ohabitation outside marriage in England and Wales has become increasingly common over recent decades, and is expected to become more prevalent in the future.'

142 Burns v Burns [1984] Ch. 317, in the course of his judgment, May LJ stated, at 345, 'I think that

she can justifiably say that fate has not been kind to her.'

¹⁴³ Scottish Law Commission, 'Report on Family Law', (No 135, May 1992), available at http://www.scotlawcom.gov.uk/files/5912/8015/2668/Report%20on%20family%20law%20Report%2 0135.pdf.

perceived as being harsh and unfair.' The emphasis on vulnerable cohabitants 145 is supported by the existence of a long-standing belief in 'common-law marriage' amongst the general public. 146 The continuing prevalence of this belief in common law marriage results in many cohabiting couples incorrectly believing that their relationship will, at some stage, acquire the rights and obligations associated with a marriage. 147 Thus, the existence of these common societal misconceptions regarding common law marriage provides additional paternalistic motivations for the legal regulation of cohabitation.

4.2.A.1. Cohabitation as a 'Marriage-Like' Relationship

Wong argues that the legal regulation of cohabitation involves 'a stretching of the marriage model to accommodate the inclusion of other close personal relationship...The basis for inclusion of other types of close personal relationships such as opposite-sex and same-sex cohabitation must thus be based on similarity or sameness.' Therefore, I suggest that the law primarily regulates cohabitation on the basis that cohabiting couples share sufficient features and characteristics with married couples and therefore can be located within the central nexus of the nuclear family. In this regard, legislation often defines cohabitation by comparison with

¹⁴⁴ ibid, Para 16.1, at 115, see further the Law Commission Report 'Cohabitation: The Financial Consequences of Relationship Breakdown', (2007), Para 1.4, at 2, which similarly proposed a regime that, 'would provide economically vulnerable members of society with the private means to rebuild their lives and ensure a fairer division of assets on relationship breakdown.'

¹⁴⁵ Such language was also apparent in a written ministerial statement regarding the Bill which subsequently became the Family Law (Scotland) Act 2006, Deputy Minister for Justice, Hugh Henry, 'Response to Justice 1 Committee Stage 1 Report', (August, 2005), at 14, stated that, 'the law should safeguard the interests of those in cohabiting relationships who may be vulnerable', available at - http://www.scottish.parliament.uk/S2_Justice1Committee/Reports/SEresponseFamilyLawStage1.pdf.

Fran Wasoff and Claudia Martin, 'Scottish Social Attitudes Survey 2004: Family Module Report', (August 2005), at 41, found that 51% of respondents believed that some form of common law marriage existed in Scotland, available at - http://www.scotland.gov.uk/Resource/Doc/57346/0016386.pdf. See further Barlow, Burgoyne, Clerly and Smithson, 'Cohabitation and the Law: Myths, Money and the Media', British Social Attitudes Survey 24, who also found 51% of respondents believed that common law marriage existed in England and Wales.

¹⁴⁷ Prior to its abolition by s.3 Family Law (Scotland) Act 2006, in Scotland an 'irregular marriage' could be constituted through marriage by cohabitation with habit and repute, therefore the existence of this doctrine may have contributed to the continued prevalence of the common law marriage myth in Scottish society.

¹⁴⁸ Simone Wong, 'Cohabitation and The Law Commission's Project' (2006) 14 (2) Feminist Legal Studies 145, at 151.

marriage, as is exemplified by the Family Law (Scotland) Act 2006, which defines opposite-sex¹⁴⁹ cohabitants as, 'a man and a woman who are (or were) living together as if they were husband and wife'. Thus, cohabitation is regulated because it is constructed and understood as being a 'marriage-like' relationship¹⁵¹ on the basis of its conjugality. The same section of the basis of its conjugality.

Any legal recognition of cohabitation is premised on the basis that, as Bottomley argues, 'by bringing cohabitants into family law, a modified marriage law regime, law and social policy might be able to find ways of stabilising domestic units through a reinforcement of family values carried through the application of a marriage (like) model.' The law is able to bring cohabiting couples within the scope of its (nuclear) understanding of family by constructing cohabitation as a 'marriage-like' relationship. Thus, the regulation of cohabiting couples shows that the traditional, nuclear family remains central to the legal understanding of the family.

¹⁴⁹ A similar approach is used in relation to same-sex cohabitation, in s.25 (1) (b), which is defined as, 'two persons of the same sex who are (or were) living together as if they were civil partners', this definition ceases to have effect as a result of s.4 (4) Marriage and Civil Partnership (Scotland) Act 2014 and is replaced by 'living together as if they were married to each other'. ¹⁵⁰ s.25 (1) (a) Family Law (Scotland) Act 2006, for examples of similar definitions from English Law,

see e.g. s.1 (3) (iii) Fatal Accidents Act 1976, s.62 (1) Family Law Act 1996 and s.1 (1A) (b) Inheritance (Provision for Family and Dependants) Act 1975.

¹⁵¹ Barlow and James, 'Regulating Marriage and Cohabitation in 21st Century Britain', suggest, at 153, 'cohabitation...has taken on the functions of marriage. Conversely, the institution of marriage itself has also changed, perhaps becoming more akin to unmarried cohabitation.'

However, it is worth noting that the law still does not envisage a direct comparison between the two relationships, instead acknowledging that cohabitation shares enough characteristics with marriage to deserve some (lesser) recognition. This distinction is evident from Scottish Law Commission, 'Report on Family Law', (No 135, May 1992), Para 16.1, at 115, which states that the proposals should not, 'undermine marriage', as well as the Law Commission Report 'Cohabitation: The Financial Consequences of Relationship Breakdown', (2007), Para 1.2, which states, at 1, '[t]he scheme would not equate cohabitants with married couples or give them equivalent rights.' As such cohabitation remains understood within the law as an inferior relationship to marriage, which remains the pinnacle or 'gold standard' of legally regulated conjugal relationships. See further Simone Wong, 'Cohabitation and The Law Commission's Project', who comments, at 151, 'a distinction is still being drawn between marriage and cohabitation, which emphasises the centrality of commitment. It is, however, not just the notion of commitment but the *type* of commitment that is critical.'

Anne Bottomley, 'From Mrs Burns to Mrs Oxley: Do Co-Habiting Women (Still) Need Marriage Law' (2006) 14 (2) Feminist Legal Studies 181, at 206.

4.2.B. The Justification for the Centrality of Conjugality

Conjugality is central to legal categorisation of relationships as 'marriage-like'. ¹⁵⁴ Barker claims that, '[b]y continuing to legally privilege marriage and other conjugal relationships, the state is reinforcing this romantic mystique, drawing a boundary around sexual relationships as the "ideal". ¹⁵⁵ Through the centrality of the conjugal relationship, the ideals and values of marriage and the traditional nuclear family continue to be promoted. If an adult relationship does not include at least the possibility of conjugality or a sexual partnership, ¹⁵⁶ it will be situated outside the boundaries of the law's conception of family. ¹⁵⁷ Wong observes that, 'cohabitation is being distinguished from other (inter)dependent relationships and a key element of that distinction is that the parties' interdependence stems from being in a relationship involving commitment and intimacy. ¹⁵⁸

Non-conjugal relationships 159 that have sometimes been suggested as deserving of

1.

¹⁵⁴ See e.g. Crake v Supplementary Benefits Commission [1982] 1 All ER 498, Kimber v Kimber [2000] 1 FLR 383, Garrad v Inglis 2014 GWD 1-17 and M v T 2011 GWD 40-828, for judicial consideration of the influence of conjugality within the definition of cohabitation, under both English and Scots law.

¹⁵⁵ Nicola Barker, 'Sex and the Civil Partnership Act: The Future of (Non) Conjugality' (2006) 14 (2) Feminist Legal Studies 241, at 248.

This arises because the traditional definition of adultery is retained by the legislation, Schedule 4 Part 3 Marriage (Same-Sex Couples) Act 2013 amends s.1 Matrimonial Causes Act 1973, providing that, '[o]nly conduct between the respondent and a person of the opposite sex may constitute adultery', see also s.5 (2) Marriage and Civil Partnership (Scotland) Act 2014, which amends s.1 Divorce (Scotland) Act 1976, inserting a similar provision. Furthermore, the other statutory provisions which relate to the importance of sexual activity within marriage are simply not extended to same-sex marriages; in England and Wales Schedule 4 Part 4 Marriage (Same-Sex Couples) Act 2013 amends s.12 Matrimonial Causes Act 1973 and states that the provisions regarding non-consummation, 'do not apply to the marriage of a same sex couple' and in Scotland s.5 (1) Marriage and Civil Partnership (Scotland) Act 2014, states, '[f]or the avoidance of doubt, the rule of law which provides for a marriage to be voidable by reason of impotence has effect only in relation to a marriage between persons of different sexes.' For a critique of the legislative approach to these issues, which suggests that this approach results in the invisibility of homosexual sex, see e.g. Lucy Crompton, 'Where's the Sex in Same-Sex Marriage?' [2013] 43 (5) Fam. Law 564.

e.g. s.2 Marriage (Scotland) Act 1977 prohibits marriages between people who are related to each other within the 'forbidden degrees' of relationships, which are specified in Schedule 1 of the Act, for the similar English law rule see s.11 (a) (i) Matrimonial Causes Act 1973.

¹⁵⁸ Simone Wong, 'Shared Commitment, Interdependency and Property Relations: A Socio-Legal Project for Cohabitation' [2012] 24 (1) Child and Family Law Quarterly 60, at 61.

¹⁵⁹ Under the Rent Acts non-conjugal adult relationships were consistently held to be outside the 'de facto familial nexus' required to be considered a 'family member', see the decisions in *Ross v Collins* [1964] 1 WLR 425, *Joram Developments Ltd. v Sharratt* [1979] 1 WLR 928 and *Sefton Holdings v Cairns* [1988] 2 FLR 109, considered above in Chapter 2, subsection 2.2.A.3, 'The Role of the 'De

legal recognition¹⁶⁰ include those between adult siblings who live together¹⁶¹ and relationships between residential carers and the person cared for.¹⁶² Herring has argued 'that what might make a relationship worthy of promotion by the state is care and mutual support, rather than sex.'¹⁶³ However, regarding the relationship between adult siblings, in *Burden v UK*¹⁶⁴ the European Court of Human Rights observed that, 'the relationship between siblings is qualitatively of a different nature to that between married couples and homosexual civil partners...The fact that the applicants have chosen to live together all their adult lives...does not alter this essential difference between the two types of relationship.'¹⁶⁵ While such relationships share some of the characteristics of those that are regulated and may be deserving of some form of legal recognition, ¹⁶⁶ the judgment proceeds on the basis that there is a 'qualitative' or 'essential' difference which limits the scope for comparable legal recognition.¹⁶⁷ Such relationships, because of their inherent and essential lack of conjugality, cannot be located within the central nexus of the nuclear family model¹⁶⁸ and as a consequence these relationships are categorised differently by the law than conjugal

Facto Familial Nexus".

¹⁶⁰ See e.g. the Canadian Law Commission Report, 'Beyond Conjugality: Recognising and Supporting Close Adult Personal Relationships' (December 2001), which suggested a radical restructuring of the regulation of adult personal relationships in Canada, moving away completely from the centrality of conjugality, however these recommendations were not enacted by the Canadian government.

During debates on the Civil Partnership Act 2004, the image of the 'spinster sisters' was invoked repeatedly by opponents of the Bill.

¹⁶² See further e.g. Brian Sloan, *Informal Carers and Private Law*, (Hart, 2012) and Beverley Clough, 'What About Us? A Case for Legal Recognition of Interdependence in Informal Care Relationships' (2014) 36 (2) Journal of Social Welfare and Family Law 129.

¹⁶³ Herring, Caring and the Law, at 192.

¹⁶⁴ (2008) 47 EHRR 38, this case involved an unsuccessful challenge, under Article 1 of Protocol 1 in conjunction with Article 14, by two elderly sisters who had lived together for their entire lives, of their exclusion from the spousal/civil partner inheritance tax exemption. See further e.g. Rosemary Auchmuty, 'Beyond Couples' (2009) 17 (2) Feminist Legal Studies 205.

¹⁶⁵ ibid. at 876.

¹⁶⁶ Barker, 'Sex and the Civil Partnership Act: The Future of (Non) Conjugality', comments, at 255, that '[i]t may be that certain parts of the law relating to marriage and divorce would be helpful for many types of relationships not currently recognised, even though other parts of marriage might be deeply problematic.'

¹⁶⁷ Interestingly in *Morgan v Morgan* [1959] P 92, Mr Commissioner Latey QC, at 101, observed that, '[m]any elderly and aged people intermarry on the basis, implied or agreed upon, that they come together merely for companionship and without any thought of sexual relations.' This statement appears to suggest a judicial acknowledgement that some marriages may be formed without any intention of conjugality, although such marriages would remain voidable in terms of s.12 Matrimonial Causes Act 1973.

¹⁶⁸ Baroness Ruth Deech, 'Sisters Sisters - And Other Family Members' [2010] 40 (4) Fam. Law 375, at 380, suggests that, '[t]hus does English law continue to prefer the idle sexual partner over the deserving family member.'

4.2.B.1. The Central Nexus: 'Stability' and the Conjugal Couple Form

What then justifies the centrality of the conjugal relationship within the law? I argue that the conjugal couple form is associated with 'stability', which is understood by government as a social value worth promoting, 170 both in terms of the relationship itself and in relation to providing a stable and secure upbringing for children. This association of the conjugal couple and children reflects the central nexus between those two relationships within the nuclear family.

This construction is based upon a particular understanding of the purpose of marriage in society, which is premised upon the historical conception that marriage, and the nuclear family, provided significantly more social stability than other, alternative, family forms. The As Herring notes, '[i]f a person falls ill, or becomes unemployed, and so no longer has an income, then the financial responsibility is likely to fall on the state if that person is single, whereas spouses or civil partners would depend on each other. Judicial reliance upon similar ideas about the significance of marriage was evident as recently as the judgment of Lord Hoffmann in *Re P (Adoption: Unmarried Couples)*, who have accepted a legal commitment to each other, tend to have more stable relationships than unmarried couples. However, such stability is no longer exclusively associated with marriage; the law now recognises that stability can also be a feature of long-term, cohabiting, conjugal relationships which are non-marital. In her concurring judgment in *Re P*, Baroness Hale acknowledged that, '[s]ome

¹

¹⁶⁹ See e.g. Carl Stychin, 'Not (Quite) a Horse and Carriage' (2006) 14 (1) Feminist Legal Studies 79, who commented, in the context of the debates around civil partnership, at 81, that, 'if the state is going to recognise relationship forms outside the institution of marriage, then it should take the opportunity to consider real alternatives to the marriage model, that might be available more widely; a model in which conjugality might be deprivileged.'

¹⁷⁰ See e.g. 'The Coalition: Our Programme for Government', (May 2010), at 19.

¹⁷¹ See e.g. *Campbell v Campbell* [1977] 1 All ER 1.

¹⁷² Herring, Family Law, at 122.

¹⁷³ [2008] 2 FLR 1084.

¹⁷⁴ ibid, per Lord Hoffmann, at 1090.

unmarried relationships are much more stable than some marriages, and vice versa.'175 The law's understanding of family is now organised around the conjugal couple form (rather than marriage specifically), which, I submit, is constructed as performing the same central role that was historically exclusive to marriage. As Stychin observes, '[t]he stable couple form, it is argued, is good for the individual, for the couple and for society (and the economy) as a whole. Long-term, traditional, stable, legally recognised relationships thus become the socially preferred option.¹⁷⁶

Related to this is the emphasis placed on the stable, conjugal couple form and the nuclear family as the environment best suited to achieving the positive development of children, which illustrates the importance of the central nexus between the conjugal relationship and the parental/child relationship. Traditionally, it was marriage that was positioned as providing this stable environment for children. ¹⁷⁷ A preference for marriage was evident in the language of government as recently as 1998, when a government consultation stated that 'children need stability and security...marriage is still the surest foundation for raising children and remains the choice of the majority of people in Britain.' This understanding was also reflected in judicial language; writing extra-judicially, Elizabeth Butler-Sloss observed that '[o]ne most important aspect of a child's welfare is the need to grow up in a secure and stable environment and to be loved and cared for...The most secure family is achieved through marriage.'179 This conception was exemplified by the judgment of Sir Mark Potter P in Wilkinson v Kitzinger (No. 2), 180 who describes marriage 'as a means not only of encouraging monogamy but also the procreation of children and their development and nurture in a family unit (or "nuclear family"). 181 Therefore, I

http://webarchive.nationalarchives.gov.uk/+/http://www.nationalarchives.gov.uk/ERORecords/HO/421 /2/P2/ACU/SUPPFAM.HTM.

179 Elizabeth Butler-Sloss, 'Ethical Considerations in Family Law' (2006) Web Journal of Current

¹⁷⁵ ibid, per Baroness Hale, at 1121.

¹⁷⁶ Carl Stychin, 'Family Friendly? Rights, Responsibilities and Relationship Recognition?' in Diduck and O'Donovan (eds.), Feminist Perspectives on Family Law, at 24.

¹⁷⁷ In the 19th century ecclesiastical courts, in *D-e v A-g* (1845) Rob Ecc. 279, Dr Lushington, at 281, described the 'two principal ends of matrimony' as 'a lawful indulgence of the passions to prevent licentiousness, and the procreation of children, according to the evident design of Divine Providence.'

^{178 &#}x27;Supporting Families: A Consultation Document' (Home Office, December 1998), at Para 8, available

Legal Issues 4. ¹⁸⁰ [2007] 1 FLR 295.

¹⁸¹ ibid, per Sir Mark Potter P, at 329.

agree with Sheldon's description that, '[t]he importance of providing a "stable" base for children is frequently asserted, with "stability" standing as a shorthand for heterosexual marital monogamy. 182

However, the law has long recognised that the procreation of children, or even the possibility of such procreation, is not required for a marriage to be valid. The judiciary acknowledged this as early as the 1940s, in *Baxter v Baxter*, ¹⁸³ where Viscount Jowitt stated that, '[i]t is indisputable that the institution of marriage generally is not necessary for the procreation of children; nor does it appear to be a principal end of marriage as understood in Christendom'. 184 This has also been reflected in contemporary judicial understanding, as illustrated by the judgment of Baroness Hale in Ghaidan v Godin-Mendoza, 185 where she stated, 'the capacity to bear or beget children has never been a prerequisite of a valid marriage in English law...A marriage, let alone a relationship analogous to marriage, can exist without either the presence or the possibility of children from that relationship.'186

Of course correlation does not prove causation, and as Chan has observed: '[t]here is little evidence to suggest that stability is caused by rather than correlated with marriage.' More recently this 'stable base' has been extended from only referring to marriage to including other conjugal relationships. This expansion of the law has partly resulted from the social shift toward unmarried cohabitation. ¹⁸⁸ The extension to other conjugal relationships has also been supported by evidence from empirical research into children's outcomes, such as that undertaken by Crawford et al, who commented that '[w]e...regard our results as a strong indication that marriage plays a relatively small role, if any, in promoting children's early cognitive or socio-

¹⁸² Sheldon, 'Fragmenting Fatherhood: The Regulation of Reproductive Technologies', at 534, this statement was made in the context of the parliamentary debates on the Human Fertilisation and Embryology Act 1990.

¹⁸³ [1948] AC 274.

ibid, per Viscount Jowitt, at 286, this contrasted with the decision in *D-e v A-g* (1845) Rob Ecc. 279, as to the centrality of procreation to marriage, which had been supported, only 3 years prior to Baxter, by Pilcher J in Cowen v Cowen [1945] 2 All ER 197, who stated, at 199, that the procreation of children was, '[o]ne of the purposes for which marriage was ordained'. ¹⁸⁵ [2004] 2 AC 557.

ibid, per Baroness Hale, at 606.

¹⁸⁷ Chan, 'Cohabitation, Civil Partnership, Marriage and the Equal Sharing Principle', at 52.

¹⁸⁸ ONS, 'Short Report: Cohabitation in the UK, 2012'.

emotional development.' Thus, I suggest that it is having two parents in an ongoing, stable, monogamous, conjugal, cohabiting relationship which is understood by the law as being intrinsically favourable to children, regardless of whether that relationship is formalised as marriage. The clear association between the conjugal couple form and the upbringing of children suggests that it is the central nexus of the nuclear family between the conjugal relationship and the parent/child relationship which remains fundamental to the understanding of family within the law. 190

Herring questions the normative significance granted to conjugality in this context, arguing, '[w]hether couples are same-sex or opposite sex; married or not; sexual or not, does not, I suggest affect the contribution they make to the social good. They deserve the same support and protection.' However, regardless of this normative argument, it is apparent that the law continues to draw a distinction between relationships on the basis of conjugality and sexual partnership. As Bottomley and Wong comment, '[w]hat these emerging patterns make clear is that the extensions to cover others tend to include the drawing of lines around a central nexus of either marriage or sexual partnership. The legal regulation of non-marital conjugal relationships occurs when these relationships are constructed to some degree as emulating the social functions of marriage and reproducing its characteristics and values. I have argued that the centrality of conjugal relationships is premised upon the perceived value provided by the conjugal couple form regarding social stability

¹⁸⁹ Claire Crawford, Alissa Goodman, Ellen Greaves and Rob Joyce, 'Cohabitation, Marriage and Child Outcomes: An Empirical Analysis of the Relationship between Marital Status and Child Outcomes in the UK using the Millennium Cohort Study' [2012] 24 (2) Child and Family Law Quarterly 176, at 195.

¹⁹⁰ As discussed above in regards to 'family' under the Rent Act in Chapter 2, subsection 2.2.A.3, 'The Role of the 'De Facto Familial Nexus' and in terms of the interpretation of 'family life' in Art. 8 ECHR in Chapter 2, subsection 2.2.B, 'The Definition of 'Family Life' under Article 8 of the European Convention on Human Rights' as well throughout Chapter 2 more generally. ¹⁹¹ Herring, *Caring and the Law*, at 199.

¹⁹² It is not the intention of this thesis to assess whether the centrality of conjugality is the most appropriate normative approach to regulating adult personal relationships, nor whether an approach based upon mutuality, interdependence and 'care' would better serve the needs of a diversity of relationships. Instead this thesis is merely attempting to set out and understand the values that currently underpin the legal regulation of adult relationships.

¹⁹³ As illustrated by the decision of the ECHR in *Burden v UK* (2008) 47 EHRR 38 and by the cases concerning the Rent Acts discussed above in Chapter 2, subsection 2.2.A.3, 'The Role of the 'De Facto Familial Nexus''.

¹⁹⁴ Anne Bottomley and Simone Wong, 'Shared Households: A New Paradigm for Thinking about the Reform of Domestic Property Relations' in Diduck and O'Donovan (eds.), *Feminist Perspectives on Family Law*, at 48.

and because of the value that relationship possesses in relation to securing the healthy and happy upbringing of children, within the central nexus that comprises the nuclear family. Thus, these relationships are now understood as being as capable as marriage of comprising that central nexus of relationships within the nuclear family. On this basis, I submit that the law continues to base its understanding of 'family' around marriage (and 'marriage-like' conjugal relationships) and the traditional nuclear family.

Conclusion

In this chapter, I have considered the legal regulation of adult personal relationships and argued that such regulation is centred upon the conjugal relationship and remains influenced by traditional understandings of marriage and the nuclear family. Therefore, I endorse Diduck and Kaganas' observation that 'law's model is still the private heterosexual traditional family. While the range of people who may be allowed into it may have increased, the norms by which they must live remain the same.' 196

To begin, I considered the legal definition and understanding of marriage; I observed that the traditional understanding of marriage and the associated division of gender roles continues to influence the contemporary understanding of marriage. However, I argued that recently there has been some evidence of a shift in the legal and social understanding of marriage, moving away from elements of the historical definition, as indicated by the recent legislation extending marriage to same-sex couples. Subsequently, I contended that conjugality is the central factor in the regulation of adult relationships. I observed that cohabiting couples are subject to legal recognition and regulation because they are constructed by the law as sharing some of the features and characteristics of marriage (with particular emphasis on conjugality) and are considered to be 'marriage-like' relationships. I submitted that the conjugal

¹⁹⁵ As illustrated above in the context of the Rent Acts, in Chapter 2, subsection 2.2.A.4, 'Unmarried Cohabitants' and 2.2.A.5, 'Same-Sex Cohabitants'.

¹⁹⁶ Diduck and Kaganas, Family Law, Gender and the State, at 23.

couple form is now understood by the law as providing social stability for the parties to the relationship and constructed as the best environment for securing the upbringing and development of children, within the central nexus of the conjugal relationship and the parent/child relationship.

To conclude, I argued that the nuclear family model has been expanded to include both same-sex couples and unmarried cohabitants, *without* any fundamental revision of the traditional, nuclear family itself. The remaining two chapters will examine the influence of the nuclear family (particularly through the traditional gendered parenting roles of 'mother' and 'father') on the other relationship within the central nexus of the nuclear family; the 'parent/child' relationship.

Part 3: Parenthood

Chapter 5: The Attribution of Legal Parenthood within UK Law

Introduction

This chapter will consider the questions: (i) who does the law deem to be the parents of a child? and (ii) what factors form the basis for this determination? These two questions are intertwined and in the course of answering them I will consider the importance of both genetic parentage and social parenthood to the attribution of legal parenthood. Although in many instances the identity of the legal parent(s) may be simple and unproblematic, in many others the answer may not be as straightforward. In these more complex scenarios, I will observe that different factors appear to be determinative of legal parenthood in different factual circumstances.

Given the discussion in the previous chapters, it is unsurprising that the law is influenced by the idealised image of the nuclear family when making determinations of legal parenthood. Thus, legal parenthood generally vests in two people,² traditionally one woman ('the mother') and one man ('the father').³ Jackson argues that this approach is 'the law's principal stumbling block, namely its assumption that a child can have only two legal parents: one mother and one father.⁴ This binary, two-parent approach often requires the law to choose between competing parental claims and pronounce a 'truth' of parenthood, rather than being able to acknowledge

¹ Although both of these types of parenthood may lead to legal parenthood being deemed in certain circumstances, e.g. genetic parenthood is generally now recognised as leading to legal parenthood for fathers in cases of natural reproduction and adoption acts to create legal parenthood for social parents.

While it is possible for a child born of natural reproduction to only have one legal parent, if the father is unknown and not registered on the birth certificate, the provisions of the Human Fertilisation and Embryology Act 2008 allows for the circumstance in which a child only has the *possibility* of one legal parent, i.e. where donor sperm is used by a single woman.

³ Although the law has recently allowed for there to be two legal parents of the same sex in certain contexts: with there being the possibility of two parents of the same sex by virtue of adoption, s.50 Adoption and Children Act 2002 and s.29 Adoption and Children (Scotland) Act 2007, two female parents in the context of assisted reproduction, s.42-s.47 Human Fertilisation and Embryology Act 2008 (which will be considered in more detail below at subsection 5.2.B, 'Non-Gestational Female 'Parents') and two male parents in the context of surrogacy, s.54 HFEA 2008.

⁴ Jackson, 'What is a Parent?', in Diduck and O'Donovan (eds.), Feminist Perspectives on Family Law, at 59.

the value and status of multiple, different relationships.⁵ Bainham has suggested that, '[i]ncreasingly the question will not be whether to prefer the genetic or social parent but how to accommodate *both* on the assumption that they both have distinctive contributions to make to the life of the child.⁶ Herring has proposed a normative underpinning for determining legal parenthood which focuses on social parenting and 'care', arguing that '[p]arental status should be earned by the care and dedication to the child, something not shown simply by a biological link. It is the changing of the nappy; the wiping of the tear; and the working out of maths together that makes a parent, not the provision of an egg or sperm.⁷ While there is evidence of greater significance being accorded to social parenting in some particular contexts,⁸ in this chapter I will argue that underpinning legal parenthood there remains a binary, two-parent model, in which ideally a child has one 'mother' and one 'father'.

Throughout this chapter, I will examine how the status of legal parenthood is assigned in a variety of circumstances: natural reproduction in section 5.1, assisted reproduction in section 5.2 and surrogacy in section 5.3. I will observe that there are clear differences in the attribution of legal parenthood based upon biological sex; with motherhood being determined on the basis of gestation in all contexts, and fatherhood being determined by different factors in different factual contexts. I will conclude that legal parenthood is ordinarily premised upon a binary, two-parent model, which is derived from the traditional, heterosexual, nuclear family ideal.

⁵ Craig Lind and Tom Hewitt have suggested that, 'parental status has declined in importance in the last few decades' in, 'Law and the Complexities of Parenting: Parental Status and Parental Function' (2009) 31 (4) Journal of Social Welfare and Family Law 391, at 392.

⁶ Bainham, 'Parentage, Parenthood and Parental Responsibility: Subtle, Elusive, Yet Very Important Distinctions' in Andrew Bainham, Shelley Day Sclater and Martin Richards (eds.), *What is a Parent?:* A Socio-Legal Analysis, (Hart, 1999), at 29.

⁷ Herring, Caring and the Law, at 200.

⁸ Particularly the legislative reforms in the Adoption and Children Act 2002 and Adoption and Children (Scotland) Act 2007, which have sought to introduce greater flexibility to the adoption context through the introduction of 'Special Guardianship' (in England and Wales) and 'Permanence Orders' (in Scotland). These new legal orders provide a long term option which does not completely replace one set of parents with another and thus recognises the complexity of the interaction between legal, genetic and social parenthood in different individual circumstances. For the policy background to these reforms, see 'Adoption: a New Approach' (Department of Health, December 2000), available at - https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/263529/5017.pdf and 'Adoption: Better Choices for Our Children', Report of the Adoption Policy Review Group, (June 2005), available at - https://www.scotland.gov.uk/Resource/Doc/54357/0014208.pdf. See further e.g. Jane Lewis, 'Adoption: The Nature of Policy Shifts in England and Wales 1972-2002' (2004) 18 (2) International Journal of Law, Policy and the Family 235 and Deborah Cullen, 'Adoption - a (Fairly) New Approach' [2005] 17 (4) Child and Family Law Quarterly 475.

5.1. Natural Reproduction

Diduck and Kaganas observe that, 'while it is usually presented as a question of fact, the determination of which of the features that are normally associated with parenthood should be singled out as identifying the "real" mother or father of a child is actually a question of judgment.' The paradigm context for the determination of legal parenthood is 'natural' reproduction (that is reproduction following heterosexual intercourse). In this section, the basic presumptions that underpin legal parenthood in cases of natural reproduction will be considered, and it will be noted that the attribution of legal motherhood involves one, simple presumption while that of legal fatherhood involves the use of several related presumptions.

Where natural reproduction is concerned, there is only one woman who can be considered the legal mother of a child at birth: the woman who has gestated and given birth to the child (mater est quam gestatio demonstrat). As Lord Simon of Glaisdale observed in the Ampthill Peerage Case, ¹⁰ '[m]otherhood, although also a legal relationship, is based on a fact, being proved demonstrably by parturition.' Thus, the attribution of legal motherhood is determined by one factor alone: gestation.

The determination of the legal father of any child is self-evidently not so simple: accordingly, the law operates several presumptions in determining legal paternity. Historically the most important of these presumptions was that legal paternity is determined on the basis of the existence of a marriage to the mother of the child (pater est quem nuptiae demonstrant). In Scotland this presumption is now statutory: 'A man shall be presumed to be the father of a child - if he was married to the mother of the child at any time in the period beginning with the conception and ending with the birth of the child'. This presumption ties men to children through their

-

⁹ Diduck and Kaganas, Family Law, Gender and the State, at 125.

¹⁰ [1977] AC 547.

¹¹ ibid, at 577.

¹² s.5 (1) (a) Law Reform (Parent and Child) (Scotland) Act 1986, though the common law rule was changed with s.5 (4) which states that this presumption 'may be rebutted by proof on a balance of probabilities'. In English law this remains a common law presumption, see the statement of Lord

relationship to the children's mothers; the key factor is the existence of a marriage to the mother. This emphasis on the marital relationship is unsurprising given that historically this presumption existed to protect the 'sanctity' of marriage. Consequently, this presumption supported the traditional, nuclear family model that dominated the legal definition of the family, by assigning legal parenthood on the basis of the existence of marriage, which historically was the only legally regulated conjugal relationship. It suggests that the continued significance of this presumption should be understood as an indication of the continuing importance of marriage and the nuclear family within UK law.

An alternative statutory presumption in Scotland, which did not exist at common law, now applies in cases where the mother of the child is unmarried at both birth and at conception. In this context, a man is considered the legal father 'if both he and the mother of the child have acknowledged that he is the father and he has been registered as such'. The key factors in this context are the willingness of the man to be recognised as the father and the consequent actions of the individual man and woman, through acknowledging the man as the child's father and in registering this on the birth certificate. In these circumstances, it is the willingness to be recognised

-

Simon of Glaisdale in the *Ampthill Peerage Case* [1977] AC 547, at 577, '[f]atherhood, by contrast, is a presumption. A woman can have sexual intercourse with a number of men any of whom may be the father of her child'. In English law, the presumption is also rebuttable on the balance of probabilities, see s.26 Family Law Reform Act 1969.

¹³ The presumption could also be understood as being a presumption that the husband of the mother is the genetic father of the child. Notably s.5 (2) Law Reform (Parent and Child) (Scotland) Act 1986 states that the presumption applies, 'in the case of a void, voidable or irregular marriage as it applies in the case of a valid and regular marriage.'

¹⁴ Historically the marital status of the parents was used to distinguish between 'legitimate' and illegitimate' children, which had significant legal consequences for the child, this distinction was effectively abolished in the Family Law Reform Act 1987 and the Law Reform (Parent and Child) (Scotland) Act 1986.

¹⁵ The presumption also reflected the historical absence of direct evidence prior to the advent of blood and DNA testing and the common experiences of mankind that the majority of married men were the fathers of their wives' children.

¹⁶ As described above, in Chapter 2, in subsection 2.2.A, 'The Definition of 'Family' Under the Rent Acts' and subsection 2.2.B, 'The Definition of 'Family Life' under Article 8 of the European Convention on Human Rights'.

¹⁷ See above throughout Chapter 4, 'The Legal Regulation of Conjugal Relationships'.

¹⁸ s.5 (1) (b) Law Reform (Parent and Child) (Scotland) Act 1986, his name must have been registered in terms of s.18 (1) Registration of Births, Deaths and Marriages (Scotland) Act 1965, see s.10 (1) Births and Deaths Registration Act 1953 for the equivalent provision in English law.

as the child's father, combined with the public acknowledgment of that willingness that leads to the granting of the legal status of father.

The genetic connection between the man and the child may also determine legal fatherhood in cases of natural reproduction. This would apply in cases of 'disputed paternity', generally either where the mother was unmarried at birth and conception and no man was registered on the birth certificate, or where a man not deemed to be the father as a result of the aforementioned presumptions wishes to assert his paternity. Due to advances in DNA testing technology it is now possible to accurately establish genetic paternity through such testing. The identification of the genetic father displaces the aforementioned presumptions of legal fatherhood. Therefore, legal fatherhood can only be attributed to one man, and I argue that this exclusivity shows that these presumptions situate the attribution of legal fatherhood, in cases of natural reproduction, within the boundaries of a binary, two-parent model.

In this way, the law operates a hierarchy of factors in cases of natural reproduction, with proof of genetic relatedness, when established, having precedence over the marital relationship with the child's mother, which in turn has precedence over willingness and public acknowledgment of fatherhood. Judicial opinion has emphasised the significance of determining the genetic 'truth', ²³ as seen in *Re H and A (Children) (Paternity: Blood Test)*, ²⁴ where Thorpe LJ stated that 'the interests of justice are best served by the ascertainment of the truth and...the court should be furnished with the best available science and not confined to such unsatisfactory

1

¹⁹ s.7 (1) Law Reform (Parent and Child) (Scotland) Act 1986 provides for actions for declarator of either 'parentage' or 'non-parentage', see s.55A Family Law Act 1986 for the similar English law provisions relating to 'declarations of parentage'.

²⁰ See e.g. S v S 2014 SLT (Sh Ct) 165 for an example of a recent case where DNA testing to establish paternity was refused and the issues that resulted from the refusal.

²¹ However, see Helen Draper and Jonathan Ives, 'Paternity Testing: a Poor Test of Fatherhood' (2009) 31 (4) Journal of Social Welfare and Family Law 407, for a critique of the reliance on paternity testing alone as a method of determining 'fatherhood'.

²² s.5 (3) Law Reform (Parent and Child) (Scotland) Act 1986.

Judicial emphasis on seeking the 'truth' of paternity is apparent in the historical 'blood test' cases. In S v S [1972] AC 24, Lord Hodson observed, at 57, that, '[t]he interests of justice in the abstract are best served by the ascertainment of the truth and there must be few cases where the interests of children can be shown to be best served by the suppression of truth.' See further e.g. Re L (An Infant) [1967] 3 WLR 1645 and B v B [1968] 3 WLR 566.

²⁴ [2002] 1 FLR 1145.

alternatives as presumptions and inferences.'²⁵ This approach has been applied consistently in other recent cases²⁶ and suggests a change from prioritising the relationship with the child's mother toward prioritising the genetic connection²⁷ with the child in determining legal fatherhood.²⁸ This is notable as it runs counter to the historical protection of the nuclear family in determinations of legal parenthood. Diduck comments that, '[n]ow that we *can* know the biological - or at least genetic - "truth", it seems that in the interests of our well-being, we *must* know. And not only must we know, it has now become our right to know. We seem happy to be in the thrall to biology or nature again in a way that has been unknown for years.'²⁹ This approach is based on the understanding which has developed within the law that genetic connection, in and of itself, has significant importance and value for children.³⁰ This has led to the genetic relationship being given greater recognition and priority within the law.³¹ Sheldon has described this as 'a "geneticisation" of understandings of fatherhood.'³²

25

²⁵ ibid, at 1157, the use of the word 'truth' is particularly notable as it seems to conflate genetic relatedness and 'parenthood'.

²⁶ See e.g. Re D (A Child) (Paternity) [2007] 2 FLR 26, Re T (Paternity: Ordering Blood Tests) [2001] 2 FLR 1190 and Re H (A Minor) (Blood Tests: Parental Rights) [1996] 2 FLR 65. These cases suggest that the contemporary judicial approach is that carrying out scientific tests to establish paternity is usually in the best interests of the child.

²⁷ Judicial opinion in this regard has also been influenced by scientific developments which have improved the accuracy of testing, resulting in less need for reliance upon presumptions based on the relationship of the men with the mother of the child, see e.g. *Re H (A Minor) (Blood Test: Parental Rights)* [1996] 2 FLR 65, where Ward LJ stated, at 77, '[s]cience has now advanced. The whole truth can now be known.'

²⁸ The historical approach to 'blood test' evidence was that carrying out these tests was often not in the best interests of the child, due to the disruption to the pre-existing nuclear family, see e.g. *Re F (A Minor) (Blood Tests: Parental Rights)* [1993] Fam. 314, where Balcombe LJ stated, at 320, 'now and for the first few years of her life E.'s physical and emotional welfare are inextricably bound up with the welfare of the family unit of which she forms part: any harm to the welfare of that unit, as might be caused by an order for the taking of blood tests, would inevitably be damaging to E.'

²⁹ Diduck, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood', at 468.

³⁰ See e.g. Lord Nicholls of Birkenhead's short speech in the House of Lords in *Re G (Children)* (*Residence: Same-Sex Partner*) [2006] 2 FLR 629, particularly his observation, at 631, that, 'the court should always have in mind that in the ordinary way the rearing of a child by his or her biological parent can be expected to be in the child's best interests, both in the short term and also, and importantly, in the longer term.'

³¹ See e.g. *Re F (Paternity: Jurisdiction)* [2008] 1 FLR 225, where it was held that the courts have jurisdiction to make an order that a child should be told the truth about their genetic origins, regardless of the wishes of the primary caregiver.

³² Sally Sheldon, 'From 'Absent Objects of Blame' to 'Fathers who Want to Take Responsibility': Reforming Birth Registration Law' (2009) 31 (4) Journal of Social Welfare and Family Law 373, at 374.

Fortin sees a link between Art.7 of the United Nations Convention on the Rights of the Child (UNCRC), which states that a child has 'the right to know and be cared for by his or her parents³³ and the developing judicial understanding that an ongoing relationship with the genetic progenitor is in the best interests of the child.³⁴ On this basis, she has suggested that 'the right to know one's parents is being developed, through faith in the magical properties of biological ties, into the right to be with one's parents.'35 Bainham has commented that 'recent changes in England and elsewhere reflect a heavy emphasis on the importance of biological truth. Ascertaining the truth, if not seen as a right of the child, is certainly viewed as being in the child's best interests most of the time.'36 Thus, some academic commentators have suggested that the genetic connection has become the crucial factor in judicial determinations of legal fatherhood in cases of natural reproduction. Nevertheless, it is apparent that the determination of legal fatherhood is complex and is based on reliance upon different factors in different factual circumstances. In spite of these differences, however, I argue that the attribution of legal fatherhood, in cases of natural reproduction, is always situated within a binary, two-parent model.

5.2. Assisted Reproduction

The question of who should be considered the legal parent(s) of a child has been made significantly more complex by advances in medical science that have led to the much more widespread availability and use of 'assisted reproduction technologies', such as IVF, because these medical processes often involve the use of genetic material provided by a donor (or indeed multiple donors).³⁷ Additional complexities

³³ Article 7 (1) UNCRC states in full: 'The child shall be registered immediately after birth and shall have the right from birth to a name, the right to acquire a nationality and, as far as possible, the right to know and be cared for by his or her parents.'

³⁴ See e.g. Re H (Contact with Biological Father) [2012] 2 FLR 627, Re L (A Child) (Contact: Domestic Violence) [2000] 2 FLR 334 and White v White 2001 SLT 485. This issue will also be further considered below in Chapter 6, subsection 6.2.C.1, 'The Significance of Contact with the Non-Resident Parent'.

³⁵ Jane Fortin, 'Children's Right to Know Their Origins - Too far, Too fast?' [2009] 21 (3) Child and Family Law Quarterly 336, at 338.

³⁶ Andrew Bainham, 'Arguments About Parentage' (2008) 67 (2) Cambridge Law Journal 322, at 331.

The UK recently became the first country to allow specialist IVF treatments for mitochondrial disorders in the Human Fertilisation and Embryology (Mitochondrial Donation) Regulations 2015,

may be introduced through the use of surrogacy,³⁸ a process which inevitably involves at least one additional person (the surrogate), in addition to the 'intended parent(s)'.³⁹ Cases of assisted reproduction can give rise to complex factual scenarios which involve competing claims of legal parenthood. The effect of these situations is that the law is required to provide clear statements regarding how legal parenthood is to be assigned, because the 'correct' answer may not be readily apparent. The process of making such determinations requires the law to navigate complex factual circumstances and occasionally difficult choices,⁴⁰ and to be more explicit about the meanings of the various terms involved than is necessary in cases of natural reproduction where the attribution of parenthood is viewed as being more straightforward.

The Human Fertilisation and Embryology Act 2008 (2008 Act)⁴¹ provides the statutory framework for determining legal parenthood in this context.⁴² McCandless

opening the possibility of children having 3 'genetic parents'. See the Nuffield Council on Bioethics, 'Novel Techniques for the Prevention of Mitochondrial DNA Disorders: An Ethical Review', (June 2012), available at - http://nuffieldbioethics.org/wp-

content/uploads/2014/06/Novel_techniques_for_the_prevention_of_mitochondrial_DNA_disorders_c ompressed.pdf.

³⁸ The use of surrogacy remains quite limited in the UK due to surrogacy arrangements remaining unenforceable, s.1A Surrogacy Arrangements Act 1985 and commercial surrogacy being prohibited, s.2 Surrogacy Arrangements Act 1985. Marilyn Crawshaw, Eric Blyth & Olga van den Akker, 'The Changing Profile of Surrogacy in the UK - Implications for National and International Policy and Practice' (2012) 34 (3) Journal of Social Welfare and Family Law 267, at 269, on the basis of correspondence with the General Register Office, state that there were 149 parental orders granted in the UK in 2011. See further National Records of Scotland, 'Vital Events Reference Tables 2014, Section 2: Adoption and Re-Registrations', (August 2015), states that there were only 9 parental orders granted in Scotland in 2014, available at - http://www.nrscotland.gov.uk/files//statistics/vital-events-ref-tab-2-3.pdf.

³⁹ The particular issues raised by the factual circumstances of surrogacy will be considered in more detail below at subsection 5.3, 'Surrogacy'.

⁴⁰ These difficult judgments become apparent in some of the high-profile litigation surrounding the HFEA 2008 and its predecessor the HFEA 1990, e.g. *Evans v Amicus Healthcare Ltd.* [2004] 3 WLR 681, *Leeds Teaching Hospitals NHS Trust v A* [2003] 1 FLR 1091 and *R v Human Fertilisation and Embryology Authority Ex p. Blood* [1997] 2 WLR 806. See further, Sally Sheldon, 'Gender Equality and Reproductive Decision-Making' (2004) 12 (3) Feminist Legal Studies 303 and Sally Sheldon, 'Evans v Amicus Healthcare; Hadley v Midland Fertility Services - Revealing Cracks in the 'Twin Pillars'?' [2004] 16 (4) Child and Family Law Quarterly 437.

⁴¹ This is amending legislation, designed to 'modernise' the original Human Fertilisation and Embryology Act 1990. The reforms of the 2008 Act have been described as 'devoid of any coherent philosophical framework' in Rachel Fenton, Susan Heenan and Jane Rees, 'Finally Fit For Purpose? The Human Fertilization and Embryology Act 2008' (2010) 32 (3) Journal of Social Welfare and Family Law 275, at 285 and as a 'missed opportunity' in Marie Fox, 'The Human Fertilisation and Embryology Act 2008: Tinkering at the Margins' (2009) 17 (3) Feminist Legal Studies 333, at 334.

⁴² For assisted reproduction and surrogacy, the determination of legal parenthood is governed through the 'status provisions', found in s.33-47 Human Fertilisation and Embryology Act 2008.

and Sheldon have commented that 'the model of family underlying these provisions was barely considered by the reformers at all, not forming a core part of the reform project.'⁴³ This lack of consideration suggests that the values preferenced in the legislation's determination of legal parenthood were not the result of an explicit, conscious choice.⁴⁴ Arguably, then, the provisions simply reflect the traditional, nuclear family, which influenced its predecessor legislation; the Human Fertilisation and Embryology Act 1990 (1990 Act).⁴⁵ Writing in reference to the 1990 Act,⁴⁶ Sheldon has observed that, 'in the status provisions, Parliament attempted to foresee every possible reproductive scenario and to provide for the resulting family arrangement to conform as closely as possible to a nuclear family model.'⁴⁷ It is notable that the factors established in the legislation do not reflect the approach employed in cases of natural reproduction. This disparity has led some commentators to identify a sense of confusion and incoherence about these concepts within the law in the UK.⁴⁸

In this section, the more complex circumstances created by assisted reproduction will be considered. The distinction between the attribution of motherhood and fatherhood is also apparent in this context,⁴⁹ with the use of one, straightforward provision for determining the legal mother (subsection 5.2.A) compared to several provisions used

http://www.hfea.gov.uk/docs/Warnock_Report_of_the_Committee_of_Inquiry_into_Human_Fertilisation_and_Embryology_1984.pdf.

⁴³ Julie McCandless and Sally Sheldon, 'The Human Fertilisation and Embryology Act (2008) and the Tenacity of the Sexual Family Form' (2010) 73 (2) Modern Law Review 175, at 182.

⁴⁴ However, the Joint Committee on the Human Tissue and Embryology (Draft) Bill, 'Human Tissue and Embryos (Draft) Bill, Volume I: Report' (HL Paper 169, HC Paper 630, August 2007), stated, at Para 263, '[p]art 3 of the draft Bill seeks to take a new approach to parenthood, moving towards the concept of parenthood as a legal responsibility rather than a biological relationship.' Available at - http://www.publications.parliament.uk/pa/jt200607/jtselect/jtembryos/169/169.pdf.

⁴⁵ Support for the traditional nuclear family is apparent in the report of the Warnock Committee, which led to Human Fertilisation and Embryology Act 1990, 'Report of the Committee of Inquiry into Human Fertilisation and Embryology' (Department of Health and Social Security, July 1984), at Para 2.11, 'we believe that as a general rule it is better for children to be born into a two-parent family, with both father and mother', available at -

⁴⁶ This issue will be considered in more detail below in subsection 5.2.D, 'The Influence of the Nuclear Family Model'.

⁴⁷ Sheldon, 'Fragmenting Fatherhood: The Regulation of Reproductive Technologies', at 541.

⁴⁸ See e.g. Therese Callus, 'First 'Designer Babies', Now a La Carte Parents' [2008] 38 (2) Fam. Law 143 and Bainham, 'Arguments About Parentage'.

⁴⁹ Although it could be argued that this distinction simply reflects the factual reality of the very different gendered contributions made to human reproduction (whether 'natural' or medically assisted) by men and women.

for the legal father (subsection 5.2.C). In addition, assisted reproduction creates the possibility of there being two legal parents of the same gender, which raises issues around choice of terminology (subsection 5.2.B). I will argue that, in spite of the radical possibilities presented by assisted reproduction, the determination of legal parenthood remains firmly situated within the boundaries of the two parent model and thus the traditional nuclear family (subsection 5.2.D).

5.2.A. Motherhood

The statutory definition of 'mother' in cases of assisted reproduction reflects the common law approach used in natural reproduction.⁵⁰ The 2008 Act provides in s.33 (1) that '[t]he woman who is carrying or has carried a child as a result of the placing in her of an embryo or of sperm and eggs, and no other woman, is to be treated as the mother of the child'.⁵¹ This provision favours gestation over the genetic connection⁵² in cases where the two factors do not coincide, therefore privileging gestation in the determination of legal motherhood.⁵³ It has been argued by Jones that:

With this legislation, the birth mother is designated the legal mother, and the significance of genetic links between mothers and children is marginalised. Hence, legal discourse is able to make claims of "truth" with regard to the ascription of parenthood. Consequently, alternative

⁵⁰ As set out above in subsection 5.1, 'Natural Reproduction'.

⁵¹ s.33 (1) HFEA 2008, this provisions replicates the provisions for determining motherhood in s.27 (1) HFEA 1990.

⁵² Interestingly, in addition to preferencing gestation in the determination of motherhood, s.47 HFEA 2008 is titled, 'Woman not to be other parent merely because of egg donation'. Therefore, the legislation is explicit that the genetic connection (through egg donation) does not give rise to any parental status, see e.g. Re G (Children) (Shared Residence Order: Biological Non-Birth Mother) [2014] 2 FLR 897.

Some empirical research suggests that the legislative position is reflected in the attitudes of many of the women who participate in egg donation or egg-sharing, see e.g. H. Abdalla, F. Shenfield and E. Latarche, 'Statutory Information for the Children Born of Oocyte Donation in the UK: What Will They Be Told in 2008?' [1998] 13 (4) Human Reproduction 1106 and K. Ahuja, E. Simons, B. Mostyn and P. Bowen-Simpkins, 'An Assessment of the Motives and Morals of Egg Share Donors: Policy of 'Payments' to Egg Donors Requires a Fair Review' [1998] 13 (10) Human Reproduction 2671.

constructions of "mother" are disqualified for the purposes of legal status and rights in relation to the donor-conceived child.⁵⁴

This definition partly represents a pragmatic response; the woman who initiates the assisted reproduction process will be the gestational mother and she is also (usually) the intended social mother. More importantly however this choice to privilege gestation over the genetic connection reflects UK law's construction of motherhood as a 'natural', indivisible, process. The use of this construction reflects the fact that, as O'Donovan suggests, '[t]he distinction between maternity and motherhood... is the distinction between giving birth and being a mother. In common parlance in Britain, this distinction is not made, nor even seen. Nor is it made in law.' The law does not separate the status of legal motherhood from the process of carrying a baby and giving birth. Cases of assisted reproduction and surrogacy, which do not follow this natural process, challenge the dominant construction, but these cases nevertheless follow the gestational paradigm of motherhood through the provisions of the 2008 Act.

5.2.B. Non-Gestational Female 'Parents'

The law's reliance upon this natural, indivisible construction of motherhood is reinforced by its response to lesbian couples who have children together using assisted reproduction.⁵⁸ As a result of reforms in the 2008 Act, the non-gestational

-

⁵⁴ Caroline Jones, 'Parents in Law: Subjective Impacts and Status Implications around the Use of Licensed Donor Insemination' in Diduck and O'Donovan (eds.), *Feminist Perspectives on Family Law* at 80

⁵⁵ Except in cases of surrogacy, which are considered below at subsection 5.3, 'Surrogacy'.

⁵⁶ The construction of 'motherhood' within UK law will considered further below in Chapter 6, subsection 6.2.A, 'What is the 'Mother' in the Law?'

⁵⁷ O'Donovan, 'Constructions of Maternity and Motherhood in Stories of Lost Children' in Bridgeman and Monk (eds.), *Feminist Perspectives on Child Law*, at 73.

⁵⁸ Under the provisions of HFEA 1990 it was not possible for both members of a lesbian couple to be considered a child's legal parents at birth. The fundamental basis of these reforms have been questioned by some commentators, see e.g. Callus, 'First 'Designer Babies', Now a La Carte Parents', at 146, '[b]y recognising the status of two female parents, the child's identity is thrown into disarray because the recognition of two female parents conceals the necessary heterosexual element of human existence.'

female 'parent'⁵⁹ is to be treated as 'a parent of the child'⁶⁰ provided certain consent requirements are met.⁶¹ While both members of the couple are legal parents, it is notable that gender-neutral terminology is used to describe the parenthood of one.⁶² This is in clear contrast to the gendered language used in the legislation to describe the parental roles of 'mother'⁶³ and 'father';⁶⁴ gender-neutral terminology is only used to describe the parenthood of the non-gestational female 'parent'.⁶⁵ This use of the terminology of 'parent' to describe the legal parenthood of one member of the lesbian couple means that, as Probert says, 'there is not even a name to make her visible.'⁶⁶ As a result of this approach, Diduck has commented:

Motherhood remains exclusive; no alternative terms were attempted. Secondly, where a woman is treated as a parent, no man may be treated as the father of the child so that legal parenthood remains limited in number to two. In fact, by making it difficult for partnered lesbian women to parent alone, the proposals reinforce the traditional hetero-familial ideal of two parents; the equality imposed upon lesbian parents by the proposals in effect imposes heterosexual norms upon them.⁶⁷

⁵⁹ A variety of terms are employed to describe this role, e.g. 'Co-Mother', 'Co-Parent', 'Other Mother', 'Second Female Parent', etc. As this section will suggest, these issues of terminology are fundamental to any consideration of this role. Throughout this thesis the terms non-gestational female 'parent' and 'second female parent' will be used to describe the role, reflecting the statutory terminology.

⁶⁰ Following provisions similar to those determining 'father' detailed below: s.42 (1) HFEA 2008 for those in a civil partnership and s.43 for those not.

⁶¹ These are found in 'the agreed female parenthood conditions', s.44 HFEA 2008.

⁶² This approach shows the 'enduring grip' of the two parent model, according to Leckey in, 'Law Reform, Lesbian Parenting, and the Reflective Claim', at 338.

⁶³ s.33 HFEA 2008.

⁶⁴ s.35-37 HFEA 2008, these provisions are considered below at subsection 5.2.C, 'Fatherhood'.

⁶⁵ The implications of this terminological distinction upon the legal understanding of the parental roles of both members of lesbian couples will be considered below in Chapter 6, section 6.3, 'The Role of Lesbian 'Mothers' or 'Parents''.

⁶⁶ Rebecca Probert, 'Families, Assisted Reproduction and the Law' [2004] 16 (3) Child and Family Law Quarterly 271, at 278.

⁶⁷ Diduck, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood', at 466.

While the reforms of the 2008 Act⁶⁸ amount to an attempt to acknowledge the diversity of family forms that exist in contemporary society,⁶⁹ in the process these provisions reinforce the assumption that a child can only have one mother. Thus, the traditional construction of motherhood, and the binary, two-parent model of parenthood are supported and remain unquestioned. Notably, this exclusivity of motherhood and the absence of an obvious term to describe the non-gestational female 'parent' role is reflected in the views of lesbian parents themselves: Gabb found that 'the vast majority of "other mothers" in my study not only felt uneasy about calling themselves mothers, they also contested the use of non-reductive "special terms" such as "co-parent".'70

Fenton, Heenan and Rees suggest that 'the parentage provisions for same-sex couples are set out as a mirror image to that of heterosexual couples. It is arguable that this is, by effect if not design, a heterocentric statement based on the primacy of the biogenetic two-parent model of the sexual family.'⁷¹ It is submitted that the reforms of the 2008 Act, which extend legal parenthood to both members of a lesbian couple, do not alter the reliance upon a binary, two-parent model, based upon the traditional, heterosexual, nuclear family.⁷²

⁶⁸ The approach of the HFEA 2008 differs substantially from the HFEA 1990, which stated in s.13 (5), '[a] woman shall not be provided with treatment services...unless account has been taken of the welfare of any child who may be born as a result of the treatment (including the need of that child for a father)'.

⁶⁹ The reforms were premised upon addressing the inequality between same-sex and opposite-sex couples as regards access to assisted reproduction, see, 'Review of the Human Fertilisation and Embryology Act' (Department of Health, December 2006), at Para 2.67, '[i]n undertaking its review of the HFE Act, the Government aimed to consider the extent to which changes may be needed to better recognise the wider range of people who seek and receive assisted reproduction treatment in the early 21st Century.' Available at -

http://www.hfea.gov.uk/docs/Review HFEA Act White Paper DH.pdf.

⁷⁰ Gabb, 'Lesbian M/Otherhood: Strategies of Familial-linguistic Management in Lesbian Parent Families', at 595. See further Dunne, 'Opting into Motherhood: Lesbians Blurring the Boundaries and Transforming the Meaning of Parenthood and Kinship'.

⁷¹ Fenton, Heenan and Rees, 'Finally Fit For Purpose? The Human Fertilization and Embryology Act 2008', at 279.

⁷² Writing in the context of the HFEA 1990, Julie Wallbank, 'Reconstructing the HFEA 1990: Is Blood Really Thicker Than Water?' [2004] 16 (4) Child and Family Law Quarterly 387, stated, 'the current law is both over-simplistic and biased towards the traditional heterosexual model of family life, which is no longer wholly appropriate or realistic in terms of contemporary family formations.' While the HFEA 2008 went some way to addressing some concerns by providing recognition for lesbian-led families, it is suggested that the provisions still illustrate the dominance of the traditional, nuclear family model in the framework for determining legal parenthood.

5.2.C. Fatherhood

In contrast with the clarity of a single criterion - gestation - for determining motherhood, the 2008 Act provides for different factors to be decisive of legal fatherhood in different factual circumstances; in this sense, UK law's approach to fatherhood in the context of assisted reproduction reflects that in natural reproduction. As Sheldon and Collier observe, '[w]hile motherhood is legally firmly rooted in gestation, fatherhood is an altogether more complex and fragmented status.' In determining legal fatherhood in cases of assisted reproduction, the legislation makes a distinction between married and unmarried couples, and different statutory provisions apply depending on relationship status.

In cases where the mother is married and donor sperm is used,⁷⁴ the legislation provides that, 'the other party to the marriage is to be treated as the father of the child unless it is shown that he did not consent to [the treatment]'.⁷⁵ The key factor here is that of consent, and it is notable that, taking account of the law's privileging of marriage and the nuclear family, what is required is evidence of lack of consent: in other words, consent is assumed due to the status of the relationship, as a marriage.⁷⁶ In cases where the mother is not married and donor sperm is used,⁷⁷ the 'agreed fatherhood conditions'⁷⁸ need to be satisfied for fatherhood to be conferred. These conditions provide that a man will be deemed the father if he, 'has given the person responsible a notice stating that he consents to being treated as the father of any child

⁷³ Sheldon and Collier, *Fragmenting Fatherhood: A Socio-Legal Study*, at 87. It is also arguable that the distinction between the statutory approaches to determining legal motherhood and legal fatherhood to some extent reflects the reality of the different mechanism of proof of legal parenthood for 'mothers' and 'fathers'.

⁷⁴ If the sperm of the husband is used, legal fatherhood would be determined on the basis of the presumptions described above for cases of natural reproduction, e.g. the *pater est* presumption and additionally the genetic connection between father and child. Further, those situations involving known donors and home based insemination outside of licensed clinics will be considered below within this section.

⁷⁵ s.35 (1) HFEA 2008.

⁷⁶ Interestingly s.38 (3) HFEA 2008 provides, '[i]n Scotland, sections 35 and 36 do not apply in relation to any child who, by virtue of any enactment or other rule of law, is treated as the child of the parties to a marriage.' This effectively restates the pre-eminence of the aforementioned *pater est* presumption, enshrined in s.5 (1) (a) Law Reform (Parent and Child) (Scotland) Act 1986.

⁷⁷ If the sperm of the unmarried partner is used, legal fatherhood would be determined on the basis of the presumptions described above for cases of natural reproduction, e.g. through registration on the birth certificate and through the genetic connection between father and child.

⁷⁸ s.36 (b) and detailed in s.37 HFEA 2008.

resulting from treatment provided to W under the licence'. While consent is again the determinative factor, here it is not assumed and must take the specific form prescribed by the legislation. Dame Elizabeth Butler-Sloss P stated, in *Leeds Teaching Hospitals NHS Trust v A*, that '[t]he husband is to be treated as the father, unless it is shown that he did not consent, that he "opted out." In subsection (3) the boot is on the other foot. The acceptance of fatherhood has to be shown. The man has to show commitment and that commitment has to be demonstrated by his active involvement. In other words, he must "opt in".' The distinction in the consent requirements for married and unmarried men illustrates the importance of the existence of marriage in the attribution of legal fatherhood, because consent is effectively assumed for married men, purely on the basis of their relationships status, whereas for unmarried men there is no such assumption of consent.

Regardless of whether the mother and father are married, sperm donors are not treated as legal fathers provided the conditions in the Act are satisfied.⁸⁴ This is based upon consent; the donor has consented to the use of his genetic material on the condition that he is *not* treated as the father.⁸⁵ Those donating sperm are no longer granted anonymity:⁸⁶ legislation now allows for children conceived using donated material, after April 1st 2005,⁸⁷ to request certain identifying information about their

⁷⁹ s.37 (1) (a) HFEA 2008.

⁸⁰ In addition, the agreed fatherhood conditions include a requirement of the consent of the mother, s.37 (1) (b), to treat an unmarried man as the legal father.

⁸¹ [2003] 1 FLR 1091, this case was considering the original Human Fertilisation and Embryology Act 1990 which employed different language in relation to unmarried couples, but it is suggested that the general principles are still relevant. The interpretation of the previous legislation, particularly the reference in s.28 (3) to 'treatment together' was subject to judicial consideration, see e.g. *Re R (A Child) (IVF: Paternity of Child)* [2005] 2 AC 621 and *Re R (A Child)* [2003] Fam. 129.

⁸² ibid, per Dame Elizabeth Butler-Sloss P, at 1103, the President was referring to the language of s.28 (3) HFEA 1990.

⁸³ These provisions in relation to unmarried fathers only apply to 'treatment services' carried out by licensed UK clinics, s.36 (a) HFEA 2008 states, 'in the course of treatment services provided in the United Kingdom by a person to whom a licence applies'. Therefore, in either private 'known donor' cases or in cases where treatment (using donor sperm) is carried out at foreign clinics these provisions do not apply and the sperm donor would have a claim to legal fatherhood on the basis of the genetic connection. There is no similar requirement in relation to married men; indeed, s.35 (2) explicitly applies to artificial insemination outside the UK.

⁸⁴ s.41 (1) HFEA 2008.

⁸⁵ s.41 (1) HFEA 2008, states, '[w]here the sperm of a man who had given such consent...was used for a purpose for which such consent was required, he is not to be treated as the father of the child.'

⁸⁶ This change was as a result of the Human Fertilisation and Embryology Authority (Disclosure of Donor Information) Regulations 2004.

⁸⁷ s.31ZA HFEA 1990 as amended by s.24 HFEA 2008.

donors.⁸⁸ This change in UK law reflects the understanding, discussed above, that the genetic connection itself is now granted significant importance within the law.⁸⁹ By removing donor anonymity the law is seeking to give some recognition to the continued importance (at least in the perception of those whose origins are different from the norm) of the genetic link.⁹⁰ This partial recognition exists notwithstanding that legal parenthood is not being determined by the genetic connection in this context.⁹¹ Indeed, I argue that this approach illustrates that the attribution of legal fatherhood, in cases of assisted reproduction, is determined within the confines of a binary, two-parent model.

The emphasis on consent in preference to the genetic connection in cases of assisted reproduction runs in direct contradiction to the hierarchy of presumptions established in cases of natural reproduction. The genetic connection is privileged over social parenting when determining legal parenthood in natural reproduction, whereas in assisted reproduction the significance of the donor's genetic connection is diminished in favour of recognising the consent of the intended (social) parents. This disparity of approaches reinforces the concern expressed by some commentators (referenced above) that the law lacks a coherent, principled approach to determining legal parenthood. He are the following the consent of the intended (social) parents.

⁸⁸ s.2 (3) Human Fertilisation and Embryology Authority (Disclosure of Donor Information) Regulations 2004, details the identifying information that is required to be retained about the sperm donor: full name, date of birth, appearance and last known address.

⁸⁹ As discussed above at subsection 5.1, 'Natural Reproduction'.

⁹⁰ Prior to the change in the law, in *R* (On the Application of Rose) v Secretary of State for Health [2002] EWHC 1593 (Admin), a claim for judicial review as to the lack of availability of non-identifying information about donors, by children born as a result of anonymous sperm donation, Scott Baker J stated, at Para 14, '[t]he claimants' requests were for information about their biological fathers which went to the very heart of their identity, and to their make-up as people.'

⁹¹ See further e.g. Catherine Donovan, 'Genetics, Fathers and Families: Exploring the Implications of Changing the Law in Favour of Identifying Sperm Donors' (2006) 15 (4) Social and Legal Studies 494.

⁹² As set out above in section 5.1, 'Natural Reproduction'.

⁹³ Callus, 'First 'Designer Babies', Now a La Carte Parents', at 147, observes, 'there are competing tendencies in the law on the one hand with reliance on biological truth where no recourse to assisted conception is required, and, on the other, a complete isolation of the biological component of parenthood to take into account social parenting.'

⁹⁴ Furthermore, some commentators have been critical of the perceived lack of transparency, within law, regarding the genetic origins of children conceived using donor sperm; see e.g. Bainham, 'Arguments About Parentage', at 332. For the opposing argument, see e.g. Carol Smart, 'Family Secrets: Law and Understandings of Openness in Everyday Relationships' (2009) 38 (4) Journal of Social Policy 551.

The complexity of assigning legal parenthood is further increased in circumstances involving home-based insemination and the use of a known sperm donor. The 2008 Act applies to those in marriages and civil partnerships in all cases of 'artificial insemination', the whereby the other party to the relationship will be the child's legal parent when a known donor was used. However, for unmarried couples the legislation applies only to treatment carried out by licensed clinics, and therefore in cases where unmarried couples use home-based insemination the known donor will be considered the legal father, traced to his genetic connection with the child. Thus, in spite of the presence of this additional potential parental figure the binary, two-parent model is still applied in these situations. Evidence suggests that in practice the vast majority of known donor cases involve lesbian couples rather than opposite sex couples. Therefore, I agree with Smith's observation that, '[e]xcluding known donors from legal recognition through a system which recognises only two parents validates and protects lesbian families but also reinforces the dyadic parenting norm based on heterosexual reproduction.'

⁹⁵ This is another area where terminology has proved somewhat controversial, the judiciary has often used the term 'biological father' to describe the donors in circumstances involving lesbian couples, e.g. Re P & L (Contact) [2012] 1 FLR 1068, Re B (Role of the Biological Father) [2008] 1 FLR 1015 and Re D (Contact and Parental Responsibility: Lesbian Mothers and Known Father) [2006] 1 FCR 556. See further Julie Wallbank and Chris Dietz, 'Lesbian Mothers, Fathers and Other Animals: Is the Political Personal in Multiple Parent Families?' [2013] 25 (4) Child and Family Law Quarterly 451. The issues that arise in disputes between lesbian couples and 'known donors' are discussed more below in Chapter 6, subsection 6.3.A.2, "Known Donor' Disputes'.

⁹⁶ s.35 HFEA 2008 for married couples and s.42 for civil partners.

⁹⁷ However, for children conceived by women in civil partnerships using 'known donors', prior to the 2008 Act coming into force, the known donor would be the legal father, see e.g. *A v B and C (Role of Father)* [2012] 2 FLR 607.

⁹⁸ s.36 HFEA 2008 for opposite-sex couples and s.43 for same-sex couples.

⁹⁹ Research evidence suggests that lesbian couples were more likely to enter into 'known donor' arrangements than use anonymous donors, prior to the reforms of the 2008 Act, see e.g. Smith, 'Is Three a Crowd? Lesbian Mothers Perspectives on Parental Status in Law', at 234 and Dunne, 'Opting into Motherhood: Lesbians Blurring the Boundaries and Transforming the Meaning of Parenthood and Kinship', at 16. It is possible that the existence of legal parenthood for both members of a lesbian couple, through the reforms of the 2008 Act, will encourage greater usage of anonymous donors than under the previous regime.

¹⁰⁰ Smith, 'Tangling the Web of Legal Parenthood: Legal Responses to the Use of Known Donors in Lesbian Parenting Arrangements', at 378.

5.2.D. The Influence of the Nuclear Family Model

From all of the foregoing, it is apparent that the law's approach to determining fatherhood in cases of assisted reproduction is complex and applies different rules in different circumstances. However, there are certain universal truths that seem to be unchallenged: as Sheldon notes, for example, central to the attribution of fatherhood under the legislation has been the notion that, 'the symbolism of fatherhood and replication of the nuclear family is crucial: each child should ideally have a father who is married to her mother or, failing that, a father who is committed to the mother and intending to create a child with her. '101 Thus, I argue that the attribution of legal fatherhood in cases of assisted reproduction is determined within the boundaries of a binary, two-parent model.

Moreover, I submit that the diminishing of the significance of the genetic connection between donors and children is premised upon the continued promotion of the nuclear family, because as Diduck and Kaganas observe:

[F]atherhood by consent is contrary to law's privileging of biology and things "natural" but...[the law] contradicts this position when there is a marriage or marriage-like partnership to privilege above nature...Reliance upon this relationship, now in extended form, continues to privilege the "sexual family". 102

Thus, ignoring the genetic link confers parenthood on men who are married to mothers or in marriage-like relationships with mothers, rather than on sperm donors, who are unknown to the mother and this approach illustrates the significance of the nuclear family model. Research evidence suggests that relatively few (heterosexual) parents inform their children that they have been conceived using

¹⁰¹ Sheldon, 'Fragmenting Fatherhood: The Regulation of Reproductive Technologies', at 541, this comment was made in the context of the provisions of the 1990 Act.

¹⁰² Diduck and Kaganas, Family Law, Gender and the State, at 170.

¹⁰³ The significance given to 'marriage-like' relationships in this context reflects the discussion of the legal regulation of cohabitation on the basis that such relationships are 'marriage-like', set out above in Chapter 4, section 4.2.A.1, 'Cohabitation as a 'Marriage-Like' Relationship'.

donor sperm¹⁰⁴ and as Fortin states, 'any child born by donor conception looks as if he or she has been born into a "normal" nuclear family with one mother and one father. The legislation in this context ignores the biological connectedness between sperm donor and child.'¹⁰⁵

The report of the Warnock Committee upon which the provisions of the Human Fertilisation and Embryology Act 1990 were based made its preference for the two-parent model and the 'nuclear family' very obvious when it stated, 'we believe that as a general rule it is better for children to be born into a two-parent family, with both father and mother'. It can be seen clearly from this that the promotion and protection of the nuclear family was influential in shaping the legislative approach to determining legal parenthood across the variety of different circumstances in cases of assisted reproduction. While the reforms of the 2008 Act have widened the scope of legal parenthood to allow for the possibility of two female parents, this has been done without altering or challenging the model on which the provisions are based, namely the two-parent model, based upon the traditional, heterosexual, nuclear family. As such, I agree with Smith's observation that, 'legal parenthood remains an exclusive concept - albeit an exclusive concept that is now capable of embracing two parents of the same sex - designed to shore up nuclear families against the disruptive potential of the fragmentation of parenthood.'

¹⁰⁴ See e.g. S. Golombok, A. Braeways, M.T. Giavazzi, D. Guerra, F. MacCallum and J. Rust, 'The European Study of Assisted Reproduction Families: The Transition to Adolescence' [2002] 17 (3) Human Reproduction 830, at 836, found only 8.6% of the families in their study had told their children (aged 11-12 at the time). See also, S. Golombok, C. Murray, V. Jadva, E. Lycett, F. MacCallum and J. Rust, 'Non-Genetic and Non-Gestational Parenthood: Consequences for Parent-Child Relationships and the Psychological Well-Being of Mothers, Fathers and Children at Age 3' [2006] 21 (7) Human Reproduction 1918, at 1921, where it was observed that 46% of parents of donor conceived children (aged 3) had already decided not to inform their children at any point.

¹⁰⁵ Fortin, 'Children's Right to Know Their Origins - Too far, Too fast?', at 344.

^{&#}x27;Report of the Committee of Inquiry into Human Fertilisation and Embryology' (1984), at Para 2.11.

¹⁰⁷ Dewar, 'The Normal Chaos of Family Law', at 482, observed in the context of the 1990 Act, '[t]he definition of paternity...reflects, more than anything, the type of parents whom the state is prepared to reproduce through the provision of fertility services: namely, the two parent, heterosexual, preferably married, parents.'

¹⁰⁸ Although it should be noted that in contrast, Sheldon and Collier, *Fragmenting Fatherhood: A Socio-Legal Study*, at 80, suggest that, '[t]he fierce protection of the heterosexual nuclear family form entrenched in the 1990 legislation gives way to a more fluid and complex sense of familial relationships.'

Smith, 'Clashing Symbols? Reconciling Support for Fathers and Fatherless Families After the Human Fertilisation and Embryology Act 2008', at 70.

5.3. Surrogacy

Horsey and Sheldon have observed that, '[t]he law relating to the attribution of legal parenthood is poorly designed to respond to surrogacy arrangements.' This occurs because of the greater complexity that is introduced into the determination of legal parenthood in cases of surrogacy, where it is possible that, as Horsey identifies:

[A] child can have up to six potential "parents": two gamete providers, the gestational/birth mother and her husband or partner (if she has one) and the two intending parents, where these are different people. Notably this number is only limited to six because the law is only prepared to recognise two parents, the number could be greater were this not the case. ¹¹¹

The complex factual circumstances that may be present in cases of surrogacy sit uneasily within the general approach to legal parenthood in cases of assisted reproduction. This is due to the law's reliance upon a binary, two-parent model. The distinct issues raised by the determination of legal parenthood in cases of surrogacy are not dealt with explicitly by the 2008 Act's framework for determining legal parenthood. Simply put, as Fenton-Glynn observes, '[t]he status provisions... were not designed with surrogacy in mind'.

¹¹⁰ Kirsty Horsey and Sally Sheldon, 'Still Hazy After All These Years: The Law Regulating Surrogacy' (2012) 20 (1) Medical Law Review 67, at 87. Emily Jackson, *Regulating Reproduction: Law, Technology and Autonomy*, (Hart, 2001), similarly observes, at 283, '[t]he Human Fertilisation and Embryology Act's status provisions apply awkwardly and inappropriately to surrogacy arrangements.'

Kirsty Horsey 'Challenging Presumptions: Legal Parenthood and Surrogacy Arrangements' [2010]
 (4) Child and Family Law Quarterly 449, at 453.
 Although the provisions of s.54 HFEA 2008 that set out the procedures for 'parental orders' allow

for the possibility that the intended social reality of the parties to the surrogacy arrangement will be given effect to within six months of the child being born; s.54 (6) states that 'the court must be satisfied that both...[the surrogate mother and her husband, if he is deemed the legal father]...have freely, and with full understanding of what is involved, agreed unconditionally to the making of the order.'

¹¹³ Horsey, 'Challenging Presumptions: Legal Parenthood and Surrogacy Arrangements', suggests that consideration of surrogacy did not form a major part of the reform proposals that led to HFEA 2008, stating, at 451, 'where a child is born using surrogacy (with or without donated gametes) the law regarding parenthood was left almost entirely untouched.'

Claire Fenton-Glynn, 'The Regulation and Recognition of Surrogacy Under English Law: An Overview of the Case-Law' [2015] 27 (1) Child and Family Law Quarterly 83, at 84.

Moreover, the issues regarding surrogacy are magnified because UK law currently treats surrogacy arrangements as unenforceable between the parties 115 and organising commercial surrogacy remains prohibited; 116 consequently, surrogacy arrangements remain unregulated by the Human Fertilisation and Embryology Authority. Historically, surrogacy was not widely used within the UK. 117 More recently, however, there has been evidence of increasing use of surrogacy, 118 with particular growth observed in international surrogacy arrangements. 119 The use of surrogacy as a method of reproduction challenges some of the fundamental underpinning assumptions of parenthood, as Cook et al state:

Surrogacy is problematic for traditional notions of "mother", "father" and "family" when it introduces a third (or even fourth) party into reproduction, when it introduces contractual or "public" arrangements into "private" affairs and when it fragments motherhood. Surrogacy makes motherhood negotiable and confounds both social and biological bases of claims to parenthood. 120

In this section, the additional complexity that results from the process of surrogacy will be explored, and in particular, I will argue that the reliance upon a binary, two-

¹¹⁵ s.1A Surrogacy Arrangements Act 1985 states, '[n]o surrogacy arrangement is enforceable by or against any of the persons making it.'

s.2 Surrogacy Arrangements Act 1985. This section prohibits the operation of a commercial surrogacy agency, making it 'an offence' in terms of s.2 (2).

¹¹⁷ Official figures as to the number of surrogacies are difficult to obtain due to the unregulated nature of surrogacy. However, the non-profit organisation 'Childlessness Overcome Through Surrogacy' (COTS), state they have been involved in 855 surrogate births since 1988, available at http://www.surrogacy.org.uk/About_COTS.htm.

¹¹⁸ Crawshaw, Blyth & van den Akker, 'The Changing Profile of Surrogacy in the UK - Implications for National and International Policy and Practice', at 269, the authors present data based on correspondence with the General Register Office, which shows that there has been a sharp increase in the number of parental orders granted in the UK, with 149 in 2011 compared to 51 in 2007. See also National Records of Scotland, 'Vital Events Reference Tables 2014, Section 2: Adoption and Re-Registrations', where, curiously, parental orders after surrogacy are listed as a 'Type of Adoption', show that in the six years between 2003 (when the first parental order was made in Scotland) and 2008 there were 15 such orders made by the Scottish courts, while in the six years between 2009 (when the rules for eligibility to apply were expanded) and 2014 there were 52 such orders.

¹¹⁹ Ibid, at 271, the authors note that in 2011 26% of parental orders in England and Wales involved a child that had been born abroad, compared to only 2% in 2008.

120 Rachel Cook and Shelley Day Sclater with Felicity Kaganas (eds.), Surrogate Motherhood:

International Perspectives, (Hart, 2003), at 4.

parent model of legal parenthood fits uneasily with the factual circumstances of surrogacy; considering the attribution of both motherhood (subsection 5.3.A) and fatherhood (subsection 5.2.B).

5.3.A. Motherhood

Douglas has commented that, 'English law views the gestational mother as the legal mother because this will produce the right result in terms of parentage for all cases except for surrogacy.' The seemingly straightforward approach to maternity, which applies across contexts, creates complexity in cases of surrogacy, because there are two women with competing claims to 'motherhood'. As Cook observes, '[a] major moral concern in relation to these types of family formation is the fragmentation of parenthood. This fragmentation is more explicit in surrogacy than any other reproductive option: it separates social motherhood from gestation and genetics.'122 However, the law relies upon the binary two-parent model and thus determines motherhood on the basis of gestation, which results in the surrogate being recognised as the child's legal mother at birth. Thus, I argue that surrogacy presents a factual context where the reliance upon the traditional, indivisible, 'natural' construction of 'mother' creates issues in the attribution of legal parenthood. 123 This approach to determining legal motherhood gives limited significance or recognition to the intended social reality in surrogacy cases, where the process of gestation and giving birth is separated from the social practice of 'mothering', as they will be undertaken by different women. As a consequence of the law's reliance upon a binary, two-parent model, it is necessary for an explicit choice to be made between these competing claims to motherhood and Wallbank suggests that, 'by adopting the either/or approach it reinforces the ideology inherent in much of family law and

¹²¹ Douglas, 'The Intention to Be a Parent and the Making of Mothers', at 640.

Rachel Cook, 'Donating Parenthood: Perspectives on Parenthood from Surrogacy and Gamete Donation' in Bainham, Day Sclater and Richards (eds.), *What is a Parent?: A Socio-Legal Analysis*, at 122.

The situation of non-gestational female 'parents' provides another example of the problematic consequences of the 'natural' indivisible construction of the 'mother' upon the determination of legal parenthood, see above at, subsection 5.2.B, 'Non-Gestational Female 'Parents''. Although it should be noted that in the context of surrogacy the privileging of gestation could also be seen as a protective measure, which ensures that the woman who carried the child ('the gestational mother') always retains the option to change her mind.

social policy that children's best interests are served by being raised in the traditional two-parent family.' Thus, the legal approach to surrogacy diminishes the importance of the intention of the parties and social parenting, instead preferencing a unified construction of motherhood based upon gestation. Is suggest that this reflects the rhetorical power of the law's construction of mother as 'natural', since legal motherhood is exclusively determined on the basis of gestation and no other factors are considered. Moreover, I argue that this approach illustrates the significance of the binary, two-parent model of the nuclear family in the attribution of legal parenthood, in spite of the more complex factual circumstances which exist in cases of surrogacy.

5.3.B. Fatherhood

Surrogacy can also complicate the attribution of legal fatherhood, because the provisions of the 2008 Act apply to all forms of assisted reproduction, meaning that in most cases where the surrogate is married, it will be her husband who is likely to be presumed to be the child's father at birth. This applies regardless of whether the sperm of the commissioning father is used; in such circumstances the commissioning father is effectively in a position equivalent to that of a 'known donor' and therefore

¹²⁴ Julie Wallbank, 'Too Many Mothers? Surrogacy, Kinship and the Welfare of the Child' (2002) 10 (3) Medical Law Review 271, at 286.

However, it is notable that the significance of social parenting and 'care' are recognised in this context by 'parental orders', under s.54 HFEA 2008, which transfer parenthood to the commissioning couple after birth, these essentially operate as a form of fast-track adoptions.

¹²⁶ The approach also does not reflect the research into the perspectives of surrogates themselves, see e.g. Vasanti Jadva, Clare Murray, Emma Lycett, Fiona MacCallum and Susan Golombok, 'Surrogacy: The Experiences of Surrogate Mothers' [2003] 18 (10) Human Reproduction 2196, who find that 59% of surrogates felt they had no 'special bond' with the child and go on to find, at 2203, that 'none of the women in the present study had any doubts about their decision to hand over the child to the commissioning couple.' See also, Fiona MacCallum, Emma Lycett, Clare Murray, Vasanti Jadva and Susan Golombok, 'Surrogacy: The Experience of Commissioning Couples' [2003] 18 (6) Human Reproduction 1334.

¹²⁷ However, the provisions of s.54 (2) HFEA 2008, governing applications for 'parental orders', include civil partners and both opposite-sex and same-sex cohabitants, therefore the act is expressly allowing for situations in which a child would be legally motherless, through the granting of a parental order to two men.

¹²⁸ Except in cases where he didn't consent to his wife's treatment, s.35 (1) HFEA 2008.

has no legally-recognised parental relationship with the child at birth. 129 This result illustrates, even more clearly than the determination of legal motherhood that a binary two-parent model is an inappropriate fit with the factual circumstances of surrogacy. This is because legal parenthood is attributed to a man with no genetic relationship to the child, who also has no intention of having any social relationship with that child in the future. 130 Consequently, relationship status is being privileged over other factors; genetic connection, the intention of the parties and social parenting. In Horsey's view, 'this is a wholly unnecessary legal fiction and, while reinforcing the notion that motherhood is determined by gestation, does not mirror the way that father following other forms of assisted reproduction is regulated.¹³¹ I submit that this acutely illustrates the law's promotion of marriage, a binary, twoparent model and the idealised image of the nuclear family, regardless of the existence of complex factual circumstances which may justify alternative, more nuanced approaches. 132

Thus, the application of a unified approach to legal parenthood in cases of assisted reproduction has the consequence in some surrogacy cases of assigning legal parenthood in a manner that ignores both the genetic connections and the intentions of the parties as to future social parenting. This stands in contrast to the law's emphasis on consent and reflecting social parenting when determining legal fatherhood in other types of assisted reproduction. 133 Therefore, I argue that surrogacy sits uneasily within a legislative framework that is premised upon the exclusive, two-parent model, because the factual circumstances of surrogacy

¹²⁹ Although as this is merely a presumption of fatherhood, the intended father in such circumstances would be able to assert his genetic paternity through testing, as described above in cases of natural reproduction, at subsection 5.1, 'Natural Reproduction'. However, the determination of legal parenthood at birth remains premised upon the centrality of marriage and the nuclear family.

130 In comparison to the surrogate, whose legal parenthood is premised upon her having a gestational

connection.

¹³¹ Horsey 'Challenging Presumptions: Legal Parenthood and Surrogacy', at 450.

ibid, at 455-474, Horsey argues for the development of an approach to determining legal parenthood in cases of assisted reproduction that is premised upon intention. Indeed, it is submitted that the approach to the attribution of legal parenthood in cases of surrogacy should be reconsidered. I suggest that this should be premised upon legal parenthood in cases of surrogacy being attributed under different legislative provisions than other types of assisted reproduction. Such an alternative approach would allow the particular issues regarding legal parenthood in the context of surrogacy to be addressed by the law.

¹³³ In addition to the circumstance of unmarried fathers in cases of natural reproduction, described above at subsection 5.1, 'Natural Reproduction'.

introduce additional potential 'parents' into the 'natural' factual scenario, but this cannot be taken into account by that model.

Conclusion

In this chapter, I have argued that UK law's approach to the attribution of legal parenthood is based on a binary, two-parent model. I have observed that the legal determination of motherhood employs a simple test in all circumstances; motherhood is based upon gestation. By contrast, the legal attribution of fatherhood involves different factors being prioritised depending on the circumstances; the existence of a marriage between the mother and father, the willingness and actions of the parties and the genetic connection between the father and the child are all used to determine fatherhood in different contexts. Jackson has commented that the approach taken to the determination of legal parenthood has occurred:

Because the law has been stymied by the principle of parental exclusivity, its response to the splitting of the normal incidents of parenthood has been to try to identify a hierarchy of criteria which will result in one putative parents claim trumping the others. In so doing, it has become spectacularly confused and confusing, not least because different hierarchies operate in different circumstances. ¹³⁴

I have argued that historically, the law sought to promote the traditional nuclear family (one mother and one father, who are married) when making determinations of legal parenthood. This approach has recently been supplanted, in cases of natural reproduction, by a focus on the genetic connection between father and child. However, I have argued that these developments occurred within an overarching framework premised upon a binary, two-parent model. Moreover, I have observed

¹³⁴ Jackson, 'What is a Parent?', in Diduck and O'Donovan (eds.), Feminist Perspectives on Family Law, at 60.

¹³⁵ Through the usage of the *pater est* presumption, which was influenced by the consequences of the statuses of legitimacy and illegitimacy.

that the emphasis on genetic connection has not been evident in the contexts of assisted reproduction and surrogacy. In these contexts, the existence of marital relationships and the consent of individual men are privileged. Thus, I argued that a binary two-parent model and the nuclear family continue to underpin the framework for determining legal parenthood in cases of assisted reproduction, even although these provisions have been reformed to include legal parenthood for both members of same-sex couples.

The next chapter will build upon this identification of a binary, two-parent model within the attribution of legal parenthood by considering the legal understanding of the parental role and arguing that this understanding is premised upon the traditional, gendered roles of 'mother' and 'father' of the nuclear family model.

Chapter 6: The Legal Understanding of the Parental Role

Introduction

This chapter will contend that the legal understanding of the parental role is dependent upon the 'natural' constructions of the gendered parenting roles of 'mother' and 'father', derived from the traditional nuclear family model. The chapter will argue that the law's starting point is an overarching conception of the parental role, which encompasses all parents. However, I will argue that this overarching 'parental role' remains opaque within legal discourse and judicial interpretation; in sharp contrast to the traditional gendered parenting roles, the role of 'parent' lacks any 'natural' or 'common-sense' construction. I will argue that, consequently, it is the gendered roles which shape the legal understanding of the role of the 'parent' and through this I will show the continuing significance of the idealised image of the nuclear family.

This chapter will illustrate the influence of the gendered parenting roles of 'mother' and 'father' by examining how the law constructs the parental role. In section 6.1, I will explore the understanding of the 'parent' within the law; considering the statutory concepts of 'parental responsibilities' and 'parental rights' and suggesting

⁻

¹ See e.g. Children (Scotland) Act 1995, which details 'parental responsibilities', s.1, and 'parental rights', s.2, in an entirely gender-neutral form. See further s.3 (1) Children Act 1989 for evidence of similar gender-neutral language in English law.

² The association of these gendered roles with the nuclear family and the public/private divide was discussed above throughout Chapter 3, particularly at subsection 3.1.B, 'The Development of the 'Public/Private' Divide' and subsection 3.3.A, 'The Legal Subject and the Nuclear Family'.

³ Recent research suggests that in 2011 the proportion of families in the UK with two full time earners had risen to 29%, from 26% in 2001, while the proportion where the father works full time and the mother works part-time had fallen to 31%, from 37% in 2001. The proportion of sole male breadwinners remained stable at 22% throughout the period, whereas the total proportion of families where women were the main earner was only 12%. This suggests that the traditional gendered roles are still influencing the practices of a significant proportion of families in contemporary society, 'Modern Fatherhood: Fathers, Work and Families in the 21st Century', available at - http://www.modernfatherhood.org/themes/fathers-and-work/?view=key-facts-and-figures.

⁴ s.1 Children (Scotland) Act 1995, see also s.3 Children Act 1989.

⁵ s.2 Children (Scotland) Act 1995, see also s.3 (1) Children Act 1989.

that these provide limited specific guidance as to the role of the 'parent'. In section 6.2, I will argue that when the parental role is considered and interpreted by the judiciary, the gap created by this lack of guidance is filled by relying upon the readily identifiable and easily understandable archetypes of the traditional, gendered parenting roles of 'mother' and 'father'. Finally, in section 6.3, I will consider the legal understanding of lesbian parenthood, which provides a situation where one parental role (that of the non-gestational female 'parent') is explicitly described in legislation as the 'parent'. I will argue that the lack of exposition or detailed consideration of the role of the non-gestational female 'parent', in this context where neither gendered role is available, further illustrates the privileging of the traditional gendered parenting roles of the nuclear family.

6.1. What is the 'Parent' in the Law?

Diduck observes that '[l]aw "speaks" of parents and spouses rather than of mothers, fathers, husbands, wives.' It is evident that the government consistently emphasises the importance of parenting using gender-neutral language; for example, '[g]ood parenting therefore reduces the risks that children experience poor behavioural outcomes, criminality and anti-social behaviour' and '[p]arenting has a greater impact on children's wellbeing, learning and development than anything else.' However, such statements are generally expressed in vague terms and consequently

_

⁶ s.42-47 Human Fertilisation and Embryology Act 2008.

⁷ Diduck, *Law's Families*, at 41.

⁸ 'Supporting Families in the Foundation Years', (Department for Education, October 2011), at 36, Para 73, available at -

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/184868/DFE-01001-2011_supporting_families_in_the_foundation_years.pdf.

9 'Positive for Youth: A New Approach to Cross-Government Policy for Young People Aged 13 to 19',

⁹ 'Positive for Youth: A New Approach to Cross-Government Policy for Young People Aged 13 to 19' (Department for Education, February 2010), at 13, Para 3.7, available at https://consumption.education.gov.uk/publications/eOrderingDownload/DFE-00133-2011.pdf.

¹⁰ See e.g. 'Support for All: The Families and Relationships Green Paper', (Department for Children, Schools and Families, January 2010), for a further example of the vague, gender-neutral language used to describe parenting, at 56, Para 3.2, which stated, '[p]arents are the most profoundly important people in the world for babies and younger children and remain hugely significant to children as they grow up. Good parenting is crucial for children's outcomes and can protect them against other disadvantages.' Available at -

 $[\]underline{https://www.education.gov.uk/consultations/downloadableDocs/Families\%20Green\%20Paper\%20fina\ 1\%20pdf.pdf.}$

do not provide any substantive insight into the understanding of the role of the 'parent', nor any guidance as to the specific details and characteristics of that role.

Similarly, legislation often utilises the gender-neutral terminology of 'parent', ¹¹ most obviously through the concepts of 'parental responsibilities', ¹² and 'parental rights'. ¹³ Notably, those people who can be granted parental responsibilities and rights by the law are a wider group than merely the legal parents of a child. ¹⁴ Although legal parents ¹⁵ automatically possess parental responsibilities and rights, it is also possible for 'natural fathers', who do not have automatic responsibilities and rights to acquire them. ¹⁷ The statutory definitions and judicial interpretation of these terms should provide some indication of the legal understanding of the role and functions of the 'parent'; however, I will argue that there is an absence of clarity about the meaning of these terms and consequently about the characteristics of the parental role. I will argue, therefore, that when the law uses the gender-neutral language of the 'parent', this role is judicially interpreted by defaulting back to the 'natural' and 'common sense' constructions of the archetypical gendered parenting roles of 'mother' and 'father', derived from the traditional nuclear family.

In this section, then, I will consider the role of the 'parent', suggesting that there is not a clear understanding of this overarching parental role, independent from the

.

¹¹ See e.g. the amendment of s.1 Children Act 1989 by s.11 Children and Families Act 2014, which states, 'as respects each parent...to presume, unless the contrary is shown, that involvement of that parent in the life of the child concerned will further the child's welfare'.

¹² s.1 Children (Scotland) Act 1995, see also s.3 Children Act 1989.

¹³ s.2 Children (Scotland) Act 1995, in English law 'parental responsibility' is defined as including 'rights', s.3 (1) Children Act 1989.

¹⁴ Under English Law there is a specific procedure (s.4A Children Act 1989) for step-parents to acquire parental responsibilities and other individuals require 'the leave of the court' (s.10 Children Act 1989) to make such applications. In Scotland, under s.11 (3) (a) (i) Children (Scotland) Act 1995, the court is able to grant orders for 'parental responsibilities and rights' to any person, 'who - not having, and never having had, parental responsibilities or parental rights in relation to the child, claims an interest'.

¹⁵ 'Legal parents' have such responsibilities and rights in terms of the following provisions, s.3 (1)(a) Children (Scotland) Act 1995, for mothers, s.3 (1)(b) Children (Scotland) Act 1995 for married fathers and s.23 (2) Family Law (Scotland) Act 2006 for unmarried fathers whose name is on the birth certificate. See s.2 Children Act 1989 for the relevant English Law provision.

¹⁶ This is the term used by the legislation; it is taken to mean unmarried 'genetic' fathers without parental responsibilities and rights.

¹⁷ Fither by agreement with the delivery of th

¹⁷ Either by agreement with the child's mothers, s.4 (1) Children (Scotland) Act 1995, or through the granting of a court order to that effect, s.11 Children (Scotland) Act 1995. See s.4 (1) Children Act 1989 for the relevant English Law provisions for unmarried fathers.

traditional gendered roles. The statutory concepts of parental responsibilities and parental rights will be considered and I will suggest that these concepts are defined in vague terms and consequently offer limited guidance as to the role of the 'parent' (subsection 6.1.A). I will argue that the law relies instead upon the 'natural' gendered roles of 'mother' and 'father' when constructing the role of the 'parent' (subsection 6.1.B).

6.1.A. The Content of Parental Responsibilities and Parental Rights

The approach taken to defining the content of parental responsibilities differs under the relevant Scottish and English legislation. However, I will observe that the approaches in both jurisdictions provide a general, overarching definition of the concept, which does not provide significant specific guidance as to the role of the 'parent'.

6.1.A.1. Scotland

In Scotland, the relevant legislation defines both 'parental responsibilities' and 'parental rights'. The former are defined as including two general 'responsibilities', which are, 'to safeguard and promote the child's health, development and welfare' and, 'to provide, in a manner appropriate to the stage of development of the child - (i) direction; (ii) guidance, to the child.' These two responsibilities overlap and are expressed in a relatively vague and universal way, providing for general rather than specific guidance to parents, and are designed to cover virtually every aspect of a child's upbringing. The legislation then defines the more specific responsibility, 'if the child is not living with the parent, to maintain personal relations and direct

_

¹⁸ s.1 (1) (a) Children (Scotland) Act 1995.

¹⁹ s.1 (1) (b) Children (Scotland) Act 1995.

s.1 (4) Children (Scotland) Act 1995 states, '[t]he parental responsibilities supersede any analogous duties imposed on a parent at common law.' Thus the statutory definition required to provide for the entirety of the child's upbringing.

contact with the child on a regular basis.'²¹ I suggest that this emphasis on maintaining contact should be viewed in the context of the law's understanding that, in the event of the breakdown of the relationship between the parents, the continuation of both parent/child relationships is prima facie beneficial for the child.²²

'Parental rights' are defined as existing to, 'enable [the parent] to fulfil his [or her] parental responsibilities'. ²³ Therefore these rights seem to be conceptualised as subordinate to, and as flowing from, the associated responsibilities. ²⁴ The statute provides for the rights; '(a) to have the child living with him or otherwise to regulate the child's residence; [and] (b) to control, direct or guide, in a manner appropriate to the stage of development of the child, the child's upbringing. ²⁵ As with the definition of parental responsibility, I suggest that there seems to be a generality to the definition of these rights. The second right essentially mirrors one of the parental responsibilities, which further emphasises that the rights are conceptualised as being subsidiary to the responsibilities. ²⁶ Therefore, I suggest that the generality of these definitions does not provide significant insight into the legal understanding of the role of the 'parent'.

-

s.1 (1) (c) Children (Scotland) Act 1995, the section defines one additional specific responsibility, s.1 (1) (d) Children (Scotland) Act 1995, 'to act as the child's legal representative'. However, it is suggested that this provides a very specific responsibility which addresses the issue of the lack of legal capacity of the child.

²² See e.g. the judgment of Lord Rodger in *White v White* 2001 SLT 485, who stated at 489, '[t]he point of reference to which they have regard - and which they take because it represents the consensus of society - is that "it may normally be assumed that the child will benefit from continued contact with the natural parent".' The issue of contact will be explored in more detail below at subsection 6.2.C.1, 'The Significance of Contact with the Non-Resident Parent'.

s.2 (1) Children (Scotland) Act 1995. See further, Scottish Law Commission, 'Report on Family Law', (No 135, May 1992), Para 2.1, at 3, which states, 'it would enable the law to make it clear that parental rights were not absolute or unqualified, but were conferred in order to enable parents to meet their responsibilities'.

²⁴ And as such the more specific 'responsibilities' of subsections (c) and (d), detailed above, are repeated as 'rights'.

^{25*} s.2 (1) Children (Scotland) Act 1995, J. M. Thomson, *Family Law in Scotland*, (7th edition, Bloomsbury Professional, 2014), at 251, states that, 'the nature of a parental right alters as the child matures: beginning with the right to take decisions on the child's behalf, it becomes, in time, a right merely to give guidance to the child.'

²⁶ In *Gillick v West Norfolk and Wisbech Area Health Authority* [1986] AC 112, at 170, Lord Fraser observed, 'parental rights to control a child do not exist for the benefit of the parent. They exist for the benefit of the child and they are justified only in so far as they enable the parent to perform his duties towards the child'.

6.1.A.2. England and Wales

In England and Wales the statutory definition is even less precise; it defines 'parental responsibility' as 'all the rights, duties, powers, responsibilities and authority which by law a parent of a child has in relation to the child and his property.'²⁷ This vague definition has left the content of parental responsibilities to be established and considered by the courts,²⁸ which has resulted in contradiction and apparent divergence between competing authorities.²⁹ Black LJ has observed that, '[p]arental responsibility can be a difficult concept to grasp, particularly when it comes to the details of how it works in practice',³⁰ and Bridgeman has commented that, 'the concept of parental responsibility has been developed into a confused, contradictory concept with little meaning in relation to the responsibility of caring for children.'³¹ This confusion about the meaning of parental responsibilities is illustrated by the contrast between judicial descriptions of parental responsibility as a 'stamp of approval', or 'essentially an acknowledgment of status', and other decisions which suggest parental responsibility should not be granted unless it could be exercised effectively. This judicial approach to parental responsibilities has led Herring to

_

²⁷ s.3 (1) Children Act 1989. In this definition 'parental rights' are explicitly included as part of 'parental responsibilities'

^{&#}x27;parental responsibilities'.

28 Bainham, 'Parentage, Parenthood and Parental Responsibility: Subtle, Elusive, Yet Very Important Distinctions' in Bainham, Day Sclater and Richards (eds.), What is a Parent?: A Socio-Legal Analysis, has suggested, at 35, that this approach presupposes, 'some knowledge of the effects of being a parent which the courts have formulated over a long period of time at common law.'

²⁹ Stephen Gilmore, Jonathan Herring and Rebecca Probert, 'Introduction: Parental Responsibility - Law, Issues and Themes' in Rebecca Probert, Stephen Gilmore and Jonathan Herring (eds.), *Responsible Parents and Parental Responsibility*, (Hart, 2009), at 12, observe that in English law, 'the concept of parental responsibility is far from straightforward.'

³⁰ T v T (Shared Residence) [2011] 1 FCR 267, at Para 23.

³¹ Jo Bridgeman, 'Parental Responsibility, Responsible Parenting and Legal Regulation' in Jo Bridgeman, Craig Lind and Helen M. Keating (eds.), *Responsibility, Law and the Family*, (Ashgate, 2008), at 237.

³² Ward LJ in *Re S (A Minor) (Parental Responsibility)* [1995] 2 FLR 648, at 657, see further John Eekelaar, 'Rethinking Parental Responsibility' [2001] 31 (6) Fam. Law 426.

³³ Thorpe LJ in Re H (A Child) (Parental Responsibility) [2002] EWCA Civ 542, at Para 15.

³⁴ See e.g. *M v M (Parental Responsibility)* [1999] 2 FLR 737, where parental responsibility was denied on the basis that the father lacked the mental capacities to make decisions on behalf of the child, *Re G (A Child) (Parental Responsibility Order)* [2006] 2 FLR 1092, where the first instance judge had purported to grant 'suspended parental responsibility', which was rejected by the Court of Appeal, with Hedley J stating, at 1096, 'to make a parental responsibility order and then effectively draw all its teeth is something that would be a most unusual thing to do' and *Re B (Role of the Biological Father)* [2008] 1 FLR 1015, where parental responsibility was denied to a 'known donor' on the basis that his use of parental responsibility would diminish the parental status of the lesbian nuclear family.

observe that, 'there is a real tension in the case law as to whether parental responsibility is about real decision-making power, or whether it is of more symbolic value', 35 and Harris and George to comment that, 'it has become increasingly uncertain what the purpose and effect of parental responsibility is.'36 I contend that, as a result of the lack of clarity within the authorities as to the meaning of 'parental responsibility' and the extent of the practical decision-making power that derives from this responsibility, this statutory concept does not provide clarification of the legal understanding of the role of the 'parent'.

Regardless of the specific differences between Scots or English law, the approaches taken in both jurisdictions provide only vague definitions of 'parental responsibilities', and neither approach offers significant substantive guidance regarding the legal understanding of the role of the 'parent'.

6.1.B. The Role of the 'Parent'

Bainham, Day Sclater and Richards ask, 'can parenting ever be a truly gender-neutral activity?'³⁷ Parental responsibilities are defined using gender-neutral language and therefore do not differ between mothers and fathers, suggesting that the law's starting point is an overarching understanding of the role of the 'parent', which provides the framework for both of the gendered parental roles.³⁸ However, the lack of a clear definition of 'parental responsibilities' in either Scotland or England and Wales results in limited guidance being provided as to the nature of that overarching parental role. Reece comments that, 'parental responsibility is no longer

-

³⁵ Herring, *Family Law*, at 436, see also Helen Reece, 'The Degradation of Parental Responsibility' in Probert, Gilmore and Herring (eds.), *Responsible Parents and Parental Responsibility*, who suggests, at 102, 'there is a trend away from parental responsibility as parental authority towards parental responsibility as nothing more than official approval'.

³⁶ Peter G. Harris and Robert H. George, 'Parental Responsibility and Shared Residence Orders: Parliamentary Intentions and Judicial Interpretation' [2010] 22 (2) Child and Family Law Quarterly 151, at 170. The authors suggest that the judicial interpretation of parental responsibility has not reflected the concept as it was originally envisaged in the Children Act 1989.

³⁷ Andrew Bainham, Shelley Day Sclater and Martin Richards, 'Introduction' in Bainham, Day Sclater and Richards (eds.), *What is a Parent?: A Socio-Legal Analysis*, at 2.

³⁸ The impact of the use of gender-neutral language will be considered below, in the context of contact with the non-resident parent, in subsection 6.2.C.2, 'Contact and Gender Neutrality'.

predominantly about parental authority or decision-making.'³⁹ I argue that due to the lack of detail and specification provided by the statutory definition and judicial interpretation of 'parental responsibilities', this concept does not assist us significantly in our understanding of the law's construction of the role of the 'parent'.

Interestingly, the 'parent' is primarily defined by its abstract, statutory 'responsibilities' and 'rights', this parent is constructed as a rational and autonomous figure, arguably embodying the influence of the orthodox legal subject and its values. ⁴⁰ I suggest that the influence of these values within the construction of the 'parent' illustrates why any alternative, relational understanding of the subject, associated with the 'ethic of care', ⁴¹ does not have a substantial normative influence in the legal construction of this parental role. As mentioned above in Chapter 3, ⁴² the law is premised upon a particular understanding of the individual (the legal subject) and I suggest that it is unsurprising that the legal understanding of the role of the 'parent' reflects the values of that legal subject. However, I argue that recognition of the influence of the orthodox construction of the legal subject upon the understanding of the overarching parental role does not provide greater clarity as to the nature, features or characteristics of that role within the law, because the legal subject itself represents an abstract standard.

Moreover, it is apparent that these definitions are not designed to prescribe to parents how they should perform their parenting⁴³ or to interfere with the day to day detail of

³⁹ Reece, 'The Degradation of Parental Responsibility' in Probert, Gilmore and Herring (eds.), *Responsible Parents and Parental Responsibility*, at 94.

⁴⁰ Which were set out above and described as 'liberal values', at Chapter 3, subsection 3.1.C.1, 'The Normative Nature of the Orthodox 'Legal Subject''.

⁴¹ See e.g. Herring, 'Where are the Carers in Healthcare Law and Ethics?', Herring, 'The Disability Critique of Care' and Herring and Foster, 'Welfare Means Relationality, Virtue and Altruism'.

⁴² See Chapter 3 above, particularly subsection 3.2.A.1, 'The Legal Subject Favours Specific Individuals'.

⁴³ As the recent judgment of the Supreme Court in *Christian Institute v Lord Advocate* [2016] UKSC 51, at Para 73, observed, '[d]ifferent upbringings produce different people...Within limits, families must be left to bring up their children in their own way.' This decision held that the provisions of Part 4 of the Children and Young People (Scotland) Act 2014, concerning the Scottish Government's plan to introduce a 'named person' for every child in Scotland, were not within the legislative competence of the Scottish Parliament.

that parenting.⁴⁴ Baroness Hale of Richmond has stated that 'it is important in a free society that parents should be allowed a large measure of autonomy in the way in which they discharge their parental responsibilities.'⁴⁵ Thus, parental responsibilities and rights are understood as merely providing an overall *framework* for parenting. In *Re L (A Child) (Care: Threshold Criteria)*,⁴⁶ Hedley J stated, '[s]ociety must be willing to tolerate very diverse standards of parenting, including the eccentric, the barely adequate and the inconsistent.'⁴⁷ I suggest that the judicial acknowledgment of parental autonomy, as well as the recognition of the diversity of acceptable parenting standards and practices further contributes to the lack of clarity as to the legal understanding of the role of the 'parent'.

Thus, while the law's starting point is the overarching concept of the 'parent', I argue that the legal understanding of the role tends to default to the easily understandable, 'natural' and 'common sense' constructions of the archetypical gendered parenting roles. Diduck has observed that, '[t]he importance of father(ing) and mother(ing) to a child's welfare if not always clear for the law, is at least meaningful. The role, on the other hand, of a de-gendered "parent" is opaque and, as yet, imaginary.' Over the course of the rest of the chapter, I will argue that the legal understanding of role of the 'parent' is dependent and reliant on the 'natural' constructions of the traditional gendered roles of 'mother' and 'father'.

.

⁴⁴ This lack of interference with day-to-day parenting practices illustrates the continuing significance of the public/private divide within UK law, considered above in Chapter 3, subsection 3.1.B.1, 'The 'Public/Private' Divide within the Law'.

⁴⁵ Baroness Hale of Richmond in *R* (*On the Application of Williamson*) *v Secretary of State for Education and Employment* [2005] 2 AC 246, at 271, an application for judicial review, which considered whether the ban on corporal punishment in all schools violated the ECHR Art.9 right to religious freedom of the head teachers, teachers and parents at four independent 'Christian' schools. ⁴⁶ [2007] 1 FLR 2050.

⁴⁷ ibid, at 2063. This statement reflects the oft-quoted language of Lord Templeman in *Re KD (A Minor) (Ward: Termination of Access)* [1988] AC 806, at 812, '[i]t matters not whether the parent is wise or foolish, rich or poor, educated or illiterate, provided the child's moral and physical health are not endangered. Public authorities cannot improve on nature.'

⁴⁸ Diduck, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood', at 464.

6.2. The Gendered Parenting Roles of the Nuclear Family

As discussed above in Chapters 2 and 3, within the traditional nuclear family model the gendered parenting roles of mother and father are constructed with emphasis on their differing functions and they are understood as inherently possessing different characteristics.⁴⁹ These distinct roles were premised upon the historical separation of gender roles envisaged within the public/private divide.⁵⁰ O'Donovan has argued:

The heterosexual dyad is a union of opposites, a bi-polar model of attributes, bodies, gestures, conduct, aptitudes and expectations of what a gendered person is and what a gendered person does. The notion of parenthood is not gender neutral. In culture, in popular morality, in juridical and "psy" discourses, parents are mothers and fathers with all that these distinct terms imply.⁵¹

Under these traditional gendered constructions, the mother is conceptualised as 'the homemaker' and viewed as the primary care figure for children. Furthermore, as previously mentioned, motherhood is constructed and presented as a 'natural' and indivisible status within the law.⁵² Thus, I suggest that the significance of 'care' is understood through the prism of the gendered parenting roles and the imagery of the mother as primary carer retains normative resonance, regardless of any changes in familial caring practices.⁵³ In contrast, the law has a more 'fragmented', construction of the father, as a result of a complicated understanding of the development of that role. Historically, the role of the father was constructed

⁻

⁴⁹ See above in Chapter 2, section 2.2, 'The Law's 'Definition(s)' of 'Family''.

⁵⁰ As discussed above throughout Chapter 3, but particularly subsection 3.1.B, 'The Development of the 'Public/Private' Divide' and subsection 3.3.A, 'The Legal Subject and the Nuclear Family'.

⁵¹ O'Donovan, Family Law Matters, at 61.

⁵² See above at Chapter 5, subsection 5.2.A, 'Motherhood'.

⁵³ The empirical research, set out below in subsection 6.2.B.1, 'The Shift in the Construction of Fatherhood', suggests that men continue to spend significantly less time per day than women on both childcare and housework, see e.g. ONS, 'Time Use Survey 2005', (July 2006) and therefore the traditional gendered parenting roles continue to exert significant influence on familial practices.

⁵⁴ Sheldon and Collier, *Fragmenting Fatherhood: A Socio-Legal Study*.

⁵⁵ See e.g. Coltrane, Family Man: Fatherhood, Housework and Gender Equality, Collier, Masculinity, Law and the Family, Graeme B. Wilson, 'The Non-Resident Parental Role for Separated Fathers: A

primarily in economic terms, as 'the breadwinner'. Changing demographics and practices⁵⁶ have led to a more contradictory construction of the role,⁵⁷ with ongoing attempts to define a specific and distinct male role ('the father as carer') in the caring, nurture and development of children.⁵⁸ However, I submit that the traditional construction continues to exert significant influence and consequently any alternative understanding of the role of the 'father' has yet to be fully developed within legal understanding, nor has it substantially influenced social attitudes and familial practices.⁵⁹

The following section will consider the influence of these traditional gendered parenting roles of the mother (subsection 6.2.A) and the father (subsection 6.2.B) within the legal understanding of the role of the parent. Thereafter, this section will examine the significance of these gendered roles within the welfare test, focusing upon the support for contact with the non-resident parent after parental separation (subsection 6.2.C). I will argue that the traditional constructions of these roles retain normative significance, because they remain understood by the judiciary as the 'common-sense' and 'natural' constructions of the parenting roles.

Review' (2006) 20 (3) International Journal of Law, Policy and the Family 286 and Brid Featherstone, 'Taking Fathers Seriously' (2003) 33 (2) British Journal of Social Work 239.

^{&#}x27;Women in the Labour Market' (September 2013), available at http://www.ons.gov.uk/ons/dcp171776 328352.pdf, shows that the percentage of women aged 16-64 in employment has risen from 53% in 1971 to 67% in 2013. This corresponds to a decline in the numbers of men in work from 92% to 76% over the same time frame. Interestingly however, 42% of those women are in part-time employment compared to only 12% for men.

⁵⁷ This shift in construction has also, quite pointedly, led to a media backlash against the implied suggestion that traditional 'fathers' are no longer a necessary component for families and the development of a popular narrative suggesting that society is undergoing a 'Crisis of Masculinity'; see e.g. Glen Poole, 'How Tackling the 'Crisis of Masculinity' Creates a Crisis for Feminism', (The 15th Guardian, London, May 2013), available http://www.theguardian.com/commentisfree/2013/may/15/why-crisis-masculinity-feminism Kunal Dutta, 'Masculinity in crisis: 'Masculinity in Crisis: 'There is a Battle Going on Inside Us That is Never Discussed" (The Independent, London, 19th May 2013), available at http://www.independent.co.uk/life-style/love-sex/masculinity-in-crisis-there-is-a-battle-going-oninside-us-that-is-never-discussed-8622992.htm.

See e.g. Hodson (ed.), Making Men into Fathers: Men, Masculinities and the Social Politics of Fatherhood and Dienhart, Reshaping Fatherhood: The Social Construction of Shared Parenting. It is arguable that the development of a construction of the role as 'father as carer' shows the normative influence of the literature on 'care'.

⁵⁹ This is reflected in social attitudes, Scott and Clery, 'Gender Roles: An Incomplete Revolution?', British Social Attitudes Survey 30, (2013), observe, at 127, '[t]he overall story is that there has been very little change over the past two decades in the percentage of couple households dividing household responsibilities along traditional gender lines.

6.2.A. What is the 'Mother' in the Law?

Historically the mother was not viewed as possessing any legal role, which reflected the influence of the public/private divide⁶⁰ and women's resultant exclusion from the public sphere of society as legal subjects.⁶¹ Blackstone observed that, 'for a mother, as such, is entitled to no power, but only to reverence and respect.'⁶² This remained the legal understanding of the role,⁶³ until the Guardianship of Infants Act 1925 stated that courts, 'shall not take into consideration whether...the claim of the father...is superior to that of the mother, or the claim of the mother is superior to that of the father.'⁶⁴ This legislation first statutorily established that 'the welfare of the child'⁶⁵ was to be the paramount consideration in decisions concerning children.

6.2.A.1. The Judicial Preference for Mothers as the Carers of Children

There has been a consistently expressed judicial understanding that in a dispute between the parents over the residence of the children, the mother was the best person to raise the children, ⁶⁶ and this was considered to be particularly true in the case of younger children. ⁶⁷ This preference was judicially presented as being based upon 'common sense' or reflecting 'nature'. ⁶⁸ Over 50 years ago in *Re B (An*

⁶⁰ As set out above in Chapter 3, subsection 3.1.B, 'The Development of the 'Public/Private' Divide'.

⁶¹ Which was illustrated in the 'persons cases', discussed above in Chapter 3, subsection 3.2.C, "The Persons Cases': The Rejection of Women as Legal Subjects'.

⁶² Sir William Blackstone, Commentaries on the Laws of England, Volume 1, at 453.

⁶³ The Guardianship of Infants Act 1886 established that mothers could become sole guardians of children in the event of the death of the father.

⁶⁴ s.1 Guardianship of Infants Act 1925.

⁶⁵ ibid, the section further states that when making decisions about the 'custody and upbringing of children', courts, 'shall regard the welfare of the infant as the first and paramount consideration'. The so-called 'welfare principle' continues to govern disputes concerning children, see e.g. s.1 (1) Children Act 1989 and s.11 (7) Children (Scotland) Act.

⁶⁶ Although in *W v W and C* [1968] 1 WLR 1310, Lord Denning MR, at 1312, referred to, 'the general principle that a boy of this age, some eight years of age, is, on the whole, other things being equal, better to be with his father.' However, the existence of such a principle was rejected by the Court of Appeal in *Re C (A) (An Infant)* [1970] 1 WLR 288. See further *Re N* [2010] 1 FLR 272 for a more recent decision which dismisses the existence of such a presumption.

⁶⁷ In *Re W (A Minor) (Residence Order)* [1992] 2 FLR 332, Lord Donaldson MR in relation to very young children, stated, at 336, 'I think there is a rebuttable presumption of fact that the best interests of a baby are served by being with its mother, and I stress the word "baby".'

⁶⁸ See e.g. Donovan LJ in *Re B (An Infant)* [1962] 1 All ER 872, at 875.

Infant), 69 Lord Evershed MR stated that 'as a matter of human sense a young child is better with its mother and needs a mother's care.'70 The subsequent passage of 30 years did not lead to any significant alteration in judicial language, as illustrated in Re H (A Minor: Custody), 71 where O'Connor LJ stated 'it is well recognised that the natural person to have the care and control of a little child is the mother of the child.'72 In the Scottish House of Lords case of Brixey v Lynas, 73 Lord Jauncey of Tullichettle observed that this understanding amounted to 'recognition of a widely held belief based on practical experience and the workings of nature.⁷⁴ I suggest that invoking 'practical experience' and 'nature' to justify the preference for mothers as the primary carers of children is a reflection of the 'natural' gender roles of the traditional, nuclear family. ⁷⁵ In the more recent decision of Re T (A Child), ⁷⁶ Wall LJ stated that 'as a matter of practice, very small babies are usually cared for by their mothers.'77 The reference to 'a matter of practice' illustrates the continuing judicial recognition of the social reality⁷⁸ that the vast majority of young children simply are cared for by their mothers.⁷⁹ It seems, therefore to remain a judicial understanding that mothers are usually best suited to the role of the primary carer of children

⁶⁹ ibid.

⁷⁰ ibid, per Lord Evershed MR, at 873, see further *Re L (Infants)* [1962] 3 All ER 1.

⁷² ibid, per O'Connor LJ, at 53. For further examples of such judicial language, see *Re A (A Minor)* (Custody) [1991] 2 FLR 394, Re W (A Minor) (Residence Order) [1992] 2 FLR 332, Re S (A Minor) (Custody: Children's Welfare) [1991] 2 FLR 388. 73 1997 SC (HL) 1.

⁷⁴ ibid, per Lord Jauncey, at 6, see further e.g. Re W (Residence) [1999] 2 FLR 390 and Re A (Children: 1959 UN Declaration) [1998] 1 FLR 354. His lordship also stressed, at 6, that there was, 'neither a presumption nor a principle' in favour of mothers being awarded residence. See Jonathan Herring and Oliver Powell, 'The Rise and Fall of Presumptions Surrounding the Welfare Principle' [2013] 43 (5) Fam. Law 553, at 556, who observe that, 'any presumption that children are best cared for by mothers has been whittled away to vanishing point.'

⁷⁵ This reflects the construction of the nuclear family as 'natural', considered above in Chapter 3, subsection 3.3, 'The Nuclear Family as the Natural Model of 'Family''.

^[2005] EWCA Civ 1397.

⁷⁷ ibid, per Wall LJ, at Para 5.

⁷⁸ It could also be argued that this approach reflects the importance placed upon not disturbing the 'status quo', see s.1 (3) (c) Children Act 1989.

⁷⁹ See e.g. Joan Hunt and Alison MacLeod, 'Outcomes of Applications to Court for Contact Orders After Parental Separation or Divorce', (Ministry of Justice, September 2008), at 239, which shows that fathers accounted for 91.6% (265 out of 289) of the non-resident parents making applications for contact in the research sample, illustrating the continued preference for mothers in the vast majority of cases, available at - http://dera.ioe.ac.uk/9145/1/outcomes-applications-contact-orders.pdf.

(particularly young children)⁸⁰ because they are understood as inherently possessing characteristics and traits that men are not. This preference for mothers as the primary carer of children illustrates the continuing normative influence of the archetypical gendered parenting role of the 'mother' derived from the traditional nuclear family.

6.2.A.2. The 'Natural' Construction of Motherhood

In view of this, I observe that the law places normative significance upon the role of the mother as the 'natural' primary carer of children. The rhetorical positioning of motherhood as a 'natural' status, ⁸¹ which is primarily associated with the care of children, is evident from *Re S (A Minor) (Custody: Children's Welfare)*, ⁸² where Butler-Sloss LJ stated that, 'it is natural for young children to be with mothers.' This emphasis on the 'natural' construction of motherhood is supported by Lord Jauncey's observation in *Brixey v Lynas*, ⁸⁴ that, '[n]ature has endowed men and women with very different attributes and it so happens that mothers are generally better fitted than fathers to provide for the needs of very young children.' ⁸⁵

The significance given to the role of the mother was made apparent in the House of Lords decision in *Re G (Children) (Residence: Same-Sex Partner)*, ⁸⁶ particularly in Lord Scott's comment that, '[m]others are special'⁸⁷ and Baroness Hale's statement that, '[h]er contribution to the welfare of the child is unique.'⁸⁸ Indeed, her ladyship

84 1997 SC (HL) 1.

See e.g. Re K (Residence Order: Securing Contact) [1999] 1 FLR 583, where Hirst LJ, at 529, stated that, 'I fully recognise that to award a residence order of a 2-year-child in the father's favour is somewhat unusual'.

⁸¹ O'Donovan and Marshall, 'After Birth: Decisions about Becoming a Mother', in Diduck and O'Donovan (eds.), *Feminist Perspectives on Family Law*, provide a useful exploration of the feminist literature on the construction of motherhood. See above in Chapter 5, subsection 5.2.A, 'Motherhood'. ⁸² [1991] 2 FLR 388.

⁸³ ibid, at 390.

⁸⁵ ibid, per Lord Jauncey of Tullichettle, at 6.

⁸⁶ [2006] 2 FLR 629.

⁸⁷ ibid, per Lord Scott of Foscote, at 631.

⁸⁸ ibid, per Baroness Hale, at 641. Indeed, given the particular factual circumstances of this case (a dispute between a separated lesbian couple) it would not be terminologically inaccurate to state that two women were 'mothering' the children. Therefore, I suggest that a deliberately narrow construction of 'mother' was being employed within the judgments.

reflected the 'natural', gestational construction of motherhood earlier in her judgment in $Re\ G$, where she stated that:

[I]t also recognises a deeper truth: that the process of carrying a child and giving him birth (which may well be followed by breast-feeding for some months) brings with it, in the vast majority of cases, a very special relationship between mother and child, a relationship which is different from any other.⁸⁹

This construction of motherhood reflects back to the public/private divide and the traditional nuclear family, with mothers understood primarily in terms of their care for their (young) children. Diduck has powerfully observed that, '[t]he romantic "good" mother is self-sacrificing in her care for her child. She is most fundamentally a nurturer, because nature (or God) made her that way.'90 Within legal understanding, mothers remain constructed as part of the private sphere, viewed as the 'natural'91 carers of children, as wives, as the gate-keepers of the nuclear family and therefore as mothers, 'in every sense of that term'. 92 Consequently, women are generally constructed as 'mothers', rather than as 'parents', by the law. Indeed, it is for this reason I observe that the 'mother' does not share the characteristics of the rational, autonomous legal subject. 93 This reflects Diduck's suggestion that the, 'mother-child dyad resides firmly in the romantic private domain of instinct or love and lies in contrast with the rational legal world. 94 Thus, the law is content to affirm 'motherhood' as a 'natural', quasi-mystical relationship and view this solely as a result of gender. When constructing the role of the mother, the judiciary has tended

_

⁸⁹ Re G (Children) (Residence: Same-Sex Partner) [2006] 2 FLR 629, per Baroness Hale, at 641.

⁹⁰ Diduck, Law's Families, at 84.

⁹¹ This reflects the positioning of the nuclear family as the 'natural' model of family, discussed above in Chapter 3, subsection 3.3, 'The Nuclear Family as the Natural Model of 'Family'', see further e.g. Richard Collier, 'A Hard Time to Be a Father?: Reassessing the Relationship Between Law, Policy, and Family (Practices)' (2001) 28 (4) Journal of Law and Society 520, at 526, who has suggested that, '[m]otherhood has appeared at once both "natural" and yet also deeply problematic and dangerous in the way it has been constituted as in need of surveillance, regulation and discipline.'

⁹² In the words of Baroness Hale in Re G (Children) (Residence: Same-Sex Partner) [2006] 2 FLR 629, at 643.

⁹³ As discussed above in Chapter 3, subsection 3.1.C.1, 'The Normative Nature of the Orthodox 'Legal Subject'.

⁹⁴ Diduck, *Law's Families*, at 84.

to view the gendered role as offering a more valuable caring role, particularly in relation to the care of young children, 95 than is entailed by the gender-neutral language of 'parent'. 96 It is submitted that this repeated and consistent judicial emphasis on 'mothers' as the 'natural' carers of children illustrates the continuing significance of the gendered parenting roles, instead of an overarching understanding of the parental role.

6.2.B. What is the 'Father' in the Law?

Historically the role of the father was understood in terms of his power and authority over his wife and children and the 'rights of the father' were emphasised. 97 In the late 19th century in Re Agar-Ellis⁹⁸ Brett MR observed that, 'the father has control over the person, education, and conduct of his children until they are twenty-one years of age. That is the law.'99 This view of the complete authority of the father was combined with an understanding of the role as embracing what Blackstone referred to as the 'natural obligation of the father to provide for his children.' The role of the 'father' was the traditional gendered role of head of the household 101 and the 'breadwinner'. Thus, I observe that the gendered division of the roles of mother and father, derived from the public/private divide and the traditional nuclear family historically underpinned the legal understanding of the role of the father. While this common law authority of the father was effectively eliminated by the reforms of the Guardianship of Infants Act 1925, subsequent to the passage of this legislation the law continued to construct the role of the father on the basis of the traditional, gendered role of the 'breadwinner'. 102

⁹⁵ See e.g. Re W (A Minor) (Residence Order) [1992] 2 FLR 332.

⁹⁶ See below at section 6.3, 'The Role of Lesbian 'Mothers' or 'Parents", for how the 'natural' construction of the 'mother' impacts upon the understanding of the parenting of lesbian couples, where one member is the 'mother' and the other is explicitly labelled as the 'parent' by legislation.

⁹⁷ Historically this power of the father was over his legitimate children, it should be noted that a distinction between the role of married and unmarried fathers has been made by law consistently throughout history.

⁹⁸ (1883) 24 Ch. 317.

⁹⁹ ibid, per Brett MR, at 326.

¹⁰⁰ Sir William Blackstone, Commentaries on the Laws of England, Volume 1, at 447.

Reflecting to a lesser extent the Roman law concept of pater families, discussed above at Chapter 3, subsection 3.1.A.1, 'The Classical Societies'.

102 See e.g. Wachtel v Wachtel (No 2) [1973] 2 WLR 366 and other cases noted above in Chapter 3,

6.2.B.1. The Shift in the Construction of Fatherhood

More recently there has been an apparent shift in the construction of the role of the father, 103 which has been evident in government policy, 104 reflecting the understanding that, '[f]athers want - and increasingly are - becoming more involved in caring for their children'. Under this construction greater emphasis is being placed on the caring features of the role of father, 106 in addition to the traditional breadwinner role. The perception of this shift in the understanding of fatherhood is evident in judicial language, 107 as illustrated by the observation of Lord Nicholls in *White v White* 108 that:

Traditionally, the husband earned the money, and the wife looked after the home and the children. This traditional division of labour is no longer the order of the day. Frequently both parents work. Sometimes it is the wife who is the money-earner, and the husband runs the home and cares for the children during the day.¹⁰⁹

subsection 3.3.B.1, 'The Nuclear Family in Historical Judicial Reasoning'.

¹⁰³ See Lupton and Barclay, *Constructing Fatherhood: Discourses and Experiences*, for a consideration of the development of the role of fatherhood, both through an examination of the sociological and psychological literature, but also through analysis of popular representations of fatherhood in the media. See further e.g. Coltrane, *Family Man: Fatherhood, Housework and Gender Equality* and Collier, *Masculinity, Law and the Family*.

This shift was initially associated with the reforms and ideals of the New Labour government. It is suggested that this construction of 'involved fatherhood' has been largely reflected by the policies of subsequent governments.

^{105 &#}x27;Support for All: The Families and Relationships Green Paper', (January 2010), at 94, Para 5.12.

¹⁰⁶ See e.g. Equal Opportunities Commission, 'Fathers: Balancing Work and Family' (March 2003), where it was stated, at 2, 'fatherhood is in a state of change. The everyday, traditional role of providing economic support for the family now takes place alongside activities previously regarded as maternal', available at - http://www.fatherhoodinstitute.org/uploads/publications/285.pdf. See further e.g. Margaret O'Brien, 'Shared Caring: Bringing Fathers into the Frame', (Equal Opportunities Commission, 2005), available at - http://dera.ioe.ac.uk/5299/1/1.73363!shared_caring_wp18.pdf.

¹⁰⁷ See e.g. the observation of Lord Donaldson MR in *Re S (A Minor) (Custody: Children's Welfare* [1991] 2 FLR 388, at 392, that, '[w]hat is clear is that there is a change in the social order, in the organisation of society, whereby it is much more common for fathers to look after young children than in bygone days.'

¹⁰⁸ [2001] 1 AC 596.

¹⁰⁹ Ibid, per Lord Nicholls, at 605. Subsequently, Baroness Hale employed similar language in *Ghaidan v Godin-Mendoza* [2004] 2 AC 557, at 607, stating, '[t]he law now differentiates between husband and wife in only a very few and unimportant respects. Husbands and wives decide for themselves who will go out to work and who will do the homework and child care. Mostly each does some of each. The roles are interchangeable.'

However, in spite of this rhetoric suggesting that there has been fundamental change in family practice, relatively recent research evidence indicates that even though the traditional breadwinner/homemaker division is no longer the dominant form of family, 110 in the vast majority of families men are still the main earner 111 and women remain the primary carers of children. 112 As Scott and Clery comment, 'the public retains a view that there should be a gender divide in terms of caring responsibilities: the shift has been in accepting the idea that a mother works part-time, rather than not at all. 113 Therefore, I agree with Sheldon and Collier when they say that, '[t]he "father as breadwinner" model, and the masculinities with which it has been associated have not been supplanted in law. Rather, they exist alongside, and in tension with, the new ideology of "father as carer". 114 Smart has drawn upon Tronto's distinction between 'caring for' and 'caring about' to note that this shift in normative construction is not necessarily based upon a shift in the caring practices of fathers, observing that 'when fathers articulated their care about their children, even if they had never really cared for them, their utterances seemed to reverberate

1 .

¹¹⁰ 'Modern Fatherhood: Fathers, Work and Families in the 21st Century', referenced above at 'Introduction', suggests that the sole male breadwinner model only exists in 22% of families, see further e.g. Man Yee Kan, Oriel Sullivan and Jonathan Gershuny, 'Gender Convergence in Domestic Work: Discerning the Effects of Interactional and Institutional Barriers from Large-Scale Data' (2011) 45 (2) Sociology 234.

^{&#}x27;Trends in Fathers Work-Family Arrangements and Fathers Working Hours (2001-2011)', in the sample of families in this study, the mother was the 'main earner' in only 12% of families, available at http://www.modernfatherhood.org/wp-content/uploads/2014/01/Nuffield-PRESENTATION-FATHERS-FINAL-2 pdf

FATHERS-FINAL-2.pdf.

112 ONS, 'Time Use Survey 2005', (July 2006), suggests that men spend significantly less time per day than women on both childcare and housework, see further Crompton and Lyonette, 'Who Does the Housework? The Division of Labour within the Home', British Social Attitudes Survey 24. This research contains a survey on the extent to which social attitudes in the UK about the roles of men and women in the family have shifted in recent years.

¹¹³ Scott and Clery, 'Gender Roles: An Incomplete Revolution?', British Social Attitudes Survey 30, at 124. Indeed, at 125, the authors further note that 31% of people think that a male full time breadwinner, women full time homemaker is the best way to organise the family when children are under school age and 38% believe that father working full time while the mother works part-time is the best method, in contrast only 4% of people favour both parents working full-time.

¹¹⁴ Sheldon and Collier, Fragmenting Fatherhood: A Socio-Legal Study, at 136. See also Collier, 'A Hard Time to Be a Father?: Reassessing the Relationship Between Law, Policy, and Family (Practices)', Hodson (ed.), Making Men into Fathers: Men, Masculinities and the Social Politics of Fatherhood and Dienhart, Reshaping Fatherhood: The Social Construction of Shared Parenting.

¹¹⁵ See e.g. Joan Tronto, 'Women and Caring: What Can Feminists Learn about Morality from Caring?' in Alison Jaggar and Susan Bordo (eds.), *Gender/Body/Knowledge: Feminist Reconstructions of Being and Knowing*, (Rutgers University Press, 1989) and Tronto, *Moral Boundaries: a Political Argument for an Ethic of Care*.

around the courts with a deafening significance.' Thus, I contend that the increased emphasis on the idealised 'father as carer' role remains significantly influenced by replicating particular normative roles, rather than reflecting the reality of societal parenting practices.

6.2.B.2. Fatherhood Outside of the Traditional Nuclear Family

As a result of increased divorce, ¹¹⁷ relationship breakdown among cohabiting couples and the rise in lone-parent families, ¹¹⁸ the vast majority of which are headed by single mothers, ¹¹⁹ more fathering is now being performed outside the traditional two-parent, nuclear family ¹²⁰ and consequently without day-to-day contact with children. ¹²¹ On this point, Simpson, Jessop and McCarthy have observed that:

Whereas in conventional family settings fathers may be physically close to their children their role is apt to leave them emotionally distant from them. Conversely, and somewhat paradoxically, divorce usually results in fathers being physically separated but does create the possibility that kinds of emotional closeness can develop which might not have been possible in marriage. 122

¹¹⁶ Carol Smart, 'Losing the Struggle for Another Voice: The Case for Family Law' (1995) 18 (2) Dalhousie Law Journal 173, at 177. This will be further explored below at subsection 6.2.C.2, 'Contact and Gender Neutrality'.

¹¹⁷ See e.g. ONS, 'Divorces in England and Wales, 2012', showing the fluctuation of divorces from the 1930s onwards.

¹¹⁸ See e.g. ONS, 'Social Trends: Households and Families' (2011) No. 41, at 7, showing the increase of single-parent families to 16.2% of all families in 2010.

ibid, the research shows that lone-mother families amount to 14.1% of all families, therefore around 87% of single-parent families were headed by women.

In 2004 it was suggested that there were around 2 million non-resident fathers in the UK, Joan Hunt with Ceridwen Roberts, 'Family Policy Briefing 3: Child Contact with Non-Resident Parents', University of Oxford, available at - http://www.spig.clara.net/reports/hunt.pdf.

¹²¹ 'Judicial and Court Statistics 2011', (Ministry of Justice, June 2012), for England and Wales, state that there were 109,656 children involved in private law applications in 2011, of which around one third were applications relating to the residence and contact of those children, available at - https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/217494/judicial-court-stats-2011.pdf.

Bob Simpson, Julie A. Jessop and Peter McCarthy, 'Fathers After Divorce' in Andrew Bainham, Bridget Lindley, Martin Richards and Liz Trinder (eds.), *Children and Their Families: Contact, Rights and Welfare*, (Hart, 2003), at 215.

This change has resulted because non-resident fathers have no option but to take on the role of the primary carer in their more limited time with their children, ¹²³ rather than defaulting to the traditional gendered roles that still exert influence within parenting by heterosexual couples. ¹²⁴ However, the late 20th and early 21st century increase in the number of non-resident fathers also led to a renewed focus on the financial obligations of those fathers, ¹²⁵ which resulted in the controversial provisions of the (frequently restructured) Child Support Act 1991. ¹²⁶ Collier comments that '[t]he experience of "being a father" continues to involve, for most men, a temporal and spatial trade-off between the domains of work and family. ¹²⁷

Thus, I observe that the traditional construction of the role of the father as the 'breadwinner' still possesses significant normative influence, even on those fathers who are now living and 'fathering' outside the boundaries of the nuclear family. Moreover, I argue that this focus upon understanding and constructing a specific role for the 'father', in the day-to-day upbringing of children, shows that it is the gendered role which is shaping legal understanding, not an overarching conception of the role of the 'parent'.

6.2.C. The Gendered Parenting Roles within the Welfare Principle

The statutory language of the welfare principle provides that 'the child's welfare shall be the court's paramount consideration', 128 in any decision concerning the

-

¹²³ ibid, see further e.g. Carol Smart, 'The New Parenthood: Fathers and Mothers after Divorce' in Silva and Smart (eds.), *The New Family?*, S. Kielty, 'Similarities and Differences in the Experiences of Non-Resident Mothers and Non-Resident Fathers' (2006) 20 (1) International Journal of Law Policy and the Family 74 and Wilson, 'The Non-Resident Parental Role for Separated Fathers: A Review'.

¹²⁴ See e.g. Jacqueline Scott, 'Family and Gender Roles: How Attitudes are Changing?' (GeNet Working Paper, No 21, 2006), available at - http://www.genet.ac.uk/workpapers/GeNet2006p21.pdf and Clare Lyonette and Rosemary Crompton, 'Sharing the Load? Partners Relative Earnings and the Division of Domestic Labour' (2015) 29 (1) Work, Employment and Society 23.

Which reflects the historical 'obligations' of the 'father', considered above at subsection 6.2.B, 'What is the 'Father' in the Law?'

¹²⁶ For a detailed consideration of the legal regulation of child support and maintenance obligations, see Nicholas Wikeley, *Child Support: Law and Policy*, (Hart, 2006).

¹²⁷ Collier, 'A Hard Time to Be a Father?: Reassessing the Relationship Between Law, Policy, and Family (Practices)', at 533.

s.1 (1) Children Act 1989, the equivalent Scottish provision, found in s.11 (7) (a) Children

'child's upbringing'. ¹²⁹ The welfare principle appears to be 'child-centred', ¹³⁰ placing the focus of decision-making on the 'best interests' of children, with those interests being preferenced to all other considerations, including the rights of parents. ¹³¹ However, as O'Donovan has observed, '[b]ehind the word "welfare" lies a claim to knowledge of what is in the child's interests. ¹³² Thus, an examination of the judicial interpretation of 'welfare' should provide further insight into the law's construction of the role of the parent. ¹³³ I will argue that the law's understanding of the welfare of the child is influenced by the gendered parenting roles of 'mother' and 'father' derived from the traditional nuclear family.

The welfare principle has been subjected to significant and sustained critique. ¹³⁴ Freeman notes that '[c]ritics have emphasised its indeterminacy, its vagueness, its values and its absence of normative content. ¹³⁵ I will focus upon only one aspect of this critique, the lack of normative content; I will argue that this has resulted in the influence of the gendered parenting roles of 'mother' and 'father' upon the understanding of the role of the parent within the welfare principle. While the welfare principle appears to allow some scope for judicial discretion, the principle

(Scotland) Act 1995, is worded slightly differently, stating that the court, 'shall regard the welfare of the child concerned as its paramount consideration.'

s.1 (1) (a) Children Act 1989 and similar language is employed in different statutes, see e.g. in the context of adoption, s.1 (2) Adoption and Children Act 2002 and s.14 (3) Adoption and Children (Scotland) Act 2007.

Nhite Paper, 'Scotland's Children: Proposals for Child Care Policy and Law' (The Scottish Office, August 1993), Para 2.2, at 5, states that children, 'should be viewed as individuals in their own right, who's wants and needs must be taken seriously.' Available at https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/271973/2286.pdf

¹³¹ See the judgment of Lord MacDermott in J v C [1970] AC 668, where he stated, at 715, 'the rights and wishes of parents, whether unimpeachable or otherwise, must be assessed and weighed in their bearing on the welfare of the child'.

¹³² O'Donovan, Family Law Matters, at 92.

¹³³ Herring, *Caring and the Law*, proposes an alternative normative construction of the 'welfare principle' underpinned by care, mutuality and interdependence, which he describes as 'relationship-based welfare', he states, at 203, '[a] care-centred approach would require us to consider the child in the network of relationships within which they live. Relationship-based welfare argues that children should be brought up in relationships which overall promote their welfare.'

¹³⁴ See e.g. Robert Mnookin, 'Child-Custody Adjudication: Judicial Functions in the Face of Indeterminancy' (1975) 39 (3) Law and Contemporary Problems 226, Helen Reece, 'The Paramountcy Principle: Consensus or Construct?' (1996) 49 (1) Current Legal Problems 267 and Adrienne Barnett, 'The Welfare of the Child Re-Visited: In Whose Best Interests? Part 1' [2009] 39 (1) Fam. Law 50 and 'Part 2' [2009] 39 (2) Fam. Law 135.

Michael Freeman, 'Feminism and Child Law', in Bridgeman and Monk (eds.), *Feminist Perspectives on Child Law*, at 30, see also Stephen Parker, 'The Best Interests of the Child - Principles and Problems' (1994) 8 (1) International Journal of Law and the Family 26 and John Eekelaar, 'Beyond the Welfare Principle' [2002] 14 (3) Child and Family Law Quarterly 237.

lacks specified normative content, at least as set out by statute. ¹³⁶ As a result of this, the judiciary has developed several shared understandings, ¹³⁷ which provide the normative content used in determining the best interests of the child. ¹³⁸ Montgomery has described the welfare principle as an 'ideological construction', ¹³⁹ and observed that, 'the extent to which proposals for the care of children are seen by the courts to promote the welfare of those children is determined by the degree to which they diverge from establishment expectations of "normal" family life. ¹⁴⁰ I argue that some of these shared judicial understandings of the welfare of the child are underpinned by the idealised image of the nuclear family, ¹⁴¹ and its construction of separate gendered parenting roles, because that archetype of family has been positioned as the 'natural', ¹⁴² model of family. ¹⁴³

The reliance upon these central factors is apparent both from judicial language 144 and

10

¹³⁶ Nevertheless, judgments often claim that the welfare of the child is the *only* principle driving the outcome, see e.g. the Supreme Court decision in $Re\ B\ (A\ Child)\ [2010]\ 1\ FLR\ 551$ and the statement of Baroness Hale in $Re\ J\ (A\ Child)\ (Custody\ Rights:\ Jurisdiction)\ [2006]\ 1\ AC\ 80$, at 94, that, 'we do not have any fixed concept of what will be in the best interests of the individual child.'

¹³⁷ These 'shared understandings' include: (i) the understanding that children are best cared for by their natural parents, see e.g. *J v C* [1970] AC 668 and *Re KD (A Minor) (Ward: Termination of Access)* [1988] AC 806, (ii) the preference for mothers as the primary carers of children after relationship breakdown, see e.g. *Re B (An Infant)* [1962] 1 All ER 872 and *Brixey v Lynas* 1997 SC (HL) 1, (iii) the proposition that siblings should continue to reside together after parental separation, see e.g. *C v C (Minors: Custody)* [1988] 2 FLR 291, (iv) the support for ongoing contact with the non-resident parent, see e.g. *White v. White* 2001 SLT 485 and *Re O (Contact: Imposition of Conditions)* [1995] 2 FLR 124 and (v) the contrary proposition in favour of allowing the reasonable proposals of resident parents in relocation applications, see e.g. *Payne v Payne* [2001] 1 FLR 1052.

¹³⁸ There is significant academic debate surrounding whether judicial 'presumptions' exist within the welfare test, see e.g. Herring and Powell, 'The Rise and Fall of Presumptions Surrounding the Welfare Principle', Jonathan Herring, 'The Welfare Principle and the Children Act: Presumably it's About Welfare?' (2014) 36 (1) Journal of Social Welfare and Family Law 14 and Christine Piper, 'Assumptions about Children's Best Interests' (2000) 22 (3) Journal of Social Welfare and Family Law 261. It is not the intention of this thesis (nor is it necessary to advance the central argument of the thesis) to determine, one way or the other, whether or not these shared judicial understandings amount to 'presumptions'.

¹³⁹ Jonathan Montgomery, 'Rhetoric and 'Welfare'' (1989) 9 (3) Oxford Journal of Legal Studies 395, at 396.

¹⁴⁰ ibid.

¹⁴¹ The preference for mothers, discussed above at subsection 6.2.A.1, 'The Judicial Preference for Mothers as the Carers of Children' and the support for contact with the non-resident parent, which will be discussed below at subsection 6.2.C.1, 'The Significance of Contact with the Non-Resident Parent'.

¹⁴² See e.g. *Re B (An Infant)* [1962] 1 All ER 872, per Donovan LJ, at 875.

Discussed above in Chapter 3, subsection 3.3, 'The Nuclear Family as the Natural Model of 'Family''.

¹⁴⁴ Various terms have been employed by the judiciary, including: 'presumption', *Re W (A Minor)* (*Residence Order*) [1992] 2 FLR 332, 'assumption', *Re L (Contact: Domestic Violence)*; *Re V*

from recent legislative provisions. ¹⁴⁵ In *White v White*, ¹⁴⁶ Lord Rodger observed that, 'when Parliament says that judges are to have regard to the welfare of the child, it must consider that judges will, by and large, have a common conception of what that involves - of what will advance the welfare of children in regard to these matters.' ¹⁴⁷ Therefore the interpretation of the welfare principle is premised upon judicial understandings about what serves the best interests of the child. ¹⁴⁸ Diduck and Kaganas comment that the interpretation of the welfare principle 'owes less to scientific "truths" than to understandings of the welfare of children that accord with prevailing beliefs about how families should be structured and what the roles of family members should be. ¹⁴⁹ I will argue that the judicial interpretation of the best interests of the child is premised upon the distinct gendered parenting roles, mother as 'homemaker' and father as 'breadwinner', derived from the traditional, nuclear family, but that these values are hidden behind references to 'nature' and simple 'common sense' ¹⁵¹ rationality within the welfare principle.

6.2.C.1. The Significance of Contact with the Non-Resident Parent

Following the breakdown of a relationship between parents (or where the parents were never in a cohabiting relationship), in the absence of agreement between the parents, ¹⁵² it is often necessary for a court to make an order ¹⁵³ regarding which parent

(Contact: Domestic Violence); Re M (Contact: Domestic Violence; Re H (Contact: Domestic Violence) [2000] 2 FLR 334, 'strong supposition', Re H (A Minor) (Custody: Interim Care and Control) [1991] 2 FLR 109, 'policy', Re B (Contact: Stepfathers Opposition) [1997] 2 FLR 579 and 'consideration', Re S (A Minor) (Custody: Children's Welfare) [1991] 2 FLR 388.

¹⁴⁵ s.11 (2) Children and Families Act 2014 amends s.1 Children Act 1989, inserting a presumption in favour of contact with both parents. Kaganas, 'A Presumption that 'Involvement of Both Parents is Best: Deciphering Law's Messages', at 288, describes this as 'symbolic legislation'.

 ^{146 2001} SLT 485.
 147 ibid, per Lord Rodger, at 489.

¹⁴⁸ ibid, per Lord McCluskey, at 494, '[t]he judge who approaches the issues raised in an application for a s.11 order does not do so value free.'

¹⁴⁹ Diduck and Kaganas, Family Law, Gender and the State, at 373.

¹⁵⁰ See e.g. *Brixey v Lynas* 1997 SC (HL) 1, per Lord Jauncey, at 6.

¹⁵¹ See e.g. Latey J in *M v M (Child: Access)* [1973] 2 All ER 81, at 88.

¹⁵² The vast majority of separated parents reach agreement as to residence and contact without any involvement from the courts. In England and Wales there were 111,302 contact orders made in 2011, 'Judicial and Court Statistics 2011', at 26. Estimates for the number of non-resident fathers vary, e.g. in 'What Do We Know about Nonresident Fathers?' (Modern Fatherhood), at 2, it was estimated that there are around 980,000 non-resident fathers in the UK, available at http://www.modernfatherhood.org/wp-content/uploads/2013/11/Briefing-paper-Non-resident-

the child will live with¹⁵⁴ and detailing the contact¹⁵⁵ between the other parent and the child(ren). In determining such disputes, judicial reasoning has generally observed that it is in the best interests of a child to continue to have a relationship with both parents; which is reflected in judicial support for ongoing contact with the 'non-resident parent'. ¹⁵⁶

In an early decision 'access' (as it was then known), was described as 'the basic right of any parent'. This language was quickly altered to reflect 'a basic right in the child rather than a basic right in the parent. Subsequent judicial formulations have generally reflected the observation of Sir Thomas Bingham MR in *Re O (Contact: Imposition of Conditions)*, that 'where parents of a child are separated and the child is in the day-to-day care of one of them, it is almost always in the

<u>fathers.pdf</u>, whereas Hunt with Roberts, 'Family Policy Briefing 3: Child Contact with Non-Resident Parents', University of Oxford, suggested, in 2004, that there may be up to 2 million non-resident fathers.

¹⁵³ Known in England and Wales as a 'Child Arrangements Order' subsequent to s.12 Children and Families Act 2014, which amended s.8 (1) Children Act 1989, replacing both the 'residence order' and the 'contact order'.

s.8 (1) Children Act 1989 defines a 'child arrangements order' as, 'an order regulating arrangements relating to any of the following - (a) with whom a child is to live, spend time or otherwise have contact, and (b) when a child is to live, spend time or otherwise have contact with any person', see s.11 (2) (c) Children (Scotland) Act 1995 for the Scottish provision relating to 'residence orders'.

¹⁵⁵ s.11 (2) (d) Children (Scotland) Act 1995 provides the Scottish provision for 'contact orders'.

¹⁵⁶ In spite of the removal of 'residence orders' in the 2014 Act, this thesis will to continue use the term 'non-resident parent' to describe the parent who is not the children's primary carer.

¹⁵⁷ S v S & P [1962] 2 All ER 1, per Wilmer LJ, at 3. In Scotland, Lord Dunpark stated in *Porchetta v Porchetta* 1986 SLT 105, at 105, '[a] father does not have an absolute right to access to his child. He is only entitled to access if the court is satisfied that this is in the best interests of the child.' See further e.g. Joseph Thomson, 'Whither the 'Right' of Access?' 1989 SLT (News) 109.

¹⁵⁸ M v M (Child: Access) [1973] 2 All ER 81, per Wrangham J, at 85. Judicial descriptions of contact as a right of the child continued into the early 1990s, see e.g. Re S (Minors) (Access) [1990] 2 FLR 166, Re W (A Minor) (Contact) [1994] 2 FLR 441 and Re J (A Minor) (Contact) [1994] 1 FLR 729.

¹⁵⁹ In Re M (Contact: Welfare Test) [1995] 1 FLR 274, at 279, Wilson J referred to 'the strong presumption in favour of contact', see further Re A (Contact: Domestic Violence) [1998] 2 FLR 171 and Re P (Contact: Discretion) [1998] 2 FLR 696. However, Thorpe LJ rejected the use of the term presumption in Re L (Contact: Domestic Violence); Re V (Contact: Domestic Violence); Re M (Contact: Domestic Violence; Re H (Contact: Domestic Violence) [2000] 2 FLR 334. There continues to be academic argument about whether there is a presumption in favour of contact, see e.g. Stephen Gilmore, 'Disputing Contact: Challenging Some Assumptions' [2008] 20 (3) Child and Family Law Quarterly 285, who argues against the existence of such a presumption and Andrew Bainham, 'Contact as a Right and Obligation' in Bainham, Lindley, Richards and Trinder (eds.), Children and Their Families: Contact, Rights and Welfare, who observes, at 61, 'those who assert that there is no right or presumption of contact are not merely misguided, but are plainly wrong.'

interests of the child that he or she should have contact with the other parent.' ¹⁶¹ In the relatively recent decision in *Re W (Children) (Direct Contact)*, ¹⁶² this understanding was re-iterated by McFarlane LJ, who referred to, 'the well known case law which stresses the benefit children will normally gain from maintaining a meaningful relationship with both of their parents following a split in the family.' ¹⁶³ Research into outcomes of court applications, conducted in 2008, identified that, 'while there is no statutory presumption, the courts operate on a *de facto* presumption that unless there were good reasons to the contrary there should be contact.' ¹⁶⁴ Indeed, the judicial preference for contact is illustrated by the fact that contact orders ¹⁶⁵ are issued in the majority of cases. ¹⁶⁶

Various justifications have been employed by the judiciary in support of contact;¹⁶⁷ I will focus upon judicial language which suggests that the support for contact is based

1 /

¹⁶¹ ibid, per Sir Thomas Bingham MR, at 128, see further e.g. the Scottish House of Lords decision *Sanderson v McManus* 1997 SC (HL) 55, where Lord Hope stated, at 64, '[i]t may normally be assumed that the child will benefit from continued contact with the natural parent'.

¹⁶² [2013] 1 FLR 494.

¹⁶³ ibid, per McFarlane LJ, at Para 35, see further e.g. *Re K (A Child)* [2010] EWCA Civ 478 and *Re H (Contact with Biological Father)* [2012] 2 FLR 627.
¹⁶⁴ Hunt and MacLeod, 'Outcomes of Applications to Court for Contact Orders After Parental

Separation or Divorce', September 2008, at 189, similarly Herring and Powell, 'The Rise and Fall of Presumptions Surrounding the Welfare Principle' refer to the 'wide academic consensus' of a 'de facto presumption in favour of contact', at 556.

¹⁶⁵ Jonathan Herring, *Family Law*, (6th edition, Pearson Longman, 2013), at 550, notes that, in England and Wales, '[i]n 2011 there were 111,302 applications for a contact order heard and in 108,557 cases were contact orders made, and in only 333 were contact orders refused.'

¹⁶⁶ Hunt and MacLeod, 'Outcomes of Applications to Court for Contact Orders After Parental Separation or Divorce', September 2008, at 11, found that 70% of cases in their sample resulted in a contact order, while 16% of applications were withdrawn, 7% resulted in no order and only 7% were dismissed. See further the 'Judicial and Court Statistics 2011', at 26, which states that there were 183,718 children involved in disposals of private law applications in which 178,517 orders were made.

Notably, there have been judicial references to the benefit of contact being demonstrated by 'scientific' evidence, external to law. See e.g. *Re L (Contact: Domestic Violence)* [2000] 2 FLR 334 and *Re G (Contact)* [2006] 1 FLR 1663, where Ward LJ stated, at 1672, 'every psychiatrist that has ever given evidence to me on this question will have told me the serious consequences if that contact is denied.' These judicial references to evidence from experts are especially notable because there is far from a consensus in the research that contact itself is beneficial for children, see e.g. Stephen Gilmore, 'Contact/Shared Residence and Child Well-Being: Research Evidence and its Implications for Legal Decision-Making' (2006) 20 (3) International Journal of Law, Policy and the Family 344, Paul Amato and Joan Gilbreth, 'Nonresident Fathers and Children's Well-Being: A Meta-Analysis' (1999) 61 (3) Journal of Marriage and the Family 557 and Judy Dunn, Helen Cheng, Thomas G. O'Connor, and Laura Bridges, 'Children's Perspectives on their Relationships with their Nonresident Fathers: Influences, Outcomes and Implications' (2004) 45 (3) Journal of Child Psychology and Psychiatry 55.

upon reflecting 'common sense' or 'nature'. ¹⁶⁸ In *White v White*, ¹⁶⁹ Lord Rodger referred to 'a working hypothesis born of human experience'. ¹⁷⁰ Kaganas suggests that the courts are 'simply stating the proposition as axiomatic; in this way it becomes indisputable', ¹⁷¹ and indeed in *Re O (Contact: Imposition of Conditions)*, ¹⁷² Sir Thomas Bingham observed that, '[t]he reason for this scarcely needs spelling out.' ¹⁷³ By presenting the benefits as self-evident, this approach illustrates the reliance upon 'nature' and 'common sense' as founding the judicial support for contact. ¹⁷⁴

Moreover, the understanding that contact is beneficial is now adduced in the language of government reports; for example, '[c]hildren generally do better if both their parents are involved in their lives, irrespective of whether the parents live together' and '[w]e believe that children normally benefit from the continued involvement of both parents in their lives'. In England and Wales, s.11 Children and Families Act 2014 amends s.1 Children Act 1989 by inserting the provision that the court is, 'to presume, unless the contrary is shown, that involvement of [the] parent in the life of the child concerned will further the child's welfare.

1

¹⁶⁸ This reflects the positioning of the nuclear family as the 'natural' model of 'family' discussed above in Chapter 3, section 3.3, 'The Nuclear Family as the Natural Model of 'Family' and the judicial reliance upon the 'natural' construction of motherhood, discussed above in subsection 6.2.A.2, 'The 'Natural' Construction of Motherhood'.

¹⁶⁹ 2001 SLT 485.

ibid, per Lord Rodger, at 490.

¹⁷¹ Felicity Kaganas, 'Regulating Emotion: Judging Contact Disputes' [2011] 23 (1) Child and Family Law Quarterly 63, at 69.

¹⁷² [1995] 2 FLR 124.

ibid, per Sir Thomas Bingham MR, at 128.

This again reflects the association of the nuclear family model and 'nature', described above in Chapter 3, subsection 3.3, 'The Nuclear Family as the Natural Model of 'Family'.

¹⁷⁵ 'Positive for Youth: A New Approach to Cross-Government Policy for Young People Aged 13 to 19', (February 2010), at 14, Para 3.7.

¹⁷⁶ The Government Response to the Family Justice Review: A System with Families and Children at its Heart', (Ministry of Justice and Department for Education, February 2012), at 18, Para 57, available at -

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/177097/CM-8273.pdf. 177 s.1 (2A) Children Act 1989.

¹⁷⁸ s.11 (2), this provision also inserts s.1 (2B) into the Children Act 1989, which provides that "involvement" means involvement of some kind, either direct or indirect, but not any particular division of a child's time.' This legislative support for parental 'involvement' proved controversial, because the insertion of this provision went directly against the recommendations of the independent, 'Family Justice Review: Final Report', (November 2011), at 139-141, Paras 4.29-4.40, available at https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/217343/family-justice-review-final-report.pdf.

presumption of 'parental involvement' was to some extent premised upon the government seeking, 'to dispel the perception that there is an in-built legal bias towards fathers or mothers.¹⁷⁹ Given the social reality of mothers as primary carers in the vast majority of cases, ¹⁸⁰ Kaganas has observed that '[t]he new presumption makes a statement about the importance of fathers and so symbolically restores their status'. 181 I submit that this legislative approach is underpinned by support for 'fathering', which is constructed as providing a different form of parenting than that provided by the mother. 182 Thus, I contend it is not the capacity of the father to provide 'care' or social parenting that is being granted normative significance within legal understanding, but rather it is the symbolic importance of the 'father' as completing the gendered archetype of the nuclear family. Kaganas and Diduck comment that, '[t]he nuclear family has long been accorded an idealised status and the "good" post-separation family has been moulded in its image. 183 Thus, I suggest that the support for contact is premised upon the need for continued male parental involvement, which reflects the distinct gendered parenting roles of the nuclear family.

6.2.C.2. Contact and Gender Neutrality

Kaganas and Day Sclater observe that 'contact, despite law's propensity to cast parenting as a gender-neutral activity, remains very much a gender issue and the meaning of a contact dispute to parents varies with gender.' The gendered dimension of contact, with mothers being the resident parent and fathers the non-

_

¹⁷⁹ Department for Education and Ministry of Justice Consultation, 'Co-operative Parenting Following Family Separation: Proposed Legislation on the Involvement of Both Parents in a Child's Life', (July 2012), at Para 4.3, available at - http://dera.ioe.ac.uk/14682/.

¹⁸⁰ As discussed above in subsection 6.2.A, 'What is the 'Mother' in the Law?', see e.g. Hunt and MacLeod, 'Outcomes of Applications to Court for Contact Orders After Parental Separation or Divorce', September 2008, at 239, which shows that fathers accounted for 91.6% (265 out of 289) of the non-resident parents making applications for contact.

¹⁸¹ Kaganas, 'A Presumption that 'Involvement of Both Parents is Best: Deciphering Law's Messages', at 293.

¹⁸² As discussed above in subsection 6.2.B, 'What is the 'Father' in the Law?'

¹⁸³ Felicity Kaganas and Alison Diduck, 'Incomplete Citizens: Changing Images of Post-Separation Children' (2004) 67 (6) Modern Law Review 959, at 981.

¹⁸⁴ Kaganas and Day Sclater, 'Contact Disputes: Narrative Constructions of 'Good' Parents', at 6.

resident parent in the vast majority of cases, 185 is often obscured by the genderneutral language employed by the law, 186 particularly through the language of the statutory provisions, 187 which refer to 'parents' 188 rather than to mothers and fathers. 189 Interestingly, in spite of the apparent gender neutrality of the law 190 some judges have felt it necessary to acknowledge the gendered context in which contact disputes occur; for example in Re O (Contact: Withdrawal of Application)¹⁹¹ Wall J commented that, '[t]he courts are not anti-father and pro-mother or vice-versa.' 192

Various academic commentators have contended that judicial considerations of contact are influenced by the different constructions of the gendered parenting roles of mothers and fathers within the traditional, nuclear family. 193 In this regard, Trinder suggests, 'underpinning the discussion of the respective rights of resident and nonresident parents are, of course, sets of ideas about the respective rights of mothers and fathers, despite the gender neutral use of "parent", 194 Collier claims that '[t]he good father and the good mother continue to be seen as having different albeit overlapping roles' while Diduck observes that 'fathers' expressions of care about

¹⁸⁵ Hunt and MacLeod, 'Outcomes of Applications to Court for Contact Orders After Parental Separation or Divorce', September 2008, at 239.

As discussed above in section 6.1, 'What is the 'Parent' in Law?'

See e.g. the definitions of 'parental responsibilities' and 'parental rights' in s.1-2 Children (Scotland) Act 1995 and s.3 Children Act 1989.

¹⁸⁸ See e.g. Re M (Contact: Welfare Test) [1995] 1 FLR 274 and Re P (Contact: Discretion) [1998] 2 FLR 696 for judicial examples of the use of such gender-neutral language.

¹⁸⁹ Although for examples of the judicial use of gendered terminology in the context of contact, see e.g. Wall J's reference in Re O (Contact: Withdrawal of Application) [2004] 1 FLR 1258, at 1260, that 'the courts recognise the vital importance of the role of non-resident fathers in the lives of their children' and *Re G (Contact)* [2006] 1 FLR 1663.

¹⁹⁰ This reflects the political and cultural power that the 'father's rights' organisations (e.g. Fathers4Justice) have achieved. On the influence of father's rights campaigners within legal debates, see e.g. Richard Collier and Sally Sheldon (eds.), Fathers Rights Activism and Law Reform in Comparative Perspective, (Hart, 2006) and Carol Smart, 'The Ethic of Justice Strikes Back: Changing Narratives of Fatherhood' in Diduck and O'Donovan (eds.), Feminist Perspectives on Family Law. ¹⁹¹ [2004] 1 FLR 1258

ibid, at 1260, see further e.g. Wall J's similar comments in Re M (Intractable Contact Dispute: Interim Care Order [2003] 2 FLR 636, at 641.

¹⁹³ The different conceptions of these gendered parenting roles were discussed above in Chapters 2 and 3, as well as earlier in this Chapter at subsections 6.2.A, 'What is the 'Mother' in the Law?' and 6.2.B. 'What is the 'Father' in the Law?'

¹⁹⁴ Liz Trinder, 'Introduction' in Bainham, Lindley, Richards and Trinder (eds.), Children and Their Families: Contact, Rights and Welfare, at 10.

¹⁹⁵ Richard Collier, 'Anxious Parenthood, the Vulnerable Child and the 'Good Father': Reflections on the Legal Regulation of the Relationship between Men and Children', in Bridgeman and Monk (eds.), Feminist Perspectives on Child Law, at 112.

the children are valued more highly than mothers' actions of caring for them.' ¹⁹⁶ Moreover, Smart observes, on the basis of empirical research into the language of parents themselves, that 'mothers, when they spoke about the work they did in caring for their children and the sacrifices they made, were hardly acknowledged. These actions were seen as being as normal as breathing and thus as worthy of as much acknowledgement as such taken for granted activities usually generate. ¹⁹⁷ Thus, the actual caring activities undertaken by the mother as primary carer are diminished and understood as part of her 'natural' role, yet when a father performs 'caring' activities, this is in addition to his 'natural' role, thus demonstrating that he is a 'good' father. ¹⁹⁸ Indeed, this distinction between the expectations placed upon men and women in their parenting clearly illustrates the influence of the gendered parenting roles.

In addition, mothers who do not embrace the dominant understanding that contact with the non-resident father is beneficial to their children can be the subject of judicial criticism; Wallbank observes that, 'women who are constructed as wilfully depriving their children of the right to contact with their father are subjected to the court's often vehement disapproval.' Historically this led to repeated judicial references to the 'implacably hostile' mother. While such language has subsequently been the subject of judicial criticism of while such language has employed, milder judicial criticism of mothers resisting giving effect to judicial decisions is still evident, as illustrated by the comment of Ward LJ, in *Re G (Contact)*, ²⁰² that '[t]his mother has got to appreciate that contact with father is in the

¹⁹⁶ Diduck, Law's Families, at 86.

¹⁹⁶ [2003] 2 AC 467.

Smart, 'Losing the Struggle for Another Voice: The Case for Family Law', at 177.

¹⁹⁸ This reflects Tronto's distinction between 'caring for' and 'caring about', See e.g. Tronto, 'Women and Caring: What Can Feminists Learn about Morality from Caring?' in Jaggar and Bordo (eds.), Gender/Body/Knowledge: Feminist Reconstructions of Being and Knowing and Tronto, Moral Boundaries: a Political Argument for an Ethic of Care, which is mentioned above in subsection 6.2.B.1, 'The Shift in the Construction of Fatherhood'.

¹⁹⁹ Julie Wallbank, 'Castigating mothers: The Judicial Response to 'Wilful' Women in Disputes over Paternal Contact in English Law' (1998) 20 (4) Journal of Social Welfare and Family Law 357, at 358. ²⁰⁰ See e.g. Re D (A Minor) (Contact: Mother's Hostility) [1993] 2 FLR 1, Re P (A Minor) (Contact) [1994] 2 FLR 374, Re P (Minors) (Contact: Parental Hostility) [1996] 2 FLR 314 and Re J (A Minor) (Contact) [1994] 1 FLR 729.

²⁰¹ See e.g. Hale J in *Re D (Contact: Reasons for Refusal)* [1997] 2 FLR 48, who stated, at 53, '[i]t is important to bear in mind that the label 'implacable hostility' is sometimes imposed by the law reporters and can be misleading.'

²⁰² [2006] 1 FLR 1663.

best interests of her son. It may not be pleasant for either of them at this particular point in time.' The judicial language in contact cases has prompted Kaganas and Day Sclater to observe:

For the most part, it remains the case that fathers have to do very little to qualify as "good" fathers. And, for the most part, it remains the case that mothers who oppose contact for reasons other than "genuine" and "reasonable" fears of physical violence are considered selfish and as harming their children.²⁰⁴

I would adopt the analysis of these authors and thus make the point that the law expects more from resident parents than non-resident parents within the contact discourse and, given that resident parents are overwhelmingly mothers, it is submitted that the law generally expects more from mothers than fathers in relation to contact. Wallbank suggests 'that where disputes arise between mothers and fathers, because of the strength of the presumption in favour of contact, mothers will inevitably have higher hurdles to jump in order to have their fears about it taken seriously.'205 I submit that this distinction is based upon reference to the differing constructions of the archetypical gendered parenting roles, with mothers presented as the 'natural' carers of children,²⁰⁶ derived from the traditional, nuclear family. Thus, I argue that the judicial support for contact provides further illustration of the influence of the gendered roles of 'mother' and 'father' and the nuclear family model upon the legal understanding of parenthood.

_

²⁰³ ibid, Ward LJ, at 1672.

Felicity Kaganas and Shelley Day Sclater, 'Contact and Domestic Violence - The Winds of Change?' [2000] 30 (9) Fam. Law 630, at 635.

Julie Wallbank, 'Getting Tough on Mothers: Regulating Contact and Residence' (2007) 15 (2) Feminist Legal Studies 189, at 216.

²⁰⁶ As discussed above in subsection 6.2.A, 'What is the 'Mother' in the Law?'

6.3. The Role of Lesbian 'Mothers' or 'Parents'

This section considers the legal understanding of lesbian parents, focusing on the role of non-gestational female 'parent', which provides an example of people whose parental role has been created and explicitly defined by legislation as the 'parent'. ²⁰⁷ This section will argue that the lack of legal engagement with the role of the non-gestational female 'parent' reflects the dominance of the 'natural' gendered parental roles of the nuclear family within the legal understanding of the role of the 'parent'.

In this section, I will consider the legal understanding of the role of lesbian parents, focusing upon the constructions of the role of the non-gestational female 'parent' in two contexts: disputes between the members of a lesbian couple (subsection 6.3.A.1), and disputes between lesbian couples and (male) 'known donors' (subsection 6.3.A.2). I will conclude that the lack of a clear legal understanding of the role of the non-gestational female 'parent' reflects the rhetorical power of the traditional gendered constructions of 'mother' and 'father' (subsection 6.3.B).

6.3.A. The Judicial Approach to Lesbian Mothers

The UK legal system first encountered lesbian mothers in the early 1980s in the context of child 'custody' disputes involving mothers who had entered lesbian relationships after the breakdown of relationships with male partners. The initial judicial response was characterised by hostility towards these mothers, and references to their 'sexual deviance'. As recently as 1991, Glidewell LJ expressed the view that 'a lesbian relationship between two adult women is an unusual background in which to bring up a child. Beresford observes that 'legal discourse

Judicial hostility was also evident in regards to the fatherhood of gay men, see e.g. Lord Kilbrandon's reference in *Re D (An Infant) (Adoption: Parent's Consent)* [1977] AC 602, at 641, to, 'those whose sexual abnormalities have denied them the possibility of a normal family life.'

²⁰⁷ s.42-47 Human Fertilisation and Embryology Act 2008. See above in Chapter 5, subsection 5.2.B, 'Non-Gestational Female 'Parents''.

²⁰⁸ As it was then known.

²¹⁰ This term is used by both Orr LJ in S v S (Custody of Children) [1980] 1 FLR 143 and Watkins LJ in Re P (A Minor) (Custody) [1983] 4 FLR 401.

²¹¹ C v C (A Minor) (Custody Appeal) [1991] 1 FLR 223, per Glidewell LJ, at 228.

largely presents motherhood as having essentialist, and naturalistic qualities, [and] a lesbian mother appears in the court room without those qualities. The approach of the judiciary to lesbian motherhood in these early cases is exemplified by the statement of Watkins LJ in $Re\ P\ (A\ Minor)\ (Custody)^{213}$ that:

I accept that it is not right to say that a child should in no circumstances live with a mother who is carrying on a lesbian relationship with a woman who is also living with her, but I venture to suggest that it can only be countenanced by the court when it is driven to the conclusion that there is in the interests of the child no other acceptable alternative form of custody.²¹⁴

While lesbian mothers were not prevented in these cases from being the primary carer for their children, their role was distinguished from the heterosexual, 'natural' ideal of motherhood. Consequently, their parenting was considered unable to offer the same qualities as the traditional, gendered nuclear family. The At this time, the 'lesbian mother' could not be positioned within the idealised image of the (heterosexual) nuclear family. Therefore, it was not the actual care she provided that was granted normative significance, but rather her differences from the traditional, gendered construction of the 'mother' as the primary carer. The gendered, binary, two-parent model of the nuclear family remained the model of parenting preferred by the judiciary, as made clear by Glidewell LJ's observation in $C \ v \ C^{217}$ that 'I regard it as axiomatic that the ideal environment for the upbringing of a child is the home of loving, caring and sensible parents, her father and her mother.'

_

²¹² Beresford, 'Get Over Your (Legal) 'Self': A Brief History of Lesbians, Motherhood and the Law', at 104.

²¹³ [1983] 4 FLR 401.

ibid, per Watkins LJ, at 405.

²¹⁵ Calman J in *B v B (Minors) (Custody, Care and Control)* [1991] 1 FLR 402, distinguished between 'militant lesbians' and 'lesbians in private', noting, at 410, that the lesbian couple in this case were, 'private persons who both do not believe in advertising their lesbianism and acting in the public field in favour of promoting lesbianism'.

²¹⁶ ibid, at 408, however it is interesting to note that Calman J emphasised that, '[t]he mother has been a blameless, faultless mother so far as care of her children is concerned.'

²¹⁷ C v C (A Minor) (Custody Appeal) [1991] 1 FLR 223.

²¹⁸ ibid, at 228.

Subsequent legislative developments, including the legal regulation of lesbian relationships through the Civil Partnership Act 2004, followed by the more recent same-sex marriage legislation, ²¹⁹ and the recognition of lesbian parenting in relation to assisted reproduction ²²⁰ and adoption ²²¹ suggest that there has been an increasing acceptance of lesbian mothers within the law and a corresponding movement away from these early negative judicial attitudes. ²²²

6.3.A.1. Disputes Between Lesbian Couples

Legal recognition of the existence of a parental relationship between a non-gestational female 'parent' and her children has only been achieved relatively recently, both through the status of legal parenthood in assisted reproduction²²³ and in terms of judicial acknowledgement that hers is a parental role.²²⁴ However, as discussed above in Chapter 5, in the attribution of legal parenthood the term 'mother' remains exclusive to the *gestational* mother, based on the dominant construction of motherhood as 'natural' and indivisible.²²⁵ Both in legislation²²⁶ and in the courts²²⁷ non-gestational female 'parents' are usually referred to using the gender-neutral terminology of 'parents' rather than as 'mothers'.²²⁸ I argue that the use of this

2 1

Marriage (Same Sex Couples) Act 2013 and the Marriage and Civil Partnership (Scotland) Act 2014

Described above in Chapter 5, at subsection 5.2.B, 'Non-Gestational Female 'Parents''.

s.144 (4) (b) Adoption and Children Act 2002 and s.29 (3) Adoption and Children (Scotland) Act 2007 allowed same-sex couples to adopt jointly, see also the decisions in *Re E (Adoption: Freeing Order)* [1995] 1 FLR 382, *T, Petitioner* 1997 SLT 724 and *Re W (A Minor) (Adoptions: Homosexual Adopter)* [1997] 3 WLR 768, which allowed adoption by individual homosexual applicants.

More recent judicial language has been much more supportive of same-sex families, see e.g. Girvan LJ's statement in *Northern Ireland Human Rights Commission's Application* [2013] NICA 37, at Para 38, that 'the status of civil partners is closely analogous to that of married partners' and Baroness Hale's statement in *Hall v Bull* [2013] 1 WLR 3741, at 3752, that 'same sex couples can enter into a mutual commitment which is the equivalent of marriage.'

²²³ As set out above in Chapter 5, subsection 5.2.B, 'Non-Gestational Female 'Parents''. In addition, s.4ZA was inserted into the Children Act 1989 to provide for the acquisition of parental responsibilities by the 'second female parent'.

See e.g. the judgment of Hedley J in Re P & L (Contact) [2012] 1 FLR 1068.

Described above at subsection 6.2.A, 'What is the 'Mother' in the Law?'

²²⁶ s.42-47 Human Fertilisation and Embryology Act 2008.

Although the courts will still not always even use terminology which gives recognition to the role as being parental, e.g. Lord Nicholls referred to the 'former partner' of the mother in *Re G (Children)* (*Residence: Same-Sex Partner*) [2006] 2 FLR 629, at 631, see also Thorpe LJ referring to the 'long-term lesbian partner', in *A v B and C (Role of Father)* [2012] 2 FLR 607, at Para 2.

²²⁸ For an exception to the judicial use of this terminological distinction, see *Re G (Children) (Shared Residence Order: Biological Non-Birth Mother)* [2014] 2 FLR 897, where Black LJ, at Para 52, uses

terminology is significant, because as Everett and Yeatman comment, '[t]he words we choose to use to describe relationships have power and we must choose them with care.'²²⁹ In spite of these non-gestational female 'parents' now being granted recognition *as* legal parents, the terminological distinction made between the two roles seems to suggest that the law takes there to be essential differences between this parenting role and that of the 'mother'. As Diduck points out, this approach leads to a situation where, "[d]oing" parenting may make lesbian parents "parents", but it is not enough to make them mothers and the difficulties presented by the limitation of language for that form of parenthood are clear.'²³⁰ This could result in a hierarchy of parenting roles being created within lesbian couples, because of the widely different idealised constructions of these two roles. As described above, the law has an understanding of the 'natural' role of the mother which underpins its construction of motherhood.²³¹ In contrast, the gender-neutral terminology of 'parent' possesses less cultural resonance because it lacks a similar readily identifiable and easily understandable archetypical role.²³²

This distinction between the two roles is evidenced in the House of Lords judgment in *Re G (Children) (Residence: Same-Sex Partner)*,²³³ where in the context of a residence dispute between two lesbians who were formerly a couple, Baroness Hale states that, 'the issues arising are just the same as those which may arise between heterosexual couples. The legal principles are also the same.'²³⁴ I suggest that this is a problematic approach, because it explicitly applies the standards of the traditional, heterosexual, nuclear family to the circumstances of a lesbian-led family, which had been purposely designed outside that framework.²³⁵ The judgment also includes the

-

the term 'biological mother' to describe a non-gestational female 'parent' whose gametes had been used in the creation of the child. Although, it is notable that even in this judgment the gender-neutral terminology of 'genetic parent' was her ladyships preferred descriptor.

²²⁹ Everett and Yeatman, 'Are Some Parents More Natural Than Others?', at 306.

²³⁰ Diduck, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood', at 465.

²³¹ As discussed above in Chapter 5, subsection 5.2.A, 'Motherhood'.

²³² See above at subsection 6.1.B, 'The Role of the 'Parent''.

²³³ [2006] 2 FLR 629, see further Elizabeth Woodcraft, 'Re G: A Missed Opportunity' [2007] 37 (1) Fam. Law 53.

²³⁴ ibid, per Baroness Hale, at 632.

²³⁵ See e.g. Fiona Kelly, 'Nuclear Norms or Fluid Families? Incorporating Lesbian and Gay Parents and Their Children into Canadian Family Law' (2004) 21 (1) Canadian Journal of Family Law 133.

observation of Baroness Hale that, '[t]he fact [CG] is the natural mother of these children in every sense of that term, while raising no presumption in her favour, is undoubtedly an important and significant factor in determining what will be best for them now and in the future.'236 This emphasis on the importance of the 'natural mother' to the child's welfare seems to contradict her description of the case as identical to one that arises with opposite-sex couples, and also seems to create a defacto presumption in favour of genetic relationships. 238 Such a presumption would be problematic for lesbian couples where, unlike in most heterosexual relationships, ²³⁹ the mother will usually have a genetic connection with the children which the non-gestational female 'parent' will never possess. 240 Beresford suggests that 'the case firmly restated so-called traditional values. It stated that the biological connection is of fundamental importance and it did this by appealing to the rhetorical importance of that which is "natural". 241 It is submitted that through the attempt to analogise a lesbian couple with a heterosexual couple, the law reflects the aforementioned hierarchy of parenting roles within lesbian relationships; with the 'parent' constructed as a second class parental role in comparison with the 'mother'; thus illustrating the normative significance of the gendered parenting roles. I argue that this is needlessly problematic and the courts' choice to view lesbian family

²³⁶ Re G (Children) (Residence: Same-Sex Partner) [2006] 2 FLR 629, per Baroness Hale, at 643.

²³⁷ Baroness Hale defines a 'natural parent' as encompassing 'genetic', 'gestational' and 'social and psychological' parenthood, therefore she distinguishes the 'natural mother' from other 'natural parents', including the non-gestational female 'parent', who generally only possesses 'psychological parenthood'.

parenthood'. ²³⁸ Subsequently the Supreme Court in *Re B (A Child)* [2010] 1 FLR 551 made clear that there exists no 'presumption' and that all decisions must be made on the basis of the 'welfare test'. However, the factual circumstances were very different, involving a dispute between a genetic father and maternal grandmother as opposed to dispute between lesbian couple. Therefore, the differing gendered parenting roles this case involved may have influenced this different approach. See further e.g. Andrew Bainham, 'Rowing Back from *Re G*? Natural Parents in the Supreme Court' [2010] 40 (4) Fam. Law 394 and Everett and Yeatman, 'Are Some Parents More Natural Than Others'.

However, the implications of the judgment in $Re\ G$ may also be problematic for fathers in disputes with mothers, because they can only possess 'genetic' and 'psychological' parenthood, whereas only the mother can have 'gestational' parenthood and therefore possess all three characteristics.

²⁴⁰ In cases of assisted reproduction where the eggs of the non-gestational female 'parent' are used, both parties would possess two of the three incidences of 'natural parenthood' described above by Baroness Hale. However, it is suggested that the influence of the traditional construction of motherhood, premised around gestation, would result in the gestational mother retaining the exclusive term 'mother', over the claim of the 'genetic mother'. See further *Re G (Children) (Shared Residence Order: Biological Non-Birth Mother)* [2014] 2 FLR 897.

²⁴¹ Beresford, 'Get Over Your (Legal) 'Self': A Brief History of Lesbians, Motherhood and the Law', at 102.

disputes through a heterosexual prism once again illustrates the influence of the traditional, nuclear family model.

6.3.A.2. 'Known Donor' Disputes

This hierarchy of roles becomes even more problematic for the non-gestational female 'parent' in the context of disputes between lesbian couples and male 'known donors'. 242 It is notable that the terms 'father' and 'biological father' are used to describe the men in these cases. 243 In *A v B and C (Role of Father)*, 244 Black LJ observed that, '[t]he practice has grown up of referring to the father in circumstances such as this as a "donor"...it seems to me that the label might merit reconsideration...as it is capable of conveying the impression that the father is giving his child away and that is misleading. 245 In a commentary on this case Zanghellini describes the decision as being based upon a series of 'heteronormative assumptions'. 46 I want to suggest that such assumptions feature throughout the known donor decisions and consequently position the donor in the readily identifiable gendered role of the 'father', to the significant diminishment of the role of the non-gestational female 'parent', a role which lacks the rhetorical power possessed by the gendered parenting roles. 247

Boyd has observed that, '[t]he problem is that the legal system still seems tempted to impose a father figure on families that are headed, and sometimes carefully designed,

-

²⁴² Prior to the reforms of the Human Fertilisation and Embryology Act 2008 the non-gestational female 'parent' could not possess the status of legal parenthood; instead the 'known donor' would have been the legal parent on the basis of the genetic connection, making her position even weaker. All but one of the decisions discussed in this section concerned children born prior to the legislative reforms.

²⁴³ See e.g. Re P and L (Contact) [2012] 1 FLR 1068, T v T (Shared Residence) [2011] 1 FCR 267, R v E and F (Female Parents: Known Father) [2010] 2 FLR 383, Re B (Role of the Biological Father) [2008] 1 FLR 1015 and Re D (Contact and Parental Responsibility: Lesbian Mothers and Known Father) [2006] 1 FCR 556.

²⁴⁴ A v B and C (Role of Father) [2012] 2 FLR 607.

²⁴⁵ ibid, per Black LJ, at Para 40.

²⁴⁶ Aleardo Zanghellini, 'A v B and C [2012] EWCA Civ 285 - Heteronormativity, Poly-Parenting, and the Homo-Nuclear Family' [2012] 24 (4) Child and Family Law Quarterly 475.

Fox, 'The Human Fertilisation and Embryology Act 2008: Tinkering at the Margins', suggests at 338, that, 'adopting the terminology of "fathers" and "female parents" tends to suggest that the two roles are not equally valorised by law'.

by women.'²⁴⁸ The role of the father, and therefore the specifically male perspective, seems to be understood by the courts to be a necessary and valuable part of a child's life, on the basis of the traditional gendered construction of the role derived from the nuclear family. This understanding of the importance of the role of the 'father' is evident from the statement of Black J in *Re D (Contact and Parental Responsibility: Lesbian Mothers and Known Father)*²⁴⁹ that '[p]erhaps most importantly of all, I am considerably influenced by the reality that [Mr B] is [D's] father. Whatever new designs human beings have for the structure of their families, that aspect of nature cannot be overcome.'²⁵⁰ As Smith observes, this creates a problematic dichotomy because, 'whereas lesbian parents tend to emphasise that genetic fathers have less authority and significance than co-parents, law takes the opposite approach.'²⁵¹

The use of the terminology of 'biological fathers' is also notable because research has found that a variety of different terms are used within lesbian-led families who involve known donors in their parenting, Donovan has observed that '[for] the biological father the terms can range from "donor" through to "daddy", "uncle" or "adult friend".'252 I argue that the judicial use of the term 'biological father' results in greater normative significance being placed on these relationships than would be likely if the term 'known donor' was used in these cases, because of the resonance of the 'natural' role of the 'father' Smith states that the approach of the courts could be 'just as easily interpreted as a regressive effort to insert identifiable fathers into lesbian families. Such capitulation to the prevailing ideology of essential fathers

 $^{^{248}}$ Boyd, 'Gendering Legal Parenthood: Bio-Genetic Ties, Intentionality and Responsibility', at 92. 249 [2006] 1 FCR 556.

²⁵⁰ ibid, at Para 89.

²⁵¹ Smith, 'Is Three a Crowd? Lesbian Mothers Perspectives on Parental Status in Law', at 244.

²⁵² Catherine Donovan, 'Who Needs a Father? Negotiating Biological Fatherhood in British Lesbian Families Using Self-Insemination' (2000) 3 (2) Sexualities 149, at 156, see also the accounts of similar families in Kathryn Almack, 'Seeking Sperm: Accounts of Lesbian Couples' Reproductive Decision-Making and Understandings of the Needs of the Child' (2006) 20 (1) International Journal of Law, Policy and the Family 1.

²⁵³ Kelly, '(Re)Forming Parenthood: The Assignment of Legal Parentage Within Planned Lesbian Families', at 205, refers to the accounts of lesbian mothers and notes their acknowledgment of the, "utter failure" of language to capture the identities created', in 'known donor' situations.

²⁵⁴ Smith was writing of the judicial approach prior to the reforms of the Human Fertilisation and Embryology Act 2008, although it is notable that the language of 'biological fathers' was still employed by Baker J in *Re G, Re Z (Children: Sperm Donors: Leave to Apply for Children Act Orders)* [2013] 1 FLR 1334, the only reported decision subsequent to the HFEA 2008 where the lesbian couple were both legal parents. The donors in this case were granted leave to apply for contact orders, but no substantive judgment was made as to the merits of their application.

demeans the parenting capacities of lesbian couples and devalues the integrity of their families.' I submit that the approach of the courts in these 'known donor' cases, particularly the consistent use within the judgments of the language of 'biological fathers' further illustrates the influence of the gendered parenting roles upon the legal understanding of the parental role, to the disadvantage of the nongestational female 'parent'.

6.3.B. The Construction of the Role of the Non-Gestational Female 'Parent'

It is submitted that there is a lack of judicial consideration throughout all of these different types of cases as to the role performed by the non-gestational female 'parent', ²⁵⁶ which reflects the lack of clarity of the legal understanding of the overarching parental role. ²⁵⁷ Everett and Yeatman observe that '[I]esbian families are in essence different from heterosexual families, not only because they defy the primacy of the biological link as the defining factor in parenthood, but also because they have two mothers. ²⁵⁸ However, given that the legislation explicitly makes the terminological distinction between 'parent' and 'mother', ²⁶⁰ it is clear that the nongestational female 'parent' is not treated as an 'additional mother'. Although there is no distinction between the legal (parental) status of the 'mother' and the 'parent', I contend that the explicit difference in terminology influences judicial understandings of the two roles and this, as illustrated by the cases considered above, appears to be influencing judicial outcomes. ²⁶¹ This is underscored by the exclusivity inherent in

²⁵⁵ Smith, 'Tangling the Web of Legal Parenthood: Legal Responses to the Use of Known Donors in Lesbian Parenting Arrangements', at 377.

²⁵⁶ It is arguable that this lack of consideration reflects the limited normative significance accorded to social parenting and 'care' within the legal understanding of parenthood in contrast to reproducing the gendered roles of the nuclear family.

²⁵⁷ As set out above in section 6.1, 'What is the 'Parent' in Law?'

²⁵⁸ Everett and Yeatman, 'Are Some Parents More Natural Than Others?', at 304.

²⁵⁹ s.33 HFEA 2008.

²⁶⁰ s.42-47 HFEA 2008.

²⁶¹ Both in the context of disputes between lesbian couples, e.g. Re G (Children) (Residence: Same-Sex Partner) [2006] 2 FLR 629 and Re G (Children) (Shared Residence Order: Biological Non-Birth Mother) [2014] 2 FLR 897 and in the 'known donor' cases, e.g. A v B and C (Role of Father) [2012] 2 FLR 607 and Re G, Re Z (Children: Sperm Donors: Leave to Apply for Children Act Orders) [2013] 1 FLR 1334.

the traditional, 'natural' construction of mother,²⁶² with its fundamental link to the process of gestation, which the 'parent' self-evidently does not and cannot possess.

Alternatively, it is possible that the role of the non-gestational female 'parent' could be understood as essentially replicating the role of father within the traditional nuclear family. 263 Hedley J in Re P & L (Contact) did describe the role as 'second parent which in a traditional family would have been fulfilled by the father.'265 This understanding of parent could be read as having the radical implication that fatherhood does not actually have distinctly 'male' features and therefore that it is not necessary for children to have (male) fathers. 266 However, the approach adopted by the courts in the known donor decisions does not support this construction of the role of 'parent'. These decisions²⁶⁷ suggest that the father fulfils a specific, significant and gendered role which is understood as being additional to the day-today parenting performed by the lesbian couple; as Millbank has commented, '[1]esbian mothers may be a functional family but they are not a complete family.'268 I argue that the presence of a (male) father, completing the gendered binary of parenting roles within the nuclear family, possesses greater normative resonance in judicial reasoning than the practical contributions of the non-gestational female 'parent' in caring for the children. Thus, in its legal construction, the second female 'parent' is, as Golombok observes, 'more likely to be viewed as an additional parent than as a replacement parent.'269 It is my contention that, within the law, the role of

²⁶² As set out above in Chapter 5, subsection 5.2.A, 'Motherhood'.

²⁶³ It is suggested that the possibility of the role being viewed as replacing that of the father could be implied by the exclusion of the possibility of their being a legal 'father' when there is a 'parent' in s.45 (1) HFEA 2008, which states, '[w]here a woman is treated by virtue of section 42 or 43 as a parent of the child, no man is to be treated as the father of the child.'

²⁶⁴ [2012] 1 FLR 1068.

²⁶⁵ ibid, per Hedley J, at Para 4.

Such an approach, which explicitly treats fatherhood as secondary to motherhood would be in contradiction with the wider policy agenda emphasising the importance of fatherhood, discussed above at subsection 6.2.B, 'What is the 'Father' in the Law?'

²⁶⁷ See e.g. Re P and L (Contact) [2012] 1 FLR 1068, T v T (Shared Residence) [2011] 1 FCR 267, R v E and F (Female Parents: Known Father) [2010] 2 FLR 383, Re B (Role of the Biological Father) [2008] 1 FLR 1015.

²⁶⁸ Millbank, 'The Limits of Functional Family: Lesbian Mother Litigation in the Era of the Eternal Biological Family', at 162.

²⁶⁹ Susan Golombok, 'Lesbian Mother Families' in Bainham, Day Sclater and Richards (eds.), *What is a Parent?: A Socio-Legal Analysis*, at 172.

the non-gestational female 'parent' is not being constructed as an equivalent to either mother *or* father.

While there has been scant specific consideration of the latter role within the judicial decisions, sociological research suggests that lesbian couples do not necessarily reflect the traditional gendered parenting roles of the nuclear family. Instead, lesbian couples have been observed to adopt a more egalitarian form of parenting, with Gabb observing that '[f]eminity and masculinity are practised without regard to their cultural referents of female and male bodies, as lesbians raise children as both mothers and fathers - beyond gender'270 and Dunne commenting that, '[w]ithout exception, respondents believed that they approached and experienced parenting in ways that were very different from the heterosexual norm.'271 In spite of this, as illustrated above, within judicial decisions there has been little or no engagement with the radical possibilities created by lesbian-led families, who may aim to parent outside of the boundaries of the heterosexual nuclear family.

The importance of language in this context should also be emphasised; the absence of an obvious ordinary language term for the role of the non-gestational female 'parent' results in law employing the gender-neutral term 'parent'. This term lacks the easily understood, traditional, 'common-sense' construction possessed by the gendered roles of mother and father. Thus, the social parental role undertaken and the caring activities performed by the 'parent' cannot grant the status of 'mother', which remains exclusively premised upon gestation. This illustrates the tremendous power possessed by language, represented through the traditional nuclear family, to exclude those who do not fit within its boundaries. I argue that the lesbian-led family is constantly being judged against heteronormative standards of parenting and the

Gabb, 'Lesbian M/Otherhood: Strategies of Familial-linguistic Management in Lesbian Parent Families', at 592.

Dunne, 'Opting into Motherhood: Lesbians Blurring the Boundaries and Transforming the Meaning of Parenthood and Kinship', at 25.

²⁷² As considered above in Chapter 5, at subsection 5.2.B, 'Non-Gestational Female 'Parents', see e.g. Diduck, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood' and Probert, 'Families, Assisted Reproduction and the Law'.

²⁷³ See e.g. Jones, 'Parents in Law: Subjective Impacts and Status Implications around the Use of Licensed Donor Insemination' in Diduck and O'Donovan (eds.), *Feminist Perspectives on Family Law*.

nuclear family ideal with which it is not and cannot be truly analogous, because as Diduck points out:

[L]esbian co-parenthood is like nothing else. It is unlike stepparenthood, heterosexual parenthood by assisted reproduction or even single parenthood, all of which bear some similarity to it but are practised within a heterosexual norm that gives meaning to the concepts and practices of motherhood and fatherhood.²⁷⁴

I submit that it is because of this lack of understanding of the reality of lesbian parenting that the role of the non-gestational female 'parent' has not been fully explored or articulated within the law. The positioning of this role illustrates the continuing normative power of the gendered parenting roles of 'mother' and 'father' upon the legal understanding of the role of the 'parent'.

Conclusion

This chapter has argued that the gendered parenting roles of 'mother' and 'father', derived from the traditional nuclear family model, underpin the legal understanding of the role of the 'parent'.

It has been argued that law's starting point is an overarching conception of parenting which applies equally to all parents, and this is illustrated by legislative usage of the gender-neutral language of 'parent'. ²⁷⁵ I have suggested, however, that the legal understanding of the role of this 'parent' is unclear. In judicial interpretation of the parental role, significant reliance is placed upon the archetypical gendered parenting roles of 'mother' and 'father', which are understood as the 'natural' and 'common sense' parenting roles. I have argued that these roles are understood by the law on the

Through the definitions of 'parental responsibilities' and 'parental rights' in s.1-2 Children (Scotland) Act 1995 and s.3 Children Act 1989.

²⁷⁴ 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood', at 473.

basis of their distinct functions within the traditional, nuclear family: the 'mother' is constructed as the 'natural' primary carer for children, whereas the 'father' is understood as the breadwinner. I have argued that the understanding of the role of the 'parent' which is evident within the judicial interpretation of the welfare principle illustrated the influence of these gendered parental roles. Considering the law's approach to the role of the non-gestational female 'parent' in lesbian couples, it has been observed that this parental role is not fully explored or developed within the case law. Moreover, the gender-neutral term 'parent' lacks the kind of 'natural' or 'common-sense' construction possessed by the gendered roles of mother and father, which reinforces the significance of those traditional, gendered parenting roles.

My overarching claim in this Chapter has been that the legal understanding of the role of the 'parent', regardless of the usage of gender-neutral terminology, is substantially underpinned by the distinct, gendered parenting roles of 'mother' and 'father' derived from the nuclear family.

Chapter 7: Conclusion

This thesis began by observing that no single, universal definition of 'the family' exists within Scots law, English law, academic literature, or lay discourse within the United Kingdom. Instead, I have argued that a particular, idealised image of family dominates our cultural and legal understandings: the nuclear family, comprising the nexus of the conjugal relationship and the 'parent/child' relationship.

The impetus for this research was provided by the major reforms that have taken place within family law over the past several decades and the significant evolution in familial demographics within UK society during the same period. My concern has been to examine the extent to which the traditional archetype of family continues to exert rhetorical and normative influence upon the legal understanding of the 'family' despite the apparent radicalism of the recent changes.

7.1. Thesis Summary

First, I noted the absence of an agreed social or legal definition of the 'family'. A number of alternative definitions of 'family' were identified within the law. An examination of these definitions and their judicial interpretation revealed the centrality of the idealised image of the traditional, nuclear family. I argued for an understanding of the nuclear family as encompassing the nexus of the conjugal relationship and the 'parent/child' relationship (Chapter 2).

I observed that this idealised image of family had recurred consistently throughout history and had come to be positioned as the 'natural' and 'common sense' model of the 'family'. I contended that the nuclear family is underpinned *both* by the orthodox, liberal construction of the 'legal subject' as rational, autonomous and self-interested *and* by the particular liberal understanding of the 'public/private' divide which, historically, envisaged separate and distinct roles for men (as 'father' and 'breadwinner') and women (as 'mother' and 'homemaker') (Chapter 3).

Throughout, I have argued that this idealised image of family exerts significant influence upon the *legal* understanding of what 'family' means. This argument has been advanced both by examining various legal definitions of 'family', in Chapter 2, and by exploring how law constructs the two relationships identified as forming the nexus which comprises the nuclear family; the conjugal relationship, in Chapter 4 and the 'parent/child' relationship, in Chapters 5 and 6.

I observed the influence of the idealised image of the family upon the legal regulation of conjugal relationships, arguing that legal recognition and regulation of adult relationships is extended only to those relationships which can be situated within the boundaries of the nuclear family model. I noted that the traditional understanding of marriage retains normative significance within the legal understanding of marriage, despite recent legislation opening marriage to same-sex couples. I argued that legal recognition and regulation has been extended to other, non-marital adult conjugal relationships on the basis that they are constructed as being sufficiently 'marriage-like' to fulfil the same functions as marriage within the nuclear family (Chapter 4).

I noted the influence of the nuclear family model and its construction of the distinct, gendered parenting roles of 'mother' and 'father' upon the attribution of legal parenthood (Chapter 5) and in the legal construction of the role of the 'parent' (Chapter 6). In Chapter 5, I argued that legal parenthood is premised around a binary, two-parent model, which ideally comprises one mother and one father. I contended that this approach was particularly problematic for determining legal parenthood in the contexts of medically assisted reproduction and surrogacy, because of the additional factual complexity introduced by those circumstances.

In Chapter 6, I argued that the impact of these gendered roles upon the legal construction and understanding of the role of the 'parent' was evident. I observed that despite the use of the gender-neutral terminology of 'parent', the judicial understanding of parenthood and parenting continues to be underpinned by the

traditional, *gendered* parenting roles of 'mother' and 'father', as derived from the idealised image of the nuclear family.

7.2. Key Conclusions

The key conclusions of this thesis are that the understanding of 'the family' within the law remains premised upon the traditional, nuclear family, comprised of the nexus of the conjugal relationship and the 'parent/child' relationship, and that the continuing centrality of the nuclear model sits uneasily against the complex and diverse family forms, practices and structures within 21st century UK society. I concluded that the extension of legal regulation to adult relationships which possess 'marriage-like' conjugality illustrates the significance of this idealised image of the nuclear family within the legal understanding of the family. I further concluded that the influence of this archetype of family was evident from the continued normative significance of the gendered parenting roles of 'mother' and 'father' in the legal understanding of the role of the 'parent'. Ultimately, I have sought to establish, by detailed and systemic analysis, that, in spite of the aforementioned law reforms and changes in social and familial demographics, the traditional nuclear family retains its centrality within the legal understanding and construction of the 'family', while its core ideals and values have remained virtually unchanged and unchallenged.

My overall conclusion then is that the law reforms are less radical than they might appear at first sight and that their reflection of changes in demographics and changes in social acceptability of family forms must not blind us to the continuing gravitational pull of the traditional nuclear family model. Family law has not shifted its focus to a care-based, or a multi-layered, model. The parameters of the legal understanding of the 'family' were set long ago and while family law now regulates more family forms than it did fifty years ago, the parameters themselves remain steadfast.

7.3. The Family of the Law in the Future

In 1997, Mrs Justice Hale (as she then was), writing extra-judicially, asked of family law, '[c]an it evolve further to meet its new role: of encouraging families to go on meeting the responsibilities which we all need them to meet, despite a rapidly changing pattern of family relationships?' I regard this question as being as relevant now as it was then, given the demographic, social, and legislative changes that have occurred since then. Diduck has commented that '[f]amily law determines the responsibilities of individuals to each other and by extension, the responsibilities of families and the state and the community to each other.' I agree that family law will continue to serve these same functions in response to additional legislative reform, as well as to further evolutions in social practices and attitudes. As has been described here, social demographics and familial structures continue to change; parenting practices and the roles of mothers and fathers are still evolving, and medical and scientific progress continues to open up additional possibilities in the context of assisted reproduction. Smart has commented that:

As family life has become more fluid and diverse, with mixtures of cohabitation, marriage, remarriage, children from different relationships, a foreseeable rise in donor conceived children and

-

¹ Rt Hon Mrs Justice Hale, 'Private Lives and Public Duties: What is Family Law For?' (1998) 20 (2) Journal of Social Welfare and Family Law 125, at 126.

² Diduck, 'What is Family Law For?', at 292.

Notably the Legal Aid, Sentencing and Punishment of Offenders Act 2012 has resulted in the availability of legal aid in family cases in England and Wales being drastically reduced, it is now not available unless there is evidence of domestic violence (Schedule 1 Para 12) or harm to the child (Schedule 1 Para 13). The impact of these reforms is ongoing, but it is already clear that the reforms have had an effect on the operation of the family court process, with a significant increase in self-representation. While a full consideration of these reforms is outside the scope of this thesis, it is apparent that this reduction in the availability of legal aid will have a substantial influence upon the legal approach to regulating 'the family' in the future. See further e.g. Stephen Cobb, 'Legal Aid Reform: Its Impact on Family Law' (2013) 35 (1) Journal of Social Welfare and Family Law 3, Chris Bevan, 'Self-Represented Litigants: The Overlooked and Unintended Consequence of Legal Aid Reform' (2013) 35 (1) Journal of Social Welfare and Family Law 43, David Emmerson and John Platt, 'Legal Aid, Sentencing and Punishment of Offenders Act 2012: LASPO Reviewed' [2014] 44 (4) Fam. Law 515 and Kirsteen Mackay, 'A Plea from Scotland: Preserving Access to Courts in Private Law Child Contact Disputes' [2013] 25 (3) Child and Family Law Quarterly 294.

⁴ A child being born with 3 'genetic parents' is now a scientific possibility, see e.g. Nuffield Council on Bioethics, 'Novel Techniques for the Prevention of Mitochondrial DNA Disorders: An Ethical Review'. Indeed, the UK became the first country to allow such research as a result of the Human Fertilisation and Embryology (Mitochondrial Donation) Regulations 2015.

surrogacy, and more transnational relationships, family's law purpose is increasingly to give recognition and legitimacy to different forms of relationship and kinship.⁵

How will the legal understanding of 'the family' respond and develop as a result of these changes? Eekelaar observes that, '[t]he very fluidity and complexity of "real-life" family problems means that neat solutions cannot always be found, and it may take time to move to work towards the most satisfactory (or sometimes the "least worst") solutions. Sometimes there will be no solutions. '6 I anticipate that the law, due to its abstract and objective nature, will continue to encounter such difficulties in responding to family problems and resolving familial disputes. As Sir James Munby has noted extra-judicially, 'past judicial utterances that we now find almost absurd should serve as a terrible warning of how history will no doubt come, in due course, to judge the present generation.' With this in mind, I am conscious that it is impossible to predict how the law might react to unknown future social developments.

Nevertheless, as I have argued throughout this thesis, despite the recent major legislative reforms and substantial changes in social demographics, the traditional, nuclear family model, comprising the nexus of the conjugal relationship and the 'parent/child' relationship, has retained its normative significance as the 'common sense' and 'natural' idealised image of 'the family'. I have found no evidence to suggest that the substantial normative and rhetorical endurance of the nuclear family, this nexus of two relationships, is diminishing. It may well continue to underpin the legal understanding of 'family' for a long time to come.

⁵ Smart, 'Law and Family Life: Insights from 25 Years of Empirical Research', at 29.

⁶ Eekelaar, 'Then and Now - Family Law's Direction of Travel', at 422.

⁷ Munby, 'Years of Change: Family Law in 1987, 2012 and 2037', at 279.

References

Books

Austin, John, *Lectures on Jurisprudence or the Philosophy of Positive Law*, revised and edited by Robert Campbell, (3rd edition, John Murray, 1869)

Barry, John C., William Hay's Lectures on Marriage, (Stair Society, 1967)

Bauman, Zygmunt, Postmodern Ethics, (Blackwell, 1993)

Bauman, Zygmunt, Liquid Love: On the Frailty of Human Bonds, (Polity Press, 2003)

Bernardes, Jon, Family Studies: An Introduction, (Routledge, 1997)

Blackstone, Sir William, Commentaries on the Laws of England, Volume 1, (4th edition, J. Exshaw, 1771)

Bottomley, Stephen and Bronitt, Simon, Law in Context, (4th edition, Federation Press, 2012)

Calhoun, Cheshire, Feminism, the Family, and the Politics of the Closet: Lesbian and Gay Displacement, (Oxford University Press, 2000)

Cane, Peter, Responsibility in Law and Morality, (Hart, 2002)

Carty, Anthony (ed.), *Post-Modern Law: Enlightenment, Revolution and the Death of Man*, (Edinburgh University Press, 1990)

Cassirer, Ernst, *The Philosophy of the Enlightenment*, (Beacon Press, 1955, translated from the original German edition, 1932)

Chambers, Deborah, A Sociology of Family Life: Change and Diversity in Intimate Relationships, (Polity Press, 2012)

Cheal, David, Sociology of Family Life, (Palgrave, 2002)

Collier, Richard, Masculinity, Law and the Family, (Routledge, 1995)

Collier, Richard and Sheldon, Sally (eds.), Fathers Rights Activism and Law Reform in Comparative Perspective, (Hart, 2006)

Collins, Hugh, Marxism and Law, (Oxford University Press, 1982)

Coltrane, Scott, Family Man: Fatherhood, Housework and Gender Equality, (Oxford University Press, 1996)

Coltrane, Scott, Gender and Families, (Rowman and Littlefield, 1998)

Connell, R.W., *Masculinities*, (University of California Press, 2005)

Cook, Rachel and Day Sclater, Shelley with Kaganas, Felicity (eds.), *Surrogate Motherhood: International Perspectives*, (Hart, 2003)

Cotterrell, Roger, *The Sociology of Law: An Introduction*, (Butterworths, 1992)

Davies, Margaret, Asking the Law Question, (3rd edition, Lawbook Co., 2008)

Dickenson, Donna, Property, Women and Politics: Subjects or Objects?, (Polity Press, 1997)

Diduck, Alison, Law's Families, (Markham, 2003)

Diduck, Alison and Kaganas, Felicity, Family Law, Gender and the State, (3rd edition, Hart, 2012)

Diduck, Alison and O'Donovan, Katherine (eds.), Feminist Perspectives on Family Law, (Abingdon, 2006)

Dienhart, Anna, Reshaping Fatherhood: The Social Construction of Shared Parenting, (SAGE, 1998)

Douzinas, Costas and Geary, Adam, *Critical Jurisprudence: The Political Philosophy of Justice*, (Hart, 2005)

Douzinas, Costas and Warrington, Ronnie, *Justice Miscarried: Ethics, Aesthetics and the Law*, (Harvester Wheatsheaf, 1994)

du Plessis, Paul, Borkowski's Textbook on Roman Law, (Oxford University Press, 2010)

Elliot, Faith Robertson, The Family: Change of Continuity?, (MacMillan, 1986)

France, Anatole, Le Lys Rouge, (Calmann-Levy, 1894)

Fraser, Patrick, *Treatise on Husband and Wife, According to the Law of Scotland*, (2nd edition, T & T Clark, 1876-78)

Friedman, Marilyn, *Liberating Care*, (Cornell University Press, 1993)

Frier, Bruce W. and McGinn, Thomas A.J., *A Casebook on Roman Family Law*, (Oxford University Press, 2004)

Gaius, 'Institutes of Gaius', Francis de Zulueta (ed.), (Clarendon Press, 1946-53)

Gay, Peter, The Enlightenment: An Interpretation - The Rise of Modern Paganism, (W.W. Norton, 1995)

Gilligan, Carol, In a Different Voice: Psychological Theory and Women's Development, (Harvard University Press, 1982)

Gittins, Diana, *The Family in Question: Changing Households and Familiar Ideologies*, (2nd edition, MacMillan, 1993)

Goodsell, Willystine, *A History of Marriage and the Family*, (2nd edition, MacMillan, 1934)

Govier, Trudy, Dilemmas of Trust, (McGill's-Queen's University Press, 1998)

Grassby, Richard, Kinship and Capitalism: Marriage, Family and Business in the English-Speaking World, 1580-1740, (Cambridge University Press, 2001)

Groenhout, Ruth, Connected Lives: Human Nature and an Ethics of Care, (Rowman and Littlefield, 2004)

Hand, Sean (ed.), The Levinas Reader, (Blackwell, 1989)

Harris, C.C., *The Family: An Introduction*, (Allen and Unwin, 1969)

Harris, D.J, O'Boyle M., Bates E.P. and Buckley, M., Law of the European Convention on Human Rights, (Oxford University Press, 2009)

Harrison, A.W., *The Law of Athens: The Family and Property*, (Oxford University Press, 1968)

Heaphy, Brian, Smart, Carol and Einarsdottir, Anna, Same Sex Marriages: New Generations, New Relationships, (Palgrave MacMillan, 2013)

Held, Virginia, *The Ethics of Care*, (Oxford University Press, 2006)

Hennon, Charles B. and Wilson, Stephan M., Families in a Global Context, (Routledge, 2008)

Herring, Jonathan, Caring and the Law, (Hart, 2013)

Herring, Jonathan, *Family Law*, (6th edition, Pearson Longman, 2013)

Herring, Jonathan, Family Law, (7th edition, Pearson Longman, 2015)

Hobbes, Thomas, *Leviathan*, (First published 1651, reprinted by Cambridge University Press, 1991)

Hodson, Barbara (ed.), Making Men into Fathers: Men, Masculinities and the Social Politics of Fatherhood, (Cambridge University Press, 2002)

Houlbrooke, Ralph A., The English Family: 1450-1700, (Longman, 1984)

Hume, David, *A Treatise of Human Nature*, (First published 1739, reprinted by Clarendon Press, 1896)

Jackson, Emily, Regulating Reproduction: Law, Technology and Autonomy, (Hart, 2001)

Jaggar, Alison, Feminist Politics and Human Nature, (Rowman and Littlefield, 1988)

James, Susan and Palmer, Stephanie (eds.), Visible Women: Essays on Feminist Legal Theory and Political Philosophy, (Hart, 2002)

Kelsen, Hans, General Theory of Law and State, (Russell and Russell, 1945)

Kelsen, Hans, *Pure Theory of Law*, (University of California Press, 1967)

Lacey, Nicola, Unspeakable Subjects: Feminist Essays in Legal and Social Theory, (Hart, 1998)

Lacey, W.K., *The Family in Classical Greece*, (The Author, 1980)

Leeder, Elaine, The Family in Global Perspective: A Gendered Journey, (SAGE, 2003)

Levinas, Emmanuel, *Alterity and Transcendence*, (Linton, 1999)

Levinas, Emmanuel, *Totality and Infinity*, (Duquesne University Press, 1999)

Lukes, Steven, *Individualism*, (Blackwell, 1973)

Lupton, Deborah and Barclay, Lesley, Constructing Fatherhood: Discourses and Experiences, (SAGE, 1997)

Mac Chombaich De Colquhoun, Patrick, *A Summary of the Roman Civil Law*, Volume 1, (V. and R. Stevens and Sons, 1849)

MacFarlane, Alan, *The Origins of English Individualism: The Family, Property and Social Transition*, (Blackwell, 1978)

MacFarlane, Alan, Marriage and Love in England: Modes of Reproduction, 1300-1840, (Blackwell, 1986)

McKie, Linda and Callan, Samantha, *Understanding Families: A Global Introduction*, (SAGE, 2012)

McPherson, C.B., The Political Theory of Possessive Individualism: Hobbes to Locke, (Clarendon Press, 1962)

Mill, John Stuart, *On Liberty and Other Writings*, Stefan Collins (ed.), (Cambridge University Press, 1989)

Moller Okin, Susan, Women in Western Political Thought, (Virago, 1980)

Morgan, David, Family Connections: An Introduction to Family Studies, (Polity Press, 1996)

Naffine, Ngaire, Law and the Sexes: Explorations of Feminist Jurisprudence, (Allen and Unwin, 1990)

Naffine, Ngaire, Law's Meaning of Life: Philosophy, Religion, Darwin and the Legal Person, (Hart, 2009)

Naffine, Ngaire and Owens, Rosemary J. (eds.), *Sexing the Subject of Law*, (LBC Information Services, 1997)

Nekam, Alexander, *The Personality Conception of the Legal Entity*, (Harvard University Press, 1938)

Noddings, Nel, Caring: A Feminine Approach to Ethics and Moral Education, (University of California Press, 1984)

Noddings, Nel, *Starting at Home: Caring and Social Policy*, (University of California Press, 2002)

O'Donovan, Katherine, Sexual Divisions in Law, (Weidenfeld and Nicholson, 1985)

O'Donovan, Katherine, Family Law Matters, (Pluto Press, 1993)

Pashukanis, Evgenii, *Law and Marxism: A General Theory*, Chris Arthur (ed.), (Pluto Press, 1989)

Pateman, Carole, *The Sexual Contract*, (Polity, 1988)

Perkin, Joan, Women and Marriage in Nineteenth-Century England, (Routledge, 1988)

Rousseau, Jean-Jacques, Of the Social Contract, Or Principles of Political Right, (First published 1762, reprinted by Dent, 1973)

Sachs, Albie and Hoff Wilson, Joan, Sexism and the Law: A Study of Male Beliefs and Legal Bias in Britain and the United States, (Martin Robertson, 1978)

Sealey, Raphael, *The Justice of the Greeks*, (University of Michigan Press, 1994)

Sheldon, Sally, and Collier, Richard, Fragmenting Fatherhood: A Socio-Legal Study, (Hart, 2008)

Sloan, Brian, Informal Carers and Private Law, (Hart, 2012)

Smart, Carol, Personal Life: New Directions in Sociological Thinking, (Polity Press, 2007)

Smart, Carol, Neale, Bren and Wade, Amanda, *The Changing Experience of Childhood: Families and Divorce*, (Polity, 2001)

Stone, Lawrence, *The Family, Sex and Marriage in England 1500-1800*, (Penguin, 1979)

Taylor, Mark C., *Deconstruction in Context: Literature and Philosophy*, (University of Chicago Press, 1986)

Thomson, J. M., Family Law in Scotland, (7th edition, Bloomsbury Professional, 2014)

Torrant, Julie P., *The Material Family*, (Sense, 2011)

Tronto, Joan, Moral Boundaries: a Political Argument for an Ethic of Care, (Routledge, 1993)

Unger, Roberto, *Knowledge and Politics*, (Free Press, 1975)

Vincent, Andrew, *Modern Political Ideologies*, (3rd edition, Wiley-Blackwell, 2009)

Weeks, Jeffrey, Heaphy, Brian and Donovan, Catherine, Same Sex Intimacies: Families of Choice and Other Life Experiences, (Routledge 2001)

White, Robin C.A. and Ovey, Clare, *Jacobs, White and Ovey: The European Convention on Human Rights*, (5th edition, Oxford University Press, 2010)

Wikeley, Nicholas, Child Support: Law and Policy, (Hart, 2006)

Williams, Fiona, *Rethinking Families*, (Calouste Gulbenkian Foundation, 2004)

Young, Iris Marion, *Intersecting Voices*, (Princeton University Press, 1997)

Chapters in Edited Collections

Auchmuty, Rosemary, 'Dissolution or Disillusion: The Unravelling of Civil Partnerships' in Nicola Barker and Daniel Monk (eds.), *From Civil Partnership to Same-Sex Marriage: Interdisciplinary Reflections*, (Routledge, 2015)

Bainham, Andrew, 'Parentage, Parenthood and Parental Responsibility: Subtle, Elusive, Yet Very Important Distinctions' in Andrew Bainham, Shelley Day Sclater and Martin Richards (eds.), What is a Parent?: A Socio-Legal Analysis, (Hart, 1999)

Bainham, Andrew 'Contact as a Right and Obligation' in Andrew Bainham, Bridget Lindley, Martin Richards and Liz Trinder (eds.), *Children and Their Families:* Contact, Rights and Welfare, (Hart, 2003)

Andrew Bainham, Shelley Day Sclater and Martin Richards, 'Introduction' in Andrew Bainham, Shelley Day Sclater and Martin Richards (eds.), *What is a Parent?: A Socio-Legal Analysis*, (Hart, 1999)

Barron, Anne, 'Legal Discourse and the Colonisation of the Self in the Modern State' in Anthony Carty (ed.), *Post-Modern Law: Enlightenment, Revolution and the Death of Man*, (Edinburgh University Press, 1990)

Bottomley, Anne and Wong, Simone, 'Shared Households: A New Paradigm for Thinking about the Reform of Domestic Property Relations' in Alison Diduck and Katherine O'Donovan (eds.), *Feminist Perspectives on Family Law*, (Abingdon, 2006)

Boyd, Susan, 'Challenging the Public/Private Divide: An Overview' in Susan Boyd (ed.), *Challenging the Public/Private Divide: Feminism, Law and Public Policy*, (Toronto University Press, 1997)

Bridgeman, Jo, 'Parental Responsibility, Responsible Parenting and Legal Regulation' in Jo Bridgeman, Craig Lind and Helen M. Keating (eds.), Responsibility, Law and the Family, (Ashgate, 2008)

Collier, Richard, 'Anxious Parenthood, the Vulnerable Child and the 'Good Father': Reflections on the Legal Regulation of the Relationship between Men and Children' in Jo Bridgeman and Daniel Monk (eds.), *Feminist Perspectives on Child Law*, (Cavendish, 2000)

Cook, Rachel, 'Donating Parenthood: Perspectives on Parenthood from Surrogacy and Gamete Donation' in Andrew Bainham, Shelley Day Sclater and Martin Richards (eds.), *What is a Parent?: A Socio-Legal Analysis*, (Hart, 1999)

Derham, D.P., 'Theories of Legal Personality' in Leicester Webb (ed.), *Legal Personality and Political Pluralism*, (Melbourne University Press, 1958)

Freeman, Michael, 'Feminism and Child Law' in Jo Bridgeman and Daniel Monk (eds.), Feminist Perspectives on Child Law, (Cavendish, 2000)

Gilmore, Stephen, Herring, Jonathan and Probert, Rebecca, 'Introduction: Parental Responsibility - Law, Issues and Themes' in Rebecca Probert, Stephen Gilmore and Jonathan Herring (eds.), *Responsible Parents and Parental Responsibility*, (Hart, 2009)

Golombok, Susan, 'Lesbian Mother Families' in Andrew Bainham, Shelley Day Sclater and Martin Richards (eds.), *What is a Parent?: A Socio-Legal Analysis*, (Hart, 1999)

Gough, Kathleen E., 'Is the Family Universal? - The Nayar Case' in Norman W. Bell and Ezra F. Vogel (eds.), *A Modern Introduction to The Family*, (Revised edition, Free Press, 1968)

Jackson, Emily, 'What is a Parent?' in Alison Diduck and Katherine O'Donovan (eds.), Feminist Perspectives on Family Law, (Abingdon, 2006)

Jones, Caroline, 'Parents in Law: Subjective Impacts and Status Implications around the Use of Licensed Donor Insemination' in Alison Diduck and Katherine O'Donovan (eds.), Feminist Perspectives on Family Law, (Abingdon, 2006)

Lacey, Nicola, "Philosophical Foundations of the Common Law": Social not Metaphysical" in Jeremy Horder (ed.), *Oxford Essays in Jurisprudence*, (4th series, Oxford University Press, 2000)

Laster, Kathy and Ranan, Padma, 'Law for One and One for All? An Intersectional Legal Subject', in Ngaire Naffine and Rosemary J. Owens (eds.), *Sexing the Subject of Law*, (LBC Information Services, 1997)

Linton, Ralph, 'The Natural History of the Family' in Ruth Nanda Anshen (eds.), *The Family: Its Function and Destiny*, (Revised edition, Harper, 1959)

Muncie, John and Sapsford, Roger, 'Issues in the Study of 'The Family' in John Muncie, et all (eds.), *Understanding the Family*, (2nd edition, SAGE, 1997)

Murdock, George P., 'The Universality of the Nuclear Family' in Norman W. Bell and Ezra F. Vogel (eds.), *A Modern Introduction to The Family*, (Revised edition, Free Press, 1968)

Naffine, Ngaire, 'Can Women Be Legal Persons?' in Susan James and Stephanie Palmer (eds.), Visible Women: Essays on Feminist Legal Theory and Political Philosophy, (Hart, 2002)

Naffine, Ngaire and Owens, Rosemary J., 'Introduction: Sexing Law' in Ngaire Naffine and Rosemary J. Owens (eds.), *Sexing the Subject of Law*, (LBC Information Services, 1997)

O'Donovan, Katherine, 'With Sense, Consent, or Just a Con?: Legal Subjects in the Discourse of Autonomy', in Ngaire Naffine and Rosemary J. Owens (eds.), *Sexing the Subject of Law*, (LBC Information Services, 1997)

O'Donovan, Katherine, 'Constructions of Maternity and Motherhood in Stories of Lost Children' in Jo Bridgeman and Daniel Monk (eds.), *Feminist Perspectives on Child Law*, (Cavendish, 2000)

O'Donovan, Katherine and Marshall, Jill, 'After Birth: Decisions about Becoming a Mother' in Alison Diduck and Katherine O'Donovan (eds.), *Feminist Perspectives on Family Law*, (Abingdon, 2006)

Rawson, Beryl, 'The Roman Family' in Beryl Rawson (ed.), *The Family in Ancient Rome: New Perspectives*, (Croom Helm, 1986)

Reece, Helen, 'The Degradation of Parental Responsibility' in Rebecca Probert, Stephen Gilmore and Jonathan Herring (eds.), *Responsible Parents and Parental Responsibility*, (Hart, 2009)

Silva, Elizabeth B. and Smart, Carol, 'The 'New' Practices and Politics of Family Life' in Elizabeth B. Silva and Carol Smart (eds.), *The New Family?*, (Thousand Oaks, 1999)

Simpson, Bob, Jessop, Julie A. and McCarthy, Peter, 'Fathers After Divorce' in Andrew Bainham, Bridget Lindley, Martin Richards and Liz Trinder (eds.), *Children and Their Families: Contact, Rights and Welfare*, (Hart, 2003)

Smart, Carol, 'The New Parenthood: Fathers and Mothers after Divorce' in Elizabeth B. Silva and Carol Smart (eds.), *The New Family?*, (Thousand Oaks, 1999)

Smart, Carol, 'The Ethic of Justice Strikes Back: Changing Narratives of Fatherhood' in Alison Diduck and Katherine O'Donovan (eds.), *Feminist Perspectives on Family Law*, (Abingdon, 2006)

Stoddard, Thomas B., 'Why Gay People Should Seek the Right to Marry' in Mark Blasius and Shane Phelan (eds.), We Are Everywhere: A Historical Sourcebook of Gay and Lesbian Politics, (Routledge, 1997)

Stychin, Carl, 'Family Friendly? Rights, Responsibilities and Relationship Recognition?' in Alison Diduck and Katherine O'Donovan (eds.), Feminist Perspectives on Family Law, (Abingdon, 2006)

Sugarman, David and Rubin, G.R., 'Introduction: Towards a New History of Law and Material Society in England 1750-1914' in G.R. Rubin and David Sugarman (eds.), Law, Economy and Society, 1750-1914: Essays in the History of English Law, (Professional Books, 1984)

Trinder, Liz, 'Introduction' in Andrew Bainham, Bridget Lindley, Martin Richards and Liz Trinder (eds.), *Children and Their Families: Contact, Rights and Welfare*, (Hart, 2003)

Tronto, Joan, 'Women and Caring: What Can Feminists Learn about Morality from Caring?' in Alison Jaggar and Susan Bordo (eds.), *Gender/Body/Knowledge: Feminist Reconstructions of Being and Knowing*, (Rutgers University Press, 1989)

Tur, Richard, 'The 'Person' in Law' in Arthur Peacocke and Grant Gillett (eds.), *Persons and Personality in Contemporary Inquiry*, (Basil Blackwell, 1987)

Journal Articles

Abdalla, H., Shenfield, F. and Latarche, E., 'Statutory Information for the Children Born of Oocyte Donation in the UK: What Will They Be Told in 2008?' [1998] 13 (4) Human Reproduction 1106

Ahuja, K., Simons, E., Mostyn, B. and Bowen-Simpkins, P., 'An Assessment of the Motives and Morals of Egg Share Donors: Policy of 'Payments' to Egg Donors Requires a Fair Review' [1998] 13 (10) Human Reproduction 2671

Almack, Kathryn, 'Seeking Sperm: Accounts of Lesbian Couples' Reproductive Decision-Making and Understandings of the Needs of the Child' (2006) 20 (1) International Journal of Law, Policy and the Family 1

Amato, Paul and Gilbreth, Joan, 'Nonresident Fathers and Children's Well-Being: A Meta-Analysis' (1999) 61 (3) Journal of Marriage and the Family 557

Auchmuty, Rosemary, 'What's So Special About Marriage? The Impact of Wilkinson v Kitzinger' [2008] 20 (4) Child and Family Law Quarterly 475

Auchmuty, Rosemary, 'Beyond Couples' (2009) 17 (2) Feminist Legal Studies 205

Auchmuty, Rosemary, 'Law and the Power of Feminism: How Marriage Lost its Power to Oppress Women' (2012) 20 (2) Feminist Legal Studies 71

Auchmuty, Rosemary, 'The Experience of Civil Partnership Dissolution: Not 'Just Like Divorce' (2016) 38 (2) Journal of Social Welfare and Family Law 152

Bainham, Andrew, 'Arguments About Parentage' (2008) 67 (2) Cambridge Law Journal 322

Bainham, Andrew, 'Rowing Back from *Re G*? Natural Parents in the Supreme Court' [2010] 40 (4) Fam. Law 394

Bamforth, Nicholas, "The Benefits of Marriage in all but Name'? Same-Sex Couples and the Civil Partnership Act 2004' [2007] 19 (2) Child and Family Law Quarterly 133

Barker, Nicola, 'Sex and the Civil Partnership Act: The Future of (Non) Conjugality' (2006) 14 (2) Feminist Legal Studies 241

Barlow, Anne, 'Cohabitation Law Reform - Messages from Research' (2006) 14 (2) Feminist Legal Studies 167

Barlow, Anne, 'Solidarity, Autonomy and Equality: Mixed Messages for the Family?' [2015] 27 (3) Child and Family Law Quarterly 223

Barlow, Anne and James, Grace, 'Regulating Marriage and Cohabitation in 21st Century Britain' (2004) 67 (2) Modern Law Review 143

Barnett, Adrienne, 'The Welfare of the Child Re-Visited: In Whose Best Interests? Part 1' [2009] 39 (1) Fam. Law 50

Barnett, Adrienne, 'The Welfare of the Child Re-Visited: In Whose Best Interests? Part 2' [2009] 39 (2) Fam. Law 135

Bendall, Charlotte, 'A Break Away from the (Hetero)norm?: *Lawrence v Gallagher* [2012] 1 FCR 557; [2012] EWA Civ 394' (2013) 21 (3) Feminist Legal Studies 303

Beresford, Sarah, 'Get Over Your (Legal) 'Self': A Brief History of Lesbians, Motherhood and the Law' (2008) 30 (2) Journal of Social Welfare and Family Law 95

Bevan, Chris, 'Self-Represented Litigants: The Overlooked and Unintended Consequence of Legal Aid Reform' (2013) 35 (1) Journal of Social Welfare and Family Law 43

Bottomley, Anne, 'From Mrs Burns to Mrs Oxley: Do Co-Habiting Women (Still) Need Marriage Law' (2006) 14 (2) Feminist Legal Studies 181

Boyd, Susan, 'What is a 'Normal' Family? C v C (A Minor) (Custody: Appeal)' (1992) 55 (2) Modern Law Review 269

Boyd, Susan, 'Gendering Legal Parenthood: Bio-Genetic Ties, Intentionality and Responsibility' (2007) 25 Windsor Yearbook of Access to Justice 63

Brook, Heather, 'Zombie Law: Conjugality, Annulment and the (Married) Living Dead' (2014) 22 (1) Feminist Legal Studies 49

Butler-Sloss, Elizabeth, 'Ethical Considerations in Family Law' (2006) Web Journal of Current Legal Issues 4

Callus, Therese, 'First 'Designer Babies', Now a La Carte Parents' [2008] 38 (2) Fam. Law 143

Chan, Winnie, 'Cohabitation, Civil Partnership, Marriage and the Equal Sharing Principle' (2013) 33 (1) Legal Studies 46

Clarke, Victoria, 'Lesbian and Gay Marriage: Transformation or Normalisation?' (2003) 13 (4) Feminism and Psychology 519

Clough, Beverley, 'What About Us? A Case for Legal Recognition of

Interdependence in Informal Care Relationships' (2014) 36 (2) Journal of Social Welfare and Family Law 129

Cobb, Stephen, 'Legal Aid Reform: Its Impact on Family Law' (2013) 35 (1) Journal of Social Welfare and Family Law 3

Collier, Richard, 'A Hard Time to Be a Father?: Reassessing the Relationship Between Law, Policy, and Family (Practices)' (2001) 28 (4) Journal of Law and Society 520

Crawford, Claire, Goodman, Alissa, Greaves, Ellen and Joyce, Rob, 'Cohabitation, Marriage and Child Outcomes: An Empirical Analysis of the Relationship between Marital Status and Child Outcomes in the UK using the Millennium Cohort Study' [2012] 24 (2) Child and Family Law Quarterly 176

Crawshaw, Marilyn, Blyth, Eric and van den Akker, Olga, 'The Changing Profile of Surrogacy in the UK - Implications for National and International Policy and Practice' (2012) 34 (3) Journal of Social Welfare and Family Law 267

Cretney, Stephen M. and Reynolds, F.M.B., 'Limits of the Judicial Function' (2000) 116 (Apr) Law Quarterly Review 181

Crompton, Lucy, 'Where's the Sex in Same-Sex Marriage?' [2013] 43 (5) Fam. Law 564

Cullen, Deborah, 'Adoption - a (Fairly) New Approach' [2005] 17 (4) Child and Family Law Quarterly 475

Dawson, John, 'The Changing Legal Status of Mentally Disabled People' (1994) 2 Journal of Law and Medicine 38

Deech, Baroness Ruth, 'Sisters Sisters - And Other Family Members' [2010] 40 (4) Fam. Law 375

Dewar, John, 'The Normal Chaos of Family Law' (1998) 61 (4) Modern Law Review 467

Diduck, Alison, 'Shifting Familiarity' (2005) 58 (1) Current Legal Problems 235

Diduck, Alison, 'If Only we can Find the Appropriate Terms to Use the Issue Will Be Solved: Law, Identity and Parenthood' [2007] 19 (4) Child and Family Law Quarterly 458

Diduck, Alison, 'What is Family Law For?' (2011) 64 (1) Current Legal Problems 287

Dietz, Mary, 'Context is All: Feminism and Theories of Citizenship' (1987) 116 (4) Daedalus 1

Donovan, Catherine, 'Who Needs a Father? Negotiating Biological Fatherhood in British Lesbian Families Using Self-Insemination' (2000) 3 (2) Sexualities 149

Donovan, Catherine, 'Genetics, Fathers and Families: Exploring the Implications of Changing the Law in Favour of Identifying Sperm Donors' (2006) 15 (4) Social and Legal Studies 494

Douglas, Gillian, 'The Intention to Be a Parent and the Making of Mothers' (1994) 57 (4) Modern Law Review 636

Draper, Helen and Ives, Jonathan, 'Paternity Testing: a Poor Test of Fatherhood' (2009) 31 (4) Journal of Social Welfare and Family Law 407

Dunn, Judy, Cheng, Helen, O'Connor, Thomas G. and Bridges, Laura, 'Children's Perspectives on their Relationships with their Nonresident Fathers: Influences, Outcomes and Implications' (2004) 45 (3) Journal of Child Psychology and Psychiatry 55

Dunne, Gillian, 'Opting into Motherhood: Lesbians Blurring the Boundaries and Transforming the Meaning of Parenthood and Kinship' (2000) 14 (1) Gender and Society 11

Eekelaar, John, 'Rethinking Parental Responsibility' [2001] 31 (6) Fam. Law 426

Eekelaar, John, 'Beyond the Welfare Principle' [2002] 14 (3) Child and Family Law Quarterly 237

Eekelaar, John, 'Then and Now - Family Law's Direction of Travel' (2013) 35 (4) Journal of Social Welfare and Family Law 415

Eekelaar, John and MacLean, Mavis, 'Marriage and the Moral Bases of Personal Responsibility' (2004) 31 (4) Journal of Law and Society 510

Emmerson, David and Platt, John, 'Legal Aid, Sentencing and Punishment of Offenders Act 2012: LASPO Reviewed' [2014] 44 (4) Fam. Law 515

Everett, Kim and Yeatman, Lucy, 'Are Some Parents More Natural Than Others?' [2010] 22 (3) Child and Family Law Quarterly 290

Featherstone, Brid, 'Taking Fathers Seriously' (2003) 33 (2) British Journal of Social Work 239

Fenton, Rachel, Heenan, Susan and Rees, Jane, 'Finally Fit For Purpose? The Human Fertilization and Embryology Act 2008' (2010) 32 (3) Journal of Social Welfare and Family Law 275

Fenton-Glynn, Claire, 'The Regulation and Recognition of Surrogacy Under English Law: An Overview of the Case-Law' [2015] 27 (1) Child and Family Law Quarterly

Fortin, Jane, 'Children's Right to Know Their Origins - Too far, Too fast?' [2009] 21 (3) Child and Family Law Quarterly 336

Fox, Marie, 'The Human Fertilisation and Embryology Act 2008: Tinkering at the Margins' (2009) 17 (3) Feminist Legal Studies 333

French, Peter, 'The Corporation as a Moral Person' (1979) 16 (3) American Philosophical Quarterly 207

Gabb, Jacqui, 'Lesbian M/Otherhood: Strategies of Familial-linguistic Management in Lesbian Parent Families' (2005) 39 (4) Sociology 585

Gaffney-Rhys, Ruth, 'EU Directive on Freedom of Movement' (2006) International Family Law 65

Gardiner, Michael, 'Alterity and Ethics: A Dialogical Perspective' (1996) 13 (2) Theory, Culture & Society 121

Gilmore, Stephen, 'Bellinger v Bellinger - Not Quite Between the Ears and Between the Legs - Transsexualism and Marriage in the Lords' [2003] 15 (3) Child and Family Law Quarterly 295

Gilmore, Stephen, 'Contact/Shared Residence and Child Well-Being: Research Evidence and its Implications for Legal Decision-Making' (2006) 20 (3) International Journal of Law, Policy and the Family 344

Gilmore, Stephen, 'Disputing Contact: Challenging Some Assumptions' [2008] 20 (3) Child and Family Law Quarterly 285

Glennon, Lisa, 'Fitzpatrick v Sterling Housing Association Ltd. - An Endorsement of the Functional Family?' (2000) 14 (3) International Journal of Law, Policy and the Family 226

Glennon, Lisa, 'Displacing the 'Conjugal Family' in Legal Policy - A Progressive Move' [2005] 17 (2) Child and Family Law Quarterly 141

Golombok, S., Braeways, A., Giavazzi, M.T., Guerra, D., MacCallum, F. and Rust, J., 'The European Study of Assisted Reproduction Families: The Transition to Adolescence' [2002] 17 (3) Human Reproduction 830

Golombok, S., Murray, C., Jadva, V., Lycett, E., MacCallum, F. and Rust, J., 'Non-Genetic and Non-Gestational Parenthood: Consequences for Parent-Child Relationships and the Psychological Well-Being of Mothers, Fathers and Children at Age 3' [2006] 21 (7) Human Reproduction 1918

Hale, Brenda, 'Equality and Autonomy in Family Law' (2011) 33 (1) Journal of

Social Welfare and Family Law 3

Hale, Rt Hon Mrs Justice, 'Private Lives and Public Duties: What is Family Law For?' (1998) 20 (2) Journal of Social Welfare and Family Law 125

Harding, Rosie, 'Sir Mark Potter and the Protection of the Traditional Family: Why Same Sex Marriage is (Still) a Feminist Issue' (2007) 15 (2) Feminist Legal Studies 223

Harding, Rosie, 'Recognising (and Resisting) Regulation: Attitudes to the Introduction of Civil Partnership' (2008) 11 (6) Sexualities 740

Harris, Peter G. and George, Robert H., 'Parental Responsibility and Shared Residence Orders: Parliamentary Intentions and Judicial Interpretation' [2010] 22 (2) Child and Family Law Quarterly 151

Herring, Jonathan, 'Where are the Carers in Healthcare Law and Ethics?' (2007) 27 (1) Legal Studies 51

Herring, Jonathan, 'The Legal Duties of Carers' (2010) 18 (2) Medical Law Review 248

Herring, Jonathan, 'The Disability Critique of Care' (2014) 8 Elder Law Review 1

Herring, Jonathan, 'The Welfare Principle and the Children Act: Presumably it's About Welfare?' (2014) 36 (1) Journal of Social Welfare and Family Law 14

Herring, Jonathan and Foster, Charles, 'Welfare Means Relationality, Virtue and Altruism' (2012) 32 (3) Legal Studies 480

Herring, Jonathan and Powell, Oliver, 'The Rise and Fall of Presumptions Surrounding the Welfare Principle' [2013] 43 (5) Fam. Law 553

Hibbs, Mary, Barton, Chris and Beswick, Joanne, 'Why Marry? - Perceptions of the Affianced' [2001] 31 (3) Fam. Law 197

Horsey, Kirsty, 'Challenging Presumptions: Legal Parenthood and Surrogacy Arrangements' [2010] 22 (4) Child and Family Law Quarterly 449

Horsey, Kirsty and Sheldon, Sally, 'Still Hazy After All These Years: The Law Regulating Surrogacy' (2012) 20 (1) Medical Law Review 67

Jadva, Vasanti, Murray, Clare, Lycett, Emma, MacCallum, Fiona and Golombok, Susan, 'Surrogacy: The Experiences of Surrogate Mothers' [2003] 18 (10) Human Reproduction 2196

James, Nicholas, 'Separate Legal Personality: Legal Reality and Metaphor' (1993) 5 (2) Bond Law Review 217

Kaganas, Felicity, 'Regulating Emotion: Judging Contact Disputes' [2011] 23 (1) Child and Family Law Quarterly 63

Kaganas, Felicity, 'A Presumption that 'Involvement' of Both Parents is Best: Deciphering Law's Messages' [2013] 25 (3) Child and Family Law Quarterly 270

Kaganas, Felicity and Day Sclater, Shelley, 'Contact and Domestic Violence - The Winds of Change?' [2000] 30 (9) Fam. Law 630

Kaganas, Felicity and Day Sclater, Shelley, 'Contact Disputes: Narrative Constructions of 'Good' Parents' (2004) 12 (1) Feminist Legal Studies 1

Kaganas, Felicity and Diduck, Alison, 'Incomplete Citizens: Changing Images of Post-Separation Children' (2004) 67 (6) Modern Law Review 959

Kan, Man Yee, Sullivan, Oriel and Gershuny, Jonathan, 'Gender Convergence in Domestic Work: Discerning the Effects of Interactional and Institutional Barriers from Large-Scale Data' (2011) 45 (2) Sociology 234

Kelly, Fiona, 'Nuclear Norms or Fluid Families? Incorporating Lesbian and Gay Parents and Their Children into Canadian Family Law' (2004) 21 (1) Canadian Journal of Family Law 133

Kelly, Fiona, '(Re)Forming Parenthood: The Assignment of Legal Parentage Within Planned Lesbian Families' (2009) 40 (2) Ottawa Law Review 185

Kielty, S., 'Similarities and Differences in the Experiences of Non-Resident Mothers and Non-Resident Fathers' (2006) 20 (1) International Journal of Law Policy and the Family 74

Lawson, F.H., 'The Creative Use of Legal Concepts' (1957) 32 (5) New York University Law Review 909

Leckey, Robert, 'Law Reform, Lesbian Parenting, and the Reflective Claim' (2011) 20 (3) Social and Legal Studies 331

Leckey, Robert, 'Must Equal Mean Identical? Same-sex Couples and Marriage' (2014) 10 (1) International Journal of Law in Context 5

Lewis, Jane, 'Adoption: The Nature of Policy Shifts in England and Wales 1972-2002' (2004) 18 (2) International Journal of Law, Policy and the Family 235

Lind, Craig and Hewitt, Tom, 'Law and the Complexities of Parenting: Parental Status and Parental Function' (2009) 31 (4) Journal of Social Welfare and Family Law 391

Lyonette, Clare and Crompton, Rosemary, 'Sharing the Load? Partners Relative Earnings and the Division of Domestic Labour' (2015) 29 (1) Work, Employment and Society 23

MacCallum, Fiona, Lycett, Emma, Murray, Clare, Jadva, Vasanti and Golombok, Susan, 'Surrogacy: The Experience of Commissioning Couples' [2003] 18 (6) Human Reproduction 1334

MacFarlane, Alan, 'Review of Stone' (1979) 18 (1) History and Theory 103

MacKay, Kirsteen, 'A Plea from Scotland: Preserving Access to Courts in Private Law Child Contact Disputes' [2013] 25 (3) Child and Family Law Quarterly 294

McCandless, Julie and Sheldon, Sally, 'The Human Fertilisation and Embryology Act (2008) and the Tenacity of the Sexual Family Form' (2010) 73 (2) Modern Law Review 175

McGlynn, Clare, 'The Europeanisation of Family Law' [2001] 13 (1) Child and Family Law Quarterly 35

Miles, Jo, Wasoff, Fran and Mordaunt, Enid, 'Reforming Family Law - The Case of Cohabitation: 'Things May Not Work Out as You Expect' (2012) 34 (2) Journal of Social Welfare and Family Law 167

Millbank, Jenni, 'The Limits of Functional Family: Lesbian Mother Litigation in the Era of the Eternal Biological Family' (2008) 22 (2) International Journal of Law, Policy and the Family 149

Mnookin, Robert, 'Child-Custody Adjudication: Judicial Functions in the Face of Indeterminancy' (1975) 39 (3) Law and Contemporary Problems 226

Montgomery, Jonathan, 'Rhetoric and 'Welfare' (1989) 9 (3) Oxford Journal of Legal Studies 395

Munby, Sir James, 'Years of Change: Family Law in 1987, 2012 and 2037' [2013] 43 (3) Fam. Law 278

Naffine, Ngaire, 'Who are Law's Persons: From Cheshire Cats to Responsible Subjects' (2003) 66 (3) Modern Law Review 346

Parker, Stephen, 'The Best Interests of the Child - Principles and Problems' (1994) 8 (1) International Journal of Law and the Family 26

Piper, Christine, 'Assumptions about Children's Best Interests' (2000) 22 (3) Journal of Social Welfare and Family Law 261

Poulter, Sebastian, 'The Definition of Marriage in English Law' (1979) 42 (4) Modern Law Review 40

Probert, Rebecca, 'Cohabitation in Twentieth Century England and Wales: Law and Policy' (2004) 26 (1) Law and Policy 13

Probert, Rebecca, 'Families, Assisted Reproduction and the Law' [2004] 16 (3) Child and Family Law Quarterly 271

Probert, Rebecca, 'Hyde v Hyde: Defining or Defending Marriage?' [2007] 19 (3) Child and Family Law Quarterly 322

Probert, Rebecca, 'Common-law Marriage: Myths and Misunderstandings' [2008] 20 (1) Child and Family Law Quarterly 1

Probert, Rebecca, 'The Evolving Concept of Non-Marriage' [2013] 25 (3) Child and Family Law Quarterly 314

Reece, Helen, 'The Paramountcy Principle: Consensus or Construct?' (1996) 49 (1) Current Legal Problems 267

Sandland, Ralph, 'Not 'Social Justice': The Housing Association, The Judges, The Tenant and His Lover' (2000) 8 (2) Feminist Legal Studies 227

Sheldon, Sally, 'Evans v Amicus Healthcare; Hadley v Midland Fertility Services - Revealing Cracks in the 'Twin Pillars'?' [2004] 16 (4) Child and Family Law Quarterly 437

Sheldon, Sally, 'Gender Equality and Reproductive Decision-Making' (2004) 12 (3) Feminist Legal Studies 303

Sheldon, Sally, 'Fragmenting Fatherhood: The Regulation of Reproductive Technologies' (2005) 68 (4) Modern Law Review 523

Sheldon, Sally, 'From 'Absent Objects of Blame' to 'Fathers who Want to Take Responsibility': Reforming Birth Registration Law' (2009) 31 (4) Journal of Social Welfare and Family Law 373

Smart, Carol, 'Losing the Struggle for Another Voice: The Case for Family Law' (1995) 18 (2) Dalhousie Law Journal 173

Smart, Carol, 'Family Secrets: Law and Understandings of Openness in Everyday Relationships' (2009) 38 (4) Journal of Social Policy 551

Smart, Carol, 'Law and Family Life: Insights from 25 Years of Empirical Research' [2014] 26 (1) Child and Family Law Quarterly 14

Smith, Bryant, 'Legal Personality' (1928) 37 (3) Yale Law Journal 283

Smith, Leanne, 'Is Three a Crowd? Lesbian Mothers Perspectives on Parental Status in Law' [2006] 18 (2) Child and Family Law Quarterly 231

Smith, Leanne, 'Clashing Symbols? Reconciling Support for Fathers and Fatherless Families After the Human Fertilisation and Embryology Act 2008' [2010] 22 (1) Child and Family Law Quarterly 46

Smith, Leanne, 'Tangling the Web of Legal Parenthood: Legal Responses to the Use of Known Donors in Lesbian Parenting Arrangements' (2013) 33 (3) Legal Studies 355

Stafford, Helen, 'Concepts of Family Under EU Law - Lessons from the ECHR' (2002) 16 (3) International Journal of Law Policy and the Family 410

Stychin, Carl, 'Not (Quite) a Horse and Carriage' (2006) 14 (1) Feminist Legal Studies 79

Thomson, Joseph, 'Whither the 'Right' of Access?' 1989 SLT (News) 109

Waaldijk, Kees, 'Standard Sequences in the Legal Recognition of Homosexuality - Europe's Past, Present and Future' (1994) 4 Australasian Gay and Lesbian Law Journal 50

Wallbank, Julie, 'Castigating Mothers: The Judicial Response to 'Wilful' Women in Disputes over Paternal Contact in English Law' (1998) 20 (4) Journal of Social Welfare and Family Law 357

Wallbank, Julie, 'Too Many Mothers? Surrogacy, Kinship and the Welfare of the Child' (2002) 10 (3) Medical Law Review 271

Wallbank, Julie, 'Reconstructing the HFEA 1990: Is Blood Really Thicker Than Water?' [2004] 16 (4) Child and Family Law Quarterly 387

Wallbank, Julie, 'Getting Tough on Mothers: Regulating Contact and Residence' (2007) 15 (2) Feminist Legal Studies 189

Wallbank, Julie and Dietz, Chris, 'Lesbian Mothers, Fathers and Other Animals: Is the Political Personal in Multiple Parent Families?' [2013] 25 (4) Child and Family Law Quarterly 451

Wikeley, Nick, 'Fitzpatrick v Sterling Housing Association Ltd.: Same-Sex Partners and Succession to Rent Act Tenancies' [1998] 10 (2) Child and Family Law Quarterly 191

Wilson, Graeme, 'The Non-Resident Parental Role for Separated Fathers: A Review' (2006) 20 (3) International Journal of Law, Policy and the Family 286

Woodcraft, Elizabeth, 'Re G: A Missed Opportunity' [2007] 37 (1) Fam. Law 53

Woods, Lorna, 'Family Rights in the EU - Disadvantaging the Disadvantaged?' [1999] 11 (1) Child and Family Law Quarterly 17

Wong, Simone 'Cohabitation and The Law Commission's Project' (2006) 14 (2) Feminist Legal Studies 145

Wong, Simone, 'Shared Commitment, Interdependency and Property Relations: A Socio-Legal Project for Cohabitation' [2012] 24 (1) Child and Family Law Quarterly 60

Zanghellini, Aleardo, 'A v B and C [2012] EWCA Civ 285 - Heteronormativity, Poly-Parenting, and the Homo-Nuclear Family' [2012] 24 (4) Child and Family Law Quarterly 475

Reports

Adoption Policy Review Group, 'Adoption: Better Choices for Our Children', (June 2005)

Barlow, Anne, Burgoyne, Carole, Clerly, Elizabeth and Smithson, Janet, 'Cohabitation and the Law: Myths, Money and the Media', British Social Attitudes Survey 24, (SAGE, 2008)

Canadian Law Commission, 'Beyond Conjugality: Recognising and Supporting Close Adult Personal Relationships', (December 2001)

Crompton, Rosemary and Lyonette, Clare, 'Who Does the Housework? The Division of Labour within the Home', British Social Attitudes Survey 24, (SAGE, 2008)

Department for Children, Schools and Families, 'Support for All: The Families and Relationships Green Paper', (January 2010)

Department for Education, 'Positive for Youth: A New Approach to Cross-Government Policy for Young People Aged 13 to 19', (February 2010)

Department for Education, 'Supporting Families in the Foundation Years', (October 2011)

Department for Education and Ministry of Justice Consultation, 'Co-operative Parenting Following Family Separation: Proposed Legislation on the Involvement of Both Parents in a Child's Life', (July 2012)

Department of Health, 'Adoption: a New Approach', (December 2000)

Department of Health, 'Review of the Human Fertilisation and Embryology Act', (December 2006)

Department of Health and Social Security, 'Report of the Committee of Inquiry into Human Fertilisation and Embryology', (July 1984)

Deputy Minister for Justice, Hugh Henry, 'Response to Justice 1 Committee Stage 1 Report', (August 2005)

Equal Opportunities Commission, 'Fathers: Balancing Work and Family', (March 2003)

Family Justice Review, 'Final Report', (November 2011)

HM Government, 'The Coalition: Our Programme for Government', (May 2010)

HM Government, 'Equal Marriage: The Government's Response', (December 2012)

Home Office, 'Supporting Families: A Consultation Document', (December 1998)

Hunt, Joan and MacLeod, Alison, 'Outcomes of Applications to Court for Contact Orders After Parental Separation or Divorce', (Ministry of Justice, September 2008)

Hunt, Joan with Roberts, Ceridwen, 'Family Policy Briefing 3: Child Contact with Non-Resident Parents', (University of Oxford, 2004)

Joint Committee on the Human Tissue and Embryology (Draft) Bill, 'Human Tissue and Embryos (Draft) Bill, Volume I: Report', HL Paper 169, HC Paper 630, (August 2007)

Law Commission, 'Cohabitation: The Financial Consequences of Relationship Breakdown', Law Com No 307, (July 2007)

Ministry of Justice, 'Judicial and Court Statistics 2011', (June 2012)

Ministry of Justice and Department for Education, 'The Government Response to the Family Justice Review: A System with Families and Children at its Heart', (February 2012)

National Records of Scotland, 'Divorces Time Series Data', (March 2013)

National Records of Scotland, 'Marriage and Civil Partnership Time Series Data', (August 2015)

National Records of Scotland, 'Vital Events Reference Tables 2014, Section 2: Adoption and Re-Registrations', (August 2015)

Nuffield Council on Bioethics, 'Novel Techniques for the Prevention of Mitochondrial DNA Disorders: An Ethical Review', (June 2012)

O'Brien, Margaret, 'Shared Caring: Bringing Fathers into the Frame', (Equal Opportunities Commission, 2005)

Office for National Statistics (ONS), 'Time Use Survey 2005', (July 2006)

ONS, 'Social Trends: Households and Families', (No. 41, February 2011)

ONS, 'Short Report: Cohabitation in the UK, 2012', (November 2012)

ONS, 'Women in the Labour Market', (September 2013)

ONS, 'Divorces in England and Wales, 2012', (February 2014)

ONS, 'Marriages in England and Wales (Provisional), 2012', (June 2014)

ONS Release, 'Marriages in England and Wales (Provisional), for Same Sex Couples Q1 and Q2 2014', (August 2014)

Parliamentary Under-Secretary of State, Ministry of Justice, Jonathan Djanogly, (6th September 2011, Hansard HC Col 16WS)

Park, Alison and Rhead, Rebecca, 'Personal Relationships: Changing Attitudes Towards Sex, Marriage and Parenthood', British Social Attitudes Survey 30, (2013)

Registrar General for Scotland, 'Scotland's Population 2011 - The Registrar General's Annual Review of Demographic Trends 157th edition', (August 2012)

Registrar General for Scotland, 'Scotland's Population 2013 - The Registrar General's Annual Review of Demographic Trends 159th edition', (August 2014)

Scott, Jacqueline, 'Family and Gender Roles: How Attitudes are Changing?', (GeNet Working Paper, No 21, 2006)

Scott, Jacqueline and Clery, Elizabeth, 'Gender Roles: An Incomplete Revolution?', British Social Attitudes Survey 30, (2013)

Scottish Law Commission, 'Report on Family Law', (No 135, May 1992)

The Scottish Office, 'Scotland's Children: Proposals for Child Care Policy and Law', (August 1993)

Wasoff, Fran and Martin, Claudia, 'Scottish Social Attitudes Survey 2004: Family Module Report', (August 2005)

Newspaper Articles

Dutta, Kunal, 'Masculinity in Crisis: 'There is a Battle Going on Inside Us That is Never Discussed', (*The Independent*, London, 19th May 2013)

Hope, Christopher, 'We Will Legalise Gay Marriage by 2015, says David Cameron', (*Daily Telegraph*, London, 24th July 2012)

Poole, Glen 'How Tackling the 'Crisis of Masculinity' Creates a Crisis for Feminism', (*The Guardian*, London, 15th May 2013)

Internet Resources

'Book of Common Prayer: The Form of Solemnization of Matrimony' - https://www.churchofengland.org/prayer-worship/worship/book-of-common-prayer/the-form-of-solemnization-of-matrimony.aspx

Childlessness Overcome Through Surrogacy - http://www.surrogacy.org.uk/About_COTS.htm

'Modern Fatherhood: Fathers, Work and Families in the 21st Century' - http://www.modernfatherhood.org/themes/fathers-and-work/?view=key-facts-and-figures

'Trends in Fathers Work-Family Arrangements and Fathers Working Hours (2001-2011)' - http://www.modernfatherhood.org/wp-content/uploads/2014/01/Nuffield-PRESENTATION-FATHERS-FINAL-2.pdf

'What Do We Know about Nonresident Fathers?'http://www.modernfatherhood.org/wp-content/uploads/2013/11/Briefing-paper-Nonresident-fathers.pdf

List of Cases

A v B and C (Role of Father) [2012] 2 FLR 607

Ackerman v Ackerman [1972] 2 WLR 1253

Ampthill Peerage Case [1977] AC 547

B v B [1968] 3 WLR 566

B v B (Minors) (Custody, Care and Control) [1991] 1 FLR 402

Balfour v Balfour [1919] 2 KB 571

Baxter v Baxter [1948] AC 274

Bebb v Law Society [1914] 1 Ch. 286

Bellinger v Bellinger [2001] 2 FLR 1048

Bellinger v Bellinger [2003] 2 AC 467

Berrehab v Netherlands (1989) 11 EHRR 322

Beresford-Hope v Lady Sandhurst (1889) 23 QBD 79

Bibi v Chief Adjudication Officer [1998] 1 FLR 375

Blyth v Birmingham Waterworks Co (1856) 11 Ex. 781

Boughameni v France (1996) 22 EHRR 228

Boyle v UK (1995) 19 EHRR 179

Brixey v Lynas 1997 SC (HL) 1

Brock v Wollams [1949] 2 KB 388

Brown v Ingram (1868) 7 M 281

Burden v UK (2008) 47 EHRR 38

Burns v Burns [1984] Ch. 317

C v C (A Minor) (Custody Appeal) [1991] 1 FLR 223

C v C (Minors: Custody) [1988] 2 FLR 291

Campbell v Campbell [1977] 1 All ER 1

Charter v Race Relations Board [1973] AC 868

Chios Property Investment Co Ltd. v Lopez (1988) 20 HLR 120

Chorlton v Lings (1868) LR 4 CP 374

Christian Institute v Lord Advocate [2016] UKSC 51

Clarke v Clarke [1943] 2 All ER 540

Collier v Hollinshead (1984) 272 EG 941

Corbett v Corbett [1970] 2 All ER 33

Cowen v Cowen [1945] 2 All ER 197

Crake v Supplementary Benefits Commission [1982] 1 All ER 498

D and Sweden v Council of the European Union [2003] 3 CMLR 9 (Joined Cases C-122/99P and C-125/99P)

Dart v Dart [1996] 2 FLR 286

D-e v A-g (1845) Rob Ecc. 279

Dean v Dean [1923] P 172

De Souza v Cobden [1891] 1 QB 687

Diatta v Land Berlin [1986] 2 CMLR 164 (Case 267/83)

Diwell v Farnes [1959] 1 WLR

Duchense v Duchense [1951] P 101

Durham v Durham (1885) 10 PD 80

Dyson Holdings Ltd. v Fox [1976] 2 QB 503

Edwards v Attorney General Canada [1930] AC 124

Evans v Amicus Healthcare Ltd. [2004] 3 WLR 681

Fitzpatrick v Sterling Housing Association Ltd. [2001] 1 AC 27

Gammans v Ekins [1950] 2 KB 328

Garrad v Inglis 2014 GWD 1-17

Ghaidan v Godin-Mendoza [2004] 2 AC 557

Gilbey v Gilbey [1927] P 197

Gillick v West Norfolk and Wisbech Area Health Authority [1986] AC 112

Gould v Gould [1970] 1 QB 275

Hall v Bull [2013] 1 WLR 3741

Hall v Incorporated Society of Law Agents (1901) 9 SLT 150

Harrogate Borough Council v Simpson (1985) 17 HLR 205

Hawes v Evenden [1953] 1 WLR 1169

Helby v Rafferty [1979] 1 WLR 13

Hyde v Hyde (1866) LR 1 P&D 130

J v C [1970] AC 668

J v J 1978 SLT 128

Jex-Blake v Senatus of University of Edinburgh (1873) 11 M 784

Jia v Migrationsverket [2007] QB 545 (Case C-1/05)

Johnston v Ireland (1987) 9 EHRR 203

Jones v Kernott [2012] 1 FLR 45

Jones v Padavatton [1969] 1 WLR 328

Jones v Trueman [1949] EGD 277

Jones v Whitehill [1950] 2 KB 204

Joram Developments Ltd. v Sharratt [1979] 1 WLR 928

Karner v Austria (2004) 38 EHRR 24

Keegan v Ireland (1994) 18 EHRR 342

Kimber v Kimber [2000] 1 FLR 383

L v L 1931 SC 477

Langdon v Horton [1951] 1 KB 666

Leeds Teaching Hospitals NHS Trust v A [2003] 1 FLR 1091

M v M (Child: Access) [1973] 2 All ER 81

M v M (Parental Responsibility) [1999] 2 FLR 737

M v T 2011 GWD 40-828

MacLeod v MacLeod [2010] 1 AC 298

Marckx v Belgium (1979) 2 EHRR 330

Mata Estevez v Spain [2001] ECHR 896

Merritt v Merritt [1970] 1 WLR 1211

Midland Bank Trust Co Ltd. v Green [1982] Ch. 529

Morgan v Morgan [1959] P 92

Moustaquim v Belgium (1991) 13 EHRR 802

Nairn v University of St. Andrews and Others [1909] AC 147

Netherlands v Reed [1987] 2 CMLR 448 (Case 59/85)

Northern Ireland Human Rights Commission's Application [2013] NICA 37

Parker v Clark [1960] 1 WLR 286

Payne v Payne [2001] 1 FLR 1052

Perry v Dembrowski [1951] 2 KB 420

Pettitt v Pettitt [1969] 2 WLR 966

Porchetta v Porchetta 1986 SLT 105

Porter v Porter [1969] 1 WLR 1155

Pretty v Pretty [1911] P 83

Price v Gould [1930] All ER 389

Price v Rhondda Urban District Council (1923) 2 Ch. 372

R v E and F (Female Parents: Known Father) [2010] 2 FLR 383

R v Human Fertilisation and Embryology Authority Ex p. Blood [1997] 2 WLR 806

R v R [1952] 1 All ER 1194

R v R (Rape: Marital Exemption) [1992] 1 FLR 217

R (On the Application of Rose) v Secretary of State for Health [2002] EWHC 1593 (Admin)

R (On the Application of Williamson) v Secretary of State for Education and Employment [2005] 2 AC 246

R v Secretary of State for the Home Department, Ex p. Bugdaycay [1987] AC 514

R v Secretary of State for the Home Department, Ex p. Sivakumaran and Conjoined Appeals [1988] AC 958

Re A (A Minor: Custody) [1991] 2 FLR 394

Re A (Children: 1959 UN Declaration) [1998] 1 FLR 354

Re A (Contact: Domestic Violence) [1998] 2 FLR 171

Re Agar-Ellis (1883) 24 Ch. 317

Re B (A Child) [2010] 1 FLR 551

Re B (An Infant) [1962] 1 All ER 872

Re B (Contact: Stepfathers Opposition) [1997] 2 FLR 579

Re B (Role of the Biological Father) [2008] 1 FLR 1015

Re Bethell (1887) 38 Ch. D 220

Re C (A) (An Infant) [1970] 1 WLR 288

Re D (A Child) (Paternity) [2007] 2 FLR 26

Re D (A Minor) (Contact: Mother's Hostility) [1993] 2 FLR 1

Re D (An Infant) (Adoption: Parent's Consent) [1977] AC 602

Re D (Contact and Parental Responsibility: Lesbian Mothers and Known Father)

[2006] 1 FCR 556

Re D (Contact: Reasons for Refusal) [1997] 2 FLR 48

Re E (Adoption: Freeing Order) [1995] 1 FLR 382

Re F (A Minor) (Blood Tests: Parental Rights) [1993] Fam. 314

Re F (Paternity: Jurisdiction) [2008] 1 FLR 225

Re G (A Child) (Parental Responsibility Order) [2006] 2 FLR 1092

Re G (Children) (Residence: Same-Sex Partner) [2006] 2 FLR 629

Re G (Children) (Shared Residence Order: Biological Non-Birth Mother) [2014] 2 FLR 897

Re G (Contact) [2006] 1 FLR 1663

Re G, Re Z (Children: Sperm Donors: Leave to Apply for Children Act Orders) [2013] 1 FLR 133

Re H (A Child) (Parental Responsibility) [2002] EWCA Civ 542

Re H (A Minor) (Blood Tests: Parental Rights) [1996] 2 FLR 65

Re H (A Minor: Custody) [1990] 1 FLR 51

Re H (A Minor) (Custody: Interim Care and Control) [1991] 2 FLR 109

Re H (Contact with Biological Father) [2012] 2 FLR 627

Re H and A (Children) (Paternity: Blood Test) [2002] 1 FLR 1145

Re J (A Child) (Custody Rights: Jurisdiction) [2006] 1 AC 80

Re J (A Minor) (Contact) [1994] 1 FLR 729

Re K (A Child) [2010] EWCA Civ 478

Re K (Residence Order: Securing Contact) [1999] 1 FLR 583

Re KD (A Minor) (Ward: Termination of Access) [1988] AC 806

Re L (A Child) (Care: Threshold Criteria) [2007] 1 FLR 2050

Re L (An Infant) [1967] 3 WLR 1645

Re L (Contact: Domestic Violence); Re V (Contact: Domestic Violence); Re M (Contact: Domestic Violence; Re H (Contact: Domestic Violence) [2000] 2 FLR 334

Re L (Infants) [1962] 3 All ER 1

Re M (Contact: Welfare Test) [1995] 1 FLR 274

Re M (Intractable Contact Dispute: Interim Care Order [2003] 2 FLR 636

Re N [2010] 1 FLR 272

Re O (Contact: Imposition of Conditions) [1995] 2 FLR 124

Re O (Contact: Withdrawal of Application) [2004] 1 FLR 1258

Re P (Adoption: Unmarried Couples) [2008] 2 FLR 1084

Re P (A Minor) (Contact) [1994] 2 FLR 374

Re P (A Minor) (Custody) [1983] 4 FLR 401

Re P (Contact: Discretion) [1998] 2 FLR 696

Re P (Minors) (Contact: Parental Hostility) [1996] 2 FLR 314

Re P & L (Contact) [2012] 1 FLR 1068

Re R (A Child) [2003] Fam. 129

Re R (A Child) (IVF: Paternity of Child) [2005] 2 AC 621

Re S (A Minor) (Custody: Children's Welfare) [1991] 2 FLR 388

Re S (A Minor) (Parental Responsibility) [1995] 2 FLR 648

Re S (Minors) (Access) [1990] 2 FLR 166

Re T (A Child) [2005] EWCA Civ 1397

Re T (Paternity: Ordering Blood Tests) [2001] 2 FLR 1190

Re W (A Minor) (Adoptions: Homosexual Adopter) [1997] 3 WLR 768

Re W (A Minor) (Contact) [1994] 2 FLR 441

Re W (A Minor) (Residence Order) [1992] 2 FLR 332

Re W (Children) (Direct Contact) [2013] 1 FLR 494

Re W (Residence) [1999] 2 FLR 390

Ross v Collins [1964] 1 WLR 425

S v HM Advocate 1989 SLT 469

S v S [1972] AC 24

S v S 2014 SLT (Sh Ct) 165

S v S (Custody of Children) [1980] 1 FLR 143

S v S & P [1962] 2 All ER 1

Salomon v Salomon & Co Ltd. [1897] AC 22

Salter v Lask [1925] 1 KB 584

Sanderson v McManus 1997 SC (HL) 55

Schalk and Kopf v Austria (2011) 53 EHRR 20

Schlesinger v Schlesinger [1960] P 191

Sefton Holdings v Cairns [1988] 2 FLR 109

Semayne's Case (1604) 77 ER 194

Sheffield City Council v E and S [2005] 1 FLR 965

Sheffield City Council v Personal Representatives of Wall [2011] 1 WLR 1342

Short v Poole Corporation (1926) Ch. 66

Simpson v United Kingdom (1986) 47 D&R 274

Spellman v Spellman [1961] 1 WLR 921

Stack v Dowden [2007] 1 FLR 1858

T, Petitioner 1997 SLT 724

TvT (Shared Residence) [2011] 1 FCR 267

Vermeire v Belgium (1993) 15 EHRR 488

Vervaeke v Smith [1982] 2 All ER 144

Viscountess Rhondda's Claim [1922] 2 AC 339

W v W and C [1968] 1 WLR 1310

Wachtel v Wachtel (No 2) [1973] 2 WLR 366

Watson v Lucas [1980] 1 WLR 1493

White v White [2001] 1 AC 596

White v White 2001 SLT 485

Wilkinson v Kitzinger (No. 2) [2007] 1 FLR 295

Wilson v Town Clerk of Salford (1868) LR 4CP 398

Wood v Wood [1891] P 272

X, Y and Z v UK (1997) 24 EHRR 143

List of Legislation

Domestic Legislation

Administration of Estates Act 1925

Adoption and Children Act 2002

Adoption and Children (Scotland) Act 2007

Births and Deaths Registration Act 1953

Children Act 1989

Children and Families Act 2014

Children and Young People (Scotland) Act 2014

Children (Scotland) Act 1995

Child Support Act 1991

Civil Partnership Act 2004

Criminal Justice (Scotland) Act 1980

Divorce Reform Act 1969

Divorce (Scotland) Act 1976

Family Law Act 1986

Family Law Act 1996

Family Law Reform Act 1969

Family Law Reform Act 1987

Family Law (Scotland) Act 1985

Family Law (Scotland) Act 2006

Fatal Accidents Act 1976

Gender Recognition Act 2004

Guardianship of Infants Act 1886

Guardianship of Infants Act 1925

Housing Act 1980

Housing Act 1985

Housing Act 1988

Housing (Scotland) Act 1988

Human Fertilisation and Embryology Act 1990

Human Fertilisation and Embryology Act 2008

Human Fertilisation and Embryology Authority (Disclosure of Donor Information) Regulations 2004

Human Fertilisation and Embryology (Mitochondrial Donation) Regulations 2015

Human Rights Act 1998

Increase of Rent and Mortgage Interest (Restrictions) Act 1920

Inheritance (Provision for Family and Dependants) Act 1975

Inheritance Tax Act 1984

Interpretation Act 1850

Law Reform (Husband and Wife) (Scotland) Act 1984

Law Reform (Parent and Child) (Scotland) Act 1986

Legal Aid, Sentencing and Punishment of Offenders Act 2012

Local Government Act 1988

Marriage Act 1836

Marriage Act 1949

Marriage and Civil Partnership (Scotland) Act 2014

Marriage (Same Sex Couples) Act 2013

Marriage (Scotland) Act 1939

Marriage (Scotland) Act 1977

Married Women's Property Act 1870

Married Women's Property Act 1882

Married Women's Property Act 1884

Married Women's Property Act 1893

Married Women's Property (Scotland) Act 1881

Married Women's Property (Scotland) Act 1920

Matrimonial Causes Act 1857

Matrimonial Causes Act 1973

Matrimonial Homes (Family Protection) (Scotland) Act 1981

Municipal Corporations Act 1882

Registration of Births, Deaths and Marriages (Scotland) Act 1965

Rent Act 1977

Rent (Scotland) Act 1984

Representation of the People Act 1867

Representation of the People Act 1918

Representation of the People (Equal Franchise) Act 1928

Representation of the People (Scotland) Act 1868

Sex Disqualification Removal Act 1919

Sexual Offences Act 1967

Succession (Scotland) Act 1964

Surrogacy Arrangements Act 1985

Taxation of Chargeable Gains Act 1992

Tenant's Rights (Scotland) Act 1980

International Legislation

Basic Law for the Federal Republic of Germany

Constitution of Ireland

European Convention on Human Rights

European Economic Community Regulation 1662/68/EEC

European Union Directive 2004/38/EC

United Nations Convention on the Rights of the Child